UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM	10-Q
-------------	------

WI 10-Q
OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 iod ended October 1, 2010
Or DR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
rom to
le Number) 000-30419
TOR CORPORATION ant as specified in its charter) 36-3840979 (I.R.S. Employer Identification No.)
AcDowell Road ix, AZ 85008) 244-6600 (ding area code, of principal executive offices)
ed to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the file such reports), and (2) has been subject to such filing requirements for the past 90
and posted on its corporate Web site, if any, every Interactive Data File required to be apter) during the preceding 12 months (or for such shorter period that the registrant was
accelerated filer, a non-accelerated filer, or a smaller reporting company. See definition Rule 12b-2 of the Exchange Act.
Accelerated filer \square Smaller reporting company \square
d in Rule 12b-2 of the Exchange Act). Yes \square No \boxtimes
of the close of business on October 29, 2010:
Number of Shares 432 649 179

INDEX

Part 1: Financial information	
Item 1. Financial Statements (unaudited)	1
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	41
Item 3. Quantitative and Qualitative Disclosures About Market Risk	66
Item 4. Controls and Procedures	67
Part II: Other Information	
<u>Item 1. Legal Proceedings</u>	68
Item 1A, Risk Factors	71
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	72
Item 3. Defaults Upon Senior Securities	72
Item 4. (Removed and Reserved)	72
Item 5. Other Information	72
Item 6. Exhibits	73
<u>Signatures</u>	74
Exhibit Index	

PART I: FINANCIAL INFORMATION

Item 1. Financial Statements (unaudited)

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEET (in millions, except share and per share data) (unaudited)

	October 1, 2010	December 31, 2009
Assets		
Cash and cash equivalents	\$ 562.9	\$ 525.7
Short-term investments	_	45.5
Receivables, net	314.3	260.9
Inventories, net	350.8	269.9
Other current assets	52.9	51.5
Deferred income taxes, net of allowances	14.8	15.1
Total current assets	1,295.7	1,168.6
Restricted cash	_	5.9
Property, plant and equipment, net	820.5	705.5
Goodwill	198.2	175.4
Intangible assets, net	318.6	298.7
Other assets	61.5	60.2
Total assets	\$ 2,694.5	\$ 2,414.3
Liabilities, Stockholders' Equity and Minority Interests		
Accounts payable	\$ 250.9	\$ 172.9
Accrued expenses	149.1	135.5
Income taxes payable	2.4	5.0
Accrued interest	4.5	0.9
Deferred income on sales to distributors	134.5	98.8
Current portion of long-term debt	135.8	205.9
Total current liabilities	677.2	619.0
Long-term debt	650.2	727.6
Other long-term liabilities	46.4	49.3
Deferred income taxes, net of allowances	18.1	13.8
Total liabilities	1,391.9	1,409.7
Commitments and contingencies (See Note 10)		
ON Semiconductor Corporation stockholders' equity:		
Common stock (\$0.01 par value, 750,000,000 shares authorized, 480,967,475 and 474,427,706 shares issued, 432,436,177 and		
427,254,100 shares outstanding, respectively)	4.8	4.7
Additional paid-in capital	2,987.7	2,916.6
Accumulated other comprehensive loss	(59.6)	(64.9)
Accumulated deficit	(1,274.9)	(1,504.4)
Less: treasury stock, at cost; 48,531,298 and 47,173,606 shares, respectively	(377.2)	(367.0)
Total ON Semiconductor Corporation stockholders' equity	1,280.8	985.0
Minority interests in consolidated subsidiaries	21.8	19.6
Total equity	1,302.6	1,004.6
Total liabilities and equity	\$ 2,694.5	\$ 2,414.3

See accompanying notes to consolidated financial statements

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE INCOME

(in millions, except per share data) (unaudited)

	Quarter	Nine Months Ended		
	October 1, 2010	October 2, 2009	October 1, 2010	October 2, 2009
Revenues	\$ 600.7	\$ 472.9	\$1,734.2	\$1,271.8
Cost of revenues	354.2	297.1	1,015.8	845.7
Gross profit	246.5	175.8	718.4	426.1
Operating expenses:				
Research and development	63.3	53.8	188.6	148.1
Selling and marketing	35.5	30.0	107.6	87.4
General and administrative	31.2	27.1	98.0	84.4
Amortization of acquisition-related intangible assets	7.9	7.3	23.8	21.8
Restructuring, asset impairments and other, net	0.9	7.9	7.0	25.6
Total operating expenses	138.8	126.1	425.0	367.3
Operating income	107.7	49.7	293.4	58.8
Other income (expenses), net:				
Interest expense	(14.1)	(15.8)	(45.0)	(49.2)
Interest income	0.2	0.1	0.4	0.7
Other	(0.8)	(1.5)	(7.0)	(4.2)
Loss on debt repurchase			(0.7)	(3.1)
Other income (expenses), net	(14.7)	(17.2)	(52.3)	(55.8)
Income before income taxes	93.0	32.5	241.1	3.0
Income tax provision	(4.6)	(1.9)	(9.4)	(8.1)
Net income (loss)	88.4	30.6	231.7	(5.1)
Net income attributable to minority interests	(0.6)	(0.7)	(2.2)	(1.9)
Net income (loss) attributable to ON Semiconductor Corporation	\$ 87.8	\$ 29.9	\$ 229.5	\$ (7.0)
Comprehensive income (loss):				
Net income (loss)	\$ 88.4	\$ 30.6	\$ 231.7	\$ (5.1)
Foreign currency translation adjustments	3.5	3.7	5.3	(10.5)
Amortization of prior service costs of defined benefit plan	_	_	_	0.1
Effects of cash flow hedge	<u> </u>		<u> </u>	0.1
Comprehensive income (loss)	91.9	34.3	237.0	(15.4)
Comprehensive income attributable to minority interest	(0.6)	(0.7)	(2.2)	(1.9)
Comprehensive income (loss) attributable to ON Semiconductor Corporation	\$ 91.3	\$ 33.6	\$ 234.8	\$ (17.3)
Net income (loss) per common share attributable to ON Semiconductor Corporation:				
Basic	\$ 0.20	\$ 0.07	\$ 0.53	\$ (0.02)
Diluted	\$ 0.20	\$ 0.07	\$ 0.52	\$ (0.02)
Weighted average common shares outstanding:				
Basic	431.6	423.3	430.0	419.2
Diluted	439.8	439.1	439.8	419.2
2 marca		100.1		110.2

See accompanying notes to consolidated financial statements

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CASH FLOWS (in millions) (unaudited)

	Nine Mont	ths Ended
	October 1, 2010	October 2, 2009
Cash flows from operating activities:		
Net income (loss)	\$ 231.7	\$ (5.1
Adjustments to reconcile net income (loss) to net cash provided by operating activities:		
Depreciation and amortization	122.4	117.6
Gain on sale and disposal of fixed assets	(5.2)	(2.2
Non-cash portion of loss on debt repurchase	0.7	0.7
Amortization of debt issuance costs and debt discount	1.9	2.4
Provision for excess inventories	2.7	15.7
Non-cash impairment charges	_	5.6
Non-cash stock compensation expense	41.3	42.3
Non-cash interest expense	25.3	26.6
Deferred income taxes	5.3	(0.9
Other	(1.5)	(1.6
Changes in assets and liabilities (exclusive of the impact of acquisitions):		
Receivables	(43.0)	(74.1
Inventories	(66.8)	53.6
Other assets	(1.4)	10.0
Accounts payable	32.8	(5.0
Accrued expenses	9.9	(3.8
Income taxes payable	(2.6)	2.6
Accrued interest	3.6	3.4
Deferred income on sales to distributors	35.7	(12.6
Other long-term liabilities	(0.8)	(0.6
Net cash provided by operating activities	392.0	174.6
Cash flows from investing activities:		
Purchases of property, plant and equipment	(145.9)	(46.3
Purchases of businesses, net of cash acquired	(91.1)	_
Funds received (deposits utilized) for purchases of property, plant and equipment	1.2	(1.7
Purchase of held-to-maturity securities	_	(89.9
Proceeds from sale of held-to-maturity securities	45.5	14.9
Net cash used in investing activities	(190.3)	(123.0
Cash flows from financing activities:		
Proceeds from issuance of common stock under the employee stock purchase plan	5.1	3.7
Proceeds from debt issuance	79.0	18.7
Proceeds from exercise of stock options	7.9	13.9
Payment of capital lease obligation	(24.8)	(21.6
Purchase of treasury stock	(10.2)	(5.0
Repayment of long-term debt	(222.5)	(125.1
Net cash used in financing activities	(165.5)	(115.4
Effect of exchange rate changes on cash and cash equivalents	1.0	0.3
Net increase (decreases) in cash and cash equivalents	37.2	(63.5
Cash and cash equivalents, beginning of period	525.7	458.7
Cash and cash equivalents, end of period	\$ 562.9	\$ 395.2
1 1	- 502.5	ψ 555,2
Supplementary disclosure of non-cash investing and financing activities	\$ —	\$ 28.5
Common stock issuance for debt repurchase	\$ —	\$ 28.5

See accompanying notes to consolidated financial statements

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (unaudited)

Note 1: Background and Basis of Presentation

ON Semiconductor Corporation, together with its wholly and majority-owned subsidiaries (the "Company"), is a premier supplier of high performance, silicon solutions for energy efficient electronics. The Company's broad portfolio of power and signal management, logic, discrete and custom devices helps customers efficiently solve their design challenges in automotive, communications, computing, consumer, industrial, LED lighting, medical, military/aerospace and power applications.

On January 27, 2010, the Company completed the purchase of California Micro Devices Corporation, a Delaware corporation ("CMD"), whereby CMD became a wholly-owned subsidiary of the Company (See Note 4: "Acquisitions" for further discussion).

On June 9, 2010, the Company completed the purchase of Sound Design Technologies, Ltd. ("SDT"), whereby SDT became a wholly-owned subsidiary of the Company (see Note 4: "Acquisitions" for further discussion).

The accompanying unaudited financial statements as of October 1, 2010, and for the three months and nine months ended October 1, 2010 and October 2, 2009, respectively, have been prepared in accordance with generally accepted accounting principles for interim financial information. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for audited financial statements. In the opinion of the Company's management, the interim information includes all adjustments, consisting only of normal recurring adjustments, necessary for a fair statement of the results for the interim periods. The footnote disclosures related to the interim financial information included herein are also unaudited. Such financial information should be read in conjunction with the consolidated financial statements and related notes thereto for the year ended December 31, 2009, included in the Company's Annual Report on Form 10-K. The results for the interim periods are not necessarily indicative of the results of operations that may be expected for the full year.

The preparation of financial statements in accordance with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amount of assets and liabilities at the date of the financial statements and the reported amount of revenues and expenses during the reporting period. Significant estimates have been used by management in conjunction with the measurement of valuation allowances relating to trade and tax receivables, inventories and deferred tax assets; estimates of future payouts for customer incentives, warranties, and restructuring activities, assumptions surrounding future pension obligations and related trust returns; the fair value of stock options and of financial instruments (including derivative financial instruments); and future cash flows associated with long-lived assets and goodwill impairment charges. Actual results could differ from these estimates.

Note 2: Goodwill and Intangible Assets

Goodwill

Goodwill represents the excess of the purchase price over the estimated fair value of the net assets acquired in the Company's acquisitions (see Note 4: "Acquisitions" for further discussion).

Goodwill is evaluated for potential impairment on an annual basis or whenever events or circumstances indicate that impairment may have occurred using a two-step process. The first step of the goodwill impairment test, used to identify potential impairment, compares the estimated fair value of the reporting unit containing goodwill with the related carrying amount. If the estimated fair value of the reporting unit exceeds its carrying

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

amount, the reporting unit's goodwill is not considered to be impaired and the second step of the impairment test is unnecessary. If the reporting unit's carrying amount exceeds its estimated fair value, the second step of the test must be performed to measure the amount of the goodwill impairment loss, if any. The second step of the test compares the implied fair value of the reporting unit's goodwill, determined in the same manner as the amount of goodwill recognized in a business combination, with the carrying amount of such goodwill. If the carrying amount of the reporting unit's goodwill exceeds the implied fair value of that goodwill, an impairment loss is recognized in an amount equal to that excess. The Company is currently performing its annual impairment analysis as of the first day of the current fiscal fourth quarter. Adverse changes in operating results and/or unfavorable changes in economic factors used to estimate fair values could result in a non-cash impairment charge in the future.

The Company has determined that its product families, which are components of its operating segments, constitute reporting units for purposes of allocating and testing goodwill; because they are one level below the operating segments, they constitute individual businesses and the Company's segment management controllers regularly review the operating results of each product family. As of each acquisition date, all goodwill was assigned to the product families that were expected to benefit from the synergies of the respective acquisition. The amount of goodwill assigned to each reporting unit was the difference between the fair value of the reporting unit and the fair value of identifiable assets and liabilities allocated to the reporting unit as of the acquisition date. The Company determined the fair value of a reporting unit using the income approach, which is based on the present value of estimated future cash flows using management's assumptions and forecasts as of the acquisition date.

A reconciliation of the cost of the goodwill from each of the below acquisition transactions to the carrying value as of October 1, 2010 and December 31, 2009 for each reporting unit that contains goodwill, is as follows, in millions:

			Balance as of December 31, 2009			For the Nine Months Ended October 1, 2010							
<u>Acquisiti</u> on	Operating Segment	Reporting Unit	Goodwill	Accumulate Amortizatio		Carrying Value	Goodwill Acquired	Purchase Price Adjustments	Impairment Losses	Goodwill	Accumulated Amortization	Accumulated Impairment Losses	Carrying Value
Cherry acquisition:													
	Automotive	& Power Group:	A 24.0		D) #	A 450	Φ.	Φ.	Φ.	A 24.0	ф (4.D)	•	A 450
		Analog Automotive	\$ 21.8	\$ (4.	2) \$ —	\$ 17.6	\$ —	\$ —	\$ —	\$ 21.8	\$ (4.2)	\$ —	\$ 17.6
	Computing &	& Consumer Products:	20.1	(5	(1)	22.5				20.1	(F.C)		22.5
Leshan additional in		Signal & Interface	29.1	(5.	6) —	23.5	_	_	_	29.1	(5.6)	_	23.5
Lesnan adaitional ii		1 (
	Standard Pro		2.0			2.0				2.0			2.0
43.670		Small Signal	3.8	_	_	3.8				3.8			3.8
AMIS acquisition:	D: :: 10 M	. 1											
	Digital & Mi Signal Pi	xea- roduct Group:											
		Industrial	238.7	_	(214.7)	24.0	_	_	_	238.7	_	(214.7)	24.0
		Foundry	146.2	_	(131.4)	14.8	_	_	_	146.2	_	(131.4)	14.8
		Medical	79.7	_	(59.9)		_	_	_	79.7	_	(59.9)	
		Military/Aerospace	44.8	_		44.8	_	_	_	44.8	_		44.8
Catalyst acquisition													
	Standard Pro	ducts:											
		Memory Products	14.1	_	_	14.1	_	_	_	14.1	_	_	14.1
PulseCore acquisition													
	Digital & Mi Signal Pi	xed- roduct Group:											
	Ü	Protection Products	13.0	_	_	13.0	_	(4.1) —	8.9	_	_	8.9
CMD acquisition:									^				
•	Standard Pro	ducts:											
		Filter Products	_	_	_	_	20.3	(0.2	_	20.1	_	_	20.1
SDT acquisition								,					
·	Digital & Mi Group:	xed-Signal Product											
	•	Medical					5.9	0.9		6.8			6.8
			\$ 591.2	\$ (9.	8) \$ (406.0)	\$ 175.4	\$ 26.2	\$ (3.4) \$ —	\$ 614.0	\$ (9.8)	\$ (406.0)	\$ 198.2

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Certain adjustments to goodwill were recorded during the quarter and nine months ended October 1, 2010 for the finalization of contingent tax liabilities, receipt of claims on escrow from certain acquisitions and inventory valuation issues.

Intangible Assets

The Company's acquisitions resulted in intangible assets consisting of values assigned to intellectual property, assembled workforce, customer relationships, noncompete agreements, patents, developed technology, trademarks, acquired software and in-process research and development. These are stated at cost less accumulated amortization and are amortized over their economic useful life ranging from 1 to 18 years using the straight-line method and are reviewed for impairment when facts or circumstances suggest that the carrying value of these assets may not be recoverable.

Intangible assets, net were as follows as of October 1, 2010 and December 31, 2009 (in millions):

			October 1, 2010		
	Original <u>Cost</u>	Accumulated Amortization	Foreign Currency Translation Adjustment	Carrying Value	Useful Life (in Years)
Intellectual property	\$ 13.9	\$ (6.6)	\$ —	\$ 7.3	5-12
Assembled workforce	6.7	(5.7)	_	1.0	5
Customer relationships	250.3	(46.4)	(27.2)	176.7	5-18
Non-compete agreements	0.5	(0.5)	_		1-3
Patents	16.7	(3.9)	_	12.8	12
Developed technology	107.8	(19.8)	_	88.0	5-12
Trademarks	11.0	(1.5)	_	9.5	15
In-process research and development	23.4	(0.1)	_	23.3	8
Acquired software	1.0	(1.0)	_	_	2
Total intangibles	\$431.3	\$ (85.5)	\$ (27.2)	\$ 318.6	
			December 31, 2009		
	Original Cost	Accumulated Amortization	December 31, 2009 Foreign Currency Translation Adjustment	Carrying Value	Useful Life (in Years)
Intellectual property			Foreign Currency Translation		
Intellectual property Assembled workforce	Cost	Amortization	Foreign Currency Translation Adjustment	Value	(in Years)
	Cost 13.9	Amortization \$ (5.3)	Foreign Currency Translation Adjustment	\frac{Value}{\\$ 8.6}	(in Years) 5-12
Assembled workforce	** 13.9 6.7	Amortization \$ (5.3) (4.7)	Foreign Currency Translation Adjustment \$	Value \$ 8.6 2.0	(in Years) 5-12 5
Assembled workforce Customer relationships	Cost \$ 13.9 6.7 244.8	Amortization \$ (5.3) (4.7) (33.2)	Foreign Currency Translation Adjustment \$	Value \$ 8.6 2.0 184.4	(in Years) 5-12 5 5-18
Assembled workforce Customer relationships Non-compete agreements	Cost \$ 13.9 6.7 244.8 0.5	Amortization \$ (5.3) (4.7) (33.2) (0.3)	Foreign Currency Translation Adjustment \$	Value	(in Years) 5-12 5 5-18 1-3
Assembled workforce Customer relationships Non-compete agreements Patents	Cöst \$ 13.9 6.7 244.8 0.5 16.7	Amortization \$ (5.3) (4.7) (33.2) (0.3) (2.9)	Foreign Currency Translation Adjustment \$	Value \$ 8.6 2.0 184.4 0.2 13.8	(in Years) 5-12 5 5-18 1-3 12
Assembled workforce Customer relationships Non-compete agreements Patents Developed technology	Cöst \$ 13.9 6.7 244.8 0.5 16.7 89.4	Amortization \$ (5.3) (4.7) (33.2) (0.3) (2.9) (12.2)	Foreign Currency Translation Adjustment \$	Value \$ 8.6 2.0 184.4 0.2 13.8 77.2	(in Years) 5-12 5 5-18 1-3 12 5-12
Assembled workforce Customer relationships Non-compete agreements Patents Developed technology Trademarks	Cöst \$ 13.9 6.7 244.8 0.5 16.7 89.4 11.0	Amortization \$ (5.3) (4.7) (33.2) (0.3) (2.9) (12.2) (0.9)	Foreign Currency Translation Adjustment \$	Value \$ 8.6 2.0 184.4 0.2 13.8 77.2 10.1	(in Years) 5-12 5 5-18 1-3 12 5-12 15

Amortization expense for intangible assets amounted to \$8.5 million and \$25.4 million for the quarter and nine months ended October 1, 2010, of which \$0.6 million and \$1.8 million was included in cost of revenues; and \$7.9 million and \$23.5 million for the quarter and nine months ended October 2, 2009, of which \$0.6 million and \$1.7 million was included in cost of revenues. During the quarter ended October 1, 2010 the Company began

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

amortizing one project upon its completion worth \$5.1 million over its useful life of 11 years. Amortization expense for intangible assets, with the exception of the remaining \$18.3 million of in-process research and development assets that will be amortized once the corresponding projects have been completed, is expected to be as follows over the next five years, and thereafter (in millions):

	lectual perty	mbled kforce	Rela	istomer itionships Assets	compete ements	Patents	eloped nology	Trade mark		Software	Res	earch and opment	Total
Remainder of 2010	\$ 0.4	\$ 0.4	\$	4.5	\$ 	\$ 0.3	\$ 2.7	\$ 0.	.2	\$ —	\$	0.1	\$ 8.6
2011	1.1	0.6		17.9	_	1.3	10.7	0.	.8	_		0.5	32.9
2012	0.7	_		17.9	_	1.3	10.7	0.	.8	_		0.5	31.9
2013	0.7	_		13.2	_	1.3	10.7	0.	.8	_		0.5	27.2
2014	0.7	_		13.2	_	1.3	10.5	0.	.8	_		0.4	26.9
Thereafter	3.7	_		110.0	_	7.3	42.7	6.	.1	_		3.0	172.8
Total estimated amortization expense	\$ 7.3	\$ 1.0	\$	176.7	\$ 	\$ 12.8	\$ 88.0	\$ 9.	.5	\$ —	\$	5.0	\$300.3

Note 3: New Accounting Pronouncements Adopted

Adoption of Accounting Standards Update No. 2010-06, "Fair Value Measurements and Disclosures (Topic 820): Improving Disclosures about Fair Value Measurements"

In January 2010, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") No. 2010-06, "Fair Value Measurements and Disclosures (Topic 820): Improving Disclosures about Fair Value Measurements" ("ASU 2010-06"), which amends the disclosure guidance with respect to fair value measurements. Specifically, the new guidance requires disclosure of amounts transferred in and out of Levels 1 and 2 fair value measurements, a reconciliation presented on a gross basis rather than a net basis of activity in Level 3 fair value measurements, greater disaggregation of the assets and liabilities for which fair value measurements are presented and more robust disclosure of the valuation techniques and inputs used to measure Level 2 and 3 fair value measurements. ASU 2010-06 is effective for interim and annual reporting periods beginning after December 15, 2009, with the exception of the new guidance around the Level 3 activity reconciliations, which is effective for fiscal years beginning after December 15, 2010. The adoption of this pronouncement did not have a material impact on the Company's consolidated financial statements.

Note 4: Acquisitions

Pending Acquisition of SANYO Semiconductor Co., Ltd.

On July 14, 2010, the Company announced it had entered into a definitive purchase agreement with SANYO Electric Co., Ltd, a Japanese corporation ("SANYO Electric"), providing for the acquisition of SANYO Semiconductor Co., Ltd., a Japanese corporation and subsidiary of SANYO Electric ("SANYO Semiconductor"), and other assets related to SANYO Electric's semiconductor business (the "SANYO Transaction"). The purchase price is approximately ¥33.0 billion (\$366.0 million), subject to adjustment at closing. The Company has been working with SANYO Electric and SANYO Semiconductor on the transaction closing conditions, and given the time table required to complete a number of these closing conditions, the Company currently believes the transaction will close in the beginning of the first quarter of 2011, rather than the fourth quarter of 2010 as originally estimated. The transaction is subject to regulatory approvals and closing conditions. The Company does not currently believe it will need to access the capital markets to finance the transaction, nor does it currently anticipate the need to issue any new shares of common stock to finance the transaction.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

The Company believes that this acquisition will provide it with access to market-leading Japanese and Asian customers, while providing SANYO Semiconductor customers with access to advanced front-end mixed-signal and analog manufacturing, and ultra high volume back-end facilities. Ultimately, the Company believes that the combination of SANYO Semiconductor operations with its own will provide it with highly complementary products, customers and geographic regions.

The purchase price is payable in Yen and therefore the actual US dollar value of the purchase price will be determined based on the prevailing exchange rate on the date of closing. All US dollar amounts reflected above are based on ¥90.2 to \$1 USD exchange rate on the date of the announcement. On October 1, 2010 the exchange rate was ¥83.75 to \$1 USD and had the transaction closed on that date, the consideration payable would have been approximately \$394.0 million.

Acquisition of Sound Design Technologies LTD

On June 9, 2010, the Company completed the purchase of SDT, whereby SDT became a wholly-owned subsidiary of the Company. The Company paid approximately \$22.0 million in cash for all outstanding stock, and has recorded a contingent liability of \$1.8 million representing the estimated fair value pursuant to its obligations under an earnout agreement if SDT is able to meet certain revenue objectives in 2010 through 2012. The range of potential earn-out payments during the period from 2010 to 2012 is from zero to \$10.0 million. SDT is a leading designer and manufacturer of ultra-low-power semiconductor solutions for hearing aids and portable, battery-powered DSP applications, and a leading provider of advanced high density interconnected technologies used in custom minimizing packages. SDT's advanced manufacturing expertise in chip-scale capacitors and high density packaging will expand the Company's capabilities in delivering advanced, highly miniaturized packaging technology. SDT's results of operations have been included in the consolidated financial statements since the date of the acquisition.

The following table presents the allocation of the purchase price of SDT, to the assets acquired based on their estimated fair values (in millions):

\$ 3.3
7.8
1.0
2.7
5.9
2.4
2.8
2.0
2.0 27.9 \$ (2.2)
\$ (2.2)
(1.9)
(4.1) \$23.8
\$23.8

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Of the \$5.2 million of acquired intangible assets, \$2.8 million was assigned to in-process research and development ("IPRD") assets that will be amortized over the useful life upon successful completion of the projects or expensed if impaired. The value assigned to IPRD was determined by considering the importance of products under development to the overall development plan, estimating costs to develop the purchased IPRD into commercially viable products, estimating the resulting net cash flows from the projects when completed and discounting the net cash flows to their present value. The fair value of IPRD was determined using the income approach. The income approach recognizes that the current value of an asset or liability is premised on the expected receipt or payment of future economic benefits generated over its remaining life. A discount rate of 9.0% was used in the present value calculations, and was derived from a weighted-average cost of capital analysis, adjusted to reflect the risks inherent in the acquired research and development operations.

The remaining \$2.4 million of acquired intangible assets have a weighted-average useful life of approximately 10 years. The intangible assets that make up the amount include: customer relationships of \$1.7 million (15.5-year weighted average useful life) and developed technology of \$0.7 million (5-year weighted average useful life).

Of the total purchase price of approximately \$23.8 million, approximately \$5.9 million was initially allocated to goodwill. Goodwill represents the excess of the purchase price of an acquired business over the fair value of the underlying net tangible and intangible assets. Among the factors that contributed to a purchase price in excess of the fair value of the net tangible and intangible assets were the potential synergies expected to be derived from combining SDT's design and manufacturing business with the Company's medical business. The Company expects these relationships to provide the capability of selling advanced technology of next generation products to the market place. Goodwill will not be amortized but instead tested for impairment at least annually (more frequently if certain indicators are present). The \$6.8 million of goodwill as of October 1, 2010 was assigned to the digital and mixed signal product group, none of which is expected to be deductible for tax purposes.

The initial allocation of purchase price is based on management estimates and assumptions, and other information compiled by management, which utilized established valuation techniques appropriate for the high technology industry, which were the income approach, cost approach or market approach, depending upon which was the most appropriate based on the nature and reliability of the data available. The income approach is predicated upon the value of future cash flows that an asset is expected to generate. The cost approach takes into account the cost to replace (or reproduce) the asset and the effects on the asset's value of physical, functional and/or economic obsolescence that has occurred with respect to the asset. The market approach is a technique used to estimate value from an analysis of actual transactions or offerings for economically comparable assets available as of the valuation date. As of October 1, 2010, management of the Company had not received all information necessary to finalize the allocation of the purchase price. Management expects to complete this during the fourth quarter of the 2010 fiscal year. Such adjustments are not expected to be material.

The Company has determined that pro forma results of operations for SDT are not significant for inclusion.

Acquisition of California Micro Devices Corporation

On January 27, 2010, the Company completed the purchase of CMD, whereby CMD became a wholly-owned subsidiary of the Company. At the effective time of the merger, the Company purchased all of CMD's issued and outstanding shares of common stock at a purchase price of \$4.70 per share, for a total cash payment of approximately \$109.5 million and \$3.7 million of estimated fair value of stock options and restricted stock for total consideration of \$113.2 million. Total acquisition-related costs were approximately \$2.0 million. CMD's

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

business is primarily engaged in application specific integrated passive (ASIP) devices in the wireless, computing and consumer electronics end-markets. In addition, CMD's expertise in protection solutions for the high brightness LED (HBLED) market, and its strengths in inductor capacitor-based EMI (electromagnetic interface) filtering and low capacitance ESD (electrostatic discharge) protection, complement the Company's existing portfolio of protection and lighting solutions. During the third quarter of 2010 the Company integrated CMD's processes into the Company's own systems and control environment.

The following table presents the allocation of the purchase price of CMD, to the assets acquired based on their estimated fair values (in millions):

	ф. 42.0
Cash and cash equivalents	\$ 42.8
Receivables, net	5.0
Inventory	9.0
Other current assets	2.0
Property, plant and equipment	1.7
Goodwill	20.3
Intangible assets	21.7
In-process research and development	18.6
Other non-current assets	0.1
Total assets acquired	121.2
Accounts payable	(6.2)
Other current liabilities	(1.6)
Long-term accrued liabilities	(0.2)
Total liabilities assumed	(8.0)
Net assets acquired	\$113.2

Of the \$40.3 million of acquired intangible assets, \$18.6 million was assigned to IPRD assets that will be amortized over the useful life upon successful completion of the projects or expensed if impaired. The value assigned to IPRD was determined by considering the importance of products under development to the overall development plan, estimating costs to develop the purchased IPRD into commercially viable products, estimating the resulting net cash flows from the projects when completed and discounting the net cash flows to their present value. The fair value of IPRD was determined using the income approach. The income approach recognizes that the current value of an asset or liability is premised on the expected receipt or payment of future economic benefits generated over its remaining life. A discount rate of 13.2% was used in the present value calculations, and was derived from a weighted-average cost of capital analysis, adjusted to reflect additional risks inherent in the acquired research and development operations. Total IPRD is composed of four primary projects, with approximately \$1.0 million of costs expected to be incurred from acquisition date until completion. The expected completion date is 2011.

The remaining \$21.7 million of acquired intangible assets have a makeup of: (i) developed technology of \$17.7 million (8-year weighted-average useful life) and (ii) customer relationships of \$4.0 million (10-year weighted average useful life).

Of the total purchase price paid of \$113.2 million, approximately \$20.3 million was allocated to goodwill. Goodwill represents the excess of the purchase price of an acquired business over the fair value of the underlying net tangible and intangible assets. Among the factors that contributed to a purchase price in excess of the fair

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

value of the net tangible and intangible assets was the acquisition of an assembled workforce of experienced semiconductor engineers. The Company expects these experienced engineers to provide the capability of developing and integrating advanced technology into next generation products. Goodwill will not be amortized but instead will be tested for impairment at least annually (more frequently if certain indicators are present). The \$20.3 million of goodwill was assigned to the standard products group, none of which is expected to be deductible for tax purposes.

The initial allocation of the purchase price is based on management estimates and assumptions, and other information compiled by management, which utilized established valuation techniques appropriate for the high-technology industry, which were either the income approach, cost approach or market approach, depending upon which was the most appropriate based on the nature and reliability of the data available. The income approach is predicated upon the value of the future cash flows that an asset will generate over its economic life. The cost approach takes into account the cost to replace (or reproduce) the asset and the effect on the asset's value of physical, functional and/or economic obsolescence that has occurred with respect to the asset. The market approach is a technique used to estimate value from an analysis of actual transactions or offerings for economically comparable assets available as of the valuation date. The Company has determined that pro forma results of operations for CMD are not significant for inclusion.

Acquisition of AMIS Holdings, Inc. ("AMIS")

On March 17, 2008, the Company completed the purchase of AMIS, whereby AMIS became a wholly-owned subsidiary of the Company.

The Company had \$12.4 million of accrued liabilities for estimated costs to exit certain activities of AMIS, of which \$0.7 million were for employee separation costs and \$11.7 million were for exit costs outstanding as of December 31, 2009. During the nine months ended October 1, 2010, the Company paid employee separation costs related to the involuntary termination or relocation of employees performing overlapping or duplicative functions throughout AMIS by \$0.5 million. Additionally, the Company paid exit costs associated with the decommissioning costs resulting from the shutdown of the fabrication facility of \$1.6 million.

The following is a rollforward of the accrued liabilities for estimated costs to exit certain activities of AMIS from December 31, 2009 through October 1, 2010:

	December 31,			October 1,
	2009	Adjustments	Usage	2010
Estimated employee separation costs:				
December 31, 2009 through October 1, 2010	\$ 0.7	<u> </u>	<u>\$(0.5)</u>	\$ 0.2
Estimated costs to exit:				
December 31, 2009 through October 1, 2010	<u>\$ 11.7</u>	<u>\$ —</u>	\$(1.6)	\$ 10.1

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Note 5: Restructuring, Asset Impairments and Other, Net

The activity related to the Company's restructuring, asset impairments and other, net for programs that were either initiated in 2010 or had not been completed as of December 31, 2009, are as follows:

Restructuring

Restructuring Activities Related to the 2010 Acquisition of SDT

Cumulative charges of \$2.2 million, net of adjustments have been recognized through October 1, 2010 related to the announced plans to integrate and restructure the overlapping operations of SDT and the Company, in part for cost savings purposes (see Note 4: "Acquisitions" for further discussion).

Cumulative employee separation charges of \$2.2 million, net of adjustments have been recognized though October 1, 2010. A total of 36 employees, including three former executive officers of SDT, were notified that their positions were being eliminated or consolidated, of which all individuals have been terminated. During the quarter and nine months ended October 1, 2010, the Company recorded employee separation charges of approximately \$0.2 million and \$2.2 million, respectively, related to these terminations. These charges have been included in restructuring, asset impairment and other, net on the consolidated statement of operations for the quarter and nine months ended October 1, 2010. As of October 1, 2010, all terminations and related payments associated with these plans were completed during the third quarter of fiscal 2010.

	Balance at Beginning of Period	Charges	Usage	Adjustments	Balance at End of <u>Period</u>
Cash employee separation charges:					
Nine Months Ended October 1, 2010	<u>\$</u>	\$ 2.2	<u>\$(2.2)</u>	<u>\$</u>	<u>\$</u>

Restructuring Activities Related to the 2010 Acquisition of CMD

Cumulative charges of \$3.6 million, net of adjustments have been recognized through October 1, 2010, related to the January 2010 announced plans to integrate and restructure the overlapping operations of CMD and the Company, in part for cost savings purposes (see Note 4: "Acquisitions" for further discussion).

Cumulative employee separation charges of \$3.5 million, net of adjustments have been recognized through October 1, 2010. A total of 25 employees, including five former executive officers of CMD, were notified during the first nine months of 2010 that their positions were being eliminated or consolidated, of which 19 of these individuals have been terminated. During the quarter and nine months ended October 1, 2010, the Company recorded employee separation charges of approximately \$0.7 million and \$3.5 million, respectively, related to these terminations. These charges have been included in restructuring, asset impairment and other, net on the consolidated statement of operations for the quarter and nine months ended October 1, 2010. All terminations associated with this plan are expected to be completed by the end of the fourth quarter of fiscal 2010, with the related termination benefits paid out by the end of the first quarter of fiscal 2012.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Cumulative exit costs of \$0.1 million have been recognized through October 1, 2010, related to charges incurred to terminate certain lease agreements. During the quarter and nine months ended October 1, 2010, the Company recognized zero and \$0.1 million, respectively, in charges on the statement of operations related to this activity. All payments related to these exit activities are expected to be completed by the end of the third quarter of fiscal 2011.

	Balance at Beginning of Period	Charges	Usage	Adjustments	Balance at End of Period
Cash employee separation charges:					
Nine Months Ended October 1, 2010	<u>\$ —</u>	\$ 3.5	\$(1.8)	<u> </u>	\$ 1.7
Exit Costs:					
Nine Months Ended October 1, 2010	<u>\$ —</u>	\$ 0.1	<u>\$—</u>	<u> </u>	\$ 0.1

Restructuring Activities Related to the 2009 Design Centers Closures

Cumulative charges of \$1.5 million, net of adjustments, have been recognized through October 1, 2010, related to the 2009 Design Center closures. During the third quarter of 2009, the Company announced plans to consolidate into fewer product development centers for cost savings purposes by closing several design centers. A total of 47 employees were notified during the third quarter of 2009 that their positions with the Company were being terminated. Additionally, during the first nine months ended October 1, 2010, 16 employees were notified that their positions with the Company were being eliminated or consolidated, of which all had exited as of October 1, 2010. During the quarter and nine months ended October 1, 2010, the Company recorded no employee separation charges and \$0.3 million in exit costs related to this activity, respectively. These charges have been included in restructuring, asset impairment and other, net on the consolidated statement of operations for the quarter and nine months ended October 1, 2010. All terminations and related payments associated with these plans were completed during the second quarter of fiscal 2010.

	Balance at Beginning <u>of Period</u>	Charges	<u>Usage</u>	Adjustments	Balance at End of Period
Cash employee separation charges:					
Nine Months Ended October 1, 2010	\$ 0.3	\$ 0.2	\$(0.5)	<u> </u>	<u>\$</u>
Exit Costs:					
Nine Months Ended October 1, 2010	\$ 0.1	\$ 0.1	\$(0.2)	\$	\$ —

Restructuring Activities Related to the 2009 Global Workforce Reduction

Cumulative employee separation charges of \$13.0 million, net of adjustments, have been recognized through October 1, 2010, related to the first quarter of 2009 announced plans to reduce worldwide personnel for cost savings purposes. A total of 570 employees were notified during 2009, of which 569 of these individuals have been terminated. These charges have been included in restructuring, asset impairment and other, net on the consolidated statement of operations for the quarter and nine months ended October 1, 2010. All terminations associated with this plan are expected to be completed by the end of the fourth quarter of 2010, with substantially all related termination benefits paid out by the end of the fourth quarter of fiscal 2010.

	Balance at Beginning				Balance at End of
	of Period	Charges	Usage	Adjustments	Period
Cash employee separation charges:					
Nine Months Ended October 1, 2010	\$ 0.7	<u>\$ —</u>	\$(0.6)	\$ 0.1	\$ 0.2

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Other

The Company made a \$0.8 million cash payment in settlement of various litigation matters with the former minority interest shareholders of a Czech subsidiary acquired by the Company. These settlement charges have been included in restructuring, asset impairment and other, net in the consolidated statement of operations for the quarter and nine months ended October 1, 2010.

A reconciliation of the activity in the tables above to the "Restructuring, asset impairments and other, net" caption on the consolidated statement of operations for the quarter and nine months ended October 1, 2010, is as follows (in millions):

	Quarter Ended October 1, 2010	Nine Months Ended October 1, 2010
Restructuring		
Charges:		
Cash employee separation charges	\$ 0.9	\$ 5.8
Exit costs	_	0.3
Less: net adjustments to reserves	-	0.1
Other		
Charges	-	0.8
	\$ 0.9	\$ 7.0

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Note 6: Balance Sheet Information

	October 1, 	December 31, 2009
Receivables, net:		
Accounts receivable	\$ 321.9	\$ 270.2
Less: Allowance for doubtful accounts	(7.6)	(9.3)
	\$ 314.3	\$ 260.9
Inventories, net:		
Raw materials	\$ 47.5	\$ 35.4
Work in process	203.2	151.6
Finished goods	100.1	82.9
	\$ 350.8	\$ 269.9
Property, plant and equipment, net:		
Land	\$ 48.7	\$ 42.0
Buildings	459.0	429.7
Machinery and equipment	1,587.2	1,420.2
Total property, plant and equipment	2,094.9	1,891.9
Less: Accumulated depreciation	(1,274.4)	(1,186.4)
	\$ 820.5	\$ 705.5
Accrued expenses:		
Accrued payroll	\$ 63.5	\$ 55.9
Sales related reserves	35.5	32.7
Restructuring reserves	12.3	13.6
Accrued pension liability	0.2	0.2
Other	37.6	33.1
	\$ 149.1	\$ 135.5
Accumulated other comprehensive loss:		
Foreign currency translation adjustments	\$ (59.3)	\$ (64.6)
Unrealized prior service cost of defined benefit pension plan	(0.1)	(0.1)
Prior service cost from pension legal plan amendment	(0.2)	(0.2)
	\$ (59.6)	\$ (64.9)

The activity related to the Company's warranty reserves for the nine months ended October 1, 2010 and October 2, 2009, respectively is as follows (in millions):

	Nine Mont	hs Ended
	October 1, 	October 2, 2009
Beginning Balance	\$ 3.2	\$ 3.9
Provision	0.7	0.6
Usage	(0.2)	(0.7)
Ending Balance	\$ 3.7	\$ 3.8

Total net periodic pension cost

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

The Company maintains defined benefit plans for some of its foreign subsidiaries. The Company recognizes the aggregate amount of all overfunded plans as an asset and the aggregate amount of all underfunded plans as a liability in its financial statements. As of October 1, 2010 and December 31, 2009, the total accrued pension liability for underfunded plans was \$21.4 million and \$19.0 million, respectively, of which the current portion of \$0.6 million and \$0.3 million, respectively, was classified as accrued expenses. As of October 1, 2010 and December 31, 2009, the total pension asset for overfunded plans was \$14.8 million and \$16.2 million, respectively. The components of the Company's net periodic pension expense for the quarters and nine months ended October 1, 2010 and October 2, 2009, respectively, are as follows (in millions):

		Qua	arter Ended	
	Octob	oer 1, 2010	9	October 2, 2009
Service Cost	\$	1.0	\$	6.0
Interest cost		8.0		0.5
Expected return on plan assets		(8.0)		(0.3)
Amortization of prior service cost		0.1		0.1
Other (Gains)/Losses		_		0.6
Total net periodic pension cost	\$	1.1	3	5 1.7
			=	
		Nine I	Months Ended	
	Octol	oer 1, 2010	(October 2, 2009
Service Cost	\$	3.0	9	5 2.5
Interest cost		2.4		1.5
Expected return on plan assets		(2.4)		(1.0)
Amortization of prior service cost		0.3		0.2
Other (Gains)/Losses		_		1.2

3.3

4.4

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Note 7: Long-Term Debt

Long-term debt consists of the following (in millions):

	October 1, 2010	December 31, 2009
Senior Bank Facilities:		
Term Loan, interest payable monthly at 2.00063%	\$ —	\$ 170.2
Zero Coupon Convertible Senior Subordinated Notes due 2024 (1)	85.9	96.9
1.875% Convertible Senior Subordinated Notes due 2025 (2)	80.7	76.5
2.625% Convertible Senior Subordinated Notes due 2026 (3)	404.7	389.0
2.25% Loan with Japanese bank due 2010, interest payable semi-annually	_	3.6
Loan with Philippine banks due 2010 through 2012, interest payable quarterly at 1.52681% and 1.28438%, respectively	16.3	18.7
Loan with Philippine bank due 2010 through 2013, interest payable quarterly at 1.05375% and 1.00563%, respectively	9.3	10.5
Loan with Philippine bank due 2010 through 2013, interest payable quarterly at 1.54188% and 1.50375%, respectively	5.2	5.9
Loan with Philippine banks due 2010 through 2014, interest payable quarterly at 6.03063% prepaid in Q2 2010	_	10.3
Loan with Philippine bank due 2010 through 2015, interest payable quarterly at 2.2925%	20.0	_
Short-term loan with Chinese bank due 2010, interest payable quarterly at 3.36938% and 3.2725%, respectively	7.0	7.0
Short-term loan with Chinese bank due 2010, interest payable quarterly at 2.7825%	_	7.0
Short-term loan with Chinese bank due 2010, interest payable quarterly at 2.7825%	_	6.0
Short-term loan with Chinese bank due 2011, interest payable quarterly at 4.09013% and 5.25063%, respectively	7.0	7.0
Short-term loan with Chinese bank due 2011, interest payable quarterly at 4.08938% and 4.28063%, respectively	7.0	7.0
Short-term loan with Chinese bank due 2010, interest payable quarterly at 2.55481% and 2.2525%, respectively	12.0	12.0
Loan with British finance company, interest payable monthly at 1.85% and 1.75%, respectively	24.4	23.1
Loan with Hong Kong bank, interest payable weekly at 2.02297%	40.0	_
1.875% Loan with Japanese bank due 2010 through 2013, interest payable semi-annually	2.5	2.6
Short-term loan with Japanese bank due 2010, interest payable monthly at .96% and 1.06%, respectively	1.8	1.6
Capital lease obligations	62.2	78.6
	786.0	933.5
Less: Current maturities	(135.8)	(205.9)
	650.2	727.6

⁽¹⁾ The Zero Coupon Convertible Senior Subordinated Notes due 2024 may be put back to the Company at the option of the holders of the notes on April 15 of 2012, 2014 and 2019 or called at the option of the Company on or after April 15, 2012.

⁽²⁾ The 1.875% Convertible Senior Subordinated Notes due 2025 may be put back to the Company at the option of the holders of the notes on December 15 of 2012, 2015 and 2020 or called at the option of the Company on or after December 20, 2012.

⁽³⁾ The 2.625% Convertible Senior Subordinated Notes due 2026 may be put back to the Company at the option of the holders of the notes on December 15 of 2013, 2016 and 2021 or called at the option of the Company on or after December 20, 2013.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Annual maturities relating to the Company's long-term debt as of October 1, 2010 are as follows (in millions):

	<u>Maturities</u>
Remainder 2010	\$54.3
2011	89.9
2012	204.9
2013	420.1
2014	3.6
Thereafter	13.2
Total	\$ 786.0

Loss on Debt Repurchase

During the nine months ended October 1, 2010, the Company incurred a loss on debt repurchase of \$0.8 million as a result of the write-off of unamortized capitalized closing costs, related to the \$169.8 million prepayment of the senior bank facilities and a \$0.1 million gain as a result of the modification of the Zero Coupon Convertible Senior Subordinated Notes due 2024.

September 2010 Chinese Loans

In September 2010, one of the Company's Chinese subsidiaries entered into two new short-term loan agreements with a Chinese bank to finance the purchase of raw materials. These short-term loans refinanced two loans with the same bank that matured in 2010. The loans, which had a balance of \$14.0 million as of October 1, 2010, bear interest payable quarterly based on 3-month LIBOR plus 3.8% per annum. The full amount of the loans is due in September 2011.

September 2010 Philippine Loan

In September 2010, the Company's Philippine subsidiary entered into a five-year loan agreement with a Philippine bank to finance capital expenditures and other general corporate purposes. The loan, which had a balance of \$20.0 million as of October 1, 2010, bears interest payable quarterly based on 3-month LIBOR plus 2.0% per annum. Sixty percent of the total loan amount will be repaid in nineteen equal quarterly installments with the balance due in September 2015.

June 2010 Hong Kong Loan

In June 2010, one of the Company's Asian subsidiaries entered into a loan with a Hong Kong bank, pursuant to which the bank purchased accounts receivables, with recourse. In accordance with Generally Accepted Accounting Principles in the United States the purchased assets remained on the Company's balance sheet as of October 1, 2010. The loan, which had a balance of \$40.0 million as of October 1, 2010, bears interest payable weekly at 2-month LIBOR plus 1.75%. The loan amount is subject to an eligible borrowing calculation as defined in the loan agreement.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Modification of the Zero Coupon Convertible Senior Subordinated Notes due 2024

In April 2010, the Company amended the Indenture for its Zero Coupon Convertible Senior Subordinated Notes due 2024.

The amendments include:

- One additional opportunity to require the Company to purchase the notes on April 15, 2012. The terms of this put option are otherwise identical to preexisting terms of the notes whereby holders of the notes had the option to require the Company to purchase the notes on April 15, 2010; and
- Terms eliminating the Company's ability to redeem the notes at its option from April 15, 2010 until April 15, 2012.

In accordance with the right of the holders of the notes to require the Company to purchase the notes on April 15, 2010, approximately \$3.2 million of the \$99.4 million par value of notes then outstanding were purchased by the Company. In accordance with ASC 470 — Debt, the amendment was considered a substantial modification for accounting purposes, therefore the \$96.2 million original remaining debt was deemed to be extinguished, resulting in a \$0.1 million gain, and new convertible debt with a fair value of \$98.5 million was deemed to be issued.

ASC 470, requires the issuer of convertible debt instruments with cash settlement features to separately account for the liability and equity components of the instrument. Thus, the liability component of the new convertible debt was recognized at the present value of its cash flows discounted using a discount rate equivalent to the borrowing rate at the date of the modification of the Convertible Notes for similar debt instruments without conversion feature. The equity component of the new convertible debt was recorded as additional paid in capital and represents the difference between the fair value of the modified Convertible Notes and the liability component. It also requires an accretion of the debt discount resulting from the allocation of a portion of the modified fair value to equity over the life of the Convertible Notes, which is expected to be the next put date of April 15, 2012.

As a result, the Company recognized \$13.3 million of debt discount, which will be amortized through April 2012.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Debt Guarantees

The Company is the sole issuer of the Zero Coupon Convertible Senior Subordinated Notes due 2024, the 1.875% Convertible Senior Subordinated Notes due 2025 and the 2.625% Convertible Senior Subordinated Notes due 2026 (collectively, the "Notes"). The Company's domestic subsidiaries, except those domestic subsidiaries acquired through the acquisitions of AMIS, Catalyst Semiconductor, Inc. ("Catalyst"), PulseCore Holdings (Cayman) Inc. ("PulseCore"), and CMD (collectively, the "Guarantor Subsidiaries"), fully and unconditionally guarantee on a joint and several basis the Company's obligations under the Notes. The Guarantor Subsidiaries include SCI LLC, Semiconductor Components Industries of Rhode Island, Inc., as well as holding companies whose net assets consist primarily of investments in the joint venture in Leshan, China and equity interests in the Company's other foreign subsidiaries. The Company's other remaining subsidiaries (collectively, the "Non-Guarantor Subsidiaries") are not guarantors of the Notes. Condensed consolidated financial information for the issuer of the Notes, the Guarantor Subsidiaries and the Non-Guarantor Subsidiaries is as follows (in millions):

	Issuer ON Semiconductor Corporation (1)		Guarantor SCI Other LLC Subsidiaries		Non-Guarantor Subsidiaries		Eliminations	Total	
As of October 1, 2010									
Cash and cash equivalents	\$	_	\$ 366.3	\$	0.1	\$	196.5	\$ —	\$ 562.9
Short-term investments		_	_		_		_	_	
Receivables, net		_	54.6		_		259.7	_	314.3
Inventories, net		_	42.2		_		298.2	10.4	350.8
Other current assets		_	8.3		_		44.6	_	52.9
Deferred income taxes		<u> </u>	5.6				9.2		14.8
Total current assets		_	477.0		0.1		808.2	10.4	1,295.7
Property, plant and equipment, net		_	200.5		2.6		620.9	(3.5)	820.5
Deferred income taxes		_			_		—	_	_
Goodwill and other intangible assets		_	298.1		37.2		216.8	(35.3)	516.8
Investments and other assets		1,856.6	1,070.2		50.8		832.3	(3,748.4)	61.5
Total assets	\$	1,856.6	\$2,045.8	\$	90.7	\$	2,478.2	\$ (3,776.8)	\$2,694.5
Accounts payable	\$		\$ 46.5	\$	0.1	\$	204.3	\$ —	\$ 250.9
Accrued expenses and other current liabilities		4.3	93.3		8.0		191.7	1.7	291.8
Deferred income on sales to distributors		_	33.1		_		101.4	_	134.5
Total current liabilities		4.3	172.9		0.9		497.4	1.7	677.2
Long-term debt		571.2	26.9		_		52.1	_	650.2
Other long-term liabilities		_	22.0		0.4		24.0	_	46.4
Deferred Income Taxes		_	6.4		_		11.7		18.1
Intercompany		0.3	(187.0)		(42.6)		23.8	205.5	
Total liabilities		575.8	41.2		(41.3)		609.0	207.2	1,391.9
Total ON Semiconductor Corporation stockholders' equity									
(deficit)		1,280.8	2,004.6		132.0		1,869.2	(4,005.8)	1,280.8
Minority interests in consolidated subsidiaries		_			_		—	21.8	21.8
Total stockholders' equity		1,280.8	2,004.6		132.0		1,869.2	(3,984.0)	1,302.6
Total liabilities and stockholders' equity	\$	1,856.6	\$2,045.8	\$	90.7	\$	2,478.2	\$ (3,776.8)	\$2,694.5

	Issuer ON Semiconductor		Guarantor SCI Other			Non-Guarantor		Eliminations		
As of December 31, 2009	Cor	poration ⁽¹⁾	LLC	Sut	osidiaries	St	ıbsidiaries	Elin	inations	Total
Cash and cash equivalents	\$		\$ 286.0	\$	_	\$	239.7	\$	_	\$ 525.7
Short-term investments	Ψ	_	— <u>— </u>	Ψ	_	Ψ	45.5	Ψ	_	45.5
Receivables, net		_	47.9		_		213.0		_	260.9
Inventories, net		_	34.2		_		230.5		5.2	269.9
Other current assets		_	7.1		_		44.4		_	51.5
Deferred income taxes		_	5.5		_		9.6		_	15.1
Total current assets	-		380.7	_		_	782.7	_	5.2	1,168.6
Restricted cash		_	_		_		5.9		_	5.9
Property, plant and equipment, net		_	149.7		2.8		557.0		(4.0)	705.5
Goodwill and intangible assets, net		_	197.1		37.3		278.1		(38.4)	474.1
Investments and other assets		1,548.3	1,210.7		45.8		29.0	(2	2,773.6)	60.2
Total assets	\$	1,548.3	\$1,938.2	\$	85.9	\$	1,652.7	\$ (2	2,810.8)	\$2,414.3
Accounts payable	\$	_	\$ 25.0	\$	0.1	\$	147.8	\$		\$ 172.9
Accrued expenses and other current liabilities		97.6	74.3		8.0		172.9		1.7	347.3
Deferred income on sales to distributors			26.9	_			71.9			98.8
Total current liabilities		97.6	126.2		0.9		392.6		1.7	619.0
Long-term debt		465.4	213.0		_		49.2		_	727.6
Other long-term liabilities		_	20.0		0.4		28.9		_	49.3
Deferred Income Taxes		_	7.6		_		6.2		_	13.8
Intercompany		0.3	(127.5)		(52.4)		(25.9)		205.5	
Total liabilities		563.3	239.3		(51.1)		451.0		207.2	1,409.7
Total ON Semiconductor Corporation stockholders' equity (deficit)		985.0	1,698.9		137.0		1,201.7	(:	3,037.6)	985.0
Minority interests in consolidated subsidiaries		<u> </u>							19.6	19.6
Total stockholders' equity		985.0	1,698.9		137.0		1,201.7	(3,018.0)	1,004.6
Total liabilities and stockholders' equity	\$	1,548.3	\$1,938.2	\$	85.9	\$	1,652.7	\$ (2	2,810.8)	\$2,414.3

	ON Ser	ssuers miconductor poration ⁽¹⁾		Other sidiaries	Non-Guarantor Subsidiaries		Eliminations Eliminations	Total
For the Quarter ended October 1, 2010								
Revenues	\$	_	\$160.3	\$ _	\$	719.8	\$ (279.4)	\$600.7
Cost of revenues		<u> </u>	144.2	 0.2		494.0	(284.2)	354.2
Gross profit			16.1	(0.2)		225.8	4.8	246.5
Research and development		_	(18.8)	2.6		79.5	_	63.3
Selling and marketing			19.2	0.3		16.0	_	35.5
General and administrative		_	3.2	0.2		27.8	_	31.2
Amortization of acquisition related intangible assets		_	4.7	_		4.3	(1.1)	7.9
Restructuring, asset impairments and other, net			0.3	 		0.6		0.9
Total operating expenses			8.6	 3.1		128.2	(1.1)	138.8
Operating income (loss)		_	7.5	(3.3)		97.6	5.9	107.7
Interest expense, net		(12.5)	0.4	_		(1.8)	_	(13.9)
Other		_	2.9	_		(3.7)	_	(0.8)
Equity in earnings		100.3	505.6	 1.2			(607.1)	
Income (loss) before income taxes and minority interests		87.8	516.4	(2.1)		92.1	(601.2)	93.0
Income tax provision			12.4	 		(17.0)		(4.6)
Net income (loss)		87.8	528.8	(2.1)		75.1	(601.2)	88.4
Net income (loss) attributable to minority interest		<u> </u>				(0.2)	(0.4)	(0.6)
Net income (loss) attributable to ON Semiconductor				_				
Corporation	\$	87.8	\$528.8	\$ (2.1)	\$	75.1	\$ (601.6)	\$ 87.8

						Non-Guarantor Subsidiaries		Elin	ninations	Total
For the quarter ended October 2, 2009				540	<u> </u>		<u> </u>			
Revenues	\$	—	\$148.2	\$	3.5	\$	612.5	\$	(291.3)	\$472.9
Cost of revenues		<u> </u>	107.7		0.5		453.1		(264.2)	297.1
Gross profit			40.5		3.0		159.4		(27.1)	175.8
Research and development		_	5.8		2.0		46.0		_	53.8
Selling and marketing		_	12.4		0.1		17.5		_	30.0
General and administrative		_	(0.4)		0.1		27.4		_	27.1
Amortization of acquisition related intangible assets		—	3.7		_		4.6		(1.0)	7.3
Restructuring, asset impairments and other, net		<u> </u>	0.1				7.8			7.9
Total operating expenses			21.6		2.2		103.3		(1.0)	126.1
Operating income (loss)		_	18.9		8.0		56.1		(26.1)	49.7
Interest expense, net		(12.5)	(1.8)		_		(1.4)		_	(15.7)
Other		_	0.9		_		(2.4)		_	(1.5)
Equity in earnings		42.4	41.8		1.5		_		(85.7)	_
Income (loss) before income taxes and minority interests		29.9	59.8		2.3		52.3		(111.8)	32.5
Income tax provision		_	(1.7)		_		(0.2)			(1.9)
Minority interests		_	_		_		_		(0.7)	(0.7)
Net income (loss)	\$	29.9	\$ 58.1	\$	2.3	\$	52.1	\$	(112.5)	\$ 29.9

		suers	Sul	iarantor osidiaries						
		iconductor ration ⁽¹⁾	SCI LLC		ther idiaries		Guarantor osidiaries	Elin	ninations	Total
For the nine months ended October 1, 2010										
Revenues	\$	_	\$ 500.0	\$	_	\$	2,140.9	\$	(906.7)	\$1,734.2
Cost of revenues			375.3		1.5		1,550.9		(911.9)	1,015.8
Gross profit		_	124.7		(1.5)		590.0		5.2	718.4
Research and development		_	4.8		7.3		176.5			188.6
Selling and marketing		_	48.8		0.7		58.1		_	107.6
General and administrative		_	11.2		0.6		86.2		_	98.0
Amortization of acquisition related intangible assets		_	12.9		_		14.0		(3.1)	23.8
Restructuring, asset impairments and other, net			0.3				6.7			7.0
Total operating expenses			78.0		8.6		341.5		(3.1)	425.0
Operating income (loss)		_	46.7		(10.1)		248.5		8.3	293.4
Interest expense, net		(38.0)	(4.2)		`— ´		(2.4)		_	(44.6)
Gain (loss) on debt prepayment			(0.7)		_				_	(0.7)
Other		_	2.9		_		(9.9)		_	(7.0)
Equity in earnings		267.5	625.7		5.0				(898.2)	
Income (loss) before income taxes and minority interests		229.5	670.4		(5.1)		236.2		(889.9)	241.1
Income tax provision			5.1				(14.5)			(9.4)
Net income (loss)		229.5	675.5		(5.1)		221.7		(889.9)	231.7
Net income (loss) attributable to minority interest							(0.1)		(2.1)	(2.2)
Net income (loss) attributable to ON Semiconductor Corporation	\$	229.5	\$ 675.5	\$	(5.1)	\$	221.6	\$	(892.0)	\$ 229.5
Net cash provided by operating activities	\$		\$ 347.2	\$	0.2	\$	44.6	\$		\$ 392.0
Cash flows from investing activities:	-		<u> </u>	*		-		-		<u> </u>
Purchases of property, plant and equipment		_	(48.6)		(0.1)		(97.2)		_	(145.9)
Funds deposited for purchases of property, plant and equipment		_	_		_		1.2		_	1.2
Proceeds from sales of held-to-maturity securities		_	_		_		45.5		_	45.5
Purchase of a business, net of cash acquired		_	_		_		(91.1)		_	(91.1)
Proceeds from sales of property, plant and equipment										
Net cash used in investing activities		_	(48.6)		(0.1)		(141.6)		_	(190.3)
Cash flows from financing activities:			<u> </u>							
Intercompany loans		_	(503.6)		_		503.6		_	
Intercompany loan repayments		_	478.6		_		(478.6)		_	_
Proceeds from debt issuance		_	_		_		79.0			79.0
Proceeds from issuance of common stock under the employee stock purchase plan		_	5.1		_		_		_	5.1
Proceeds from exercise of stock options			7.9		_		_		_	7.9
Repurchase of Treasury Stock		_	(10.2)		_		_		_	(10.2)
Payment of capital lease obligation		_	(22.8)		_		(2.0)		_	(24.8)
Repayment of long term debt			(173.3)				(49.2)			(222.5)
Net cash used in financing activities			(218.3)				52.8			(165.5)
Effect of exchange rate changes on cash and cash equivalents					_		1.0		_	1.0
Net increase (decrease) in cash and cash equivalents		_	80.3		0.1		(43.2)		_	37.2
Cash and cash equivalents, beginning of period		_	286.0		_		239.7		_	525.7
Cash and cash equivalents, end of period	\$	_	\$ 366.3	\$	0.1	\$	196.5	\$	_	\$ 562.9

ON Semiconductor Corporation (1) SCI Other Subsidiaries Non-Guarantor Subsidiaries Eliminations	\$1,271.8 845.7 426.1 148.1
For the nine months ended October 2, 2009 Revenues \$ - \$356.6 \$ 8.9 \$ 1,646.0 \$ (739.7)	\$1,271.8 845.7 426.1
Revenues \$ — \$356.6 \$ 8.9 \$ 1,646.0 \$ (739.7)	845.7 426.1
	845.7 426.1
	426.1
Gross profit	1/10/1
Research and development — 35.4 6.3 106.5 (0.1)	
Selling and marketing — 34.2 0.7 52.5 —	87.4
General and administrative — (32.0) 0.4 73.6 42.4	84.4
Amortization of acquisition related intangible assets — 8.5 — 15.2 (1.9)	21.8
Restructuring, asset impairments and other, net	25.6
Total operating expenses — 48.7 7.4 270.8 40.4	367.3
Operating income (loss) — 15.1 0.3 96.8 (53.4)	58.8
Interest expense, net (39.6) (4.2) — (4.8) 0.1	(48.5)
Other — 0.1 — (4.3) —	(4.2)
Loss on dept prepayment (3.1) — — — — —	(3.1)
Equity in earnings 35.7 37.0 4.3 119.4 (196.4)	
Income (loss) before income taxes, and minority interests (7.0) 48.0 4.6 207.1 (249.7)	3.0
Income tax provision — 0.6 — (8.7) —	(8.1)
Minority interests	(1.9)
Net income (loss) \$ (7.0) \$ 48.6 \$ 4.6 \$ 198.4 \$ (251.6)	\$ (7.0)
Net cash provided by (used in) operating activities \$ - \$(177.4) \$ - \$ 352.0 \$ -	\$ 174.6
Cash flows from investing activities:	
Purchases of property, plant and equipment — (8.9) — (37.4) —	(46.3)
Deposits utilized for purchases of property, plant and equipment — — — — — — (1.7) —	(1.7)
Purchases of held to maturity securities — — — (89.9)	(89.9)
Proceeds from sales of held to maturity securities — — — 14.9 —	14.9
Net cash provided by (used in) investing activities (8.9) (114.1)	(123.0)
Cash flows from financing activities:	
Intercompany loans — 6.5 — (6.5) —	_
Intercompany loan repayments — 230.0 — (230.0) —	_
Proceeds from debt issuance – – 18.7 – 18.7 –	18.7
Proceeds from issuance of common stock under the employee stock purchase	
plan — 3.7 — — —	3.7
Proceeds from exercise of stock options and warrants — 13.9 — — — —	13.9
Repurchase of treasury stock — (5.0) — — — —	(5.0)
Payment of capital lease obligation — (16.9) — (4.7) —	(21.6)
Repayment of long term debt	(125.1)
Net cash provided by (used in) financing activities	(115.4)
Effect of exchange rate changes on cash and cash equivalents	0.3
Net increase (decrease) in cash and cash equivalents — (20.9) — (42.6) —	(63.5)
Cash and cash equivalents, beginning of period	458.7
Cash and cash equivalents, end of period \$	\$ 395.2

⁽¹⁾ The Company is a holding Company and has no operations apart from those of its operating subsidiaries. Additionally, the Company does not maintain a bank account; rather SCI LLC, its primary domestic operating subsidiary, processes all of its cash receipts and disbursements on its behalf.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

See also Note 10: "Commitments and Contingencies—Other Contingencies" for further discussion of the Company's guarantees.

Note 8: Equity

Income per share calculations for the quarter and nine months ended October 1, 2010 and October 2, 2009, respectively, are as follows (in millions, except per share data):

	Quarte	r Ended	Nine Months Ended		
	October 1, 2010	October 2, 2009	October 1, 2010	October 2, 2009	
Net income (loss) applicable to ON Semiconductor Corporation	\$ 87.8	\$ 29.9	\$ 229.5	\$ (7.0)	
Diluted net income (loss) applicable to ON Semiconductor Corporation	\$ 87.8	\$ 29.9	\$ 229.5	\$ (7.0)	
Basic weighted average common shares outstanding	431.6	423.3	430.0	419.2	
Add: Incremental shares for:					
Dilutive effect of stock compensation	7.8	14.9	9.4	_	
1.875% convertible senior subordinated notes	0.4	0.9	0.4	_	
Diluted weighted average common shares outstanding	439.8	439.1	439.8	419.2	
Income (loss) per common share attributable to ON Semiconductor Corporation					
Basic:	\$ 0.20	\$ 0.07	\$ 0.53	\$ (0.02)	
Diluted:	\$ 0.20	\$ 0.07	\$ 0.52	\$ (0.02)	

Basic income (loss) per share is computed by dividing net income by the weighted average number of common shares outstanding during the period.

The number of incremental shares from the assumed exercise of stock options is calculated by applying the treasury stock method. For the nine months ended October 2, 2009, the effects of stock option shares and restricted stock units were not included because the related impact would have been anti-dilutive, as the Company generated a net loss. Common shares relating to employee stock options where the exercise price exceeded the average market price of the Company's common shares or the assumed exercise would have been anti-dilutive were also excluded from the diluted earnings per share calculation. The excluded option shares were 24.0 million and 18.7 million for the quarters ended October 1, 2010 and October 2, 2009, respectively, and 17.4 million and 27.9 million for the nine months ended October 1, 2010 and October 2, 2009, respectively.

For the quarter and nine months ended October 1, 2010 and October 2, 2009, the assumed conversion of the zero coupon convertible senior subordinated notes was also excluded in determining diluted net income per share. The zero coupon convertible senior subordinated notes are convertible into cash up to the par value of \$96.2 million and \$99.4 million, based on a conversion price of \$9.82 per share at October 1, 2010 and October 2, 2009, respectively. The excess of fair value over par value is convertible into stock. As of October 1, 2010 and October 2, 2009, the Company's common stock traded below \$9.82; thus, the effects of an assumed conversion would have been anti-dilutive and therefore were excluded.

For the quarter and nine months ended October 1, 2010 and October 2, 2009, the assumed conversion of the 2.625% convertible senior subordinated notes was also excluded in determining diluted net income per share. The 2.625% convertible senior subordinated notes are convertible into cash up to the par value of \$484.0 million,

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

based on a conversion price of \$10.50 per share. The excess of fair value over par value is convertible into stock. As of October 1, 2010 and October 2, 2009, the Company's common stock traded below \$10.50; thus, the effects of an assumed conversion would have been anti-dilutive and therefore were excluded.

Additionally, warrants held by non-employees to purchase 5.3 million shares of the Company's common stock, which were obtained from the AMIS acquisition, were outstanding as of October 1, 2010 and October 2, 2009, but were not included in the computation of diluted net income per share as the effect would have been anti-dilutive under the treasury stock method.

Treasury stock is recorded at cost and is presented as a reduction of stockholders' equity in the accompanying consolidated financial statements. Shares withheld upon the vesting of restricted stock units to pay applicable employee withholding taxes are considered common stock repurchases. Upon vesting, the Company currently does not collect the applicable employee withholding taxes from employees. Instead, the Company automatically withholds, from the restricted stock units that vest, the portion of those shares with a fair market value equal to the amount of the employee withholding taxes due, which is accounted for as a repurchase of common stock. The Company then pays the applicable withholding taxes in cash. The amounts remitted in the quarter ended October 1, 2010 were \$3.2 million for which the Company withheld 465,171 shares of common stock that were underlying the restricted stock units that vested. None of these shares had been reissued or retired as of October 1, 2010, but may be reissued or retired by the Company at a later date.

At December 31, 2009, the minority interest balance was \$19.6 million. This balance increased to \$21.8 million at October 1, 2010 due to the minority interest's \$0.6 million and \$2.2 million share of the earnings for the quarter and nine months, respectively, which has been reflected in the Company's consolidated statement of operations for the quarter and nine months ended October 1, 2010.

At December 31, 2008, the minority interest balance was \$17.3 million. This balance increased to \$19.2 million at October 2, 2009 due to the minority interest's \$0.7 million and \$1.9 million share of the earnings for the quarter and nine months, respectively, which has been reflected in the Company's consolidated statement of operations for the quarter and nine months ended October 2, 2009.

See Note 4: "Acquisitions" for discussion of potential issuance of common stock relating to the pending SANYO Semiconductor acquisition.

Note 9: Employee Stock Benefit Plans

At December 31, 2009, there was an aggregate of 20.6 million shares of common stock available for grant under the Company's stock option and restricted stock unit plans which subsequently expired on February 17, 2010. An amended and restated stock incentive plan was approved by the Company's shareholders at the annual shareholder meeting on May 18, 2010. At October 1, 2010, there was an aggregate of 25.5 million shares of common stock available for grant under the Company's new amended and restated stock incentive plan.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Stock Options

The weighted-average estimated fair value of stock options granted during the quarter and nine months ended October 1, 2010 was \$2.85 per share and \$2.92 per share, respectively. The weighted-average estimated fair value of stock options granted during the quarter and nine months ended October 2, 2009 was \$3.27 per share. The weighted-average assumptions associated with the stock options granted during the periods are as follows:

	Qua	rter Ended
	October 1, 2010	October 2, 2009
Volatility	49.3%	50.3%
Risk-free interest rate	1.8%	2.7%
Expected term	4.9 years	5.3 years
	Nino N	Months Ended
	October 1,	October 2,

	October 1, 2010	October 2, 2009
Volatility	48.7%	58.8%
Risk-free interest rate	1.9%	2.4%
Expected term	4.9 years	5.2 years

Pre-vesting forfeitures were estimated to be approximately 12% for the quarter and nine months ended October 1, 2010, and 12% for the quarter and nine months ended October 2, 2009, based on historical experience.

A summary of stock option transactions for all stock option plans is as follows (in millions, except per share and term data):

		Nine Months Ended October 1, 2010						
	Number of Shares	Weighted- Average Exercise Price	Weighted - Average Remaining Contractual Term (in years)	Aggregate Intrinsic Value (In-The- Money)				
Outstanding at December 31, 2009	29.4	\$ 7.48						
Assumed in acquisitions	2.9	8.23						
Grants	1.0	6.65						
Exercised	(1.4)	5.62						
Cancellations	(2.2)	11.74						
Outstanding at October 1, 2010	29.7	\$ 7.29	5.0	\$ 29.1				
Exercisable at October 1, 2010	23.0	\$ 7.37	4.3	\$ 23.3				

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Additional information about stock options outstanding at October 1, 2010 with exercise prices less than or above \$7.23 per share, the closing price of the Company's common stock at October 1, 2010, is as follows (number of shares in millions):

	Exercisable		Unexero	isable	Total		
	Number of	Weighted Average Exercise	Number of	Weighted Average Exercise	Number of	Weighted Average Exercise	
Exercise Prices	Shares	Price	Shares	Price	Shares	Price	
Less than \$7.23	13.7	\$ 5.53	4.5	\$ 5.93	18.2	\$ 5.63	
Above \$7.23	9.3	\$ 10.06	2.2	\$ 9.31	11.5	\$ 9.92	
Total outstanding	23.0	\$ 7.37	6.7	\$ 7.04	29.7	\$ 7.29	

Restricted Stock Units and Awards

Restricted stock units that vest over two to four years with service-based requirements as well as restricted stock units that vest based on performance-based requirements are payable in shares of the Company's common stock upon vesting. The following table presents a summary of the status of the Company's restricted stock units granted to certain officers, directors, and employees of the Company as of October 1, 2010, and changes during the quarter and nine months ended October 1, 2010 (number of shares in millions):

		nded October 1, 10	
	Number of Shares	Weighte Averag Grant D Fair Va	ge Oate
Nonvested shares of restricted stock units at December 31, 2009	15.1	\$ 4	4.17
Granted	3.6		7.77
Released	(4.2)	4	4.26
Forfeited	(0.5)	Į.	5.26
Nonvested shares of restricted stock units at October 1, 2010	14.0	\$!	5.02

During the nine months ended October 1, 2010, the Company granted 0.1 million shares in restricted stock awards with a weighted average grant date fair value of \$8.16 per share to members of the Board of Directors. The awards vested and were issued immediately upon the effective date of the grant.

Employee Stock Purchase Plans

As of October 1, 2010, there were 5.5 million shares available for issuance under the 2000 Employee Stock Purchase Plan ("ESPP"). The weighted-average fair value of shares issued under the ESPP during the quarters and nine months ended October 1, 2010 and October 2, 2009 were \$1.44 per share and \$1.94 per share, and \$1.76 per share and \$1.53 per share, respectively. The weighted-average assumptions used in the pricing model are as follows:

	Quarter 1	Ended	Nine Months	s Ended
Employee Stock Purchase Plan	October 1, 2010	October 2, 2009	October 1, 2010	October 2, 2009
Expected life (in years)	0.25	0.25	0.25	0.25
Risk-free interest rate	0.2%	0.2%	0.1%	0.2%
Volatility	38.0%	62.0%	40.0%	96.0%

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Share-Based Compensation Expense

Total share-based compensation expense, related to the Company's stock options, restricted stock units, restricted stock awards and employee stock purchase plan, recognized for the quarter and nine months ended October 1, 2010 and October 2, 2009, respectively, was comprised as follows (in millions):

	Quarter Ended				Nine Months E			s Ended		
	October	October 1, 2010 Oc		October 2, 2009		October 2, 2009 October 1,		er 1, 2010	Octo	ber 2, 2009
Cost of revenues	\$	4.5	\$	3.5	\$	11.1	\$	10.6		
Research and development		2.8		2.7		8.1		8.2		
Selling and marketing		2.9		2.1		7.5		6.9		
General and administrative		2.0		5.2		14.3		16.6		
Restructuring		0.1		_		0.5		_		
Share-based compensation expense before income taxes	,	12.3		13.5		41.5		42.3		
Related income tax benefits (1)		_		_		_		_		
Share-based compensation expense, net of taxes	\$	12.3	\$	13.5	\$	41.5	\$	42.3		

⁽¹⁾ Most of the Company's share-based compensation relates to its domestic subsidiaries, which have historically experienced recurring net operating losses; therefore, no related tax benefits are recorded.

At October 1, 2010, total unrecognized share-based compensation expense, net of estimated forfeitures, related to non-vested stock options and non-vested restricted stock units granted prior to that date was \$13.5 million and \$25.4 million, respectively. The total intrinsic value of stock options exercised during the quarter and nine months ended October 1, 2010 was \$0.2 million and \$3.1 million respectively. The Company recorded cash received from the exercise of stock options of \$0.6 million and \$7.9 million and cash from issuance of shares under the ESPP of \$1.8 million and \$5.1 million and recorded no related tax benefits during the quarter and nine months ended October 1, 2010, respectively. Upon the exercise of stock options, restricted stock units vesting or the completion of a purchase under the ESPP, the Company issues new shares of stock.

Note 10: Commitments and Contingencies

Leases

The following is a schedule by year of future minimum lease obligations under non-cancelable operating leases as of October 1, 2010 (in millions):

Remainder of 2010	\$ 7.2
2011	18.6
2012	15.7
2013	12.4
2014	8.1
Thereafter	38.2
Total (1)	\$100.2

⁽¹⁾ Minimum payments have not been reduced by minimum sublease rentals of \$0.1 million due in the future under subleases.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Other Contingencies

The Company's headquarters and manufacturing facility in Phoenix, Arizona is located on property that is a "Superfund" site, a property listed on the National Priorities List and subject to clean-up activities under the Comprehensive Environmental Response, Compensation, and Liability Act. Motorola is actively involved in the cleanup of on-site solvent contaminated soil and groundwater and off-site contaminated groundwater pursuant to consent decrees with the State of Arizona. As part of the August 4, 1999 recapitalization, Motorola has retained responsibility for this contamination, and has agreed to indemnify the Company with respect to remediation costs and other costs or liabilities related to this matter.

In the Czech Republic the Company has ongoing remediation projects to respond to releases of hazardous substances that occurred during the years that this facility was operated by government-owned entities. In each case, the remediation project consists primarily of monitoring groundwater wells located on-site and off-site with additional action plans developed to respond in the event activity levels are exceeded at each of the respective locations. The government of the Czech Republic has agreed to indemnify the Company and the respective subsidiaries, subject to specified limitations, for remediation costs associated with this historical contamination. Based upon the information available, total future remediation costs to the Company are not expected to be material.

The Company's design center in East Greenwich, Rhode Island has adjoining property that has localized soil contamination. In connection with the purchase of the facility, the Company entered into a Settlement Agreement and Covenant Not to Sue with the State of Rhode Island. This agreement requires that remedial actions be undertaken and a quarterly groundwater monitoring program be initiated by the former owners of the property. Based on the information available, any costs to the Company in connection with this matter are not expected to be material.

As a result of the acquisition of AMIS, the Company is a "primary responsible party" to an environmental remediation and cleanup at AMIS' former corporate headquarters in Santa Clara, California. Costs incurred by AMIS include implementation of the clean-up plan, operations and maintenance of remediation systems, and other project management costs. However, AMIS' former parent company, a subsidiary of Nippon Mining, contractually agreed to indemnify AMIS and the Company for any obligation relating to environmental remediation and cleanup at this location. The Company has not offset the receivable from Nippon Mining's subsidiary against the estimated liability on the consolidated balance sheets. Therefore, a receivable from Nippon Mining's subsidiary is recorded on the accompanying consolidated balance sheets as of October 1, 2010 related to this matter for approximately \$0.1 million. The Company does not believe that the liability and receivable amounts are material to the Company's consolidated financial position, results of operations or cash flows.

A bank guarantee issued on behalf of the Company under a non-reusable commitment credit with the bank had an outstanding amount of \$4.2 million as of October 1, 2010. The Belgian bank that issued the guarantee has the right to create a mortgage on the real property of one of the Company's European subsidiaries in the amount of \$3.0 million but had not done so as of October 1, 2010. The Company also has outstanding guarantees and letters of credit outside of its non-reusable commitment credit totaling \$13.0 million as of October 1, 2010.

As part of securing financing in the normal course of business, the Company issued guarantees related to its capital lease obligations which totaled approximately \$17.1 million as of October 1, 2010. For its operating leases, the Company expects to make cash payments and similarly incur expenses totaling \$100.2 million as future payments are made. The Company had not recorded any liability in connection with these operating leases, letters of credit and guarantee arrangements.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Based on historical experience and information currently available, the Company believes it will not be required to make payments under the standby letters of credit or guarantee arrangements.

Indemnification Contingencies

The Company is a party to a variety of agreements entered into in the ordinary course of business pursuant to which it may be obligated to indemnify the other parties for certain liabilities that arise out of or relate to the subject matter of the agreements. Some of the agreements entered into by the Company require it to indemnify the other party against losses due to intellectual property infringement, property damage including environmental contamination, personal injury, failure to comply with applicable laws, the Company's negligence or willful misconduct, or breach of representations and warranties and covenants related to such matters as title to sold assets.

The Company faces risk of exposure to warranty and product liability claims in the event that its products fail to perform as expected or such failure of its products results, or is alleged to result, in bodily injury or property damage (or both). In addition, if any of the Company's designed products are alleged to be defective, the Company may be required to participate in their recall. Depending on the significance of any particular customer and other relevant factors, the Company may agree to provide more favorable indemnity rights to such customers for valid warranty claims.

The Company has, from time to time, been active in merger and acquisition activity. In connection with these mergers or acquisitions, we have agreed to indemnify the other party or parties to the merger or acquisition agreement for certain things, limited in most instances, by time and/or monetary amounts.

The Company and its subsidiaries provide for indemnification of directors, officers and other persons in accordance with limited liability agreements, certificates of incorporation, by-laws, articles of association or similar organizational documents, as the case may be. The Company maintains directors' and officers' insurance, which should enable it to recover a portion of any future amounts paid.

In addition to the above, from time to time the Company provides standard representations and warranties to counterparties in contracts in connection with sales of its securities and the engagement of financial advisors and also provides indemnities that protect the counterparties to these contracts in the event they suffer damages as a result of a breach of such representations and warranties or in certain other circumstances relating to the sale of securities or their engagement by the Company.

While the Company's future obligations under certain agreements may contain limitations on liability for indemnification, other agreements do not contain such limitations and under such agreements it is not possible to predict the maximum potential amount of future payments due to the conditional nature of the Company's obligations and the unique facts and circumstances involved in each particular agreement. Historically, payments made by the Company under any of these indemnities have not had a material effect on the Company's business, financial condition, results of operations or cash flows. Additionally, the Company does not believe that any amounts that it may be required to pay under these indemnities in the future will be material to the Company's business, financial position, and results of operations or cash flows.

Legal Matters

The Company currently is involved in a variety of legal matters that arise in the normal course of business. Based on information currently available, management does not believe that the ultimate resolution of these matters, including the matters described or referred to in the next paragraphs, will have a material effect on the

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Company's financial condition, results of operations or cash flows. However, because of the nature and inherent uncertainties of litigation, should the outcome of these actions be unfavorable, the Company's business, consolidated financial position, results of operations or cash flows could be materially and adversely affected.

Securities Class Action Litigation

During the period July 5, 2001 through July 27, 2001, the Company was named as a defendant in three shareholder class action lawsuits that were filed in federal court in New York City against the Company and certain of its former officers, current and former directors and the underwriters of the Company's initial public offering. The lawsuits allege violations of the federal securities laws and have been docketed in the U.S. District Court for the Southern District of New York ("District Court") as: Abrams v. ON Semiconductor Corp., et al., C.A. No 01-CV-6114; Breuer v. ON Semiconductor Corp., et al., C.A. No. 01-CV-6287; and Cohen v. ON Semiconductor Corp., et al., C.A. No. 01-CV-6942. On April 19, 2002, the plaintiffs filed a single consolidated amended complaint that supersedes the individual complaints originally filed. The amended complaint alleges, among other things, that the underwriters of the Company's initial public offering improperly required their customers to pay the underwriters' excessive commissions and to agree to buy additional shares of the Company's common stock in the aftermarket as conditions of receiving shares in the Company's initial public offering. The amended complaint further alleges that these supposed practices of the underwriters should have been disclosed in the Company's initial public offering prospectus and registration statement. The amended complaint alleges violations of both the registration and antifraud provisions of the federal securities laws and seeks unspecified damages. The Company understands that various other plaintiffs have filed substantially similar class action cases against approximately 300 other publicly-traded companies and their public offering underwriters in New York City, which have all been transferred, along with the case against the Company, to a single federal district court judge for purposes of coordinated case management. The Company believes that the claims against the Company cannot guarantee that the outcome of these claims will be favora

On July 15, 2002, together with the other issuer defendants, the Company filed a collective motion to dismiss the consolidated, amended complaints against the issuers on various legal grounds common to all or most of the issuer defendants. The underwriters also filed separate motions to dismiss the claims against them. In addition, the parties have stipulated to the voluntary dismissal without prejudice of the Company's individual former officers and current and former directors who were named as defendants in the Company's litigation, and they are no longer parties to the litigation. On February 19, 2003, the District Court issued its ruling on the motions to dismiss filed by the underwriter and issuer defendants. In that ruling the District Court granted in part and denied in part those motions. As to the claims brought against the Company under the antifraud provisions of the securities laws, the District Court dismissed all of these claims with prejudice, and refused to allow plaintiffs the opportunity to re-plead these claims. As to the claims brought under the registration provisions of the securities laws, which do not require that intent to defraud be pleaded, the District Court denied the motion to dismiss these claims as to the Company and as to substantially all of the other issuer defendants as well. The District Court also denied the underwriter defendants' motion to dismiss in all respects.

In June 2003, upon the determination of a special independent committee of the Company's Board of Directors, the Company elected to participate in a proposed settlement with the plaintiffs in this litigation. Had it been approved by the District Court, this proposed settlement would have resulted in the dismissal, with prejudice, of all claims in the litigation against the Company and against any of the other issuer defendants who elected to participate in the proposed settlement, together with the current or former officers and directors of participating issuers who were named as individual defendants. This proposed issuer settlement was conditioned on, among other

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

things, a ruling by the District Court that the claims against the Company and against the other issuers who had agreed to the settlement would be certified for class action treatment for purposes of the proposed settlement, such that all investors included in the proposed classes in these cases would be bound by the terms of the settlement unless an investor opted to be excluded from the settlement in a timely and appropriate fashion.

On December 5, 2006, the U.S. Court of Appeals for the Second Circuit ("Court of Appeals") issued a decision in *In re Initial Public Offering Securities Litigation* that six purported class action lawsuits containing allegations substantially similar to those asserted against the Company could not be certified as class actions due, in part, to the Court of Appeals' determination that individual issues of reliance and knowledge would predominate over issues common to the proposed classes. On January 8, 2007, the plaintiffs filed a petition seeking rehearing *en banc* of this ruling. On April 6, 2007, the Court of Appeals denied the plaintiffs' petition for rehearing of the Court of Appeals' December 5, 2006 ruling. The Court of Appeals, however, noted that the plaintiffs remained free to ask the District Court to certify classes different from the ones originally proposed which might meet the standards for class certification that the Court of Appeals articulated in its December 5, 2006 decision. In light of the Court of Appeals' December 5, 2006 decision regarding certification of the plaintiffs' claims, the District Court entered an order on June 25, 2007 terminating the proposed settlement between the plaintiffs and the issuers, including the Company.

On August 14, 2007, the plaintiffs filed amended complaints in the six focus cases. The issuer defendants and the underwriter defendants separately moved to dismiss the claims against them in the amended complaints in the six focus cases. On March 26, 2008, the District Court issued an order in which it denied in substantial part the motions to dismiss the amended complaints in the six focus cases.

On February 25, 2009, the parties advised the District Court that they had reached an agreement-in-principle to settle the litigation in its entirety. A stipulation of settlement was filed with the District Court on April 2, 2009. On June 9, 2009, the District Court preliminarily approved the proposed global settlement. Notice was provided to the class, and a settlement fairness hearing, at which members of the class had an opportunity to object to the proposed settlement, was held on September 10, 2009. On October 6, 2009, the District Court issued an order granting final approval to the settlement. That order remains subject to appeal. Several appeals have been filed objecting to the definition of the settlement class and fairness of the settlement, and those appeals remain pending. The settlement calls for a total payment of \$586 million from all defendants, including underwriters, of which \$100 million is allocated to the approximately 300 issuer defendants. Under the settlement, the Company's insurers are to pay the full amount of settlement share allocated to the Company, and the Company would bear no financial liability. The Company, as well as the officer and director defendants (current and former) who were previously dismissed from the action pursuant to tolling agreements, are to receive complete dismissals from the case. While the Company can make no assurances or guarantees as to the outcome of these proceedings, based upon the Company's current knowledge, the Company believes that the final result of this action will have no material effect on the Company's consolidated financial position, results of operations or cash flows.

Other Litigation

On January 27, 2010, the Company completed its acquisition of all of the outstanding shares of common stock of CMD through a cash tender offer of \$4.70 per share which was then followed by the merger of Purchaser (defined below) and CMD, in accordance with the December 14, 2009 definitive merger agreement ("Merger Agreement") which the Company previously announced it had entered into with CMD ("Transaction"). Shortly after the Company signed the Merger Agreement and announced the tender offer, the Company was named as a defendant in three purported class action lawsuits, described below, filed in California and Delaware against it, CMD, CMD's Board of Directors and Pac-10 Acquisition Corporation, its direct, wholly-owned subsidiary ("Purchaser").

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

On December 14, 2009, a purported class action lawsuit was filed in the Superior Court of Santa Clara County, California ("Court") captioned *Robert Varrenti*, et al. v. Robert Dickinson, Edward Ross, John Sprague, David Wittrock, David Sear, Jon Castor, John Fichthorn, J. Michael Gullard, Kenneth Potashner, California Micro Devices, ON Semiconductor Corporation and Pac-10 Acquisition Corporation (No. 109CV159469) ("California Action"). On December 29, 2009, the plaintiff in the California Action filed an amended complaint. On December 21, 2009, a second purported class action lawsuit was filed in the Court of Chancery in the State of Delaware captioned Annamarie Medeiros et al. v. California Micro Devices, Jon S. Castor, Robert V. Dickinson, Edward C. Ross, John Fichthorn, J. Michael Gullard, Kenneth Potashner, David L. Wittrock, Pac-10 Acquisition Corporation and ON Semiconductor Corporation (No. 5159). On January 4, 2010, a third purported class action lawsuit was filed in the Court of Chancery in the State of Delaware captioned Sanjay Israni, et al. v. California Micro Devices, Robert V. Dickinson, Edward C. Ross, Jon S. Castor, John Fichthorn, J. Michael Gullard, Kenneth Potashner, David L. Wittrock, ON Semiconductor Corporation and Pac-10 Acquisition Corporation (No. 5181) (the Delaware lawsuits referred to herein as the "Delaware Actions"). All three lawsuits contain similar allegations, stating generally that the proposed Transaction is the product of a breach of fiduciary duties by CMD's Board of Directors by failing to adequately discharge their duties in negotiating and agreeing to the Transaction and that the Company and the Purchaser assisted in that breach. All three lawsuits requested an injunction enjoining the consummation of the Transaction. The Israni complaint also included a request for damages.

On January 19, 2010, the parties entered into a memorandum of understanding ("MOU") to settle the three lawsuits and on February 18, 2010, the parties entered into a stipulation of settlement ("Stipulation"). The settlement, like the MOU, calls for CMD to agree to make available to shareholders certain additional information, which has been completed, and CMD or its insurer to agree to pay plaintiffs' counsel for fees and expenses not to exceed \$495,000. The Company expects CMD's insurer to pay \$245,000 of this total amount. This payment did not affect the amount of consideration paid to the stockholders of CMD in connection with the Transaction. See the MOU document at Exhibit 10.1 to CMD's Form 8-K filed with the SEC on January 20, 2010, and see a summary of the MOU in the Company's Amendment No. 3 to Schedule TO filed with the SEC on January 20, 2010, which summary is incorporated by reference herein. The defendants maintain that the lawsuits are completely without merit. Nevertheless, the defendants agreed to the settlement in order to avoid costly litigation, eliminate the risks of delaying the closing of the Transaction, and because the only effect of the settlement on CMD stockholders was to provide additional disclosure. The Stipulation was filed with the Court on March 29, 2010 and a hearing to preliminarily approve the settlement was held on May 7, 2010. On May 25, 2010, the Court preliminarily approved the Stipulation and scheduled a settlement hearing on July 23, 2010. At that July 23, 2010 hearing, the court approved the settlement, but requested additional information from the plaintiffs regarding fees and expenses.

On August 2, 2010, the Court issued a final judgment approving the settlement set forth in the Stipulation and dismissing the California Action with prejudice. On September 8, 2010, the Delaware Court issued its order dismissing the Delaware Actions with prejudice based on the preceding order issued by the California Court. The Company believes that the final result of this action will have no material effect on the Company's consolidated financial position, results of operations or cash flows.

Intellectual Property Matters

The Company faces risk to exposure from claims of infringement of the intellectual property rights of others. In the ordinary course of business, the Company receives letters asserting that its products or components breach another party's rights. These threats may seek that the Company makes royalty payments, that it stop use of such rights, or other remedies.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Prior to the acquisition of AMIS by the Company on March 17, 2008, in January 2003, Ricoh Company, Ltd. ("Ricoh") filed in the U.S. District Court for the District of Delaware a complaint against AMIS and other parties (including Synopsys, Inc. ("Synopsys"), alleging infringement of a patent owned by Ricoh. AMIS promptly tendered the defense of this claim to Synopsys, and Synopsys agreed to assume the defense of the case on AMIS' behalf to the extent that the Synopsys software that AMIS licensed from Synopsys is alleged to constitute the basis of Ricoh's claim of infringement. The case has been transferred to the U.S. District Court for the Northern District of California. Ricoh is seeking an injunction and damages in an unspecified amount relating to such alleged infringement. The patents relate to certain methodologies for the automated design of custom semiconductors. The case was scheduled to go to trial in March 2007; however, in December 2006, the court issued an order staying the case pending a re-examination proceeding filed by Synopsys before the U.S. Patent & Trademark Office ("PTO") challenging the validity of the patent claims at issue in this case. Since that time, Synopsys filed a total of three re-examination petitions with the PTO challenging the validity of the claims at issue which the PTO granted and consolidated all three re-examinations into one proceeding before a single examiner. The re-examination proceeding was completed in September 2008, and the PTO examiner issued a final rejection of all claims in the asserted patent over prior art. Ricoh has appealed that final rejection to the PTO Board of Appeals, which held a hearing on the appeal September 29, 2010. The Board of Appeals is expected to issue its written decision on the appeal within the next several months. In April 2008, the court lifted the stay despite the ongoing re-examination proceeding in the PTO. In September 2008, the court granted defendants' request to refile a summary judgment motion on non-infringement that had been vacated as moot when the stay was imposed in December 2006. On March 6, 2009, the judge issued a ruling denying the summary judgment motion without prejudice because of a factual dispute over a patent claim element. After an exchange of briefs by the parties related to the disputed claim element, the judge held a further hearing on the matter on June 12, 2009. On October 23, 2009, the judge issued his ruling on the disputed claim element. Based on the judge's ruling, Synopsys filed another motion for summary judgment on non-infringement on January 8, 2010. A hearing on that motion was held on March 8, 2010 and on April 14, 2010 the judge granted Synopsys' motion for summary judgment. On April 28, 2010, Ricoh filed a motion for reconsideration on the summary judgment ruling. On May 28, 2010, the judge denied the Ricoh motion and entered final judgment in Synopsys' favor for non-infringement. Ricoh subsequently filed a Notice of Appeal on June 23, 2010 and served its brief on appeal August 31, 2010. Counsel for Synopsys expects to file its responsive brief by October 29, 2010 with oral arguments scheduled to occur as early as February 2011 and a decision on the appeal likely in the spring of 2011. The Company believes that the asserted claims are without merit, and even if meritorious, that the Company will be indemnified against damages by Synopsys, and that resolution of this matter will not have a material adverse effect on the Company's consolidated financial position, results of operations or cash flows.

Note 11: Recent Accounting Pronouncements

In April 2010, FASB issued ASU No. 2010-17, "Revenue Recognition—Milestone Method," which is included in ASC 605—Milestone Method of Revenue Recognition. This ASU codifies the consensus reached in EITF 08-9, "Milestone Method of Revenue Recognition" ("EITF 08-9"), and addresses the accounting when entities enter into revenue arrangements with multiple payment streams for a single deliverable or a single unit of accounting. The pronouncement shall be applied prospectively to milestones achieved in fiscal years, and interim periods within those years, beginning after June 15, 2010, with earlier application and retrospective application permitted. The Company is currently assessing the impact of ASU No. 2010-17 on its financial position and results of operations.

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Note 12: Fair Value of Financial Instruments

The following table summarizes our financial assets and liabilities measured at fair value on a recurring basis as of October 1, 2010 and December 31, 2009 (in millions):

	ance as of ber 1, 2010	ed Prices in arkets (Level 1)		nce as of er 31, 2009	ed Prices in arkets (Level 1)
<u>Description</u>					
Assets:					
Cash and cash equivalents:					
Demand and time deposits	\$ 249.1	\$ 249.1	\$	172.2	\$ 172.2
Money market funds	46.1	46.1		42.0	42.0
US government treasury obligations	267.7	267.7		311.5	311.5
Short-term investments					
Commercial paper	_	_		20.0	20.0
Treasuries and agencies	_	_		25.5	25.5
Total cash, cash equivalents and short-term	,		'		
investments	\$ 562.9	\$ 562.9	\$	571.2	\$ 571.2
Other Current Assets					
Foreign currency exchange contracts	0.6	0.6		0.4	0.4
Total financial assets	\$ 563.5	\$ 563.5	\$	571.6	\$ 571.6
Liabilities:	 	 	-		
Foreign currency exchange contracts	\$ 0.6	\$ 0.6	\$	0.5	\$ 0.5

The Company's financial assets and liabilities are valued using market prices on active markets (Level 1). Level 1 instrument valuations are obtained from real-time quotes for transactions in active exchange markets involving identical assets. Cash and cash equivalents are short-term, highly liquid investments with original or remaining maturities of three months or less when purchased.

As of October 1, 2010 the Company held no direct investments in auction rate securities, collateralized debt obligations, structured investment vehicles or mortgage-backed securities.

Long-Term Debt, Including Current Portion

The carrying amounts and fair values of the Company's long-term borrowings (excluding capital lease obligations) at October 1, 2010 and December 31, 2009 are as follows (in millions):

	October	October 1, 2010		r 31, 2009
	Carrying Amount	Fair Value	Carrying Amount	Fair Value
Long-term debt, including current portion				
Level 1	\$ 571.2	\$695.7	\$ 562.4	\$771.7
Level 2	\$ —	\$ —	\$ 170.2	\$163.8
Level 3	\$ 152.6	\$148.2	\$ 122.3	\$114.4

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

The following is a rollforward of fair value measurement using significant unobservable inputs (Level 3) of long-term debt, including current portion from December 31, 2009 to October 1, 2010 (in millions):

Beginning balance as of December 31, 2009	\$ 114.4
Gain (loss) recorded during 2010	0.5
Gain (loss) not recorded during 2010	3.6
New debt	78.9
Principal payments	(49.2)
Ending balance as of October 1, 2010	\$ 148.2

The fair value of Level 3 financial instruments was determined by discounting the remaining payments of the outstanding debt using estimated current rates ranging from 0.96% to 4.59% at October 1, 2010.

Note 13: Segment Information

For quarter ended October 1, 2010:

Revenues from external customers

Segment gross profit

Segment operating income

Revenues, gross profit and operating income for the Company's reportable segments for the three and nine months ended October 1, 2010 and October 2, 2009, respectively, are as follows (in millions):

Computing &

Consumer Group

Automotive &

Power Group

\$

\$

\$

286.0

85.6

8.0

Digital & Mixed-

Signal Product Group

Standard

Products Group

Total

\$1,271.8

\$ 454.4

\$ 135.8

Revenues from external customers
Segment operating income \$ 26.8 \$ 26.3 \$ 27.5 \$ 57.1 \$ 137.7 For quarter ended October 2, 2009: Revenues from external customers \$ 108.3 \$ 114.4 \$ 97.0 \$ 153.2 \$ 472.9 Segment gross profit \$ 38.1 \$ 41.1 \$ 54.5 \$ 49.5 \$ 183.2 Segment operating income \$ 11.5 \$ 13.4 \$ 18.8 \$ 29.5 \$ 73.2 For nine months ended October 1, 2010: Revenues from external customers \$ 409.7 \$ 392.9 \$ 353.8 \$ 577.8 \$ 1,734.2 Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
For quarter ended October 2, 2009: Revenues from external customers \$ 108.3 \$ 114.4 \$ 97.0 \$ 153.2 \$ 472.9 \$ 5egment gross profit \$ 38.1 \$ 41.1 \$ 54.5 \$ 49.5 \$ 183.2 \$ 5egment operating income \$ 11.5 \$ 13.4 \$ 18.8 \$ 29.5 \$ 73.2 \$ 18.3 \$ 1.
Revenues from external customers \$ 108.3 \$ 114.4 \$ 97.0 \$ 153.2 \$ 472.9 Segment gross profit \$ 38.1 \$ 41.1 \$ 54.5 \$ 49.5 \$ 183.2 Segment operating income \$ 11.5 \$ 13.4 \$ 18.8 \$ 29.5 \$ 73.2 For nine months ended October 1, 2010: Revenues from external customers \$ 409.7 \$ 392.9 \$ 353.8 \$ 577.8 \$ 1,734.2 Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
Segment gross profit \$ 38.1 \$ 41.1 \$ 54.5 \$ 49.5 \$ 183.2 Segment operating income \$ 11.5 \$ 13.4 \$ 18.8 \$ 29.5 \$ 73.2 Computing & Consumer Group Digital & Mixed-Signal Product Group Standard Products Group Total For nine months ended October 1, 2010: Revenues from external customers \$ 409.7 \$ 392.9 \$ 353.8 \$ 577.8 \$1,734.2 Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
Segment operating income \$ 11.5 \$ 13.4 \$ 18.8 \$ 29.5 \$ 73.2 Automotive & Power Group Computing & Consumer Group Digital & Mixed-Signal Product Group Standard Products Group Total For nine months ended October 1, 2010: 8 409.7 \$ 392.9 \$ 353.8 \$ 577.8 \$ 1,734.2 Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
Automotive & Power Group Computing & Consumer Group Digital & Mixed-Signal Product Group Standard Products Group Total For nine months ended October 1, 2010: Total Segment gross profit \$ 409.7 \$ 392.9 \$ 353.8 \$ 577.8 \$ 1,734.2 Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
Automotive & Power Group Consumer Group Digital & Mixed-Signal Product Group Standard Products Group Total For nine months ended October 1, 2010: Total
Revenues from external customers \$ 409.7 \$ 392.9 \$ 353.8 \$ 577.8 \$1,734.2 Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
Segment gross profit \$ 152.3 \$ 172.1 \$ 220.3 \$ 213.5 \$ 758.2 Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
Segment operating income (loss) \$ 66.1 \$ 78.7 \$ 95.4 \$ 132.9 \$ 373.1
For nine months ended October 2, 2009:

\$

\$

294.7

98.3

19.4

\$

\$

\$

286.7

154.6

50.8

\$

\$

\$

404.4

115.9

57.6

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Depreciation and amortization expense is included in segment operating income. Reconciliations of segment gross profit and segment operating income to the financial statements are as follows (in millions):

		r Ended
	October 1, 2010	October 2, 2009
Gross profit for reportable segments	\$ 261.2	\$ 183.2
Unallocated amounts:		
Other unallocated manufacturing costs	(14.7)	(7.4)
Gross profit	\$ 246.5	\$ 175.8
Operating income for reportable segments	\$ 137.7	\$ 73.2
Unallocated amounts:		
Restructuring and other charges	(0.9)	(7.9)
Other unallocated manufacturing costs	(14.7)	(7.4)
Other unallocated operating expenses	(14.4)	(8.2)
Operating income	\$ 107.7	\$ 49.7
		<u> </u>
		ths Ended
	October 1, 2010	October 2, 2009
Gross profit for reportable segments	\$ 758.2	\$ 454.4
Gross profit for reportable segments Unallocated amounts:	\$ 758.2	\$ 454.4
	\$ 758.2 (39.8)	\$ 454.4
Unallocated amounts:	,	
Unallocated amounts: Other unallocated manufacturing costs	(39.8)	(28.3)
Unallocated amounts: Other unallocated manufacturing costs Gross profit	(39.8) \$ 718.4	(28.3) \$ 426.1
Unallocated amounts: Other unallocated manufacturing costs Gross profit Operating income for reportable segments Unallocated amounts: Restructuring and other charges	(39.8) \$ 718.4	(28.3) \$ 426.1
Unallocated amounts: Other unallocated manufacturing costs Gross profit Operating income for reportable segments Unallocated amounts:	(39.8) \$ 718.4 \$ 373.1	(28.3) \$ 426.1 \$ 135.8
Unallocated amounts: Other unallocated manufacturing costs Gross profit Operating income for reportable segments Unallocated amounts: Restructuring and other charges	(39.8) \$ 718.4 \$ 373.1 (7.0)	(28.3) \$ 426.1 \$ 135.8 (25.6)

ON SEMICONDUCTOR CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS—(Continued) (unaudited)

Revenues by geographic location including local sales and exports made by operations within each area, based on shipments from the respective country are summarized as follows (in millions):

	Qu	arter Ended
	October 1, 2010	October 2, 2009
United States	\$ 124.9	\$ 72.6
The Other Americas	4.9	1.0
United Kingdom	90.0	78.7
Belgium	0.2	0.4
China	207.6	159.0
Singapore	137.7	108.8
The Other Asia/Pacific	35.4	52.4
	\$ 600.7	\$ 472.9

	Nine M	iontns Ended
	October 1, 2010	October 2, 2009
United States	\$ 362.0	\$ 270.5
The Other Americas	9.4	3.1
United Kingdom	276.7	180.2
Belgium	0.3	29.1
China	579.2	405.2
Singapore	401.9	267.5
The Other Asia/Pacific	104.7	116.2
	\$ 1,734.2	\$ 1,271.8

Property, plant and equipment, net by geographic location are summarized as follows (in millions):

	October 1, 	December 31, 2009
United States	\$ 203.1	\$ 170.7
China	99.8	94.9
Europe	105.2	109.5
Malaysia	131.9	114.3
The Other Asia/Pacific	160.9	105.9
Japan	72.0	68.8
Belgium	42.7	38.7
The Other Americas	4.9	2.7
	\$ 820.5	\$ 705.5

For the quarter and nine months ended October 1, 2010, one of the Company's customers accounted for 14% and 12% of the Company's total revenues, respectively. For the quarter and nine months ended October 2, 2009, one of the Company's customers accounted for 12% and 11% of the Company's total revenues, respectively.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion in conjunction with our audited historical consolidated financial statements, which are included in our Form 10-K, filed with the SEC on February 25, 2010, and our unaudited consolidated financial statements for the fiscal quarter ended October 1, 2010, included elsewhere in this Form 10-Q. Management's Discussion and Analysis of Financial Condition and Results of Operations contains statements that are forward-looking. These statements are based on current expectations and assumptions that are subject to risks, uncertainties and other factors. Actual results could differ materially because of certain factors discussed below or elsewhere in this Form 10-Q. See Part II, Item 1A. "Risk Factors" of this Form 10-Q and our most recent Form 10-K.

Company Highlights for the Quarter Ended October 1, 2010

- · Record quarterly revenues of approximately \$600.7 million, an increase of approximately 3% from the second quarter of 2010
- GAAP gross margin of 41.0%
- GAAP net income of \$0.20 per fully diluted share
- Record low net debt position of approximately \$223 million

Executive Overview

This Executive Overview presents summary information regarding our industry, markets and operating trends only. For further information regarding the events summarized herein, you should read "Management's Discussion and Analysis of Financial Condition and Results of Operations" in its entirety.

Industry Overview

We participate in unit and revenue surveys and use data summarized by the World Semiconductor Trade Statistics ("WSTS") group to evaluate overall semiconductor market trends and also to track our progress against the total market in the areas we provide semiconductor components. The most recently published estimates of WSTS project a compound annual revenue growth rate in our total addressable market of approximately 4.6% during 2010 through 2013. These are not our projections and may not be indicative of actual results, but we, like many of our competitors, use this information as useful, third-party projections and estimates.

Business and Company Overview

We are a premier supplier of high performance, silicon solutions for energy efficient electronics. Our broad portfolio of power and signal management, logic, discrete and custom devices helps customers efficiently solve their design challenges in automotive, communications, computing, consumer, industrial, LED lighting, medical, military/aerospace and power applications. We design, manufacture and market an extensive portfolio of semiconductor components that addresses the design needs of sophisticated electronic systems and products. Our power management semiconductor components control, convert, protect and monitor the supply of power to the different elements within a wide variety of electronic devices. Our custom application specific integrated circuits use analog, digital signal processing, mixed-signal and advanced logic capabilities to act as the brain behind many of our automotive, medical, military, aerospace, consumer and industrial customers' unique products. Our data management semiconductor components provide high-performance clock management and data flow management for precision computing and communications systems. Our standard semiconductor components serve as "building block" components within virtually all electronic devices. These various products fall into the logic, analog and discrete categories used by WSTS.

We serve a broad base of end-user markets, including power supply, automotive, communications, computer, consumer, medical, industrial, mobile phone and military/aerospace. Applications for our products in

these markets include portable electronics, computers, game stations, servers, automotive and industrial automation control systems, routers, switches, storage-area networks and automated test equipment.

Our extensive portfolio of devices enables us to offer advanced integrated circuits and the "building block" components that deliver system level functionality and design solutions. Our product portfolio currently comprises approximately 42,000 products and we shipped approximately 28.6 billion units during the first nine months of 2010 as compared to approximately 20.5 billion units in the first nine months of 2009. We specialize in micro packages, which offer increased performance characteristics while reducing the critical board space inside today's ever shrinking electronic devices. We believe that our ability to offer a broad range of products provides our customers with single source purchasing on a cost-effective and timely basis.

Acquisitions

On July 14, 2010, we announced we had entered into a definitive purchase agreement with SANYO Electric Co., Ltd, a Japanese corporation ("SANYO Electric"), providing for the acquisition of SANYO Semiconductor Co., Ltd., a Japanese corporation and subsidiary of SANYO Electric ("SANYO Semiconductor"), and other assets related to SANYO Electric's semiconductor business. The purchase price is approximately \(\frac{\psi}{33.0}\) billion (\(\frac{\psi}{366.0}\) million), subject to adjustment at closing. We have been working with SANYO Electric and SANYO Semiconductor on the transaction closing conditions, and given the time table required to complete a number of these closing conditions, we believe the transaction will close in the beginning of the first quarter of 2011, rather than the fourth quarter of 2010 as originally estimated. The transaction is subject to regulatory approvals and closing conditions. We do not currently believe we will need to access the capital markets to finance the transaction, nor do we currently anticipate the need to issue any new shares of common stock to finance the transaction.

The purchase price is payable in Yen and therefore the actual US dollar value of the purchase price will be determined based on the prevailing exchange rate on the date of closing. All US dollar amounts reflected above are based on a ¥90.2 to \$1 USD exchange rate on the date of the announcement. On October 1, 2010 the exchange rate was ¥83.75 to \$1 USD and had the transaction closed on that date, the consideration payable would have been \$394.0 million.

On June 9, 2010, we completed the purchase of Sound Design Technologies, Ltd. ("SDT"), whereby SDT became our wholly-owned subsidiary. The aggregate purchase price of this all cash transaction was approximately \$22.0 million. We believe that this acquisition solidifies our position as a leading supplier of ultra-low power digital processing (DSP) technology for hearing aids and audio processing applications, strengthens our talent base and adds an experienced design and applications engineering team for medical devices within our digital and mixed-signal products group. In addition, SDT's advanced manufacturing expertise in chip-scale capacitors and high density packaging will also expand our capabilities in delivering advanced, highly miniaturized packaging technology crucial for hearing aid and similarly size-constrained applications that demand medical-grade quality.

On January 27, 2010, we completed the purchase of California Micro Devices Corporation ("CMD"), pursuant to a tender offer at a price of \$4.70 per share of CMD stock, whereby CMD became our wholly-owned subsidiary. The aggregate purchase price of this all-cash transaction was approximately \$113.2 million. We believe that the combination will significantly strengthen our offering of application specific integrated passive (ASIP) devices to protect products in the wireless, computing and consumer electronics end-markets. In addition, CMD's expertise in protection solutions for the high brightness LED (HBLED) market and its strengths in inductor capacitor-based EMI (electromagnetic interference) filtering and low capacitance ESD (electrostatic discharge) protection, complement our existing portfolio of protection and lighting solutions.

Segments

We are organized into four operating segments, which also represent four reporting segments: automotive and power group, standard products, computing and consumer products and digital and mixed-signal product group. Each of our major product lines has been assigned to a segment, as illustrated in the table below, based on our operating strategy. Because many products are sold into different end markets, the total revenue reported for a segment is not indicative of actual sales in the end market associated with that segment, but rather is the sum of the revenues from the product lines assigned to that segment. From time to time we reassess the alignment of our product families and devices to our operating segments and may move product families or individual devices from one operating segment to another.

Automotive & Power Group	Computing & Consumer Products	Digital & Mixed-Signal Products Group	Standard Products
MOSFETs	DC-DC Conversion	Medical	Bipolar Power
Analog Automotive	Analog Switches	Integrated Sensor Products ("ISP")	Thyristor
Auto Power	AC-DC Conversion	Military & Aerospace	Small Signal
LDO & Vregs	Low Voltage	Industrial	Zener
Mixed-Signal Automotive	Standard Logic	Communications & High Voltage	Protection
Automotive Application Specific Standard Products ("ASSP")	Power Switching	High Frequency	Rectifier
	Signal & Interface	Foundry and Manufacturing Services	Filters
			Memory Products

We have approximately 403 direct customers worldwide, and we also service approximately 274 significant original equipment manufacturers indirectly through our distributor and electronic manufacturing service provider customers. Our direct and indirect customers include: (1) leading original equipment manufacturers in a broad variety of industries, such as Continental Automotive Systems, Delta, Samsung, Hella, LG Electronics, Motorola, Boston Scientific, Delphi, Huawei Technology, Seagate Technology, Sony Ericsson, Bosch, General Electric, Raytheon, Alcatel, Siemens, and Visteon; (2) electronic manufacturing service providers, such as Flextronics, Jabil and Sammina; and (3) global distributors, such as Avnet, World Peace, Arrow, Wintech, Yosun, EBV Elektronik, and Future.

We currently have major design operations in Arizona, Rhode Island, Idaho, California, Texas, Oregon, China, Romania, Switzerland, the Czech Republic, Korea, Belgium, Canada, Germany, India, Ireland, and France, and we currently operate manufacturing facilities in Arizona, Oregon, Idaho, Belgium, China, the Czech Republic, Japan, Malaysia, the Philippines, and Thailand.

New Product Innovation

As a result of the success of our research and development initiatives, excluding the introduction of lead-free products, we introduced 247 new product families in 2009. During the nine months ended October 1, 2010, we introduced an additional 187 new product families. Our new product development efforts continue to be focused on building solutions in power management that appeal to customers in focused market segments and across multiple high growth applications. As always, it is our practice to regularly re-evaluate our research and development spending, to assess the deployment of resources and to review the funding of high growth

technologies regularly. We deploy people and capital with the goal of maximizing our investment in research and development in order to position us for continued growth. As a result, we often invest at the greatest opportunity to refresh existing products in our commodity logic, analog, and discrete products. We invest in these initiatives when we believe there is a strong customer demand or opportunities to innovate our current portfolio in high growth markets and applications.

Cost Savings and Restructuring Activities

We continue to implement profitability enhancement programs to improve our cost structure and, as a result, we expect to rank, as compared to our primary competitors, among the lowest in terms of cost structure.

Our profitability enhancement programs will continue to focus on:

- transferring production to lower cost regions;
- increasing die manufacturing capacity in a cost-effective manner by moving production from 6" wafers to 8" wafers and smaller geometrics resulting in an increase to the number of die per wafer;
- · reducing the number of product platforms and process flows; and
- focusing production on profitable product families.

Macroeconomic Environment

While we have recognized efficiencies from implemented restructuring activities and programs and continue to implement profitability enhancement programs to improve our cost structure, the United States and global economies continue to undergo a period of unprecedented volatility. In addition, the semiconductor industry has traditionally been highly cyclical and has often experienced significant downturns in connection with, or in anticipation of, declines in general economic conditions. These macroeconomic factors have affected our customers and suppliers, which in turn has affected our business, including sales, the collection of receivables, and results of operations. Although we view many of these macroeconomic environment issues as temporary, our continuing outlook for the future will ultimately affect our future emphasis on marketing to various industries, our future research and development efforts into new product lines and our segments in general.

Outlook

Based on current booking trends, backlog levels and estimated turns levels, we anticipate that total revenues will be approximately \$565.0 million to \$585.0 million in the fourth quarter of 2010. Backlog levels at the beginning of the fourth quarter of 2010 were down slightly from backlog levels at the beginning of the third quarter of 2010 and represent over 90% of our anticipated fourth quarter 2010 revenues. We expect average selling prices for the fourth quarter of 2010 will be approximately flat compared to the third quarter of 2010. We expect cash capital expenditures of approximately \$54.0 million in the fourth quarter of 2010.

For the fourth quarter of 2010, we expect gross profit as a percentage of revenues to be approximately 40.0% to 41.0%. For the fourth quarter of 2010, we also expect total operating expenses of approximately \$137.0 million to \$141.0 million, which includes amortization of acquisition-related intangible assets, restructuring and other charges of \$10.0 million.

We anticipate interest expense, net of interest income, and other expenses will be approximately \$18.0 million, which includes non-cash interest expense of approximately \$9.0 million relating to our convertible senior subordinated notes. We expect the provision for income taxes to be approximately \$4.0 million in the fourth quarter of 2010, with cash payments of income taxes to be approximately \$3.0 million in the fourth quarter of 2010. We also expect share-based compensation expense of approximately \$12.0 million to \$13.0 million in the fourth quarter of 2010.

Results of Operations

Quarter Ended October 1, 2010 Compared to Quarter Ended October 2, 2009

The following table summarizes certain information relating to our operating results that has been derived from our unaudited consolidated financial statements for the quarters ended October 1, 2010 and October 2, 2009. The amounts in the following table are in millions:

	Quarte		
	October 1, 2010	October 2, 2009	Dollar Change
Revenues	\$ 600.7	\$ 472.9	\$ 127.8
Cost of revenues	354.2	297.1	57.1
Gross profit	246.5	175.8	70.7
Operating expenses:			
Research and development	63.3	53.8	9.5
Selling and marketing	35.5	30.0	5.5
General and administrative	31.2	27.1	4.1
Amortization of acquisition-related intangible assets	7.9	7.3	0.6
Restructuring, asset impairments and other net	0.9	7.9	$\underline{\hspace{1cm}(7.0)}$
Total operating expenses	138.8	126.1	12.7
Operating income	107.7	49.7	58.0
Other income (expenses):	·		
Interest expense	(14.1)	(15.8)	1.7
Interest income	0.2	0.1	0.1
Other	(0.8)	(1.5)	0.7
Other income (expenses), net	(14.7)	(17.2)	2.5
Income (loss) before income taxes and minority interests	93.0	32.5	60.5
Income tax provision	(4.6)	(1.9)	(2.7)
Net income (loss)	88.4	30.6	57.8
Net (income) loss attributable to minority interests	(0.6)	(0.7)	0.1
Net income (loss) attributable to ON Semiconductor Corporation	\$ 87.8	\$ 29.9	\$ 57.9

Revenues

Revenues were \$600.7 million and \$472.9 million during the quarters ended October 1, 2010 and October 2, 2009, respectively. The increase in the third quarter of 2010 as compared to the third quarter of 2009 was primarily due to an increase in volume and mix of 26% combined with increased revenues from our acquisition of PulseCore, CMD and SDT of \$20.3 million, or 4%, partially offset by decreases in average selling prices of approximately 3%. The revenues by reportable segment were as follows (dollars in millions):

	Quarter Ended October 1, 2010	As a % Revenue ⁽¹⁾	Quarter Ended October 2, 2009	As a % Revenue	Dollar Change	% Change
Automotive and Power Group	\$ 144.6	24%	\$ 108.3	23%	\$ 36.3	34%
Computing and Consumer Group	134.6	22%	114.4	24%	20.2	18%
Digital and Mixed-Signal Product Group	115.7	19%	97.0	21%	18.7	19%
Standard Products Group	205.8	34%	153.2	32%	52.6	34%
Total revenues	\$ 600.7		\$ 472.9		\$127.8	

⁽¹⁾ Certain amounts may not total due to rounding of individual amounts

Revenues from automotive power group increased \$36.3 million, or 34%, in the third quarter of 2010 as compared to the third quarter of 2009. The increase is attributed to increases in revenues from MOSFETs of 27%, mixed signal automotive products of 42%, analog automotive products of 44%, LDO and voltage regulator products of 25%, auto power products of 39% and automotive application specific standard products of 55%.

Revenues from computing and consumer products increased \$20.2 million, or 18%, in the third quarter of 2010 as compared to the third quarter of 2009. The increase is attributed to increases in revenues signal and interface products of 32%, standard logic products of 43%, power switch products of 17%, AC to DC conversion products of 21% and DC to DC conversion products of 45%, partially offset by decreases in revenues from analog switch products of 48% and low voltage power products of 1%.

Revenues from digital and mixed-signal product group increased \$18.7 million, or 19%, in the third quarter of 2010 as compared to the third quarter of 2009. The increase is attributed to increases in revenues from industrial products of 49%, high frequency products of 36%, revenues from recent acquisition of SDT, military and aerospace products of 16%, communication and high voltage products of 88%, revenues from acquisition of PulseCore, integrated sensor products of 36% and manufacturing service revenues of 5%, partially offset by decreases in revenues from foundry products.

Revenues from standard products increased \$52.6 million, or 34%, in the second quarter of 2010 as compared to the second quarter of 2009. The increase is attributed to increases in revenues from protection products of 46%, rectifier products of 29%, filter products of 74%, revenues from our acquisition of Catalyst of 49%, small signal products of 11%, zener products of 44%, bipolar power products of 34% and thyristor products of 23%.

Revenues by geographic area as a percentage of total revenues were as follows (dollars in millions):

	Oct	ter Ended tober 1, 2010	As a % Revenue	rter Ended ctober 2, 2009	As a % Revenue ⁽¹⁾
Americas	\$	129.8	22%	\$ 73.6	16%
Asia/Pacific		380.7	63%	320.2	68%
Europe		90.2	15%	79.1	17%
Total	\$	600.7	100%	\$ 472.9	100%

⁽¹⁾ Certain amounts may not total due to rounding of individual amounts.

A majority of our end customers, served directly or through distribution channels, are manufacturers of electronic devices. For the quarter ended October 1, 2010, we had one customer that accounted for approximately 14% of our total revenues. For the quarter ended October 2, 2009, we had one customer that accounted for 12% our total revenues.

Gross Profit

Our gross profit was \$246.5 million in the third quarter of 2010 compared to \$175.8 million in the third quarter of 2009. As a percentage of revenues, our gross profit was 41.0% in the third quarter of 2010 as compared to 37.2% in the third quarter of 2009. Gross profit as a percentage of revenues increased during the third quarter of 2010 as compared to the third quarter of 2009 primarily due to increases in volume driving better factory utilization, cost savings from profitability enhancement programs, combined with lower expensing of fair market value step up of inventory in the third quarter of 2010 as compared to the third quarter of 2009, partially offset by a 3% decrease in average selling prices in the third quarter of 2010. The gross profit by reportable segment in each of these quarters are as follows (dollars in millions):

	 ter Ended per 1, 2010	As a % Net <u>Revenue</u>	•	rter Ended ber 2, 2009	As a % Net Revenue	Dollar <u>Change</u>	% <u>Change</u>
Automotive & Power Group	\$ 55.6	9.3%	\$	38.1	8.1%	\$ 17.5	46%
Computing and Consumer Products	57.6	9.6%		41.1	8.7%	16.5	40%
Digital & Mixed-Signal Product Group	68.1	11.3%		54.5	11.5%	13.6	25%
Standard Products	79.9	13.3%		49.5	10.5%	30.4	61%
Gross profit by segment	261.2			183.2		\$ 78.0	
Unallocated Manufacturing	(14.7)	-2.4%		(7.4)	-1.6%		
Total gross profit	\$ 246.5	41.0%	\$	175.8	37.2%		

Gross profit from automotive and power group increased \$17.5 million, or 46%, in the third quarter of 2010 as compared to the third quarter of 2009. The increase is attributable to increases in gross profit from mixed signal automotive products of 48%, LDO and voltage regulator products of 78%, analog automotive products of 86%, MOSFETs of 21%, auto power products of 39% and automotive application specific standard products of 76%.

Gross profit from computing and consumer products increased \$16.5 million, or 40%, in the third quarter of 2010 as compared to the third quarter of 2009. The increase is attributable to increases in gross profit from signal and interface products of 75%, power switch products of 77%, standard logic products of 73%, AC to DC conversion products of 18%, DC to DC conversion products of 55% and low voltage power products of 12%, partially offset by decreases in gross margin from analog switch products of 58%.

Gross profit from digital and mixed-signal products increased \$13.6 million, or 25%, in the third quarter of 2010 as compared to the third quarter of 2009. The increase is attributable to increases in gross profit from industrial products of 59%, manufacturing service products of greater than 100%, high frequency products of 53%, military and aerospace products of 12%, communications and high voltage products of 87%, gross margin from acquisition of PulseCore and integrated sensor products of 25%, partially offset by decreases in gross margin from foundry products of 24% and medical products of 17%.

Gross profit from standard products increased \$30.4 million, or 61%, in the second quarter of 2010 as compared to the second quarter of 2009. The increase is attributable to increases in gross profit from protection products of 45%, rectifier products of 54%, small signal products of 33%, zener products of 62%, filters of 65%, gross margin from acquisition of Catalyst of 72% and bipolar power products of 49%.

Operating Expenses

Research and development expenses were \$63.3 million in the third quarter of 2010 compared to \$53.8 million in the third quarter of 2009, representing an increase of \$9.5 million, or 17.7%. Research and development expenses represented 10.5% and 11.4% of revenues in the third quarter of 2010 and the third quarter of 2009, respectively. The dollar increase in research and development expense was primarily attributed

to increased expense associated with on-going research and development activities as a result of the newly acquired PulseCore, CMD and SDT businesses, combined with increased employee salaries and wages due to the elimination in 2010 of work furloughs or short work weeks based upon local legal requirements.

Selling and marketing expenses were \$35.5 million in the third quarter of 2010 compared to \$30.0 million in the third quarter of 2009, representing an increase of \$5.5 million, or 18.3%. Selling and marketing expenses represented 5.9% and 6.3% of revenues in the third quarter of 2010 and the third quarter of 2009, respectively. The dollar increase in selling and marketing expense was primarily attributed to increased employee salaries and wages due to the elimination in 2010 of work furloughs or short work weeks based upon local legal requirements.

General and administrative expenses were \$31.2 million in the third quarter of 2010 compared to \$27.1 million in the third quarter of 2009, representing an increase of \$4.1 million, or 15.1%. General and administrative expenses represented 5.2% and 5.7% of revenues in the third quarter of 2010 and the third quarter of 2009, respectively. The increase in general and administrative expenses was primarily attributed to increased acquisition related expenses, employee salaries and wages due to the elimination in 2010 of work furloughs or short work weeks based upon local legal requirements, partially offset by decreased stock compensation expense.

Other Operating Expenses—Restructuring, Asset Impairments and Other, Net

Restructuring, asset impairment and other, net were \$0.9 million in the third quarter of 2010 compared to \$7.9 million in the third quarter of 2009.

In June 2010, we acquired SDT and announced plans to integrate the overlapping operations of SDT and the Company, in part for cost savings purposes. As part of these plans, certain duplicative positions were eliminated. During the quarter ended October 1, 2010 we recorded \$0.2 million of employee separation charges. All termination benefits were for 36 individuals, of which all had been terminated by the end of the third quarter of 2010.

In January 2010, we acquired CMD and announced plans to integrate the overlapping operations of CMD and the Company, in part for cost savings purposes. As part of these plans, certain duplicative positions were eliminated. During the quarter ended October 1, 2010, we recorded \$0.7 million of employee separation charges. These termination benefits were for 11 individuals, of which six had been terminated as of October 1, 2010. It is anticipated that all terminations will be completed by the end of fiscal 2010.

In May 2008, we announced our intention to cease production at our Piestany, Slovakia manufacturing facilities resulting in an impairment charge of \$7.9 million, which was recorded during the second quarter of 2008 The closure of the Piestany, Slovakia manufacturing facilities was completed during the quarter ended October 2, 2009. An additional impairment charge of \$5.4 million was recorded during the quarter ended October 2, 2009 that resulted from the subsequent decline in the estimated fair value of the Piestany property. Additionally, during the quarter ended October 2, 2009, we recorded a (\$0.2) million adjustment of termination benefits for approximately 430 employees which is being recognized ratably from the date of announcement to the employees' actual termination dates. During the quarter ended October 2, 2009, we recorded exit costs of \$0.7 million related to the planned shutdown of the Piestany, Slovakia manufacturing facility. The exit costs of \$0.7 million were for charges incurred relative to the cancellation of certain contracts and other relevant costs.

In August 2009, we announced plans to consolidate into fewer product development centers for cost savings purposes by closing our Hong Kong and Bulgaria design centers. During the quarter ended October 2, 2009, we recorded employee separation charges of \$1.0 million dollars and \$0.3 million dollars in exit costs related to this activity. These termination benefits were for approximately 47 individuals.

In January 2009, we announced a worldwide employee reduction program for cost savings purposes. As part of these plans, certain employee positions were eliminated. During the quarter ended October 2, 2009 we

recorded an additional employee separation charge of \$0.4 million. Those termination benefits were for approximately 14 individuals of which 10 had been terminated as of October 2, 2009.

In March 2008, we acquired AMIS Holdings, Inc. ("AMIS") and announced plans to integrate the operations of the two companies in part, for cost savings purposes. As part of these plans, certain duplicative positions were eliminated and certain overlapping or duplicative contracts with external suppliers were renegotiated with terms applicable to the combined company. During the quarter ended October 2, 2009, we recorded exit costs of \$0.3 million related to the 2008 acquisition of AMIS. The exit costs of \$0.3 million were for charges incurred relative to transfer of certain assets to other locations.

Other Operating Expenses—Amortization of Acquisition—Related Intangible Assets

Amortization of acquisition-related intangible assets was \$7.9 million and \$7.3 million for the quarters ended October 1, 2010 and October 2, 2009, respectively. The increase of \$0.6 million from the third quarter of 2009 to 2010 was primarily attributed to amortization of intangible assets associated with our acquisitions of PulseCore, CMD and SDT.

Operating Income

Information about operating income from our reportable segments for the quarters ended October 1, 2010 and October 2, 2009 is as follows, in millions:

			Com	puting &	Dig	gital &	Sta	ndard	
	Autor	notive &	Coi	nsumer	Mixe	d-Signal	Pr	oducts	
	Powe	r Group	G	Froup	Produ	ct Group	G	roup	Total
For the quarter ended October 1, 2010: Segment operating income	\$	26.8	\$	26.3	\$	27.5	\$	57.1	\$ 137.7
For the quarter ended October 2, 2009: Segment operating income	\$	11.5	\$	13.4	\$	18.8	\$	29.5	\$ 73.2

Depreciation and amortization expense is included in segment operating income. Reconciliations of segment information to financial statements follow, in millions:

	Quarter	Ended
	October 1, 2010	October 2, 2009
Operating income for reportable segments	\$ 137.7	\$ 73.2
Unallocated amounts:		
Restructuring asset impairments and other charges, net	(0.9)	(7.9)
Other unallocated manufacturing costs	(14.7)	(7.4)
Other unallocated operating expenses	(14.4)	(8.2)
Operating income	\$ 107.7	\$ 49.7

Interest Expense and Other

Interest expense decreased \$1.7 million to \$14.1 million in the third quarter of 2010 compared to \$15.8 million in the third quarter of 2009. We recorded amortization of debt discount to interest expense of \$8.3 million and \$8.3 million for the quarters ended October 1, 2010 and October 2, 2009, respectively. Our average long-term debt balance (including current maturities and net of debt discount) in the third quarter of 2010 was \$769.7 million with a weighted average interest rate of 7.3% compared to \$899.9 million and a weighted average interest rate of 7% in the third quarter of 2009. See "Liquidity and Capital Resources—Key Financing Events" below for a description of our refinancing activities.

See Note 10: "Commitments and Contingencies" of the notes to our unaudited consolidated financial statements included elsewhere in this Form 10-Q for contingencies relating to other legal proceedings and matters including intellectual property matters.

Provision for Income Taxes

Provision for income taxes was \$4.6 million in the third quarter of 2010 compared to a \$1.9 million tax provision in the third quarter of 2009. The provision for the third quarter of 2010 included \$4.0 million for income and withholding taxes of certain of our foreign and U.S. operations and \$0.6 million of new reserves and interest on existing reserves for potential liabilities in U.S. and foreign taxing jurisdictions. Provision for income taxes was \$1.9 million in the third quarter of 2009. The provision for the third quarter of 2009 included \$1.6 million for income and withholding taxes of certain of our foreign operations and \$0.3 million of new reserves and interest on existing reserves for potential liabilities in foreign taxing jurisdictions.

Due to our domestic tax losses and tax rate differentials in our foreign subsidiaries, our effective tax rate is lower than the U.S. statutory federal income tax rate. We continue to maintain a full valuation allowance on all of our domestic deferred tax assets.

Nine Months Ended October 1, 2010 Compared to Nine Months Ended October 2, 2009

The following table summarizes certain information relating to our operating results that has been derived from our unaudited consolidated financial statements for the nine months ended October 1, 2010 and October 2, 2009. The amounts in the following table are in millions:

	Nine Months Ended		
	October 1, 2010	October 2, 2009	Dollar Change
Revenues	\$1,734.2	\$1,271.8	\$ 462.4
Cost of revenues	1,015.8	845.7	170.1
Gross profit	718.4	426.1	292.3
Operating expenses:			
Research and development	188.6	148.1	40.5
Selling and marketing	107.6	87.4	20.2
General and administrative	98.0	84.4	13.6
In-process research and development	_	_	_
Amortization of acquisition-related intangible assets	23.8	21.8	2.0
Restructuring, asset impairments and other net	7.0	25.6	(18.6)
Total operating expenses	425.0	367.3	57.7
Operating income	293.4	58.8	234.6
Other income (expenses):			
Interest expense	(45.0)	(49.2)	4.2
Interest income	0.4	0.7	(0.3)
Other	(7.0)	(4.2)	(2.8)
Loss on debt repurchase	(0.7)	(3.1)	2.4
Other income (expenses), net	(52.3)	(55.8)	3.5
Income (loss) before income taxes and minority interests	241.1	3.0	238.1
Income tax (provision) benefit	(9.4)	(8.1)	(1.3)
Net income (loss)	231.7	(5.1)	236.8
Net (income) loss attributable to minority interests	(2.2)	(1.9)	(0.3)
Net income (loss) attributable to ON Semiconductor Corporation	\$ 229.5	\$ (7.0)	\$ 236.5

Revenues

Revenues were \$1,734.2 million and \$1,271.8 million during the nine months ended October 1, 2010 and October 2, 2009, respectively. The increase from the first nine months of 2010 compared to the first nine months of 2009 was primarily due to increase in volume and mix of sales of 39% combined with an increase in revenues from our acquisitions of SDT, PulseCore and CMD of \$25.8 million, or 2%, partially offset by decreases in average selling prices of 5%. The revenues by reportable segment were as follows (dollars in millions):

	 Ionths Ended ber 1, 2010	As a % <u>Revenue</u>					% <u>Change</u>
Automotive and Power Group	\$ 409.7	24%	\$	286.0	22%	\$123.7	43%
Computing and Consumer Group	392.9	23%		294.7	23%	98.2	33%
Digital and Mixed-Signal Product Group	353.8	20%		286.7	23%	67.1	23%
Standard Products Group	 577.8	33%		404.4	32%	173.4	43%
Total revenues	\$ 1,734.2		\$	1,271.8		\$462.4	

Revenues from automotive and power group increased \$123.7 million, or 43%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase is attributed to increases in MOSFETs of 34%, mixed signal automotive products of 53%, analog automotive products of 66%, LDO and voltage regulator products of 36%, auto power products of 40% and automotive application specific standard products of 64%.

Revenues from computing and consumer products group increased \$98.2 million, or 33%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase is attributable to increases in revenues from signal and interface products of 45%, AC to DC conversion products of 39%, power switch products of 34%, standard logic products of 61%, DC to DC conversion products of 41% and low voltage power products of 4%, partially offset by decreases in revenues from analog switch products of 14%.

Revenues from digital and mixed-signal product group increased \$67.1 million, or 23%, in the first nine months of 2010 as compared to the first six months of 2009. The increase in revenues is attributable to increases in revenues from industrial products of 57%, military and aerospace products of 16%, high frequency products of 27%, integrated sensor products of 90%, revenues from acquisition of SDT, revenues from acquisition of PulseCore, medical products of 3% and communication and high voltage products of 16%, partially offset by decreases in revenues from foundry products of 2% and manufacturing service revenues of less than 1%.

Revenues from standard products group increased \$173.4 million, or 43%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase is attributed to rectifier products of 45%, protection products of 45%, small signal of 22%, filters products of 64%, zener products of 54%, revenues from acquisition of Catalyst of 31%, bipolar power products of 38% and thyristor products of 49%.

Revenues by geographic area as a percentage of total revenues were as follows (dollars in millions):

	Nine Months Ended October 1, 2010		As a % Revenue	Nine Months Ended October 2, 2009		As a % <u>Revenue</u>
Americas	\$	371.4	21%	\$	273.6	22%
Asia/Pacific		1,085.7	63%		788.9	62%
Europe		277.1	16%		209.3	16%
Total	\$	1,734.2	100%	\$	1,271.8	100%

A majority of our end customers, served directly or through distribution channels, are manufacturers of electronic devices. For the nine months ended October 1, 2010, we had one customer that accounted for approximately 12% of our total revenues. For the nine months ended October 2, 2009, we had one customer that accounted for 11% of our total revenues.

Gross Profit

Our gross profit was \$718.4 million in the first nine months of 2010 compared to \$426.1 million in the first nine months of 2009. As a percentage of revenues, our gross profit was 41.4% in the first nine months of 2010 as compared to 33.5% in the first nine months of 2009. Gross profit as a percentage of revenues increased during the first nine months of 2010 as compared to the first nine months of 2009 primarily due to increases in volume driving better factory utilization, cost savings from profitability enhancement programs, combined with a lower amount of expensing of fair market value step up of inventory in the first nine months of 2010 as compared to the first nine months of 2009, offset by the 5% decrease in average selling prices in the first nine months of 2010. The gross profit by reportable segment in each of these quarters are as follows (dollars in millions):

	 nths Ended r 1, 2010	As a % of Total Company Net Revenue	Months Ended ober 2, 2009	As a % of Total Company Net Revenue	Dollar Change	% Change
Automotive & Power Group	\$ 152.3	8.8%	\$ 85.6	6.7%	\$ 66.7	78%
Computing and Consumer Products	172.1	9.9%	98.3	7.7%	73.8	75%
Digital & Mixed-Signal Product Group	220.3	12.7%	154.6	12.2%	65.7	42%
Standard Products	213.5	12.3%	115.9	9.1%	97.6	84%
Gross profit by segment	758.2		454.4		\$303.8	
Unallocated Manufacturing	(39.8)	-2.3%	(28.3)	-2.2%		
Total gross profit	\$ 718.4	41.4%	\$ 426.1	33.5%		

Gross profit from automotive and power group increased \$66.7 million, or 78%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase is attributable to increases in gross profit from mixed signal automotive products of 97%, MOSFETs of 59%, LDO and voltage regulator products of greater than 100%, analog automotive products of 73%, auto power products of 53% and automotive application specific standard products of 89%.

Gross profit from computing and consumer products increased \$73.8 million, or 75%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase can be attributable to increases in gross profit from signal and interface products of 85%, power switch products of greater than 100%, standard logic products of greater than 100%, AC to DC conversion products of 61%, DC to DC conversion products of 76% and low voltage power products of 10%, partially offset by decreases in gross profit from analog switch products of 3%.

Gross profit from digital and mixed-signal product group increased \$65.7 million, or 42%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase is attributable to increases in gross profit from industrial products of 97%, manufacturing service products of greater than 100%, high frequency products of 37%, military and aerospace products of 17%, gross profit from acquisition of PulseCore, integrated sensor products of 80% and communication and high voltage products of 12%, partially offset by decreases in gross profit from foundry products of 12% and medical products of 1%.

Gross profit from standard products increased \$97.6 million, or 84%, in the first nine months of 2010 as compared to the first nine months of 2009. The increase is attributable to increases in gross profit from rectifier products of greater than 100%, protection products of 65%, small signal products of 63%, zener products of greater than 100%, gross profit from acquisition of Catalyst of 49%, filter products of 43%, bipolar power products of 87% and thyristor products of 68%.

Operating Expenses

Research and development expenses were \$188.6 million in the first nine months of 2010 compared to \$148.1 million in the first nine months of 2009, representing an increase of \$40.5 million, or 27.3%. Research and development expenses represented 10.9% and 11.6% of revenues in the first nine months of 2010 and the first nine months of 2009, respectively. The increase in research and development expense was primarily attributable to increased expense associated with ongoing research and development activities as a result of the newly acquired SDT, PulseCore and CMD businesses, increased salaries and wages due to the elimination of work furloughs or short work weeks in 2010 based upon local requirements, and an increase in performance bonuses as a result of our achievement of certain financial goals.

Selling and marketing expenses were \$107.6 million in the first nine months of 2010 compared to \$87.4 million in the first nine months of 2009, representing an increase of \$20.2 million, or 23.1%. Selling and marketing expenses represented 6.2% and 6.9% of revenues in the first nine months of 2010 and the first nine months of 2009, respectively. The increase in selling and marketing expense was primarily attributed to increased salaries and wages due to the elimination of work furloughs or short work weeks in 2010 based upon local requirements and an increase in performance bonuses as a result of our achievement of certain financial goals.

General and administrative expenses were \$98.0 million in the first nine months of 2010 compared to \$84.4 million in the first nine months of 2009, representing an increase of \$13.6 million, or 16.1%. General and administrative expenses represented 5.7% and 6.6% of revenues in the first nine months of 2010 and in the first nine months of 2009, respectively. The increase in general and administrative expenses was primarily attributed to increased acquisition related expenses, employee salaries and wages due to the elimination of work furloughs or short work weeks in 2010 based upon local legal requirements, an increase in performance bonuses as a result of our achievement of certain financial goals, partially offset by decreased stock compensation expense.

Other Operating Expenses—Restructuring, Asset Impairments and Other, Net

Restructuring, asset impairment and other, net were \$7.0 million in the first nine months of 2010 compared to \$25.6 million in the first nine months of 2009.

In June 2010, we acquired SDT and announced plans to integrate the overlapping operations of SDT and the Company, in part for cost savings purposes. As part of these plans, certain duplicative positions were eliminated. During the nine months ended October 1, 2010 we recorded \$2.2 million of employee separation charges. Termination benefits were for approximately 36 individuals, of which all have been terminated as of October 1, 2010.

During the first quarter of 2010, we agreed to make a \$0.8 million cash payment settlement of various litigation matters with the former minority interest shareholders of a Czech subsidiary acquired by the Company. These settlement charges have been included in restructuring, asset impairment and other, net on the consolidated statement of operations for the nine months ended October 1, 2010.

In January 2010, we acquired CMD and announced plans to integrate the overlapping operations of CMD and the Company, in part for cost savings purposes. As part of these plans, certain duplicative positions were eliminated. During the nine months ended October 1, 2010 we recorded \$3.5 million of employee separation charges. These termination benefits were for 25 individuals, of which 19 were terminated as of October 1, 2010. It is anticipated that all terminations will be completed by the end of fiscal 2010. Additionally, during the nine months ended October 1, 2010 we recorded \$0.1 million of exit costs associated with the exit of certain leases.

In January 2009, we announced a worldwide employee reduction program for cost savings purposes. As part of these plans, certain employee positions were eliminated. During the nine months ended October 1, 2010, we recorded an additional employee separation charge of \$0.1 million for one employee who was terminated.

In August 2009, we announced plans to reduce the number of design centers for cost savings purposes. During the nine months ended October 1, 2010 we recorded employee separation charges of \$0.2 million and \$0.1 million of exit costs associated with this activity. All termination benefits were for approximately 16 individuals, of which all had been terminated by the end of the second quarter of 2010.

In May 2008, we announced our intention to cease production at our Piestany, Slovakia manufacturing facilities. The closure of the Piestany, Slovakia manufacturing facilities was completed during the quarter ended October 2, 2009. An additional impairment charge of \$5.4 million was recorded during the quarter ended October 2, 2009 that resulted from the subsequent decline in the estimated fair value of the Piestany property. Additionally, during the nine months ended October 2, 2009, we recorded \$1.7 million, net of adjustments of termination benefits for approximately 430 employees which were being recognized ratably from the date of announcement to the employees' termination dates. Additionally, we recorded \$1.3 million of exit costs related to the shutdown of the manufacturing facility.

In August 2009, we announced plans to reduce the number of design centers for cost savings purposes. During the nine months ended October 2, 2009, we recorded employee separation charges of \$1.0 million dollars and \$0.3 million dollars in exit costs related to this activity. These termination benefits were for approximately 47 individuals, of which one had been terminated at the end of the third quarter of 2009.

In June 2009, we recorded \$0.2 million of asset impairments. Prior to June 2009, we had capitalized certain assets used in one of our wafer fabrication facilities. The \$0.2 million of asset impairments resulted from the fact that we currently have no plans to use the assets, and management does not consider this a temporary cessation of use.

In June 2009, we announced a Belgium employee reduction program, for cost savings purposes. As part of these plans, certain employee positions were eliminated. During the nine months ended October 2, 2009, we recorded employee separation charges \$3.9 million. These termination benefits were for approximately 21 individuals of which 16 had been terminated as of October 2, 2009.

In January 2009, we announced a worldwide employee reduction program, for cost savings purposes. As part of these plans, certain employee positions were eliminated. During the nine months ended October 2, 2009, we recorded employee separation charges of \$12.3 million. These termination benefits are for approximately 570 individuals of which 566 have been terminated. During the quarter ended October 2, 2009, we recorded exit costs of \$0.4 million related to the elimination of certain contracts.

In March 2008, we acquired AMIS and announced plans to integrate the operations of the two companies in part, for cost savings purposes. As part of these plans, certain duplicative positions were eliminated and certain overlapping or duplicative contracts with external suppliers were renegotiated with terms applicable to the combined company. During the nine months ended October 2, 2009, we recorded exit costs of \$1.6 million related to the 2008 acquisition of AMIS. The exit costs of \$1.6 million were for charges incurred relative to shutdown of the manufacturing facility.

In March 2009, we recorded a net gain of \$2.5 million associated with the settlement of two legal disputes.

Other Operating Expenses—Amortization of Acquisition—Related Intangible Assets

Amortization of acquisition-related intangible assets were \$23.8 million and \$21.8 million for the nine months ended October 1, 2010 and October 2, 2009, respectively. The increase of \$2.0 million from the first nine month of 2009 to 2010 was primarily attributed to amortization of intangible assets associated with our acquisition of PulseCore, CMD and SDT.

Operating Income

Information about operating income from our reportable segments for the nine months ended October 1, 2010 and October 2, 2009, respectively, is as follows, in millions:

	motive & er Group	Coi	puting & nsumer Group	Mixe	ital & d-Signal ct Group	Standard Products Group	Total
For the nine months ended October 1, 2010: Segment operating							
income	\$ 66.1	\$	78.7	\$	95.4	\$ 132.9	\$ 373.1
For the nine months ended October 2, 2009: Segment operating							
income	\$ 8.0	\$	19.4	\$	50.8	\$ 57.6	\$ 135.8

Depreciation and amortization expense is included in segment operating income. Reconciliations of segment information to financial statements follow, in millions:

		Nine Months Ended		
	Octol	oer 1, 2010	Octo	ber 2, 2009
Operating income for reportable segments	\$	373.1	\$	135.8
Unallocated amounts:				
Restructuring asset impairments and other charges, net		(7.0)		(25.6)
Other unallocated manufacturing costs		(39.8)		(28.3)
Other unallocated operating expenses		(32.9)		(23.1)
Operating income	\$	293.4	\$	58.8
Restructuring asset impairments and other charges, net Other unallocated manufacturing costs Other unallocated operating expenses	\$	(39.8) (32.9)	\$	(28. (23.

Interest Expense and Other

Interest expense decreased \$4.2 million to \$45.0 million in the first nine months of 2010 compared to \$49.2 million in the first nine months of 2009. We recorded amortization of debt discount to interest expense of \$25.3 million and \$26.6 million for the nine months ended October 1, 2010 and October 2, 2009, respectively. Our average long-term debt balance (including current maturities and net of debt discount) in the first nine months of 2010 was \$859.7 million with a weighted average interest rate of 7.0% compared to \$952.7 million and a weighted average interest rate of 6.9% in the first nine months of 2009. See "Liquidity and Capital Resources—Key Financing Events" below for a description of our refinancing activities.

See Note 10: "Commitments and Contingencies" of the notes to our unaudited consolidated financial statements included elsewhere in this Form 10-Q for contingencies relating to other legal proceedings and matters including intellectual property matters.

Loss on Debt Repurchase

During the first nine months of 2010, we prepaid approximately \$169.8 million of our senior bank facilities and incurred a \$0.8 million loss on the prepayment, for the write off of \$0.8 million in unamortized debt issuance costs. Additionally, during the first nine months of 2010, we recognized a \$0.1 million gain on the modification of our Zero Coupon Convertible Senior Subordinated Notes Due 2024, for a net loss on debt repurchase of \$0.7 million during the first nine months of 2010.

During the first nine months of 2009, we repurchased approximately \$99.7 million of our Zero Coupon Convertible Senior Subordinated Notes due 2024 for \$64.8 million in cash and the issuance of 7.4 million shares of common stock, which had a value of \$28.5 million based on the closing price of our common stock at the time of repurchase. We reduced unamortized debt discount by \$8.7 million and recognized a \$3.1 million loss on the prepayment, which included the write-off of \$0.7 million in unamortized debt issuance costs.

Provision for Income Taxes

Provision for income taxes was \$9.4 million in the first nine months of 2010 compared to a provision of \$8.1 million in the first nine months of 2009. The provision for the first nine months of 2010 included \$9.4 million for income and withholding taxes of certain of our foreign operations and \$1.2 million of new reserves and interest on existing reserves for potential liabilities in foreign taxing jurisdictions. This is partially offset by the reversal of \$1.2 million for reserves and interest for potential liabilities in foreign taxing jurisdictions which were effectively settled or for which the statute lapsed during the first nine months of 2010. Provision for income taxes was \$8.1 million in the first nine months of 2009. The provision for the first nine months of 2009 included \$10.6 million for income and withholding taxes of certain of our foreign operations and \$1.8 million of new reserves and interest on existing reserves for potential liabilities in foreign taxing jurisdictions, partially offset by a tax benefit of \$4.3 million for the reversal of previously accrued income taxes for uncertain tax positions that have been effectively settled through examination.

Liquidity and Capital Resources

This section includes a discussion and analysis of our cash requirements, off balance sheet arrangements, contingencies, sources and uses of cash, operations, working capital, and long-term assets and liabilities.

Cash Requirements

Contractual Obligations

Our principal outstanding contractual obligations relate to our long-term debt, operating leases, pension obligations and purchase obligations. The following table summarizes our contractual obligations at October 1, 2010 and the effect such obligations are expected to have on our liquidity and cash flow in the future (in millions):

	Payments Due by Period							
	Total	Remainder of 2010	2011	2012	2013	2014	Thereafter	
Contractual obligations (3)								
Long-term debt (2)	\$ 950.0	\$ 63.5	\$109.2	\$246.0	\$513.2	\$ 4.2	\$ 13.9	
Operating leases (1)	100.2	7.2	18.6	15.7	12.4	8.1	38.2	
Purchase obligations (1):								
Capital purchase obligations	86.4	68.7	15.9	1.5	0.3	_	_	
Foundry and inventory purchase obligations	14.5	2.0	4.5	2.7	2.1	1.7	1.5	
Mainframe support	1.1	0.6	0.5	_	_	_	_	
Information technology and communication services	34.3	4.0	15.7	10.2	4.4	_	_	
Other	52.6	6.6	13.2	6.9	6.1	6.0	13.8	
Total contractual obligations	\$1,239.1	\$ 152.6	\$177.6	\$283.0	\$538.5	\$20.0	\$ 67.4	

These represent our off-balance sheet arrangements (See "Liquidity and Capital Resources—Off Balance Sheet Arrangements" for a description of our off balance sheet arrangements.)

⁽²⁾ Includes the interest portion of payments for long-term debt.

⁽³⁾ The table above does not include approximately \$16.1 million of liabilities related to unrecognized tax benefits because we are unable to reasonably estimate the timing of such liabilities.

Our long-term debt includes approximately \$96.2 million of Zero Coupon Convertible Senior Subordinated Notes due 2024 at par, approximately \$95.0 million of 1.875% Convertible Senior Subordinated Notes due 2025 at par, approximately \$48.0 million of 2.625% Convertible Senior Subordinated Notes due 2026 at par, approximately \$4.3 million of loans with two Japanese banks, approximately \$33.0 million of loans with two Chinese banks, approximately \$50.9 million of loans with three Philippine banks, approximately \$24.4 million of loans with a British finance company, approximately \$40.0 million of financing with a Hong Kong bank and approximately \$62.2 million of capital lease obligations. For purposes of the contractual obligations schedule, we have shown the convertible debt at par value maturing upon the first put date. (See Note 7: "Long-Term Debt" of the notes to our unaudited consolidated financial statements included elsewhere in this Form 10-Q).

Our other long-term contractual obligations consist of estimated payments to fund liabilities that have been accrued in our unaudited consolidated balance sheet for our foreign pension plans (see Note 6: "Balance Sheet Information" of the notes to our unaudited consolidated financial statements included elsewhere in this Form 10-Q).

On July 14, 2010 we entered into a definitive purchase agreement with SANYO Electric, providing for the acquisition of SANYO Semiconductor and other assets related to SANYO Electric's semiconductor business. The purchase price is approximately \(\xxi3.0\) billion (\(\xxi366.0\) million), subject to adjustment at closing. We have been working with SANYO Electric and SANYO Semiconductor on the transaction closing conditions, and given the time table required to complete a number of these closing conditions, we believe the transaction will close in the beginning of the first quarter of 2011, rather than the fourth quarter of 2010 as originally estimated. We do not currently believe we will need to access the capital markets to finance the transaction, nor do we currently anticipate the need to issue any new shares of common stock to finance the transaction.

The purchase price is payable in Yen and therefore the actual US dollar value of the purchase price will be determined based on the prevailing exchange rate on the date of closing. All US dollar amounts reflected above are based on a ¥90.2 to \$1 USD exchange rate on the date of the announcement. On October 1, 2010 the exchange rate was ¥83.75 to \$1 USD and had the transaction closed on that date, the consideration payable would have been approximately \$394.0 million.

Off Balance Sheet Arrangements

In the normal course of business, we enter into various operating leases for buildings and equipment including our mainframe computer system, desktop computers, communications, foundry equipment and service agreements relating to this equipment.

In the normal course of business, we provide standby letters of credit or other guarantee instruments to certain parties initiated by either our subsidiaries or us, as required for transactions such as material purchase commitments, agreements to mitigate collection risk, leases or customs guarantees. A bank guarantee issued on our behalf under a non-reusable commitment credit with the bank had an outstanding amount of \$4.2 million as of October 1, 2010. The Belgian bank that issued the guarantee has the right to create a mortgage on the real property of one of our European subsidiaries in the amount of \$3.0 million but had not done so as of October 1, 2010. We also have outstanding guarantees and letters of credit outside of our non-reusable commitment credit totaling \$13.0 million at October 1, 2010.

As part of securing financing in the normal course of business, we issued guarantees related to our capital lease obligations which totaled approximately \$17.1 million as of October 1, 2010. For our operating leases, we expect to make cash payments and similarly incur expenses totaling \$100.2 million as payments come due. We have not recorded any liability in connection with these operating leases, letters of credit and guarantee arrangements.

Based on historical experience and information currently available, we believe that in the foreseeable future we will not be required to make payments under the standby letters of credit or guarantee arrangements.

Contingencies

We are a party to a variety of agreements entered into in the ordinary course of business pursuant to which we may be obligated to indemnify other parties for certain liabilities that arise out of or relate to the subject matter of the agreements. Some of the agreements entered into by us require us to indemnify the other party against losses due to intellectual property infringement, property damage including environmental contamination, personal injury, failure to comply with applicable laws, our negligence or willful misconduct, or breach of representations and warranties and covenants related to such matters as title to sold assets.

We face risk of exposure to warranty and product liability claims in the event that our products fail to perform as expected or such failure of our products results, or is alleged to result, in bodily injury or property damage (or both). In addition, if any of our designed products are alleged to be defective, we may be required to participate in their recall. Depending on the significance of any particular customer and other relevant factors, we may agree to provide more favorable indemnity rights to such customers for valid warranty claims.

We from time to time, have been active in merger and acquisition activity. In connection with the these mergers or acquisitions, we have agreed to indemnify the other party or parties to the merger or acquisition agreement for certain claims or occurrences, limited in most instances, by time and/or monetary amounts.

We and our subsidiaries provide for indemnification of directors, officers and other persons in accordance with limited liability agreements, certificates of incorporation, by-laws, articles of association or similar organizational documents, as the case may be. We maintain directors' and officers' insurance, which should enable us to recover a portion of any future amounts paid.

In addition to the above, from time to time we provide standard representations and warranties to counterparties in contracts in connection with sales of our securities and the engagement of financial advisors and also provide indemnities that protect the counterparties to these contracts in the event they suffer damages as a result of a breach of such representations and warranties or in certain other circumstances relating to the sale of securities or their engagement by us.

While our future obligations under certain agreements may contain limitations on liability for indemnification, other agreements do not contain such limitations and under such agreements it is not possible to predict the maximum potential amount of future payments due to the conditional nature of our obligations and the unique facts and circumstances involved in each particular agreement. Historically, payments made by us under any of these indemnities have not had a material effect on our business, financial condition, results of operations or cash flows and we do not believe that any amounts that we may be required to pay under these indemnities in the future will be material to our business, financial condition, results of operations or cash flows.

See Note 10: "Commitment and Contingencies" of the notes to our unaudited consolidated financial statements, Part II, Item 1 "Legal Proceedings" of this Quarterly Report on Form 10-Q for possible contingencies related to legal matters and see Part I, Item 1 "Business—Government Regulation" of our Annual Report on Form 10-K for the fiscal year ending December 31, 2009 for information on certain environmental matters.

Sources and Uses of Cash

We require cash to fund our operating expenses and working capital requirements, including outlays for research and development, to make capital expenditures, strategic acquisitions and investments, to repurchase our stock and other Company securities, and to pay debt service, including principal and interest and capital lease payments. Our principal sources of liquidity are cash on hand, cash generated from operations and funds from

external borrowings and equity issuances. In the near term, we expect to fund our primary cash requirements through cash generated from operations and cash and cash equivalents on hand. Additionally, as part of our business strategy, we review acquisition and divestiture opportunities and proposals on a regular basis. For additional information about current cash commitments related to the SANYO Transaction, see Note 4: "Acquisitions."

We believe that the key factors that could affect our internal and external sources of cash include:

- factors that affect our results of operations and cash flows, including changes in demand for our products, competitive pricing pressures, effective management of our manufacturing capacity, our ability to achieve further reductions in operating expenses, the impact of our restructuring programs on our production and cost efficiency and our ability to make the research and development expenditures required to remain competitive in our business; and
- factors that affect our access to bank financing and the debt and equity capital markets that could impair our ability to obtain needed financing on acceptable terms or to respond to business opportunities and developments as they arise, including interest rate fluctuations, macroeconomic conditions, sudden reductions in the general availability of lending from banks or the related increase in cost to obtain bank financing; and our ability to maintain compliance with covenants under our existing debt agreements.

Our ability to service our long-term debt including our senior subordinated notes, to remain in compliance with the various covenants contained in our debt agreements and to fund working capital, capital expenditures and business development efforts will depend on our ability to generate cash from operating activities which is subject to, among other things, our future operating performance as well as to general economic, financial, competitive, legislative, regulatory and other conditions, some of which may be beyond our control.

If we fail to generate sufficient cash from operations, we may need to raise additional equity or borrow additional funds to achieve our longer term objectives. There can be no assurance that such equity or borrowings will be available or, if available, will be at rates or prices acceptable to us. We believe that cash flow from operating activities coupled with existing cash and cash equivalents will be adequate to fund our operating and capital needs including the expected SANYO transaction, as well as enable us to maintain compliance with our various debt agreements through at least the next twelve months. To the extent that results or events differ from our financial projections or business plans, our liquidity may be adversely impacted.

Operations

Our operational cash flows are affected by the ability of our operations to generate cash, and our management of our assets and liabilities, including both working capital and long-term assets and liabilities. Each of these components is discussed below.

Working Capital

Working capital fluctuates depending on end-market demand and our effective management of certain items such as receivables, inventory and payables. In times of escalating demand, our working capital requirements may increase as we purchase additional manufacturing materials and increase production. Our working capital may also be affected by restructuring programs, which may require us to use cash for severance payments, asset transfers and contract termination costs. Our working capital, including cash, was \$618.5 million at October 1, 2010. Our working capital, excluding cash, was \$55.6 million at October 1, 2010, and has fluctuated between \$101.3 million and \$13.1 million over the last eight quarter-ends.

The components of our working capital at October 1, 2010 and December 31, 2009 are set forth below (in millions), followed by explanations for changes between October 1, 2010 and December 31, 2009 for cash and cash equivalents and any other changes greater than \$5 million:

	October 1, 2010	December 31, 2009	Change
Current Assets			
Cash and cash equivalents	\$ 562.9	\$ 525.7	\$ 37.2
Short-term investments	_	45.50	(45.5)
Receivables, net	314.30	260.9	53.4
Inventories, net	350.80	269.9	80.9
Other current assets	52.90	51.5	1.4
Deferred income taxes	14.80	15.1	(0.3)
Total current assets	1,295.7	1,168.6	127.1
Current Liabilities			
Accounts payable	250.9	172.9	78.0
Accrued expenses	149.1	135.5	13.6
Income taxes payable	2.4	5.0	(2.6)
Accrued interest	4.5	0.9	3.6
Deferred income on sales to distributors	134.5	98.8	35.7
Current portion of long-term debt	135.8	205.9	(70.1)
Total current liabilities	677.2	619.0	58.2
Working capital	\$ 618.5	\$ 549.6	\$ 68.9

The increase of \$37.2 million of cash and cash equivalents is primarily due to \$392.0 million of cash provided by operating activities, partially offset by \$190.3 million of cash used in investing activities and \$165.5 million of cash used in financing activities.

The decrease of \$45.5 million in short-term investments is the result of investments maturing as cash and cash equivalents.

The increase of \$53.4 million in receivables, net is the result of increased revenues at the end of the third quarter of 2010 compared to the end of the fourth quarter of 2009.

The increase of \$80.9 million in inventories, net is the result of increased production needs to satisfy increased demand for products during the third quarter of 2010 as compared to the fourth quarter of 2009.

The increase of \$78.0 million in accounts payable is the result of increased purchase of property, plant and equipment combined with an increase in purchases for production of inventory during the third quarter of 2010 as compared with the fourth quarter of 2009.

The increase of \$13.6 million in accrued expenses is the result of increases in accrued bonuses, accrued vacation and accrued payroll.

The increase of \$35.7 million in deferred income on sales to distributors is the result of increased inventories at our distributors.

The decrease of \$70.1 million in current portion of long-term debt was primarily due to the reclassification of the principal balance of our Zero Coupon Convertible Senior Subordinated Notes to long-term debt as a result of the amendment to the notes which added an April 15, 2012 put and call option.

Long-Term Assets and Liabilities

Our long-term assets consist primarily of property, plant and equipment, intangible assets, goodwill, foreign tax receivables and capitalized debt issuance costs.

Our manufacturing rationalization plans have included efforts to utilize our existing manufacturing assets and supply arrangements more efficiently. We believe that near-term access to additional manufacturing capacity, should it be required, could be readily obtained on reasonable terms through manufacturing agreements with third parties. Cash capital expenditures were \$145.9 million during the first nine months of 2010 compared to cash capital expenditures of \$46.3 million during the first nine months of 2009. We will continue to look for opportunities to make strategic purchases in the future for additional capacity.

Our long-term liabilities, excluding long-term debt, consist of liabilities under our foreign defined benefit pension plans and contingent tax reserves. In regard to our foreign defined benefit pension plans, generally, our annual funding of these obligations is equal to the minimum amount legally required in each jurisdiction in which the plans operate. This annual amount is dependent upon numerous actuarial assumptions.

Key Financing Events

2010 Financing Events

On April 12, 2010, we unilaterally amended the Indenture for our Zero Coupon Convertible Senior Subordinated Notes due 2024.

The amendments include:

- One additional opportunity to require us to purchase the notes on April 15, 2012. The terms of this put option are otherwise identical to pre-existing trms of the notes whereby holders of the notes had the option to require us to purchase the notes on April 15, 2010; and
- Terms eliminating our ability to redeem the notes at our option from April 15, 2010 until April 15, 2012.

In accordance with the right of the holders of the notes to require us to purchase the notes on April 15, 2010, approximately \$3.2 million of the \$99.4 million par value of notes then outstanding were purchased by us. In accordance with ASC 470—Debt, the amendment was considered a substantial modification for accounting purposes therefore; the \$96.2 million original remaining debt was deemed to be extinguished, resulting in a \$0.1 million gain, and new convertible debt with fair value of \$98.5 million was deemed to be issued.

ASC 470, requires the issuer of convertible debt instruments with cash settlement features to separately account for the liability and equity components of the instrument. Thus, the liability component of the new convertible debt was recognized at the present value of its cash flows discounted using a discount rate equivalent to the borrowing rate at the date of the modification of the Convertible Notes for similar debt instruments without conversion feature. The equity component of the new convertible debt was recorded as additional paid in capital and represents the difference between the fair value of the modified Convertible Notes and the liability component. It also requires an accretion of the debt discount resulting from the allocation of a portion of the modified fair value to equity over the life of the Convertible Notes, which is expected to be the next put date. As a result, we recognized \$13.3 million of debt discount, which will be amortized through April 2012.

In June 2010, one of our Asian subsidiaries entered into a loan agreement with a Hong Kong bank pursuant to which the bank purchased accounts receivables, with recourse. In accordance with Generally Accepted Accounting Principles in the United States, the purchased assets remained on our balance sheet as of October 1, 2010. The loan, which had a balance of \$40.0 million as of October 1, 2010, bears interest payable weekly at 2-month LIBOR plus 1.75%. The loan amount is subject to an eligible borrowing calculation as defined in the loan agreement.

In September 2010, one of our Chinese subsidiaries entered into two new short-term loan agreements with a Chinese bank to finance the purchase of raw materials. These short-term loans refinanced two loans with the same bank that matured in 2010. The loans, which had a balance of \$14.0 million as of October 1, 2010, bear interest payable quarterly based on 3-month LIBOR plus 3.8% per annum. The full amount of the loans is due in September 2011.

In September 2010, our Philippine subsidiary entered into a five-year loan agreement with a Philippine bank to finance capital expenditures and other general corporate purposes. The loan, which had a balance of \$20.0 million as of October 1, 2010, bears interest payable quarterly based on 3-month LIBOR plus 2.0% per annum. Sixty percent of the total loan amount will be repaid in nineteen equal quarterly installments with the balance due in September 2015.

Debt Instruments, Guarantees and Related Covenants

The following table presents the components of long-term debt as of October 1, 2010 and December 31, 2009 (dollars in millions):

	October 1, 2010	December 31, 2009
Senior Bank Facilities:		
Term Loan, interest payable monthly at 2.00063%	\$ —	\$ 170.2
Zero Coupon Convertible Senior Subordinated Notes due 2024 (1)	85.9	96.9
1.875% Convertible Senior Subordinated Notes due 2025 (2)	80.7	76.5
2.625% Convertible Senior Subordinated Notes due 2026 (3)	404.7	389.0
2.25% Loan with Japanese bank due 2010, interest payable semi-annually	_	3.6
Loan with Philippine banks due 2010 through 2012, interest payable quarterly at 1.52681% and 1.28438%, respectively	16.3	18.7
Loan with Philippine bank due 2010 through 2013, interest payable quarterly at 1.05375% and 1.00563%, respectively	9.3	10.5
Loan with Philippine bank due 2010 through 2013, interest payable quarterly at 1.54188% and 1.50375%, respectively	5.2	5.9
Loan with Philippine banks due 2010 through 2014, interest payable quarterly at 6.03063% prepaid in Q2 2010	_	10.3
Loan with Philippine bank due 2010 through 2015, interest payable quarterly at 2.2925%	20.0	_
Short-term loan with Chinese bank due 2010, interest payable quarterly at 3.36938% and 3.2725%, respectively	7.0	7.0
Short-term loan with Chinese bank due 2010, interest payable quarterly at 2.7825%	_	7.0
Short-term loan with Chinese bank due 2010, interest payable quarterly at 2.7825%	_	6.0
Short-term loan with Chinese bank due 2011, interest payable quarterly at 4.09013% and 5.25063%, respectively	7.0	7.0
Short-term loan with Chinese bank due 2011, interest payable quarterly at 4.08938% and 4.28063%, respectively	7.0	7.0
Short-term loan with Chinese bank due 2010, interest payable quarterly at 2.55481% and 2.2525%, respectively	12.0	12.0
Loan with British finance company, interest payable monthly at 1.85% and 1.75%, respectively	24.4	23.1
Loan with Hong Kong bank, interest payable weekly at 2.02297%	40.0	_
1.875% Loan with Japanese bank due 2010 through 2013, interest payable semi-annually	2.5	2.6
Short-term loan with Japanese bank due 2010, interest payable monthly at .96% and 1.06%, respectively	1.8	1.6
Capital lease obligations	62.2	78.6
	786.0	933.5
Less: Current maturities	(135.8)	(205.9)
	650.2	727.6

- (1) The Zero Coupon Convertible Senior Subordinated Notes due 2024 may be put back to us at the option of the holders of the notes on April 15 of 2012, 2014 and 2019 or called at our option on or after April 15, 2012.
- (2) The 1.875% Convertible Senior Subordinated Notes due 2025 may be put back to us at the option of the holders of the notes on December 15 of 2012, 2015 and 2020 or called at our option on or after December 20, 2012.
- (3) The 2.625% Convertible Senior Subordinated Notes due 2026 may be put back to us at the option of the holders of the notes on December 15 of 2013, 2016 and 2021 or called at our option on or after December 20, 2013.

As of October 1, 2010, we were in compliance with the indentures relating to our Zero Coupon Convertible Senior Subordinated Notes due 2024, our 1.875% Convertible Senior Subordinated Notes due 2025 and our 2.625% Convertible Senior Subordinated Notes due 2026 and with covenants relating to other debt agreements. We believe that we will be able to comply with the various covenants and other requirements contained in such indentures and debt agreements through October 1, 2011.

Cash Management

Our ability to manage cash is limited, as our primary cash inflows and outflows are dictated by the terms of our sales and supply agreements, contractual obligations, debt instruments and legal and regulatory requirements. While we have some flexibility with respect to the timing of capital equipment purchases, we must invest in capital equipment on a timely basis to allow us to maintain our manufacturing efficiency and support our platforms of new products.

New Accounting Pronouncements Adopted

Adoption of ASU No. 2010-06, "Fair Value Measurements and Disclosures (Topic 820): Improving Disclosures about Fair Value Measurements"

In January 2010, the FASB issued ASU No. 2010-06, "Fair Value Measurements and Disclosures (Topic 820): Improving Disclosures about Fair Value Measurements" ("ASU 2010-06"), which amends the disclosure guidance with respect to fair value measurements. Specifically, the new guidance requires disclosure of amounts transferred in and out of Levels 1 and 2 fair value measurements, a reconciliation presented on a gross basis rather than a net basis of activity in Level 3 fair value measurements, greater disaggregation of the assets and liabilities for which fair value measurements are presented and more robust disclosure of the valuation techniques and inputs used to measure Level 2 and 3 fair value measurements. ASU 2010-06 is effective for interim and annual reporting periods beginning after December 15, 2009, with the exception of the new guidance around the Level 3 activity reconciliations, which is effective for fiscal years beginning after December 15, 2010. The adoption of this pronouncement did not have a material impact on our consolidated financial statements.

Critical Accounting Policies and Estimates

The accompanying discussion and analysis of our financial condition and results of operations is based upon our unaudited consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States of America. We believe certain of our accounting policies are critical to understanding our financial position and results of operations. We utilize the following critical accounting policies in the preparation of our financial statements.

Revenue. We generate revenue from sales of our semiconductor products to original equipment manufacturers, electronic manufacturing service providers and distributors. We also generate revenue, although to a much lesser extent, from manufacturing services provided to customers. We recognize revenue on sales to original equipment manufacturers and electronic manufacturing service providers and sales of manufacturing

services net of provisions for related sales returns and allowances when persuasive evidence of an arrangement exists, title and risk of loss pass to the customer (which is generally upon shipment), the price is fixed or determinable and collectability is reasonably assured. Title to products sold to distributors typically passes at the time of shipment by us so we record accounts receivable for the amount of the transaction, reduce our inventory for the products shipped and defer the related margin in our consolidated balance sheet. We recognize the related revenue and cost of revenues when the distributor informs us that they have resold the products to the end user. Inaccuracies in the sales or inventory data provided to us by our distributors can therefore result in inaccuracy in our reporting revenues. As a result of our inability to reliably estimate up front the effect of the returns and allowances with these distributors, we defer the related revenue and margin on sales to distributors. Although payment terms vary, most distributor agreements require payment within 30 days.

Taxes assessed by government authorities on revenue-producing transactions, including value added and excise taxes, are presented on a net basis (excluded from revenues) in the statement of operations.

Sales returns and allowances are estimated based on historical experience. Our original equipment manufacturer customers do not have the right to return our products other than pursuant to the provisions of our standard warranty. Sales to distributors, however, are typically made pursuant to agreements that provide return rights with respect to discontinued or slow-moving products. Under our general agreements, distributors are allowed to return any product that we have removed from our price book. In addition, agreements with our distributors typically contain standard stock rotation provisions permitting limited levels of product returns. However, since we defer recognition of revenue and gross profit on sales to distributors until the distributor resells the product, due to our inability to reliably estimate up front the effect of the returns and allowances with these distributors, sales returns and allowances have minimal impact on our results of operations. Provisions for discounts and rebates to customers, estimated returns and allowances, and other adjustments are provided for in the same period the related revenues are recognized, and are netted against revenues. Given that our revenues consist of a high volume of relatively similar products, our actual returns and allowances and warranty claims have not traditionally fluctuated significantly from period to period, and our returns and allowances and warranty provisions have historically been reasonably accurate.

We generally warrant that products sold to our customers will, at the time of shipment, be free from defects in workmanship and materials and conform to our approved specifications. Our standard warranty extends for a period that is the greater of (i) three years from the date of shipment or (ii) the period of time specified in the customer's standard warranty (provided that the customer's standard warranty is stated in writing and extended to purchasers at no additional charge). At the time revenue is recognized, we establish an accrual for estimated warranty expenses associated with our sales, recorded as a component of cost of revenues. In addition, we also offer cash discounts to customers for payments received by us within an agreed upon time, generally 10 days after shipment. We accrue reserves for cash discounts as a reduction to accounts receivable and a reduction to revenues, based on experience with each customer.

Freight and handling costs are included in the cost of revenues and are recognized as period expenses during the period in which they are incurred.

Inventories. We carry our inventories not related to an acquisition at the lower of standard cost (which approximates actual cost on a first-in, first-out basis) or market and record provisions for slow moving inventories based upon a regular analysis of inventory on hand compared to historical and projected end user demand. These provisions can influence our results from operations. For example, when demand falls for a given part, all or a portion of the related inventory is reserved, impacting our cost of revenues and gross profit. If demand recovers and the parts previously reserved are sold, we will generally recognize a higher than normal margin. However, the majority of product inventory that has been previously reserved is ultimately discarded. Although we do sell some products that have previously been written down, such sales have historically been relatively consistent on a quarterly basis and the related impact on our margins has not been material.

Inventory obtained in the purchase of a business is stated at the lower of cost or market. Upon the acquisition of a company such as Catalyst, AMIS, PulseCore, CMD or SDT, we used management estimates to determine the fair value of the inventory as of the acquisition date. The methodology involves stepping up the value of acquired finished goods and work-in-process to expected sales value less variable costs to dispose. For the nine months ended October 1, 2010 approximately \$7.4 million of the initial \$7.9 million in the inventory step up for acquisitions has been expensed to the statement of operations since the inventory was shipped to the customer, leaving \$0.5 million in inventory and inventories at distributors at October 1, 2010. As this inventory is shipped to customers, it will significantly decrease the gross margin reported on those future sales until the inventory is completely sold.

Deferred Tax Valuation Allowance. We record a valuation allowance to reduce our deferred tax assets to the amount that is more likely than not to be realized. In determining the amount of the valuation allowance, we consider estimated future taxable income as well as feasible tax planning strategies in each taxing jurisdiction in which we operate. If we determine that we will not realize all or a portion of our remaining deferred tax assets, we will increase our valuation allowance with a charge to income tax expense. Conversely, if we determine that we will ultimately be able to utilize all or a portion of the deferred tax assets for which a valuation allowance has been provided, the related portion of the valuation allowance will be released to income as a credit to income tax expense. In the fourth quarter of 2001, a valuation allowance was established for our domestic deferred tax assets and a portion of our foreign deferred tax assets. Additionally, from 2003 throughout 2009, no incremental domestic deferred tax benefits were recognized. Our ability to utilize our deferred tax assets and the continuing need for a related valuation allowance are monitored on an ongoing basis.

Impairment of Long-Lived Assets. We evaluate the recoverability of the carrying amount of our property, plant and equipment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be fully recoverable. Impairment is assessed when the undiscounted expected cash flows derived for an asset are less than its carrying amount. Impairment losses are measured as the amount by which the carrying value of an asset exceeds its fair value and are recognized in operating results. We continually apply our best judgment when applying these impairment rules to determine the timing of the impairment test, the undiscounted cash flows used to assess impairments and the fair value of an impaired asset. The dynamic economic environment in which we operate and the resulting assumptions used to estimate future cash flows impact the outcome of our impairment tests. In recent years, most of our assets that have been impaired consist of assets that were ultimately abandoned, sold or otherwise disposed of due to cost reduction activities and the consolidation of our manufacturing facilities. In some instances, these assets have subsequently been sold for amounts higher than their impaired value. When material, these gains are recorded in the restructuring, asset impairment and other, net line item in our consolidated statement of operations and disclosed in the footnotes to the financial statements.

Goodwill. We evaluate our goodwill for potential impairment on an annual basis or whenever events or circumstances indicate that an impairment may have occurred using a two-step process. The first step of the goodwill impairment test, used to identify potential impairment, compares the estimated fair value of the reporting unit containing our goodwill with the related carrying amount. If the estimated fair value of the reporting unit exceeds its carrying amount, the reporting unit's goodwill is not considered to be impaired and the second step is unnecessary. Our methodologies used for valuing goodwill have not changed.

We have determined that our product families, which are components of our operating segments, constitute reporting units for purposes of allocating and testing goodwill, because they are one level below the operating segment, they constitute individual businesses and our segment management regularly reviews the operating results of each product family. As of each acquisition date, all goodwill acquired was assigned to the product families that were expected to benefit from the synergies of the respective acquisition. The amount of goodwill assigned to each reporting unit was the difference between the fair value of the reporting unit and the fair value of identifiable assets and liabilities allocated to the reporting unit as of the acquisition date. We determined the fair value of each reporting unit using the income approach, which is based on the present value of estimated future cash flows using management's assumptions and forecasts as of the acquisition date.

We perform our annual impairment analysis as of the first day of the fourth quarter of each year. Our next annual test for impairment is expected to be performed in our fourth quarter of 2010; however, identification of a triggering event may result in the need for earlier reassessments of the recoverability of our goodwill and may result in material impairment charges in future periods.

Defined Benefit Plans. We maintain pension plans covering certain of our employees. For financial reporting purposes, net periodic pension costs are calculated based upon a number of actuarial assumptions, including a discount rate for plan obligations, assumed rate of return on pension plan assets and assumed rate of compensation increase for plan employees. All of these assumptions are based upon management's judgment, considering all known trends and uncertainties. Actual results that differ from these assumptions impact the expense recognition and cash funding requirements of our pension plans.

Contingencies. We are involved in a variety of legal matters that arise in the normal course of business. Based on the available information, we evaluate the relevant range and likelihood of potential outcomes and we record the appropriate liability when the amount is deemed probable and estimable.

Valuation of Stock Compensation. The fair value of each option grant is estimated on the date of grant using a lattice-based option valuation model. The lattice model uses: 1) a constant volatility; 2) an employee exercise behavior model (based on an analysis of historical exercise behavior); and 3) the treasury yield curve to calculate the fair value of shares issued for each option grant. We continue to use the Black-Scholes option-pricing model to calculate the fair value of shares issued under the 2000 Employee Stock Purchase Plan.

Recent Accounting Pronouncements

In April 2010, the FASB issued Accounting Standards Update ("ASU") No. 2010-17, Revenue Recognition—Milestone Method and is included in ASC 605—Milestone Method of Revenue Recognition. The ASU codifies the consensus reached in EITF 08-9, "Milestone Method of Revenue Recognition," ("EITF 08-9") addresses the accounting when entities enter into revenue arrangements with multiple payment streams for a single deliverable or a single unit of accounting. The pronouncement shall be applied prospectively to milestones achieved in fiscal years, and interim periods within those years, beginning after June 15, 2010, with earlier application and retrospective application permitted. We are currently assessing the impact of ASU No. 2010-17 on our financial position and results of operations.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We are exposed to financial market risks, including changes in interest rates and foreign currency exchange rates. To mitigate these risks, we utilize derivative financial instruments. We do not use derivative financial instruments for speculative or trading purposes.

At October 1, 2010, our long-term debt (including current maturities) totaled \$786.0 million. We have no interest rate exposure to rate changes on our fixed rate debt, which totaled \$635.9 million. We do have interest rate exposure with respect to the \$150.1 million outstanding balance on our variable interest rate debt. A 50 basis point increase in interest rates would impact our expected annual interest expense for the next twelve months by approximately \$0.8 million. However, some of this impact would be offset by additional interest earned on our cash and cash equivalents should rates on deposits and investments also increase.

A majority of our revenue, expense and capital purchasing activities are transacted in U.S. dollars. However, as a multinational business, we also conduct certain of these activities through transactions denominated in a variety of other currencies. We use forward foreign currency contracts to hedge firm commitments and reduce our overall exposure to the effects of currency fluctuations on our results of operations and cash flows. Gains and losses on these foreign currency exposures would generally be offset by corresponding losses and gains on the related hedging instruments. This strategy reduces, but does not eliminate, the short-term impact of foreign

currency exchange rate movements. For example, changes in exchange rates may affect the foreign currency sales price of our products and can lead to increases or decreases in sales volume to the extent that the sales price of comparable products of our competitors are less or more than the sales price of our products. Our policies prohibit speculation on financial instruments, trading in currencies for which there are no underlying exposures, or entering into trades for any currency to intentionally increase the underlying exposure.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures. We carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures (as defined in the Securities Exchange Act of 1934 ("Exchange Act") Rules 13a-15(e) and 15d-15(e)). Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that, as of the end of the period covered in this report, our disclosure controls and procedures were effective to ensure that information required to be disclosed in reports filed under the Exchange Act is recorded, processed, summarized and reported within the required time periods and is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Controls Over Financial Reporting. On January 27, 2010, we acquired CMD, which operated under its own set of systems and internal controls. We completed the integration of CMD's operations into our systems and control environment during the third quarter of 2010.

On June 9, 2010, we acquired SDT, which operated under its own set of systems and internal controls. We are separately maintaining SDT's systems and much of its control environment until we are able to incorporate SDT's process into our own system and control environment. We currently expect to complete the integration of SDT's operations into our systems and control environment by the second half of 2011.

Other than as discussed above, there have been no other changes to our internal controls over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) that occurred during the fiscal quarter ended October 1, 2010 which have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II: OTHER INFORMATION

Item 1. Legal Proceedings

We currently are involved in a variety of legal matters that arise in the normal course of business. Based on information currently available, management does not believe that the ultimate resolution of these matters, including the matters described or referred to in the next paragraphs will have a material effect on our financial condition, results of operations or cash flows. However, because of the nature and inherent uncertainties of litigation, should the outcome of these actions be unfavorable, our business, consolidated financial position, results of operations or cash flows could be materially and adversely affected.

Securities Class Action Litigation

During the period July 5, 2001 through July 27, 2001, we were named as a defendant in three shareholder class action lawsuits that were filed in federal court in New York City against us and certain of our former officers, current and former directors and the underwriters of our initial public offering. The lawsuits allege violations of the federal securities laws and have been docketed in the U.S. District Court for the Southern District of New York ("District Court") as: *Abrams v. ON Semiconductor Corp.*, et al., C.A. No. 01-CV-6114; Breuer v. ON Semiconductor Corp., et al., C.A. No. 01-CV-6287; and Cohen v. ON Semiconductor Corp., et al., C.A. No. 01-CV-6942. On April 19, 2002, the plaintiffs filed a single consolidated amended complaint that supersedes the individual complaints originally filed. The amended complaint alleges, among other things, that the underwriters of our initial public offering improperly required their customers to pay the underwriters' excessive commissions and to agree to buy additional shares of our common stock in the aftermarket as conditions of receiving shares in our initial public offering. The amended complaint further alleges that these supposed practices of the underwriters should have been disclosed in our initial public offering prospectus and registration statement. The amended complaint alleges violations of both the registration and antifraud provisions of the federal securities laws and seeks unspecified damages. We understand that various other plaintiffs have filed substantially similar class action cases against approximately 300 other publicly-traded companies and their public offering underwriters in New York City, which have all been transferred, along with the case against us, to a single federal district court judge for purposes of coordinated case management. We believe that the claims against us are without merit and have defended, and intend to continue to defend, the litigation vigorously. The litigation process is inherently uncertain, however, and we cann

On July 15, 2002, together with the other issuer defendants, we filed a collective motion to dismiss the consolidated, amended complaints against the issuers on various legal grounds common to all or most of the issuer defendants. The underwriters also filed separate motions to dismiss the claims against them. In addition, the parties have stipulated to the voluntary dismissal without prejudice of our individual former officers and current and former directors who were named as defendants in our litigation, and they are no longer parties to the litigation. On February 19, 2003, the District Court issued its ruling on the motions to dismiss filed by the underwriter and issuer defendants. In that ruling the District Court granted in part and denied in part those motions. As to the claims brought against us under the antifraud provisions of the securities laws, the District Court dismissed all of these claims with prejudice, and refused to allow plaintiffs the opportunity to re-plead these claims. As to the claims brought under the registration provisions of the securities laws, which do not require that intent to defraud be pleaded, the District Court denied the motion to dismiss these claims as to us and as to substantially all of the other issuer defendants as well. The District Court also denied the underwriter defendants' motion to dismiss in all respects.

In June 2003, upon the determination of a special independent committee of our Board of Directors, we elected to participate in a proposed settlement with the plaintiffs in this litigation. Had it been approved by the District Court, this proposed settlement would have resulted in the dismissal, with prejudice, of all claims in the litigation against us and against any of the other issuer defendants who elected to participate in the proposed settlement, together with the current or former officers and directors of participating issuers who were named as

individual defendants. This proposed issuer settlement was conditioned on, among other things, a ruling by the District Court that the claims against us and against the other issuers who had agreed to the settlement would be certified for class action treatment for purposes of the proposed settlement, such that all investors included in the proposed classes in these cases would be bound by the terms of the settlement unless an investor opted to be excluded from the settlement in a timely and appropriate fashion.

On December 5, 2006, the U.S. Court of Appeals for the Second Circuit ("Court of Appeals") issued a decision in *In re Initial Public Offering Securities Litigation* that six purported class action lawsuits containing allegations substantially similar to those asserted against us could not be certified as class actions due, in part, to the Court of Appeals' determination that individual issues of reliance and knowledge would predominate over issues common to the proposed classes. On January 8, 2007, the plaintiffs filed a petition seeking rehearing *en banc* of this ruling. On April 6, 2007, the Court of Appeals denied the plaintiffs' petition for rehearing of the Court of Appeals' December 5, 2006 ruling. The Court of Appeals, however, noted that the plaintiffs remained free to ask the District Court to certify classes different from the ones originally proposed which might meet the standards for class certification that the Court of Appeals articulated in its December 5, 2006 decision. In light of the Court of Appeals' December 5, 2006 decision regarding certification of the plaintiffs' claims, the District Court entered an order on June 25, 2007 terminating the proposed settlement between the plaintiffs and the issuers, including us.

On August 14, 2007, the plaintiffs filed amended complaints in the six focus cases. The issuer defendants and the underwriter defendants separately moved to dismiss the claims against them in the amended complaints in the six focus cases. On March 26, 2008, the District Court issued an order in which it denied in substantial part the motions to dismiss the amended complaints in the six focus cases.

On February 25, 2009, the parties advised the District Court that they had reached an agreement-in-principle to settle the litigation in its entirety. A stipulation of settlement was filed with the District Court on April 2, 2009. On June 9, 2009, the District Court preliminarily approved the proposed global settlement. Notice was provided to the class, and a settlement fairness hearing, at which members of the class had an opportunity to object to the proposed settlement, was held on September 10, 2009. On October 6, 2009, the District Court issued an order granting final approval to the settlement. That order remains subject to appeal. Several appeals have been filed objecting to the definition of the settlement class and fairness of the settlement, and those appeals remain pending. The settlement calls for a total payment of \$586 million from all defendants, including underwriters, of which \$100 million is allocated to the approximately 300 issuer defendants. Under the settlement, our insurers are to pay the full amount of settlement share allocated to us, and we would bear no financial liability. We, as well as the officer and director defendants (current and former) who were previously dismissed from the action pursuant to tolling agreements, are to receive complete dismissals from the case. While we can make no assurances or guarantees as to the outcome of these proceedings, based upon our current knowledge, we believe that the final result of this action will have no material effect on our consolidated financial position, results of operations or cash flows.

Other Litigation

On January 27, 2010, we completed our acquisition of all of the outstanding shares of common stock of CMD through a cash tender offer of \$4.70 per share which was then followed by the merger of Purchaser (defined below) and CMD, in accordance with the December 14, 2009 definitive merger agreement ("Merger Agreement") which we previously announced we had entered into with CMD ("Transaction"). Shortly after we signed the Merger Agreement and announced the tender offer, we were named as a defendant in three purported class action lawsuits, described below, filed in California and Delaware against us, CMD, CMD's Board of Directors and Pac-10 Acquisition Corporation, our direct, wholly-owned subsidiary ("Purchaser").

On December 14, 2009, a purported class action lawsuit was filed in the Superior Court of Santa Clara County, California ("Court") captioned *Robert Varrenti*, et al. v. Robert Dickinson, Edward Ross, John Sprague,

David Wittrock, David Sear, Jon Castor, John Fichthorn, J. Michael Gullard, Kenneth Potashner, California Micro Devices, ON Semiconductor Corporation and Pac-10 Acquisition Corporation (No. 109CV159469) ("California Action"). On December 29, 2009, the plaintiff in the California Action filed an amended complaint. On December 21, 2009, a second purported class action lawsuit was filed in the Court of Chancery in the State of Delaware captioned Annamarie Medeiros et al. v. California Micro Devices, Jon S. Castor, Robert V. Dickinson, Edward C. Ross, John Fichthorn, J. Michael Gullard, Kenneth Potashner, David L. Wittrock, Pac-10 Acquisition Corporation and ON Semiconductor Corporation (No. 5159). On January 4, 2010, a third purported class action lawsuit was filed in the Court of Chancery in the State of Delaware captioned Sanjay Israni, et al. v. California Micro Devices, Robert V. Dickinson, Edward C. Ross, Jon S. Castor, John Fichthorn, J. Michael Gullard, Kenneth Potashner, David L. Wittrock, ON Semiconductor Corporation and Pac-10 Acquisition Corporation (No. 5181) (the Delaware lawsuits referred to herein as the "Delaware Actions"). All three lawsuits contain similar allegations, stating generally that the proposed Transaction is the product of a breach of fiduciary duties by CMD's Board of Directors by failing to adequately discharge their duties in negotiating and agreeing to the Transaction and that we and the Purchaser assisted in that breach. All three lawsuits requested an injunction enjoining the consummation of the Transaction. The Israni complaint also included a request for damages.

On January 19, 2010, the parties entered into a memorandum of understanding ("MOU") to settle the three lawsuits and on February 18, 2010, the parties entered into a stipulation of settlement ("Stipulation"). The settlement, like the MOU, calls for CMD to agree to make available to shareholders certain additional information, which has been completed, and CMD or its insurer to agree to pay plaintiffs' counsel for fees and expenses not to exceed \$495,000. We expect CMD's insurer to pay \$245,000 of this total amount. This payment did not affect the amount of consideration paid to the stockholders of CMD in connection with the Transaction. See the MOU document at Exhibit 10.1 to CMD's Form 8-K filed with the SEC on January 20, 2010, and see a summary of the MOU in the Company's Amendment No. 3 to Schedule TO filed with the SEC on January 20, 2010, which summary is incorporated by reference herein. The defendants maintain that the lawsuits are completely without merit. Nevertheless, the defendants agreed to the settlement in order to avoid costly litigation, eliminate the risk of delaying the closing of the Transaction, and because the only effect of the settlement on CMD stockholders was to provide additional disclosure. The Stipulation was filed with the Court on March 29, 2010 and a hearing to preliminarily approve the settlement was held on May 7, 2010. On May 25, 2010, the Court preliminarily approved the Stipulation and scheduled a settlement hearing on July 23, 2010. At the July 23, 2010 hearing, the court approved the settlement, but requested additional information from the plaintiffs regarding fees and expenses.

On August 2, 2010, the Court issued a final judgment approving the settlement set forth in the Stipulation and dismissing the California Action with prejudice. On September 8, 2010 the Delaware Court issued its order dismissing the Delaware Actions with prejudice based on the preceding order issued by the California Court. We believe that the final result of this action will have no material effect on our consolidated financial position, results of operations or cash flows.

Intellectual Property Matters

We face risk to exposure from claims of infringement of the intellectual property rights of others. In the ordinary course of business, we receive letters asserting that our products or components breach another party's rights. These threats may seek that we make royalty payments, that we stop use of such rights, or other remedies.

Prior to the acquisition of AMIS by us on March 17, 2008, in January 2003, Ricoh Company, Ltd. ("Ricoh") filed in the U.S. District Court for the District of Delaware a complaint against AMIS and other parties (including Synopsys, Inc. ("Synopsys"), alleging infringement of a patent owned by Ricoh. AMIS promptly tendered the defense of this claim to Synopsys, and Synopsys agreed to assume the defense of the case on AMIS' behalf to the extent that the Synopsys software that AMIS licensed from Synopsys is alleged to constitute the basis of Ricoh's claim of infringement. The case has been transferred to the U.S. District Court for the Northern District of California. Ricoh is seeking an injunction and damages in an unspecified amount relating to such alleged infringement. The patents relate to certain methodologies for the automated design of custom semiconductors.

The case was scheduled to go to trial in March 2007; however, in December 2006, the court issued an order staying the case pending a re-examination proceeding filed by Synopsys before the U.S. Patent & Trademark Office ("PTO") challenging the validity of the patent claims at issue in this case. Since that time, Synopsys filed a total of three re-examination petitions with the PTO challenging the validity of the claims at issue which the PTO granted and consolidated all three re-examinations into one proceeding before a single examiner. The re-examination proceeding was completed in September 2008, and the PTO examiner issued a final rejection of all claims in the asserted patent over prior art. Ricoh has appealed that final rejection to the PTO Board of Appeals, which held a hearing on the appeal September 29, 2010. The Board of Appeals is expected to issue its written decision on the appeal within the next several months. In April 2008, the court lifted the stay despite the ongoing re-examination proceeding in the PTO. In September 2008, the court granted defendants' request to refile a summary judgment motion on non-infringement that had been vacated as moot when the stay was imposed in December 2006. On March 6, 2009, the judge issued a ruling denying the summary judgment motion without prejudice because of a factual dispute over a patent claim element. After an exchange of briefs by the parties related to the disputed claim element, the judge held a further hearing on the matter on June 12, 2009. On October 23, 2009, the judge issued his ruling on the disputed claim element. Based on the judge's ruling, Synopsys filed another motion for summary judgment on non-infringement on January 8, 2010. A hearing on that motion was held on March 8, 2010 and on April 14, 2010 the judge granted Synopsys' motion for summary judgment. On April 28, 2010 Ricoh filed a motion for reconsideration on the summary judgment ruling. On May 28, 2010, the judge denied the Ricoh motion and entered final judgment in Synopsys' favor for non-infringement. Ricoh subsequently filed a Notice of Appeal on June 23, 2010 and served its brief on appeal August 31, 2010. Counsel for Synopsys expects to file its responsive brief by October 29, 2010 with oral arguments scheduled to occur as early as February 2011 and a decision on the appeal likely in the spring of 2011. We believe that the asserted claims are without merit, and even if meritorious, that we will be indemnified against damages by Synopsys, and that resolution of this matter will not have a material adverse effect on our consolidated financial position, results of operations or cash flows.

Item 1A. Risk Factors

The risk factor below updates those risk factors included in our Form 10-Q for the fiscal quarters ended April 2, 2010 and July 2, 2010, and in our Form 10-K for the fiscal year ended December 31, 2009 ("2009 Form 10-K"). This Form 10-Q includes "forward-looking statements," as that term is defined in Section 27A of the Securities Act of 1933 and Section 21E of the Exchange Act. All statements, other than statements of historical facts, included or incorporated in this Form 10-Q could be deemed forward-looking statements, particularly statements about our plans, strategies and prospects under the heading "Management's Discussion and Analysis of Financial Condition and Results of Operations." Forward-looking statements are often characterized by the use of words such as "believes," "estimates," "expects," "projects," "may," "will," "intends," "plans," or "anticipates," or by discussions of strategy, plans or intentions. All forward-looking statements in this Form 10-Q are made based on our current expectations and estimates, and involve risks, uncertainties and other factors that could cause results or events to differ materially from those expressed in forward-looking statements. Among these factors are, as discussed more below and in other parts of this and our previously filed Form 10-Qs, our revenues and operating performance, poor economic conditions and markets (including the current credit and financial conditions), the cyclical nature of the semiconductor industry, changes in demand for our products, changes in our customers and distributors, inventories, technological and product development risks, availability of raw materials, competitors' actions, pricing and gross margin pressures, loss of key customers, order cancellations or reduced bookings, changes in manufacturing yields, control of costs and expenses, significant litigation, risks associated with acquisitions and dispositions (including the pending acquisition of SANYO Semiconductor and other assets related to SANYO Electric's semiconductor business), risks associated with our substantial leverage and restrictive covenants in our debt agreements, risks associated with our international operations including foreign employment and labor matters associated with unions and collective bargaining arrangements, the threat or occurrence of international armed conflict and terrorist activities both in the United States and internationally, risks and costs associated with regulation of corporate governance and disclosure standards (including pursuant to Section 404 of the Sarbanes-Oxley Act of 2002 ("SOX")), risks

related to new legal requirements and risks involving environmental or other governmental regulation. Additional factors that could affect our future results or events are described under Part I, Item 1A. "Risk Factors" in our 2009 Form 10-K and from time to time in our other Securities and Exchange Commission reports. You should carefully consider the trends, risks and uncertainties described in this Form 10-Q, the 2009 Form 10-K and subsequent reports filed with or furnished to the Securities and Exchange Commission before making any investment decision with respect to our securities. If any of these trends, risks or uncertainties actually occurs or continues, our business, financial condition or operating results could be materially adversely affected, the trading prices of our securities could decline, and you could lose all or part of your investment. Readers are cautioned not to place undue reliance on forward-looking statements. We assume no obligation to update such information. All forward-looking statements attributable to us or persons acting on our behalf are expressly qualified in their entirety by this cautionary statement.

The purchase price for the SANYO Transaction is payable in Yen and is subject to foreign currency exchange rate risk which could potentially increase the actual U.S. dollar value of the purchase price for this acquisition.

The purchase price for the Company's acquisition of SANYO Semiconductor and other assets related to SANYO Electric's semiconductor business (the "SANYO Transaction") is payable in Yen. As previously reported, the purchase price of the SANYO Transaction is approximately \(\frac{\text{33.0}}{33.0} \) billion (approximately \(\frac{\text{336.0}}{36.0} \) million), subject to adjustment at closing. The original estimate of the US dollar amount was based upon an exchange rate of \(\frac{\text{1}}{10.0} \) USD to \(\frac{\text{490.2}}{90.2} \) on July 14, 2010, the date of the announcement of the SANYO Transaction. However, because the purchase price has been priced in Yen the actual US dollar value of the purchase price will be adjusted based on the prevailing exchange rates on the date of closing. On October 1, 2010, the exchange rate was \(\frac{\text{1}}{10.0} \) USD to \(\frac{\text{483.75}}{83.75}, \) which would have increased the expected purchase price by approximately \(\frac{\text{28}}{20.0} \) million had the transaction closed on that date. At this time, we are unable to accurately estimate the exchange rate as of the date of closing, which is subject to factors beyond our control and which may not be foreseeable. Should the US dollar continue to weaken against the Japanese Yen, the U.S. dollar value of the realized purchase price for the SANYO Transaction would materially increase when compared to the estimate of this value as of the date of the announcement of such transaction.

For additional risk factors related to the Company's foreign currency exchange rate risk from international operations, see Part I, Item 1A, "Risk Factors—Trends, Risks and Uncertainties Related to Our Business—Our international operations subject us to risks inherent in doing business on an international level that could adversely impact our results of operations" in our 2009 Form 10-K.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Defaults Upon Senior Securities

None.

Item 4. (Removed and Reserved)

Item 5. Other Information

None.

Item 6.	Exhibits
Exhibit No. 2.1	<u>Description</u> Purchase Agreement by and among ON Semiconductor Corporation, Semiconductor Components Industries, LLC and SANYO Electric Co., Ltd. dated July 15, 2010 (1)(2)(5)
31.1	Certification by CEO pursuant to Rule 13a-14(a) or 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (2)
31.2	Certification by CFO pursuant to Rule 13a-14(a) or 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (2)
32.1	Certification by CEO and CFO pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (3)
101.INS	XBRL Instance Document (3)(4)
101.SCH	XBRL Taxonomy Extension Schema Document (3)(4)
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document (3)(4)
101.LAB	XBRL Taxonomy Extension Label Linkbase Document (3)(4)
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document (3)(4)

Schedules omitted pursuant to Item 601(b)(2) of Regulation S-K under the Securities and Exchange Act of 1934, as amended. The Company agrees to furnish a supplemental copy of any omitted schedule to the SEC upon request.

- Filed herewith.
- Furnished herewith. (3)
- In accordance with Rule 406T of Regulation S-T, the XBRL related information shall not be deemed to be "filed" for purposes of Section 18 of the Securities and Exchange Act of 1934, as amended, or otherwise subject to the liability of that section, and shall not be incorporated by reference into any registration statement or other document filed under the Securities Act of 1933, as amended, except as shall be expressly set forth by specific reference in such filing.
- The purchase agreement filed herewith has been provided solely to inform investors of its terms and does not guarantee that the transaction with SANYO Semiconductor will close when anticipated, or at all, or that it will not close subject to materially different terms than those reflected in the purchase agreement. The representations, warranties and covenants contained in this filed purchase agreement were made only for the purposes of such agreement and as of specific dates, were made solely for the benefit of the parties to the purchase agreement and may be intended not as statements of fact, but rather as a way of allocating risk to one of the parties if those statements prove to be inaccurate. In addition, such representations, warranties and covenants may have been qualified by certain disclosures not reflected in the text of the purchase agreement and may apply standards of materiality in a way that is different from what may be viewed as material by stockholders of, or other investors in, the Company. The Company's stockholders and other investors are not third-party beneficiaries under the purchase agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts of conditions of the Company or any of its respective subsidiaries or affiliates.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: November 4, 2010

ON SEMICONDUCTOR CORPORATION

(Registrant)

By: /s/ DONALD COLVIN

Donald Colvin
Executive Vice President and Chief
Financial Officer (Principal Financial
Officer, Principal Accounting Officer and
officer duly authorized to sign this report)

Exhibit No.	<u>Description</u>
2.1	Purchase Agreement by and among ON Semiconductor Corporation, Semiconductor Components Industries, LLC and SANYO Electric Co., Ltd. dated July 15, 2010 ⁽¹⁾⁽²⁾⁽⁵⁾
31.1	Certification by CEO pursuant to Rule 13a-14(a) or 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 ⁽²⁾
31.2	Certification by CFO pursuant to Rule 13a-14(a) or 15d-14(a) of the Securities Exchange Act of 1934, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 ⁽²⁾
32.1	Certification by CEO and CFO pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002(3)
101.INS	XBRL Instance Document(3)(4)
101.SCH	XBRL Taxonomy Extension Schema Document(3)(4)
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document(3)(4)
101.LAB	XBRL Taxonomy Extension Label Linkbase Document(3)(4)
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document(3)(4)

Schedules omitted pursuant to Item 601(b)(2) of Regulation S-K under the Securities and Exchange Act of 1934, as amended. The Company agrees to furnish a supplemental copy of any omitted schedule to the SEC upon request.

⁽²⁾ Filed herewith.

⁽³⁾ Furnished herewith.

In accordance with Rule 406T of Regulation S-T, the XBRL related information shall not be deemed to be "filed" for purposes of Section 18 of the Securities and Exchange Act of 1934, as amended, or otherwise subject to the liability of that section, and shall not be incorporated by reference into any registration statement or other document filed under the Securities Act of 1933, as amended, except as shall be expressly set forth by specific reference in such filing.

The purchase agreement filed herewith has been provided solely to inform investors of its terms and does not guarantee that the transaction with SANYO Semiconductor will close when anticipated, or at all, or that it will not close subject to materially different terms than those reflected in the purchase agreement. The representations, warranties and covenants contained in this filed purchase agreement were made only for the purposes of such agreement and as of specific dates, were made solely for the benefit of the parties to the purchase agreement and may be intended not as statements of fact, but rather as a way of allocating risk to one of the parties if those statements prove to be inaccurate. In addition, such representations, warranties and covenants may have been qualified by certain disclosures not reflected in the text of the purchase agreement and may apply standards of materiality in a way that is different from what may be viewed as material by stockholders of, or other investors in, the Company. The Company's stockholders and other investors are not third-party beneficiaries under the purchase agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts of conditions of the Company or any of its respective subsidiaries or affiliates.

PURCHASE AGREEMENT

by and among

ON SEMICONDUCTOR CORPORATION,

a Delaware corporation,

SEMICONDUCTOR COMPONENTS INDUSTRIES, LLC, a Delaware limited liability company, and

SANYO ELECTRIC CO., LTD.,

a Japanese corporation

Dated as of July 15, 2010

TABLE OF CONTENTS

ARTICLE I	PURCHASE AND SALE	1
1.1	Agreement to Sell and Purchase	1
1.2	Closing	2
1.3	Purchase Price; Closing Payment; Escrow	2
1.4	Purchase Price Determination and Adjustment	4
1.5	Withholding Rights	10
ARTICLE II	CLOSING DELIVERIES	10
2.1	Deliveries by Seller	10
2.2	Deliveries by Purchaser	11
ARTICLE III	REPRESENTATIONS AND WARRANTIES REGARDING COMPANY	11
3.1	Organization, Good Standing, Qualification	11
3.2	Company Group Entities	11
3.3	Capitalization	12
3.4	No Conflicts; Required Consents	13
3.5	Financial Statements	13
3.6	Absence of Undisclosed Liabilities	14
3.7	Absence of Changes	14
3.8	Ownership and Sufficiency of Business Assets	14
3.9	Accounts Receivable	15
3.10	Indebtedness	16
3.11	Inventory	16
3.12	Material Contracts	16
3.13	Intellectual Property	19
3.14	Employees	23
3.15	Employee Benefits	26
3.16	Compliance with Laws	26
3.17	Governmental Approvals	27
3.18	Proceedings and Orders	27
3.19	Environmental Matters	27
3.20	Taxes	28
3.21	Insurance	30
3.22	Real Property	30
3.23	Personal Property	32
3.24	Customers and Suppliers	32
3.25	Bank Accounts	33
3.26	Company Transaction Expenses	33
3.27	Transactions with Affiliates	34
3.28	Brokers	34
3.29	Product Specifications and Warranties	34
3.30	Product Liability	34

TABLE OF CONTENTS

(continued)

		Page
3.31	Foreign Exchange Notification	34
3.32	Demerger	35
3.33	Service Providers	35
3.34	SSV Union Approval	35
RTICLE IV	REPRESENTATIONS AND WARRANTIES REGARDING SELLER	35
4.1	Ownership of Purchased Shares	35
4.2	Authority; Validity of Contemplated Transactions	35
4.3	Proceedings	36
4.4	Brokers	36
4.5	Securities Act Matters	36
ARTICLE V	REPRESENTATIONS AND WARRANTIES REGARDING PURCHASER AND PURCHASER PARENT	37
5.1	Organization and Good Standing	37
5.2	Authority; Binding Nature of Agreements	37
5.3	Purchaser Parent Capital Structure	37
5.4	No Conflicts; Required Consents	38
5.5	SEC Filings; Financial Statements	38
5.6	Brokers	39
5.7	Proceedings	39
5.8	Non-Reliance of Purchaser and Purchaser Parent	39
RTICLE VI	PRE-CLOSING COVENANTS	40
6.1	Company's Conduct of the Business Prior to the Closing	40
6.2	Restrictions on Company's Conduct of the Business Prior to the Closing	40
6.3	Certain Notifications	42
6.4	Access to Information	43
6.5	Reasonable Best Efforts	43
6.6	Consents	44
6.7	Indebtedness and Intercompany Arrangements	46
6.8	Financial Statements and Periodic Financial Information Seller shall promptly deliver to Purchaser copies of:	46
6.9	Preparation of Pro Forma Audited Financial Statements of Company Group Entities	48
6.10	Conduct Prior to the Closing	48
6.11	Capital Expenditures	48
6.12	Certain Intellectual Property Matters	48
6.13	Updated Seller Disclosure Schedules	49
6.14	Third Party Patent License	49
6.15	Recapitalization of Seller Loan Receivables	49
6.16	July Disclosed Documents	49

TABLE OF CONTENTS (continued)

		Page
6.17	Specified Employees	50
ARTICLE VII	ADDITIONAL AGREEMENTS	50
7.1	Intellectual Property License Agreement	50
7.2	Trademark License Agreement	50
7.3	Transition Services Agreement	50
7.4	IP Assignment Agreement(s)	50
7.5	Leases	50
7.6	Service Provider Agreements	51
7.7	Secondment Agreement	51
7.8	No Solicitation of Competing Transactions	51
7.9	Non-Competition and Non-Solicitation Agreement	52
7.10	Confidentiality	54
7.11	Public Announcements	55
7.12	Current Public Information	55
7.13	Purchaser Parent's Obligations	55
7.14	Benefit of Business Assets	56
7.15	Financial Arrangements.	56
7.16	Transaction with Seller Group Entities	57
7.17	Net Operating Losses	57
7.18	Reorganization	57
7.19	Hazardous Material Waste	58
7.20	SNA Assignment and Assumption Agreement	58
7.21	SCE Assignment and Assumption Agreement	58
7.22	Transfer of SSMPE Interest	58
7.23	SLTI	58
RTICLE VIII	EMPLOYEE MATTERS	59
8.1	Employee Matters	59
8.2	Restructuring Plan	59
ARTICLE IX	CONDITIONS TO CLOSING	59
9.1	Conditions to Purchaser's Obligation to Close	59
9.2	Conditions to Fulchaser's Obligation to Close	63
9.3	Conditions to Obligations of Each Party to Close	64
ARTICLE X	POST-CLOSING OBLIGATIONS	64
10.1	Accounting and Tax	64
10.2	Environment	64
ARTICLE XI	TERMINATION	65
11.1	Circumstances for Termination	65
11.2	Effect of Termination	65

TABLE OF CONTENTS

(continued)

		Page
ARTICLE XII	INDEMNIFICATION	65
12.1	Survival of Representations, Warranties and Special Indemnities	65
12.2	Indemnification by Seller	66
12.3	Indemnification by Purchaser	69
12.4	Procedures for Indemnification	70
12.5	Limitations on Indemnification	71
12.6	Remedies Exclusive	73
12.7	Nature of Indemnification Payments	73
12.8	Offset	73
ARTICLE XIII	TAX MATTERS	74
13.1	Preparation and Filing of Tax Returns	74
13.2	Amended Returns; Tax Elections	74
13.3	Transfer Taxes	74
13.4	Straddle Period Allocations	74
13.5	Termination of Tax Sharing Agreements	75
13.6	Cooperation, Access to Information, and Record Retention	75
13.7	Tax Refund	75
ARTICLE XIV	ISSUED SHARES	76
14.1	Legends and Transfer of Issued Shares	76
14.2	Ownership of Issued Shares	76
14.3	Purchaser Parent's Right of First Offer	76
ARTICLE XV	MISCELLANEOUS PROVISIONS	78
15.1	Expenses	78
15.2	Interpretation	78
15.3	Further Assurances	78
15.4	Notices	79
15.5	Entire Agreement	80
15.6	Modifications, Amendments and Waivers	80
15.7	Successors and Assigns	80
15.8	Governing Law	80
15.9	Severability	81
15.10	Specific Performance	81
15.11	Counterparts	81
15.12	Dispute Resolution	81
15.13	Attorneys' Fees, Costs and Expenses	81
15.14	No Third-Party Beneficiaries	81

TABLE OF CONTENTS

(continued)

EXHIBITS, SCHEDULES AND ANNEXES

Exhibits

Exhibit A: Certain Defined Terms Exhibit A-1

v

Page

PURCHASE AGREEMENT

THIS PURCHASE AGREEMENT is made and entered into this 15th day of July, 2010, by and among **ON Semiconductor Corporation**, a Delaware corporation (the "*Purchaser Parent*"), Semiconductor Components Industries, LLC, a Delaware limited liability company ("*Purchaser*") and SANYO Electric Co., Ltd., a Japanese corporation (the "*Seller*"). Each of the Purchaser, Purchaser Parent and Seller is referred to as a "*party*" and collectively as the "*parties*." Certain capitalized terms used in this Agreement are defined on Exhibit A.

RECITALS

- WHEREAS, SANYO Semiconductor Co., Ltd. ("Company"), together with the other Company Group Entities, is engaged in the Business;
- WHEREAS, Seller owns all of the Purchased Shares, and the Purchased Shares evidence all of the ownership rights to, and equity interests in, Company;
- WHEREAS, Company owns all of the equity interests of the Company Group Entities;
- WHEREAS, Seller and certain of the Seller Group Entities own the Seller Loan Receivables, the Transferred IP Assets, the Transferred SNA Business and the Transferred SCE Business:
 - WHEREAS, Purchaser Parent, together with its affiliates, is engaged in the semiconductor business;
 - WHEREAS, Purchaser is a wholly-owned subsidiary of Purchaser Parent, and
- **WHEREAS**, the parties desire to enter into this Agreement pursuant to which Purchaser (or one or more designated Affiliates of Purchaser) will acquire the Purchased Business from Seller and certain other Seller Group Entities, upon the terms and subject to the conditions set forth herein.
- **NOW, THEREFORE**, in consideration of the foregoing recitals and the mutual representations, warranties, covenants and promises contained herein, the adequacy and sufficiency of which are hereby acknowledged, the parties agree as follows:

AGREEMENT

ARTICLE I

PURCHASE AND SALE

- 1.1 **Agreement to Sell and Purchase**. For the consideration hereinafter provided and upon the terms and subject to the conditions of this Agreement, at the Closing, Seller shall, and shall cause the other Seller Group Entities to, sell, assign, transfer, convey and deliver to Purchaser (or one or more designated Affiliates of Purchaser), free and clear of all Encumbrances (other than licenses in the case of Transferred IP Assets), and Purchaser (or one or more Affiliates of Purchaser) shall purchase and acquire from Seller and the other Seller Group Entities the following (collectively, the "**Purchased Business**"):
 - (a) the Purchased Shares,

- (b) the Seller Loan Receivables,
- (c) the Transferred IP Assets,
- (d) the Transferred SNA Business,
- (e) the Transferred SCE Business, and
- (f) the SSMPE Ownership Interests.
- 1.2 **Closing**. The closing of the Transaction (the "*Closing*") shall take place at the offices of Morrison & Foerster LLP, Shin Marunouchi Building 29th Floor, Marunouchi 1-5-1, Chiyoda-ku, Tokyo, 105-6529, at 10:00 A.M. on the later of (i) November 1, 2010 and (ii) the first (1st) Business Day of the month immediately following the month in which all of the conditions to closing set forth in Article IX are satisfied or waived (other than conditions that are intended to be satisfied at the Closing), or at such other date, time or place as Purchaser and Seller may agree (the "*Closing Date*").
 - 1.3 Purchase Price; Closing Payment; Escrow.
- (a) <u>Purchase Price</u>. The aggregate purchase price to be paid by Purchaser (or one or more designated Affiliates of Purchaser) to Seller for the Purchased Business (the "*Purchase Price*") shall equal:
 - (i) Agreed Enterprise Value,
 - (ii) plus Closing Cash and Cash Equivalents,
 - (iii) less Closing Outstanding Bank Debt,
 - (iv) less Closing Factoring Liabilities,
 - (v) less Closing Assumed Unfunded Benefit Liabilities, and
 - (vi) <u>less</u> Net Company Portion Seller Restructuring Plan Costs.

The Purchase Price shall be initially calculated, and shall be subject to adjustment and offset, as provided in Section 1.4.

(b) Appraisal and Allocation.

- (i) <u>Appraisal</u>. Promptly after the date of this Agreement, Seller and Purchaser shall engage an independent appraisal firm of international reputation mutually acceptable to Seller and Purchaser to determine the fair market value of the Transferred IP Assets (the "IP Appraisal") and shall cause such appraisal firm to deliver the IP Appraisal to each of Seller and Purchaser no later than sixty (60) days after the date of this Agreement.
- (ii) <u>Allocation</u>. Seller and Purchaser agree that Purchaser shall prepare and provide to Seller a draft allocation ("**Allocation**") of the Purchase Price among the Purchased Business no later than thirty (30) days prior to Closing, and that the portion of the Purchase Price allocated to the Transferred IP Assets shall be equal to the IP Appraisal set forth in <u>Section 1.3(b)(i)</u>. Seller shall notify Purchaser within ten (10) days of receipt of such draft Allocation of any objection Seller may have thereto. Seller and Purchaser agree to resolve any disagreement with respect to such Allocation in good faith. Seller and Purchaser hereby undertake and agree to use the Allocation determined pursuant to this <u>Section 1.3(b)</u> to report, act and file Tax Returns (including Internal Revenue Service Form 8594 and any amendments thereto).
 - (c) <u>Closing Payments to Seller</u>. At the Closing, Purchaser shall pay to Seller:
- (i) a portion of the Purchase Price in an amount (such amount, the "Closing Cash Payment") equal to (1) eleven billion six hundred million Yen (\pmu11,600,000,000) (the "Cash Consideration"), (2) minus the Escrow Amount, (3) minus the Closing Cash and Cash Equivalents Shortfall, by wire transfer of immediately available funds to the bank account designated by Seller no less than ten (10) Business Days prior to the Closing (the "Seller Account"), and
- (ii) a portion of the Purchase Price in an aggregate amount (the "Stock Consideration") equal to (1) the Purchase Price (as determined in accordance with Section 1.4(a)); (2) minus the Cash Consideration, (3) plus or minus, as the case may be, the Estimated Closing Net Working Capital Adjustment Amount (as determined in accordance with Section 1.4(a)(vi)), (4) plus the Closing Cash and Cash Equivalents Shortfall, to be paid by delivering to Seller such number of unregistered shares of the common stock of Purchaser Parent, par value \$0.01 per share (the "Purchaser Parent Shares"), that shall be determined by dividing the Stock Consideration by the Purchaser Parent Share Price (the quotient thereof, rounded down to the nearest whole number, the "Issued Shares"); provided, however, that Purchaser shall have the option (the "Additional Cash Option") to elect to pay any or all of the amount of the Stock Consideration in cash (such additional cash amount, the "Additional Cash Amount") and the Stock Consideration shall be deemed decreased by the Additional Cash Amount.
- (d) <u>Escrow Amount</u>. At the Closing and as partial security for Seller's indemnification obligations set forth in <u>Article XII</u>, Purchaser shall deliver four billion three hundred million Yen (¥4,300,000,000) (the "*Escrow Amount*") to the Escrow Agent for deposit into an escrow account (the "*Escrow Account*") established and maintained pursuant to the terms of the Escrow Agreement and <u>Article XII</u>.

1.4 Purchase Price Determination and Adjustment.

(a) Closing Determination.

- (i) No later than ten (10) Business Days prior to the Closing Date, Seller shall prepare and deliver to Purchaser for its review, and Seller and Purchaser shall thereafter mutually agree in good faith on, a statement setting forth a good faith estimate of (i) the Closing Net Working Capital ("Estimated Closing Net Working Capital"); (ii) Closing Cash and Cash Equivalents ("Estimated Closing Cash and Cash Equivalents"); (iii) the Closing Outstanding Bank Debt ("Estimated Closing Outstanding Bank Debt"); (iv) the Closing Factoring Liabilities ("Estimated Closing Factoring Liabilities"); and (v) the Closing Assumed Unfunded Benefit Liabilities"). In estimating the Closing Assumed Unfunded Benefit Liabilities, Seller shall use the formula set forth in Annex I.
- (ii) No later than fifteen (15) Business Days prior to the Closing Date, Seller shall prepare and deliver to Purchaser for its review, and Seller and Purchaser shall thereafter agree in good faith on, a statement setting forth a good faith estimate of: (w) the number of employees of the Seller Group Entities who will transfer to a Company Group Entity as of the Closing in accordance with <u>Annex III</u> (but excluding any Seconded Employees to be seconded after Closing) and the Unfunded Benefit Liabilities as at the Closing Adjustment Base Date associated with such employees calculated in accordance with <u>Annex I</u>, (x) the Unfunded Benefit Liabilities as at the Closing Adjustment Base Date associated with the Target Number of Transferring Gifu and Gunma Employee calculated in accordance with <u>Annex I</u>; and (y) the Unfunded Benefit Liabilities as at the Closing Adjustment Base Date associated with the Target Number of Transferring KSS Employee calculated in accordance with <u>Annex I</u> (the aggregate of such estimated amounts, the "Estimated Transferred Unfunded Benefit Liabilities"), each with a breakdown in reasonable detail.
- (iii) If Seller and Purchaser cannot agree on (u) Estimated Closing Net Working Capital; (v) Estimated Closing Cash and Cash Equivalents; (w) Estimated Closing Outstanding Bank Debt; (x) Estimated Closing Factoring Liabilities; (y) Estimated Closing Assumed Unfunded Benefit Liabilities or (z) Estimated Transferred Unfunded Benefit Liabilities, then the amount of such unagreed item shall be as set forth in <u>Annex II</u> (Illustrative Purchase Price Calculation).
 - (iv) The parties acknowledge and agree that:
- (1) in consideration of the Company's assumption of the Transferred Unfunded Benefit Liabilities associated with the Transferred Employees, Seller shall be obligated to pay Company an amount equal to the Transferred Unfunded Benefit Liabilities associated with the Transferred Employees and agree that Purchaser is to receive such amount from Seller on behalf of Company and such obligation shall be deemed to be performed and satisfied by offsetting such amount against the Stock Consideration and the making of any required payment pursuant to Section 1.4(1);
- (2) in consideration of New SSMC's assumption of the Transferred Unfunded Benefit Liabilities associated with the Transferred Gifu and Gunma Employees upon their transfer, SSMC shall owe to New SSMC the obligation to pay the amount equal to the Transferred Unfunded Benefit Liabilities associated with the Transferred Gifu and Gunma Employees and agree that (i) Seller on behalf of SSMC is to pay to Purchaser, and Purchaser on behalf of New SSMC is to receive from Seller, such amount, and (ii) such payment obligation shall be deemed to be performed and satisfied by offsetting such amount against the Stock Consideration and the making of any required payment pursuant to Section 1.4(m); and

(3) in consideration of KSS's assumption of the Transferred Unfunded Benefit Liabilities associated with the Transferred KSS Employees upon their transfer, New KSS shall owe to KSS the obligation to pay the amount equal to the Transferred Unfunded Benefit Liabilities associated with the Transferred KSS Employees and agree that (i) Seller on behalf of New KSS is to pay to Purchaser, and Purchaser on behalf of KSS is to receive from Seller, such amount, and (ii) such payment obligation shall be deemed to be performed and satisfied by offsetting such amount against the Stock Consideration and the making of any required payment pursuant to Section 1.4(m).

(v) Purchaser's obligation to pay the Stock Consideration at the Closing shall accordingly be reduced in part on a Yen to Yen basis by the aggregate amount of the Estimated Transferred Unfunded Benefit Liabilities.

(vi) The parties agree that the Purchase Price was based on a targeted Closing Net Working Capital equal to the Closing Net Working Capital Target. The Stock Consideration shall be (i) increased, Yen for Yen, by the amount by which the Estimated Closing Net Working Capital is greater than Closing Net Working Capital Target or (ii) decreased, Yen for Yen, by the amount by which the Estimated Closing Net Working Capital is less than the Closing Net Working Capital Target (the difference, positive or negative, between the Estimated Closing Net Working Capital and the Closing Net Working Capital Target, the "Estimated Closing Net Working Capital Adjustment Amount").

(vii) Net Company Portion Seller Restructuring Plan Costs. No later than ten (10) Business Days prior to the Closing Date, Seller shall deliver to Purchaser a statement setting forth the amount of the Company Portion Seller Restructuring Plan Costs, less the lesser of (x) the amount of any such Company Portion Seller Restructuring Plan Costs that any Seller Group Entity or any Company Group Entity has actually paid or incurred prior to the Closing (for the purpose of preparing the Closing Balance Sheet, any such costs incurred shall be included as a "current liability" of Company Group Entity in the Closing Balance Sheet) (the "Seller Incurred Restructuring Plan Costs") and (y) the amount of such Company Portion Seller Restructuring Plan Costs as set forth in the Seller Restructuring Plan Costs Schedule (such net amount, the "Net Company Portion Seller Restructuring Plan Costs"); provided, however, that Seller shall obtain Purchaser's consent (such consent not to be unreasonably withheld) prior to any Seller Group Entity or Company Group Entity paying or incurring any such Company Portion Seller Restructuring Plan Costs that were not contemplated to be paid or incurred by any Seller Group Entity or Company Group Entity prior to the Closing in the Seller Restructuring Plan Costs Schedule and, thereafter, shall maintain records, reasonably satisfactory to Purchaser, of the amounts expended for which credit will be sought hereunder. Seller shall keep Purchaser apprised on a periodic basis of the amounts expended hereunder.

(viii) <u>Additional Cash Option</u>. For purposes of the adjustments set forth in <u>Section 1.4(a)(iv)</u>, (v) and (vi), if Purchaser elects to exercise the Additional Cash Option, then to the extent the amount of offset to reduce the Stock Consideration pursuant to <u>Section 1.4(a)(iv)</u>, (v) and/or (vi) exceeds the Stock Consideration, the Stock Consideration shall be deemed zero and such excess amount shall be offset against the Additional Cash Amount.

(b) Closing Balance Sheet and Closing Statement. As promptly as practicable, but in no event later than one hundred twenty (120) days following the Closing Date, Purchaser shall cause the following to be prepared and delivered to Seller (collectively, the "Closing Balance Sheet"): (i) a consolidated balance sheet of the Company Group Entities as of the Closing Adjustment Base Date, prepared in accordance with J-GAAP on a basis consistent with the Unaudited Financial Statements; and (ii) a statement (the "Closing Statement") based on such Closing Balance Sheet which sets forth in detail a calculation of the Closing Net Working Capital, the Closing Cash and Cash Equivalents, the Closing Outstanding Bank Debt, and the Closing Factoring Liabilities. Purchaser shall provide Seller the relevant work papers used in the preparation of the Closing Balance Sheet. In connection with the preparation of the Closing Date Balance Sheet, unless Purchaser decides otherwise, as of the Closing Date Seller shall cause Company to take a physical inventory of all product stocks, other than those on consignment, which Purchaser's and Seller's accountants shall be permitted to observe. Any obsolete or damaged inventory shall not be included as inventory for purposes of the Closing Net Working Capital determination. Upon the request of Purchaser, Seller shall provide, and shall cause the other Seller Group Entities to provide, all reasonable assistance in connection with the preparation of the Closing Balance Sheet. Except as set forth below, the Closing Balance Sheet and the Closing Statement shall be deemed to be and shall be final, binding and conclusive on the parties for all purposes of determining the purchase prices (except for the Closing Assumed Unfunded Benefit Liabilities) and the payments required pursuant to this Section 1.4 (other than the payment pursuant to Section 1.4(1) and Section 1.4(m)) upon the earlier of (i) Seller's delivery of a written notice to Purchaser of its approval of the Closing Balance Sheet and the Closing Statement, (ii) the failure of Seller to deliver to the Purchaser, within sixty (60) days after the delivery of such documents to the Seller, a written notice, specifying that it is provided hereunder and setting forth in reasonable detail the component or components of the Closing Balance Sheet and/or Closing Statement which the Seller disputes, and the basis therefor (each a "Closing Balance Sheet Dispute Notice"), (iii) the resolution of all disputes, pursuant to Section 1.4(c), by Purchaser and Seller, and (iv) the resolution of all disputes, pursuant to Section 1.4(c), by the Independent Accounting Firm (such earlier date, the "Final Resolution Date").

(c) <u>Dispute</u>. If Seller delivers a Closing Balance Sheet Dispute Notice to Purchaser, Purchaser and Seller shall attempt to reconcile the parties' differences set forth in the Closing Balance Sheet Dispute Notice, and any resolution by them as to any disputed amounts set forth in the Closing Balance Sheet Dispute Notice shall be final, binding and conclusive on the parties for all purposes of determining the purchase prices and the payments required pursuant to this <u>Section 1.4</u>. If Purchaser and Seller are unable to reach a resolution within thirty (30) days after the delivery of the Closing Balance Sheet Dispute Notice, Purchaser and Seller shall submit their respective determinations and calculations and the items set forth in the Closing Balance Sheet Dispute Notice remaining in dispute for resolution to PricewaterhouseCoopers Aarata or another independent accounting firm of international reputation mutually acceptable to Purchaser and Seller (the "Independent Accounting Firm"). The parties shall cause the Independent Accounting Firm to submit a report to Purchaser and Seller with a determination regarding such remaining disputed items, within thirty (30) days after submission of the matter, and such report shall be final, binding and conclusive on Purchaser and Seller. In resolving any matters in dispute, the Independent Accounting Firm may only assign a value to any item in dispute within the range of the greatest value and the smallest value for such matters assigned by Purchaser or Seller. The Independent Accounting Firm's determination will be based solely on presentations by Purchaser, Seller and their respective advisors and the responses of Purchaser, Seller and their respective advisors to questions to the Independent Accounting Firm in connection with such review, and shall not involve the Independent Accounting Firm in resolving the remaining disputed items shall be borne equally by Purchaser and Seller.

- (d) Final Closing Net Working Capital. The Purchase Price shall be increased on a Yen for Yen basis by the amount by which the actual Closing Net Working Capital of the Company Group Entities finally determined in accordance with Sections 1.4(b) and 1.4(c) (the "Actual Closing Net Working Capital") exceeds the amount of the Estimated Closing Net Working Capital determined in accordance with Section 1.4(a); provided, however, that if the Actual Closing Net Working Capital exceeds the Estimated Closing Net Working Capital by one billion five hundred million yen (\xi1,500,000,000) or less, the Actual Closing Net Working Capital shall be deemed to equal the Estimated Closing Net Working Capital. The Purchase Price shall be decreased on a Yen for Yen basis by the amount by which the amount of the Estimated Closing Net Working Capital exceeds the Actual Closing Net Working Capital by one billion five hundred million yen (\xi1,500,000,000) or less, the Actual Closing Net Working Capital shall be deemed to equal the Estimated Closing Net Working Capital. Any such adjustment to the Purchase Price shall be referred to as a "Closing Net Working Capital Adjustment."
- (e) <u>Final Closing Cash and Cash Equivalents</u>. The Purchase Price shall be increased or decreased on a Yen for Yen basis by the amount by which the actual Closing Cash and Cash Equivalents finally determined in accordance with <u>Sections 1.4(b)</u> and <u>1.4(c)</u> is greater or less than the amount of the Estimated Closing Cash and Cash Equivalents determined in accordance with <u>Section 1.4(a)</u>.
- (f) <u>Final Closing Outstanding Bank Debt</u>. The Purchase Price shall be increased or decreased on a Yen for Yen basis by the amount by which the actual Closing Outstanding Bank Debt finally determined in accordance with <u>Sections 1.4(b)</u> and <u>1.4(c)</u> is less or greater than the amount of the Estimated Closing Outstanding Bank Debt determined in accordance with <u>Section 1.4(a)</u>.
- (g) <u>Final Closing Factoring Liabilities</u>. The Purchase Price shall be increased or decreased on a Yen for Yen basis by the amount by which the actual Closing Factoring Liabilities finally determined in accordance with <u>Sections 1.4(b)</u> and <u>1.4(c)</u> is less or greater than the amount of the Estimated Closing Factoring Liabilities determined in accordance with <u>Section 1.4(a)</u>.

- (h) <u>Purchase Price Adjustment Amount</u>. The amount of the increases or decreases to be made to the Purchase Price pursuant to <u>Sections 1.4(d)</u>, (e), (f) and (g) shall be netted together as one amount (such net payment amount, the "*Purchase Price Adjustment Amount*"). If the Purchase Price is to be increased by the Purchase Price Adjustment Amount, Purchaser shall pay an aggregate amount of cash equal to the Purchase Price Adjustment Amount to Seller by wire transfer of immediately available funds within five (5) Business Days of the Final Resolution Date. If the Purchaser by wire transfer of immediately available funds within five (5) Business Days of the Final Resolution Date.
- (i) Actuary Determination of Closing Assumed Unfunded Benefit Liabilities and Transferred Unfunded Benefit Liabilities Associated with the Transferred Employees. Within five (5) Business Days following the Closing Date, Seller and Purchaser shall retain Sumitomo Life Insurance Company, the pension actuary of Seller's corporate pension fund (the "Actuary"), to calculate (i) the Closing Assumed Unfunded Benefit Liabilities, and (ii) the Transferred Unfunded Benefit Liabilities associated with the Transferred Employees as of the Closing Adjustment Base Date (the statement setting forth the foregoing amounts, an "Actuary Statement"). The costs and fees payable to the Actuary shall be borne equally by Seller and Purchaser.
- (j) <u>Actuary Determination of Certain Transferred Unfunded Benefit Liabilities</u>. Within five (5) Business Days following the SSMC Transfer Date, Seller and Purchaser shall retain the Actuary to calculate (i) the Transferred Unfunded Benefit Liabilities associated with the Transferred Gifu and Gunma Employees as of the date immediately preceding the SSMC Transfer Date and (ii) the Transferred Unfunded Benefit Liabilities associated with the Transferred KSS Employees as of the date immediately preceding the KSS Transfer Date, and set forth such calculated amounts in an Actuary Statement prepared by the Actuary. The costs and fees payable to the Actuary shall be borne equally by Seller and Purchaser.
- (k) <u>Dispute of Actuary Statement</u>. If Purchaser disputes any amount set forth in an Actuary Statement, then within thirty (30) days after Purchaser's receipt of the Actuary Statement, Purchaser shall deliver to Seller a written notice setting forth in reasonable detail the amount or amounts which Purchaser disputes and the basis therefor (the "Actuary Statement Dispute Notice"). If Purchaser delivers an Actuary Statement Dispute Notice to Seller, Purchaser and Seller shall attempt to reconcile the parties' differences set forth in the Actuary Statement Dispute Notice, and any resolution by them as to any disputed amounts set forth in the Actuary Statement Dispute Notice shall be final, binding and conclusive on the parties for all purposes of determining the purchase prices and the payments required pursuant to this Section 1.4(k). If Purchaser and Seller are unable to reach a resolution within twenty (20) days after the delivery of the Actuary Statement Dispute Notice, Purchaser and Seller shall jointly retain another independent third party pension appraiser of international reputation mutually and reasonably acceptable to Purchaser and Seller (the "Independent Actuary"). The parties shall cause the Independent Actuary to submit a report to Purchaser and Seller with a determination regarding such disputed items, and such determination shall be final, binding and conclusive on Purchaser and Seller. The fees, costs and expenses of the Independent Actuary in resolving the disputed items shall be borne equally by Purchaser and Seller.

(l) Final Closing Assumed Unfunded Benefit Liabilities and Certain Transferred Unfunded Benefit Liabilities. If the sum of the actual Closing Assumed Unfunded Benefit Liabilities and the actual Transferred Unfunded Benefit Liabilities associated with the Transferred Employees as of the Closing Adjustment Base Date, each as finally determined in accordance with Sections 1.4(i) and (k), is less than the sum of the Estimated Closing Assumed Unfunded Benefit Liabilities and the Estimated Transferred Unfunded Benefit Liabilities associated with the Transferred Employees as of the Closing Adjustment Base Date, Purchaser shall pay the amount equal to such difference to Seller by wire transfer of immediately available funds to the Seller's Account within five (5) Business Days of the date on which the Actuary, or Independent Actuary, as applicable, submits the result of calculation to Seller and Purchaser. If the sum of the actual Closing Assumed Unfunded Benefit Liabilities and the actual Transferred Unfunded Benefit Liabilities associated with the Transferred Employees as of the Closing Adjustment Base Date, each as finally determined in accordance with Sections 1.4(i) and (k), is greater than the sum of the Estimated Closing Assumed Unfunded Benefit Liabilities and the Estimated Transferred Unfunded Benefit Liabilities associated with the Transferred Employees as of the Closing Adjustment Base Date, Seller shall pay the amount equal to such difference to Purchaser by wire transfer of immediately available funds to the bank account designated by Purchaser within five (5) Business Days of the date on which the Actuary, or Independent Actuary, as applicable, submits the result of calculation to Seller and Purchaser.

(m) Final Transferred Unfunded Benefit Liabilities. If the sum of the actual Transferred Unfunded Benefit Liabilities associated with the Transferred Gifu and Gunma Employees as of the date immediately preceding the SSMC Transfer Date and the actual Transferred Unfunded Benefit Liabilities associated with the Transferred KSS Employees as of the date immediately preceding the KSS Transfer Date, each as finally determined in accordance with Sections 1.4(j) and (k), is less than the sum of the Estimated Transferred Unfunded Benefit Liabilities associated with the Target Number of Transferring Gifu and Gunma Employees as of the Closing Adjustment Base Date and the Estimated Transferred Unfunded Benefit Liabilities associated with the Target Number of Transferring KSS Employees as of the Closing Adjustment Base Date, Purchaser shall pay the amount equal to such difference to Seller by wire transfer of immediately available funds to the Seller's Account within five (5) Business Days of the date on which the Actuary or Independent Actuary, as applicable, submits the result of calculation to Seller and Purchaser. If the sum of the actual Transferred Unfunded Benefit Liabilities associated with the Transferred Gifu and Gunma Employees as of the date immediately preceding the SSMC Transfer Date and the actual Transferred Unfunded Benefit Liabilities associated with the Transferred KSS Employees as of the date immediately preceding the KSS Transfer Date, each as finally determined in accordance with Sections 1.4(j) and (k), is greater than the sum of the Estimated Transferred Unfunded Benefit Liabilities associated with the Target Number of Transferring Gifu and Gunma Employees as of the Closing Adjustment Base Date and the Estimated Transferred Unfunded Benefit Liabilities associated with the Target Number of Transferring KSS Employees as of the Closing Adjustment Base Date and the Estimated Transferred Unfunded Benefit Liabilities associated with the Target Number of Transferring KSS Employees as of the Closing Adjustme

- (n) If the amount of cash or cash equivalents of Company Group Entities decreases, or the amount of the current liabilities of Company Group Entities increases, due to the payment or accrual of any costs and expenses prior to Closing that are to be borne by Purchaser or any of its Affiliates (including the Company Group Entities after the Closing) under this Agreement or any of the Transaction Agreements, and Purchaser has agreed in advance in writing to such payment or incurrence of costs and expenses prior to Closing, such decrease or increase, as the case may be, shall be deemed to have not occurred and shall be disregarded for the purpose of the determination of and adjustment to the Purchase Price set forth in Section 1.4. If the amount of any cost arising out of the Restructuring Plan that will be incurred by any of the Company Group Entities after the Closing shall be reserved on the Closing Balance Sheet as a current liability, such reserved amount shall be disregarded for the purpose of the determination of and adjustment to the Purchase Price set forth in Section 1.4.
- (o) If Purchaser elects the Additional Cash Option, Purchaser shall reimburse Seller for fifty percent (50%) of costs incurred by Seller as a result of preparing the Pro Forma Audited Financial Statements and the Interim Financial Statements in accordance with U.S. GAAP, U.S. GAAS and SAS 100 pursuant to Section 6.8 and Section 6.9, provided that in no event shall such amount of reimbursement by Purchaser, in the aggregate, exceed one hundred fifty million Yen (¥150,000,000).
- 1.5 **Withholding Rights**. Each of Purchaser and Seller shall be entitled to deduct and withhold from any amounts payable pursuant to this Agreement such amounts as are required to be deducted and withheld under any applicable Legal Requirement; *provided*, *however*, that, if any withholding Tax is imposed on the payment of the Purchase Price allocated to the Transferred IP Assets under applicable Legal Requirement in the jurisdiction from which the purchase of the Transferred IP Assets is made or in the jurisdiction of organization of such purchaser, the Purchase Price shall be grossed up by the amount equal to such withholding Tax so that the Seller will receive such amount as it would receive if such withholding Tax were not imposed. To the extent such amounts are so deducted or withheld, such amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made and shall be timely remitted by the withholding party to the appropriate Governmental Authority in compliance with all applicable Legal Requirements.

ARTICLE II

CLOSING DELIVERIES

- 2.1 Deliveries by Seller. At or prior to the Closing, Seller shall deliver or cause to be delivered to Purchaser:
- (a) <u>Shareholders' Register Request</u>. A request form in substantially the form attached hereto as <u>Exhibit C</u> ("Shareholders' Register Request"), duly executed by Seller affixing Seller's seal, to register Purchaser or a designated Affiliate of Purchaser as the holder of the Purchased Shares on the Company's shareholders' register (kabunushi-meibo); and
- (b) <u>Article 122 Document</u>. The document as provided for under Article 122 of the Companies Act of Japan (*Kaisha-hou*) and issued immediately upon submission of the Shareholders' Register Request to Company, which sets forth information included in Company's shareholders' register, certified by the Representative Director of Company.

- 2.2 **Deliveries by Purchaser**. At or prior to the Closing, Purchaser shall deliver or cause to be delivered to Seller:
- (a) <u>Issued Shares</u>. A certificate or certificates representing the Issued Shares duly endorsed in blank or transfer to Seller or accompanied by stock powers duly executed in blank; and
- (b) Wire Transfers. Wire transfers to Seller's Account in the amount of the Closing Cash Payment and to the Escrow Account in the amount equal to the Escrow Amount.

ARTICLE III

REPRESENTATIONS AND WARRANTIES REGARDING COMPANY

Except as set forth in the Seller Disclosure Schedule, Seller represents and warrants to Purchaser as follows:

3.1 **Organization, Good Standing, Qualification**. Schedule 3.1 of the Seller Disclosure Schedule sets forth, for each Company Group Entity, the jurisdiction of organization and each state or other jurisdiction in which each Company Group Entity is qualified to do business. Each Company Group Entity is or has (i) an entity duly organized, validly existing and in good standing (to the extent such a concept exists in the relevant jurisdiction) under the laws of its jurisdiction of organization; (ii) duly qualified to conduct business under the laws of each jurisdiction in which the nature of its business, the operation of its assets or the ownership or leasing of its properties requires such qualification; and (iii) full corporate power and authority required to own, lease and operate its assets and to carry on the Business.

3.2 Company Group Entities.

- (a) **Ownership of Company Group Entities**. The Company owns, directly or indirectly, all equity interests in each of the other Company Group Entities. Schedule 3.2(a) of the Seller Disclosure Schedule sets forth, with respect to each Company Group Entity, the authorized, issued and outstanding capital stock of each Company Group Entity, specifying: (i) the number of shares or other interests issued or outstanding; and (ii) the record owners of the shares or other interests, showing the number of shares or other interests held by each such owner. Except pursuant to this Agreement, neither Seller nor any Subsidiary thereof, nor, to the Knowledge of Seller, any other record owner of any shares or other interest in any Company Group Entity, has entered into any agreement to sell, transfer or otherwise dispose of any such shares or other interests in any Company Group Entity.
- (b) **Charter Documents**. For each Company Group Entity, Company has delivered or made available to Purchaser correct and complete copies of (i) the Charter Documents and (ii) the capital stock records or other similar records of equity interests, including the capital stock ledger or similar ledger. No Company Group Entity is in violation in any material respect of any of the provisions of its Charter Documents or of any of the resolutions of its stockholders, boards of directors or committees thereof, or any similar bodies, and, to the Knowledge of Seller, no condition or circumstance exists that is reasonably likely to (with or without notice or lapse of time) constitute or result in such a material violation.

- (c) Capital Stock. All issued and outstanding shares of capital stock of, or other equity interests in, the Company Group Entities have been duly authorized and validly issued, are fully paid and non-assessable, have been issued in compliance with Legal Requirements, Orders and Governmental Approvals in all material respects and are free and clear of all Encumbrances. In respect of each Company Group Entity, there are no (i) outstanding preemptive rights, subscriptions, options, calls, warrants or other rights or Company Contracts to acquire any of its securities from each such Company Group Entity or from any other Company Group Entity, Seller Group Entity or any other Person; (ii) outstanding securities, instruments or obligations issued or granted by any Company Group Entity or Seller Group Entity that are or may become convertible into or exchangeable for any Company Group Entity's securities; (iii) Company Contracts under which any Company Group Entity is or may become obligated to sell, issue or otherwise dispose of or redeem, purchase or otherwise acquire any of its securities; or (iv) stockholder agreements, voting trusts or other Contracts that may affect the exercise of voting or any other rights with respect to a Company Group Entity's capital stock to which any Company Group Entity or Seller Group Entity is a party.
- (d) **Business**. No Company Group Entity has engaged in any material respect in any business or business activity other than the Business or held any material business interests or assets other than as used in the Business.

3.3 Capitalization.

- (a) The authorized capital stock of Company consists of eight hundred thousand (800,000) shares of capital stock. Company has issued and outstanding five hundred thousand (500,000) shares of Common Stock.
- (b) <u>Schedule 3.3(b)</u> of the Seller Disclosure Schedule sets forth the name of each holder of capital stock of Company and the number and class of shares held by such holder. No other shares of capital stock of Company are issued or outstanding. All of the Purchased Shares are beneficially owned and held of record by Seller and have been duly authorized and validly issued, are fully paid and non-assessable, have been issued in full compliance with all applicable securities laws and other applicable Legal Requirements and Orders and are free and clear of all Encumbrances. Pursuant to this Agreement, Purchaser shall acquire all legal and beneficial right, title and interest in and to the Purchased Shares, free and clear of all Encumbrances.
- (c) There are no (i) outstanding preemptive rights, subscription, option, call, warrant or other rights of any kind or nature to acquire any securities of Company; (ii) outstanding security, instrument or obligation that is or may become convertible into or exchangeable for any securities of Company; (iii) Company Contracts under which Company is or may become obligated to sell, issue or otherwise dispose of or redeem, purchase or otherwise acquire any its securities; or (iv) shareholder agreements, voting trusts or other agreement, arrangement or understanding to which any Company or Seller is a party that may affect the exercise of voting or any other rights with respect to the capital stock of Company.

- 3.4 **No Conflicts; Required Consents**. The execution, delivery and performance by Seller or any Seller Group Entity of any Transaction Agreement to which it is a party does not and will not (with or without notice or lapse of time):
- (a) conflict with, violate or result in any breach of (i) any of the provisions of the Charter Documents of any such Seller Group Entity or any Company Group Entity; (ii) any Resolutions; (iii) any of the terms or requirements of any Governmental Approval held by such Seller Group Entity or any Company Group Entity; or (iv) any provision of any Material Contract, in each case, in any material respect;
- (b) give any Governmental Authority or other Person the right to (i) materially challenge or prevent the Transaction; (ii) exercise any material remedy or obtain any material relief under any material Legal Requirement or any Order to which any such Seller Group Entity or any Company Group Entity is subject; (iii) declare a material default of, exercise any material remedy under, accelerate the performance of, cancel, terminate, materially modify or receive any material payment under, any Material Contract; or (iv) revoke, suspend or materially modify any material Governmental Approval granted to any Company Group Entity;
 - (c) result in the imposition or creation of any Encumbrance upon or with respect to any material assets of any Company Group Entity; or
- (d) except for applicable requirements, if any, under any Antitrust Law, require any such Seller Group Entity or any Company Group Entity to obtain any material Consent from, or make or deliver any material filing or notice to, a Governmental Authority.

3.5 Financial Statements.

- (a) Seller has previously delivered or made available to Purchaser the unaudited consolidated balance sheets, and the profit and loss statements, of the Company Group Entities as of and for the fiscal years ended March 31 in each of 2008, 2009, and 2010 (collectively, the "*Unaudited Financial Statements*"). The unaudited consolidated balance sheet and the profit and loss statement of the Company Group Entities as of and for the fiscal year ended March 31, 2010 is referred to herein as the "*Latest Financial Statement*."
- (b) Upon delivery, the Audited Pro Forma Financial Statements (i) present fairly the financial condition of the Pro Forma Company Group Entities as of the respective dates thereof and the results of operations of the Pro Forma Company Group Entities for the periods covered thereby in accordance with U.S. GAAP; and (ii) have been prepared in accordance with U.S. GAAP, applied on a consistent basis throughout the periods covered.
- (c) Upon delivery, the Interim Financial Statements (i) present fairly the respective financial condition of the Pro Forma Company Group Entities as of the respective dates thereof and the respective results of operations of the Pro Forma Company Group Entities for the periods covered thereby in accordance with J-GAAP or in accordance with U.S. GAAP; and (ii) have been prepared in accordance with J-GAAP or U.S. GAAP, in all cases, applied on a consistent basis throughout the periods covered; *provided, however*, that (i) the Interim Financial Statements prepared in accordance with U.S. GAAP will be reviewed pursuant to SAS 100 pursuant to U.S. GAAP and (ii) the Interim Financial Statements are subject to year-end adjustments consistent with past practice (which will not be material individually or in the aggregate) and does not contain all of the footnotes required by J-GAAP or U.S. GAAP.

- (d) The Unaudited Financial Statements (i) present fairly the financial condition of the Company Group Entities as of the respective dates thereof and the results of operations of the Company Group Entities for the periods covered thereby in accordance with J-GAAP; and (ii) have been prepared in accordance with J-GAAP, applied on a consistent basis throughout the periods covered; *provided*, *however*, that the Unaudited Financial Statements do not contain all of the footnotes required by J-GAAP and are subject to normal year-end audit adjustments.
- 3.6 **Absence of Undisclosed Liabilities**. No Company Group Entity has, to the Knowledge of Seller, any material Liabilities other than (i) those set forth in the Latest Financial Statement; (ii) those incurred in the ordinary course of business; (iii) those incurred by any actions taken by Seller or Company Group Entity in order to comply with any obligation or covenant of Seller under this Agreement; (iv) those incurred by any action permitted hereunder; (v) liabilities under existing Contracts; and (vi) those caused by any actions or matters referred to in the Seller Disclosure Schedule.
- 3.7 **Absence of Changes**. Between the Latest Balance Sheet Date and the date of this Agreement, (i) the Company Group Entities have conducted the Business in the ordinary course of business; and (ii) no event or circumstance has occurred that has had a Company Material Adverse Effect. Without limiting the generality of the foregoing, since the Latest Balance Sheet Date and the date of this Agreement, no Company Group Entity has taken any of the actions described in Section 6.2(a), (b), (c), (e), (f), (n), (o), (p) and (v).
 - 3.8 Ownership and Sufficiency of Business Assets.
- (a) **Title**. The Company Group Entities will have, as of the Closing, good and valid title to (or valid leases or licenses in respect of) all the Business Assets and Services and each other assets and services held, owned or leased by or provided to a Seller Group Entity or Company Group Entity, that are material to the Business, other than the Business IP Assets and the Excluded Business Assets and Services, in each case free and clear of all Encumbrances.
 - (b) Sufficiency of Assets other than Intellectual Property. Subject to the receipt by the Company Group Entities of:
- (i) such services and benefits to be provided to any Company Group Entities pursuant to the Transition Services Agreement or other Transaction Agreements;
- (ii) such services and benefits to be provided directly or indirectly to any Company Group Entity by service providers set forth on Schedule 7.6(a) and Schedule 7.6(b);

- (iii) the Governmental Approvals listed in Schedule 3.17(a) of the Seller Disclosure Schedule; and
- (iv) the Contract Consents,

upon the Closing, the Company Group Entities will own and have the right to use (or, in the case of leased or licensed interests, will have the right to use) such assets and services, without contravention of the terms and conditions of any of such assets and services (or, with respect to assets or services provided under the Transaction Agreements or by service providers set forth on Schedule 7.6(a) and Schedule 7.6(b), the right to use assets or obtain services on the terms and subject to the conditions thereof) as are necessary to enable the Company Group Entities to conduct the Current Business in all material respects, in compliance in all material respects with all Legal Requirements, Orders, Governmental Approvals and Contracts applicable to the Current Business.

This Section 3.8(b) shall not apply to Intellectual Property Rights, which are the subject of Section 3.8(c).

- (c) **Sufficiency of Intellectual Property**. The (i) Company IP Assets; (ii) rights and licenses to Seller Group Entities' Intellectual Property Rights provided under the IP Assignment Agreement, IP License Agreement, Trademark License Agreement, and Transition Services Agreement; (iii) rights and licenses under Intellectual Property Rights granted to Company Group Entities in or under Company Contracts under which Company Group Entities continue to enjoy their rights immediately following Closing; (iv) rights and licenses under Intellectual Property Rights granted to Seller Group Entities that have been or will be assigned, sublicensed, or otherwise transferred to Company, (v) rights under Intellectual Property Rights conveyed to Seller Group Entities or Company Group Entities by operation of law (such as patent exhaustion and implied licenses) in the ordinary course of business, and (vi) Intellectual Property Rights of third parties for which an indemnity is provided by Seller pursuant to Section 12.2(a)(vii), together will constitute, immediately following the Closing Date, all of the material Intellectual Property Rights (including third party Intellectual Property Rights) used in or necessary to the Business as it currently is conducted or is planned by Seller to be conducted immediately prior to the Closing.
- (d) **Research and Development**. The Business Assets and Services and the services and benefits to be provided to the Company Group Entities by the Seller Group Entities under the Transaction Agreements and the Company Group Employees and Transferred Employees together constitute all of the assets and personnel (other than (i) those to be provided under any Company Contract and (ii) Intellectual Property Rights which shall be subject to subsection (c) above) used in or necessary for the Company Group Entities to continue, in all material respects, the research and development projects relevant to the Business as such projects are currently conducted or reasonably contemplated to be conducted.
- 3.9 **Accounts Receivable**. The Accounts Receivable are (i) valid and legally binding obligations of the account debtor enforceable in accordance with its terms; (ii) a true and correct statement of the account for merchandise actually sold and delivered to, or for services actually performed for and accepted by, such account debtor; and (iii) to the Knowledge of Seller, fully collectible in the ordinary course of business (less the amount of any provision or reserve therefore made in the books and records of the Company Group Entities).

3.10 Indebtedness.

- (a) Schedule 3.10(a)(i) of the Seller Disclosure Schedule lists all Indebtedness (other than Intercompany Indebtedness) of any Company Group Entity, including all Capital Leases ("Identified Capital Leases") as of May 31, 2010, setting forth as to each item of Indebtedness the counterparty, the principal amount outstanding, the per annum interest rate and the maturity date. No Company Group Entity is in material breach (or has received written notice of any material breach or material default that remains pending) or material default under any of the terms or conditions set forth in any loan document or other document or instrument related to such Indebtedness. Schedule 3.10(a)(ii) of the Seller Disclosure Schedule sets forth all Intercompany Indebtedness as of May 31, 2010. Complete and correct copies of all material documents relating to the Permitted Indebtedness have been delivered or made available by Seller to Purchaser.
- (b) <u>Schedule 3.10(b)</u> of the Seller Disclosure Schedule sets forth a complete and correct list as of the date of this Agreement of the Company Group Entity, Seller Group Entity, financial institution and/or factoring company involved in each of the SSMC Supplier Alternative Payment Systems and the Company Supplier Alternative Payment System.
- (c) Schedule 3.10(c) of the Seller Disclosure Schedule sets forth a complete and correct list as of the date of this Agreement of each guarantee provided by any Seller Group Entity for the benefit of the Business, any Company Group Entity or any of the employees of a Company Group Entity.
- 3.11 **Inventory**. To the Knowledge of Seller, all Inventory consist of items of a quality, quantity and condition usable and salable in the ordinary course of the business of the Company Group Entities without material discount or material reduction, conform to generally accepted standards in the industry of which the Business is a part, are free of material defects, materially meet the Company Group Entities' current standards and specifications and comply in all material respects with applicable Legal Requirements, are at levels consistent with past practices of the business of the Company Group Entities, and are physically present without material damage, material loss or material shrinkage.

3.12 Material Contracts.

- (a) All Contracts, including all material amendments, material supplements, material modifications and material waivers thereof, to which any Company Group Entity is a party or otherwise primarily relate to the Business ("Company Contracts") to which any of the descriptions set forth below may apply are referred to as "Material Contracts":
- (i) Contracts relating to profit sharing, stock option, employee stock purchase or other plan or arrangements providing for deferred compensation (including any bonuses or other remuneration and whether in cash or otherwise), to employees, former employees or consultants, or any other employee benefit plan or arrangement, or any collective bargaining agreement or any other agreement with any labor union, or severance agreements, programs, policies or arrangements;

- (ii) Contract for the employment by any Company Group Entity of any officer, individual employee or other Person on a full-time, part-time, consulting or other basis involving annual compensation in excess of ten million Yen (¥10,000,000) (or the equivalent amounts in any other currency based on the Applicable Spot Rate), other than payments that are either mandatory under applicable Legal Requirements or Orders or payable pursuant to a Plan available generally to Active Employees within a particular country, or Contracts relating to loans by any Company Group Entity to officers, directors, shareholders or Affiliates of any Company Group Entity;
 - (iii) All joint venture, partnership or other Contracts involving a sharing of profits with a third party;
- (iv) Contract or indenture relating to borrowed money or other Indebtedness or the mortgaging, pledging or otherwise placing an Encumbrance (other than any license) on any material asset or group of material assets of any Company Group Entity, or pursuant to which any Company Group Entity guarantees in any way any Contract of a third Person;
- (v) Contract under which any Company Group Entity is lessee of or holds or operates any personal property owned by any other Person, except for any lease of personal property under which the aggregate annual rental payments do not exceed one hundred million Yen (¥100,000,000);
- (vi) Contract under which any Company Group Entity is lessor or that permits any third party to hold or operate any property, real or personal, owned or controlled by any Company Group Entity, except for any Contract which involves consideration in the aggregate not exceeding or not reasonably likely to exceed one hundred million Yen (¥100,000,000) annually (or the equivalent amounts in any other currency based on the Applicable Spot Rate);
- (vii) Contract or group of related Contracts other than those specified in items (i) through (vi) and (viii) through (xvi) with the same party or group of affiliated parties (other than customers or suppliers) the performance of which involves consideration in the aggregate exceeding or reasonably likely to exceed two hundred million Yen (¥200,000,000) annually (or the equivalent amounts in any other currency based on the Applicable Spot Rate);
- (viii) Contracts that grant a license or sublicense (including but not limited to a covenant not to sue or similar undertaking) pursuant to which any Company Group Entity is authorized to use any third party Intellectual Property Right that is material to the Business, excluding generally commercially available, software programs that are not incorporated into any product marketed, sold, or licensed by a Company Group Entity ("Off the Shelf Software") and excluding Intellectual Property Rights of Seller Group Entities;
- (ix) Contracts that grant a license or sublicense (including but not limited to a covenant not to sue or similar undertaking) pursuant to which any third party is authorized to use any Company IP Asset that is material to the Business, other than nonexclusive licenses or rights granted or arising in the ordinary course of business in connection with the manufacture, distribution or sale of products;

- (x) all Contracts to acquire a business of, or equity interest of twenty percent (20%) or greater in, any Person, other than a Company Group Entity;
- (xi) Contracts prohibiting any Company Group Entity from freely engaging in any business or competing anywhere in the world, or pursuant to which any material benefit is required to be given or lost as a result of so competing;
 - (xii) any Contract to which any Seller Group Entity is a party and is material to, or otherwise primarily relates to, the Business;
 - (xiii) any other Contract that is material to any Company Group Entity and not made in the ordinary course;
- (xiv) any Contract with any of the customers or suppliers of any Company Group Entity required to be disclosed pursuant to <u>Schedule 3.24</u> of the Seller Disclosure Schedule;
- (xv) any Contract between or among any Company Group Entity, on the one hand, and any Seller Group Entity (other than a Company Group Entity);
- (xvi) any employee collective bargaining agreement or any other material Contract with any labor organization, union or association of which any Active Employee is a member;
 - (xvii) Contracts relating to any Securitization of Receivables Transaction or any Supplier Alternative Payment System; and
 - (xviii) Real Property Leases.
- (b) Seller has delivered or made available to Purchaser or its Representative complete and correct copies of each Material Contract (other than the Contracts described in (xvii) above) in effect as of the date of this Agreement. Each Material Contract is in writing, is currently valid and in full force and effect, and, to the Knowledge of Seller, is enforceable by the relevant Company Group Entity or Seller Group Entity in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles related to or limiting creditors' rights generally and by general principles of equity.
- (c) No Company Group Entity is in material breach or material default, and in the two (2) year period prior to the date of this Agreement, no party has notified any Company Group Entity in writing that it is in material breach or material default, under any Material Contract, other than breaches or defaults with respect to which the notification has been withdrawn. To the Knowledge of Seller, no event has occurred, and no circumstance or condition exists, that is reasonably likely to (with or without notice or lapse of time) (a) result in a material violation or material breach of any of the provisions of any Material Contract; (b) give any Person the right to declare a material default or exercise any material remedy under any Material Contract; (c) give any Person the right to accelerate the maturity or performance of any Material Contract or to cancel, terminate or materially modify any Material Contract; or (d) otherwise have a Company Material Adverse Effect in connection with any Material Contract. The Company has not waived any of its material rights under any Material Contract. To the Knowledge of Seller, no counter-party to any Material Contract is in material breach or material default under such Material Contract.

3.13 Intellectual Property.

- (a) Schedule 3.13(a) of the Seller Disclosure Schedule sets forth a complete and correct list (as of March 31, 2010) of all Company IP Assets that are Registered IP Assets (subject to any updates to the lists of Freely Assigned Patents and Restricted Assigned Patents set forth in Exhibits F and H to the IP Assignment Agreement made according to Section 1.6 of the IP Assignment Agreement). Schedule 3.13(a) of the Seller Disclosure Schedule specifies as to each the nature of such item, lists the jurisdictions in which each such item has been issued or registered or in which any such application for such issuance and registration has been filed, indicates any applicable registration or application number, and indicates any co- or joint owner and managing division (\(\subseteq \subseteq \subseteq \subseteq \subseteq \text{usho} \)).
- (b) All necessary registration, maintenance and renewal fees in connection with the Company IP Assets that are Registered IP Assets, which become due and payable prior to the Closing Date, will have been paid up until the Closing Date with regard to such Registered IP Assets, and documents and certificates due prior to the Closing Date will have been filed, except where any Seller Group Entity or Company Group Entity has reasonably determined in its ordinary course of business to terminate the renewal or maintenance of such registration. For each Registered IP Asset assigned to each Company Group Entity or Seller Group Entity, an assignment has been recorded in accordance with applicable Legal Requirements and in a timely manner with the relevant patent, copyright, trademark or other authorities.
- (c) As of the Closing Date, Seller Group Entities and Company Group Entities will have complied with all applicable internal rules of Seller and Company and Legal Requirements with respect to payment of inventor awards, including the obligation to pay fair compensation to inventors due under Article 35 of the Japan Patent Law to the relevant inventors of any Patents or assigned inventions within the Company IP Assets.
- (d) Each item of the Company IP Assets (other than Restricted Assigned Patents, Restricted Assigned Copyrights and Know-How, and common law trademarks) will be solely owned by a Company Group Entity as of the Closing. Each item of Company IP Assets that are Restricted Assigned Patents or Restricted Assigned Copyrights and Know-How will be owned by a Company Group Entity as of the Closing if, prior to the Closing, Seller has obtained the consent of the Third Party IP Owner to the assignment of such Restricted Assigned Patent or Restricted Assigned Copyrights and Know-How to the Company Group Entities. Each item of the Business IP Assets licensed to a Company Group Entity under any Transaction Agreement is owned by a Seller Group Entity that is authorized to so license such Business IP Assets and will be duly and validly licensed to a Company Group Entity as of the Closing for use in the manner currently used or reasonably contemplated to be used in the conduct of the Business. The Company IP Assets (other than common law trademarks) will be, as of the Closing, free and clear of any Encumbrances other than (i) any license, rights, obligations or covenants as set forth in the IP Assignment Agreement, (ii) outbound Intellectual Property Licenses disclosed in Section 3.13(f) and (iii) those non-material Encumbrances arising in ordinary course of business in connection with the Business of the Company Group Entities. As soon as reasonably practicable after the Closing and subject to the Encumbrances as described in (i), (ii) and (iii) above in the immediately preceding sentence, each of the Business IP Assets will be owned by a Company Group Entity or terms and conditions set forth in the Intellectual Property License Agreement, Trademark License Agreement and IP Assignment Agreement, or identical to those under which such Company Group Entity currently uses or practices such Business IP Assets without any affirmative act by any Company Group Entity, Purchaser or any other Person, including any assignment, release, C

- (e) In each case in which any Company Group Entity or Seller Group Entity has acquired an ownership interest in any Company IP Assets from any Person, either (i) such Company IP Assets were developed, written, or created solely by employees of a Company Group Entity or a Seller Group Entity acting within the scope of their employment, or (ii) such ownership interest was validly and irrevocably assigned to the relevant Company Group Entity or Seller Group Entity, as the case may be. Each employee (current or past) of each Company Group Entity and each employee (current or past) of each Seller Group Entity who participated in the creation of any Company IP Asset that are material to the Business has signed the Company Group Entities' or Seller Group Entities', as the case may be, standard agreement regarding proprietary information and inventions assignment, a complete and correct copy of the current form of which is set forth in Schedule 3.13(e) of the Seller Disclosure Schedule. To the Knowledge of Seller, as of the date hereof, no employee, consultant or independent contractor of any Seller Group Entity or Company Group Entity is, as a result of or in the course of such employee's, consultant's or independent contractor's engagement by any Seller Group Entity or Company Group Entity in default or breach of any material term related to the assignment of Intellectual Property Rights of any employment agreement, non-disclosure agreement, assignment of invention agreement, or similar agreement between such Persons and any Seller Group Entity or Company Group Entity.
- (f) Except for Company Contracts (i) listed in <u>Schedule 3.13(f)</u> of the Seller Disclosure Schedule as royalty-bearing and (ii) for Off the Shelf Software, there are (and upon the Closing, will be), no royalties, fees, or other payments in any of the preceding three (3) years under any Contracts by a Company Group Entity to any third party, with respect to any use of Intellectual Property Rights in connection with the conduct of the Business.
- (g) Schedule 3.13(g) of the Seller Disclosure Schedule sets forth a complete and correct list of all Contracts to which any Seller Group Entity is a party (i) granting any Intellectual Property License to Company IP Assets (other than nonexclusive licenses or rights granted or arising in the ordinary course of business in connection with the manufacture, distribution or sale of products) or (ii) otherwise limiting any Company Group Entity's ability to exploit fully any of the Company IP Assets. None of the Seller Group Entities to any such Contract is in material breach of any such Contract, and there is no material dispute under any such Contract regarding the scope of such Contract or performance under such Contract, including with respect to any payments to be made or received by any Seller Group Entity thereunder. No Contract grants to or authorizes the retention of any joint ownership interest or exclusive license of, or other exclusive right with respect to, any of the Company IP Assets by any Person other than a Company Group Entity.

- (h) The operation of the Business as it is currently conducted, or as it is currently planned by Seller to be conducted, including but not limited to the design, development, use, import, branding, advertising, promotion, marketing, manufacture and sale of the Company's products and services, does not and will not when operated in the same manner following the Closing, infringe or misappropriate any Intellectual Property Rights of any Person, or constitute unfair competition or trade practices under the laws of any jurisdiction (excluding (i) any infringement or misappropriation following the Closing for which Seller indemnifies Purchaser Parties as set forth in Schedules 12.2(a)(vii), 12.2(a)(viii), and item 1 of 12.2(a)(ix) and (ii) any infringement or misappropriation arising from or related to common law trademarks). In the three (3) years preceding the date of this Agreement, no Seller Group Entity or Company Group Entity has received written notice that could reasonably be construed as a threat of litigation or enforcement absent a license, including an "invitation to take a license" from any Person claiming that such operation or any of the Company's products and services (including any currently under development) infringes or misappropriates any Intellectual Property Rights of any Person, or constitutes unfair competition or trade practices under the laws of any jurisdiction, except as provided in Schedule 3.13(g) of the Seller Disclosure Schedule.
- (i) To the Knowledge of Seller, as of the date of this Agreement, no Person is infringing, using or disclosing without authorization, or misappropriating any Company IP Assets that are material to the Business. No Company Group Entity or Seller Group Entity has sent a notice within the three (3) years preceding the date of this Agreement to any Person kept in the records of the legal or intellectual property divisions of a Company Group Entity or Seller Group Entity alleging any such claims against such Person with respect to any such Company IP Assets.
- (j) To the Knowledge of Seller, each item of Company IP Assets (i) in the case of a Registered IP Asset (other than a pending application) is valid and in full force and effect; and (ii) has not been abandoned (except in the ordinary course of business) or passed into the public domain (except as a result of the expiration of its term). Company has no Knowledge of any facts, circumstances or information that would (x) render any Company IP Assets invalid or unenforceable or (y) adversely affect or impede the ability of a Company Group Entity to use any Business IP Assets in the conduct of the Business as it is currently conducted or as it is currently planned by Seller to be conducted. No Company Group Entity or Seller Group Entity has misrepresented, or failed to disclose, and has no Knowledge of any misrepresentation or failure to disclose, any fact or circumstances in any application for any Registered IP Assets in a manner that would constitute fraud with respect to such application or that would otherwise adversely affect the validity or enforceability of such Registered IP Assets.

- (k) The Seller Group Entities and Company Group Entities have taken reasonable steps to, including having established internal rules, safeguard and maintain the secrecy and confidentiality of all of the Trade Secrets that are material to the Business and are not otherwise protected by Patents and have complied with such internal rules in all material respects. Without limitation on the generality of the foregoing, any disclosures to third parties of Trade Secrets that are material to the Business have been made pursuant to written confidentiality agreements prohibiting disclosure, except where such third parties have any implied confidentiality obligations under Legal Requirements.
- (l) None of the execution, delivery, and performance of this Agreement, the other Transaction Agreements and the consummation of the transactions contemplated hereby and thereby in and of themselves will (i) conflict with, result in a breach of, modify, or terminate, any Intellectual Property License (except for those solely among Company Group Entities); (ii) cause the forfeiture or termination or give rise to a right of forfeiture of, termination of, or result in any loss of a Company Group Entity's right to use, any Business IP Assets, including by in any way impairing the right of any Company Group Entity to use or practice or bring any action for the infringement, unauthorized use or disclosure, or misappropriation of any Company IP Assets; or (iii) result in any Company Group Entity or Purchaser or any of its Affiliates granting to any Person any right or license to any of the Company IP Assets (other than pursuant to the Transaction Agreements) or Purchaser or any of its Affiliates, in each case other than as a result of Contracts to which Purchaser or any of its Affiliates, and not a Company Group Entity, is a party.
- (m) To the Knowledge of Seller, the electronic data processing, information, record keeping, communications, and computer systems and any other information technology systems (collectively, "*Technology Systems*") used by the Company Group Entities and the Seller Group Entities to operate the Business as of the Closing are adequate for the operation of the Business as presently conducted. In the three (3) year period preceding the date of this Agreement, there has not been any material malfunction with respect to any of the Technology Systems that resulted in the loss of any Intellectual Property Rights of Seller Group Entities or Company Group Entities material to the Business, a material interruption in the manufacturing operations of the Business, or costs and expenses exceeding one hundred million Yen (¥100,000,000) and that has not been remedied in all material respects.
- (n) For the three (3) year period preceding the date of this Agreement, the Company Group Entities and Seller Group Entities have complied in all material respects with their respective published privacy policies relating to privacy and data security (the "*Privacy Policies*") in connection with the Business, including with respect to the collection, use, disclosure and transfer (including cross-border transfer) of Personally-Identifiable Information. In connection with the Business, the Company Group Entities and Seller Group Entities use commercially reasonable technical and organizational measures to protect Personally-Identifiable Information against unauthorized access, disclosure, use, modification or other misuse or misappropriation to the extent required by their Privacy Policies and applicable Legal Requirements. To the Knowledge of Seller, except as set forth in <u>Schedule 3.1(m)</u> of the Seller Disclosure Schedule, there have been no material unauthorized or accidental accesses, disclosures, intrusions or breaches of security of Personally-Identifiable Information maintained by or on behalf of any Company Group Entity or Seller Group Entity in connection with the Business in the three (3) year period preceding the date of this Agreement, nor any complaints, notices to, or claims asserted by any Person (including any Governmental Authority) regarding the collection, use, transmission, disclosure or sharing of Personally-Identifiable Information by any Company Group Entities in connection with the Business in the three (3) year period preceding the date of this Agreement. The execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby will comply with the Privacy Policies.

- (o) The Company Group Entities' and the Seller Group Entities' use and distribution, if any, of each component of Open Source Software complies in all material respects with the applicable license agreement, and such use or distribution, if any, by the Company Group Entities and Seller Group Entities does not give rise under such license agreement to any material obligation in favor of any Person for the Company Group Entities and Seller Group Entities to disclose or distribute, in source code form, any Business IP Asset, to license for the purpose of making derivative works any Business IP Asset, or to distribute without charge any Business IP Asset.
- (p) <u>Schedule 3.13(p)</u> of the Seller Disclosure Schedule contains a true and complete list of all inbound Intellectual Property Licenses to which a Seller Group Entity is a party under which any Company Group Entity receives rights under Patents.

3.14 Employees.

(a) Schedule 3.14(a) of the Seller Disclosure Schedule lists separately for each Company Group Entity or Seller Group Entity, as applicable, each Active Employee (other than temporary employees (hakenshain)) as of June 21, 2010, in the case of Active Employees employed by an Entity in Japan and May 31, 2010 in the case of Active Employees employed by an Entity outside Japan. Schedule 3.14(a) of the Seller Disclosure Schedule shows in respect of each such Active Employee (other than temporary employees (hakenshain)): (i) status as a full-time part-time, or secondee; (ii) function, department, title and position; (iii) date of employment, date of commencement of continuous employment (if different) and date of secondment, each as applicable; (iv) the office or location in which such Active Employee is employee or to which such Active Employee is seconded and the Entity in which such Active Employee is registered; and (y) such Active Employee's annual salary or hourly compensation, as the case may be, and any other material remuneration arrangements (including equity grants, contributions payable to any private pension arrangement in respect of the employee and compensation payable pursuant to bonus, incentive, in-kind and deferred compensation, allowances or commission arrangements and identifying in each case the relevant employer and any other entity responsible for such salary or other compensation). Each Company Group Entity and, to the extent employing any Active Employees, each Seller Group Entity, is in compliance and has complied since the date falling three (3) years prior to the date of this Agreement in all material respects with all Legal Requirements, Orders and Contract terms relating to such employees, including terms relating to the calculation and payment of wages (including overtime pay, maximum hours of work and child labor restrictions), equal employment opportunity (including Legal Requirements prohibiting discrimination and/or harassment on the basis of race, color, national origin, religion, gender, disability, age, sexual orientation, or otherwise or requiring accommodation of such characteristics), affirmative action and other hiring practices, occupational safety and health, workers' compensation, unemployment compensation, the payment of social security and other Taxes, and unfair labor practices. There are no material workers' compensation, labor, or similar employment related claims pending or, to the Knowledge of Seller, being overtly threatened against any Company Group Entity or against any Seller Group Entity by any Active Employees.

- (b) Schedule 3.14(b) of the Seller Disclosure Schedule lists each Active Employee as of June 21, 2010, in the case of Active Employees employed by an Entity in Japan and May 31, 2010 in the case of Active Employees employed by an Entity outside Japan, who is required by applicable Legal Requirements to hold a temporary work authorization or a particular class of non-immigrant visa in order to work in any jurisdiction in which such employee is employed (each a "Work Permit"), and shows for each such employee the type of Work Permit held and the remaining period of validity. With respect to each Work Permit, all of the information that any Seller Group Entity or any of the Company Group Entities has provided to the relevant Governmental Entities (collectively, "Immigration Authorities") in the application for such Work Permit was true and complete in all material respects. The Company Group Entities have received the appropriate notice of approval from the Immigration Authorities with respect to each such Work Permit. Since the date falling three (3) years prior to the date of this Agreement, none of the Company Group Entities or Seller Group Entities has received any written notice from the Immigration Authorities that any Work Permit has been revoked, other than any revocation that has been cured in all material respects or with respect to which the notice was withdrawn. To the Knowledge of Seller, there is no action pending or overtly threatened to revoke or adversely modify the terms of any Work Permit. No Active Employee is a non-immigrant employee of a nationality other than that of the jurisdiction in which he or she is employed whose right to remain in such employment would terminate or otherwise be affected by the transactions contemplated by this Agreement. For each Active Employee hired after November 6, 1986 who is working in the United States, the respective Company Group Entity or Seller Group Entity that hired such Active Employee has completed and retained an Immigration and Naturalization Service
- (c) The employment of any terminated former employee of any of the Company Group Entities or Seller Group Entities who was primarily engaged in the Business has been terminated in material compliance with any applicable Contract terms and applicable Legal Requirements, and no Company Group Entity has any material liability under any Contract or applicable Legal Requirements toward any such terminated employee, except as (i) may be required under a Plan available generally to Active Employees or (ii) expressly set forth in the Restructuring Plan, that is reasonably likely to become a material liability of Purchaser or any Company Group Entity after the Closing Date. Except as otherwise expressly set forth in the Restructuring Plan, the Transactions will not (in and of themselves) cause Purchaser or any Company Group Entity to incur or suffer any material liability relating to, or material obligation to pay, severance, termination or other payment to any present or former employee.
- (d) No Company Group Entity has made any loans (except advances for business expenses in the ordinary course of business) to any Active Employee that have not been fully repaid, forgiven or otherwise satisfied.

- (e) During the three (3)-year period ending on the date hereof, no Company Group Entity and, to the extent having employed any Active Employees during such three (3)-year period, no Seller Group Entity, has experienced and, to the Knowledge of Seller, there has not been threatened, any strike, work stoppage, slowdown, lockout, picketing, boycott, or other material labor dispute, union organization attempt, demand for recognition from a labor organization or petition for representation with respect to Active Employees. No Company Group Entity and, to the extent employing any Active Employees, no Seller Group Entity, is subject to any collective bargaining agreement or obligation to engage in collective bargaining with any trade union or similar organization with respect to Active Employees, except as required by any applicable Legal Requirement. No Company Group Entity and, to the extent employing any Active Employees, no Seller Group Entity, has since the date falling three (3) years prior to the date of this Agreement, engaged in any unfair labor practices within the meaning of any applicable Legal Requirements with respect to any Active Employees. There is no pending or, to the Knowledge of Seller, overtly threatened dispute between any of the Company Group Entities or, to the extent employing any Active Employees, any Seller Group Entity, on the one hand, and any trade union, works council or similar organization, on the other hand, with respect to employees engaged in the Business. No action, claim, labor dispute or Proceeding is pending or, to the Knowledge of Seller, being overtly threatened against any of the Company Group Entities or, to the extent employing any Active Employees, any Seller Group Entity, respecting or involving any applicant for employment in the Business, any current employee or any former employee engaged in the Business, or any class of the foregoing, by or before any Governmental Authority.
- (f) No Company Group Entity is liable for any arrears of wages, salaries, bonuses or commissions owed to Active Employees except for bonuses and similar compensation earned during the period in which the Closing falls but payable on a deferred basis in accordance with the terms thereof.
- (g) There has been no lay-off or reduction in force of employees engaged in the Business undertaken by or on behalf of any of the Company Group Entities since the date falling three (3) years prior to the date of this Agreement. No material Orders, awards, improvements, prohibitions or other notices have been served upon and no other material enforcement or similar Proceedings have been taken against any of the relevant Company Group Entities since the date falling three (3) years prior to the date of this Agreement pursuant to any legislation, regulations, Orders or codes of conduct of any Governmental Authority in respect of Active Employees.
- (h) Except as expressly set forth in the Restructuring Plan, there are no plans for the future lay off, rationalization or similar measures for employees engaged in the Business undertaken by or on behalf of any of the Company Group Entities.
- (i) Except as expressly set forth in the Restructuring Plan, there are no current negotiations with any union or similar organization for any material change in the rate of remuneration or the bonus, incentives, or private pension benefits of any Active Employee or any other material change to the costs or expenses associated with Active Employee activities.

3.15 Employee Benefits.

- (a) <u>Schedule 3.15(a)</u> of the Seller Disclosure Schedule lists all material Plans. True, correct and complete copies of all material documents with respect to such Plans have been delivered or made available to Purchaser.
- (b) Except for Unfunded Benefit Liabilities as of the date of this Agreement, all Plans are validly existing in all material respects with such regulatory authorities as may be applicable and, in the three-year period preceding the date of this Agreement, no written notice of non-compliance, failure to validly exist has been received by Seller or any of the Company Group Entities from any such Governmental Authority with respect to any of the Plans. The Unfunded Benefit Liabilities as at March 31, 2010 are set forth on Schedule 3.15(b) of the Seller Disclosure Schedule. With respect to any unfunded Plan providing pension, retirement, or similar benefits for which applicable Local GAAP or other Legal Requirement mandate that reserves be recorded on a statement of financial position, reserves have been recorded on the Latest Financial Statement in a manner that is consistent in all material respects with applicable Local GAAP and other Legal Requirements. With respect to any funded Plan providing pension, retirement, or similar benefits, such plans have been funded in all material respects in accordance with applicable Legal Requirements.
- (c) No Company Group Entity has any unsatisfied material Liabilities or, to the Knowledge of Seller, is reasonably likely to incur any material Liabilities that are reasonably likely to become material Liabilities of any of the Company Group Entities or Purchaser with respect to any Plan. Each Plan to which any of the Company Group Entities contributes that is intended to qualify for any Tax benefit under applicable Legal Requirements has received any required confirmation of such qualification from an appropriate Governmental Authority and is in material compliance with all requirements of applicable Legal Requirements, as to both form and operation, necessary to maintain such qualification. Each Company Group Entity has performed all of its material obligations in relation to the Plans in accordance with the governing documentation of the Plans and the requirements of all applicable Legal Requirements.
- (d) No written undertaking or assurance (whether or not constituting a legally binding commitment) has been given to any Active Employee as to the continuation of any of the Plans after the Closing.
- (e) There are no pending criminal proceedings against, and no pending material civil, arbitration, administrative or other Proceedings or pending disputes by or against, the trustees, managers or administrators of the Plans or any of the Company Group Entities in relation to the Plans and, to the Knowledge of Seller, none is being overtly threatened.
- (f) Employees of the Seller Group Entities who are to be transferred to a Company Group Entity can remain as participants in the Seller Pension Plans after the Closing Date under current applicable Legal Requirements.
- 3.16 **Compliance with Laws**. The Company Group Entities are, and at all times since the date falling three (3) years prior to the date of this Agreement has been, in compliance in all material respects with each Legal Requirement or Order that is applicable to such Company Group Entity or any of its respective properties, assets, operations or businesses, and, to the Knowledge of Seller, no event has occurred, and no condition or circumstance exists, that is reasonably likely (with or without notice or lapse of time) to constitute, or result in, a material default under, a material breach or violation of, or a material failure to comply with, any such Legal Requirement or Order by any Company Group Entity. Neither Seller nor any Company Group Entity has received, since the date falling three (3) years prior to the date of this Agreement, any written notice from any third party regarding any actual, alleged or potential material violation of any Legal Requirement or Order by any Company Group Entity in connection with the Business, except for violations that have been cured or notices that have been withdrawn.

3.17 Governmental Approvals.

- (a) Each Company Group Entity has all material Governmental Approvals that are necessary or required in connection with the ownership and use of its properties or assets or the operation of its business. Each Company Group Entity has made all material filings with, and given all material notifications to, all material Government Authorities as required by all applicable Legal Requirements. Schedule 3.17(a) of the Seller Disclosure Schedule contains a complete and correct list and summary description of each such material Governmental Approval. Each such material Governmental Approval is valid and in full force and effect, and there is not pending or, to the Knowledge of Seller, being overtly threatened any Proceeding which is reasonably likely to result in the suspension, termination, revocation, cancellation, limitation or impairment of any such material Governmental Approval.
- (b) The Company has delivered or made available to Purchaser complete and correct copies of all of the Governmental Approvals identified in <u>Schedule</u> <u>3.17(a)</u> of the Seller Disclosure Schedule, including all renewals thereof and all amendments thereto.

3.18 Proceedings and Orders.

- (a) There is no material Proceeding pending or, to the Knowledge of Seller, being overtly threatened against any Company Group Entity. To the Knowledge of Seller, no event has occurred, and no condition or circumstance exists, that is reasonably likely to serve as a basis for the commencement of any such Proceeding. Since the date falling three (3) years prior to the date of this Agreement, no insurance company has asserted in writing to any Company Group Entity or Seller Group Entity that any such Proceeding is not covered by the applicable insurance policy held by any Company Group Entity or any Seller Group Entity in respect of the Business, other than any failure of coverage that has been cured or any assertion that has been withdrawn.
- (b) No Company Group Entity nor any of their respective properties, assets, operations or businesses, nor any of their respective rights relating to any of the foregoing, is subject to any Order that remains in effect.

3.19 Environmental Matters.

(a) The operations of the Business are, and at all times since the date falling three (3) years prior to the date of this Agreement has been, in compliance in all material respects with all Environmental and Safety Laws.

- (b) The Company and each other Company Group Entity, as applicable, has obtained and owns or possesses all material Governmental Approvals that are required under Environmental and Safety Laws to conduct the Current Business (collectively, the "Environmental Permits"), and the Company or the applicable other Company Group Entity is in compliance in all material respects with all such Environmental Permits, and all such Environmental Permits are in full force and effect, without unusual conditions or restrictions.
- (c) There has been no material Release by any Company Group Entity or, to the Knowledge of Seller, any other Person at the Property or any of the Facilities in violation of any material Legal Requirement. To the Knowledge of Seller, no Hazardous Materials are present in, on, or under, or migrating to, or from, the Property in violation of any material Legal Requirement.
- (d) There are no Proceedings pending or, to the Knowledge of Seller, being overtly threatened against any Company Group Entity or against any Seller Group Entity in respect of the Business alleging a violation of an Environmental and Safety Law by any Company Group Entity, noncompliance by any Company Group Entity with Environmental Permits, or alleging liability of any Company Group Entity for the presence of Hazardous Materials in violation of any material Legal Requirement. During the three (3) year period prior to the date of this Agreement, no Seller Group Entity nor any Company Group Entity has received any written notice from any Person or Governmental Authority alleging, and, to the Knowledge of Seller, no condition or circumstance currently exists that is reasonably likely to result in (with or without notice or lapse of time or both), a material violation of, or material failure to comply with, any material Environmental and Safety Law or Environmental Permit by any Company Group Entity or by any Seller Group Entity in respect of the Business.
- (e) To the Knowledge of Seller, none of the following exists at the Property or any of the Facilities: any asbestos-containing material in any form which is friable; PCBs; active or out-of-service or underground storage tanks or sites from which such storage tanks have been removed; or landfills, solid waste management units, surface impoundments, waste piles or land disposal areas of any kind.
- (f) The Company has delivered or made available to Purchaser complete and correct copies of all material studies, reports, surveys, assessments, audits, correspondence, investigations, analysis, tests, and other material documents similar thereto (whether in hard copy or electronic form) in any Seller Group Entity's or any Company Group Entity's possession regarding the presence or alleged presence of Hazardous Materials at, or on, or regarding any of such Entities' compliance with any Environmental and Safety Laws.

3.20 Taxes.

(a) Since April 1, 2006, each of the Company Group Entities has timely filed (or has had timely filed on its behalf) with the appropriate Governmental Authority all Tax Returns required to be filed; all such Tax Returns are true, complete and correct in all material respects; and no Company Group Entity has requested or obtained any extension of time within which to file any Tax Return, which Tax Return has not since been filed.

- (b) All Taxes of the Company Group Entities (whether or not shown on any Tax Return) that are due and payable have been timely paid in full. Each of the Company Group Entities has withheld and timely paid over (or set aside for payment when due) all Taxes required to have been withheld and paid over, and complied with all information reporting and backup withholding requirements, including maintenance of required records with respect thereto, in connection with amounts paid or owing to any employee, independent contractor, shareholder, creditor or other third party. All Active Employees have been properly classified for Tax purposes, including withholding of Taxes, in the relevant jurisdiction in which such classification is required. There are no liens for Taxes on any asset of any Company Group Entity, other than for Taxes not yet due and payable.
- (c) Each of the Company Group Entities has made available to Purchaser complete and correct copies of (i) its income or franchise (whether federal, state, local or foreign) Tax Returns for all taxable periods beginning on or after January 1, 2008; and (ii) rulings it received from, and any agreements it entered into with Governmental Authorities that are currently in effect and relate to Taxes. Since April 1, 2006, no claim has been made by a Governmental Authority in a jurisdiction where neither the Company nor any other Company Group Entity files Tax Returns that the Company or any other Company Group Entity is or may be required to pay Taxes or file Tax Returns in that jurisdiction. No Company Group Entity does business in or derives income from any state, local, territorial or foreign taxing jurisdiction other than those for which all Tax Returns have been made available to Purchaser. Since April 1, 2006, no Company Group Entity has a "permanent establishment" in any foreign country, as defined in any applicable Tax treaty between Japan and such foreign country nor has received written notice from any Governmental Authority that any Company Group Entity has a permanent establishment in any foreign country.
- (d) Since April 1, 2006, no Company Group Entity has been the subject of any Examination of Taxes. Any deficiencies and proposed adjustments with respect to Taxes of any Company Group Entity have been resolved, and no other deficiencies or proposed adjustments are expected to be asserted or made. No Examinations or other Proceedings with respect to Taxes or Tax Returns of any Company Group Entity are pending or, to the Knowledge of Seller, being overtly threatened. Since April 1, 2006, no material deficiency, underpayment, late payment or other unpaid liability has been raised in writing by any Governmental Authority with respect to Taxes of any Company Group Entity. No waiver or extension of any statute of limitations with respect to Taxes or Tax Returns of any Company Group Entity is currently in effect or has been requested.
- (e) No Company Group Entity is the party to or subject of any closing agreement, private letter ruling, technical advice memorandum, advance pricing agreement, or similar ruling or agreement relating to Taxes with any Governmental Authority.
 - (f) None of the Company Group Entities has been party to or received a Tax concession provided by a relevant Governmental Authority.
- (g) The useable net operating losses for tax purposes available to the Company Group Entities at Closing will be as set forth on Schedule 3.20(g) of the Seller Disclosure Schedule, less any amount utilized as permitted by Section 7.17. These net operating losses shall expire on the schedule attached as Schedule 3.20(g) of the Seller Disclosure Schedule.

3.21 **Insurance**. Schedule 3.21 of the Seller Disclosure Schedule sets forth a complete and correct list of all material insurance policies, self-insurance arrangements and fidelity bonds, in effect as of June 30, 2010 that insure any Company Group Entity whether or not the applicable policy is held by any Company Group Entity or any Seller Group Entity (collectively, the "**Insurance Policies**"). The Company has delivered or made available to Purchaser true, correct and complete copies of all Insurance Policies. Each Insurance Policy is valid, binding, and in full force and effect and will remain so valid, binding, and in full force and effect, on the same terms as those in effect as of the date hereof in accordance with their terms, following the consummation of the Transaction. No Company Group Entity is in material breach of any Insurance Policy, and, to the Knowledge of Seller, no event has occurred, which, with notice or the lapse of time, is reasonably likely to constitute such a material breach, or permit termination, material modification, or material acceleration, of any Insurance Policy. Since the date falling three (3) years prior to the date of this Agreement, neither Seller, any Seller Group Entity, nor any Company Group Entity has received any written notice of cancellation or non-renewal of any Insurance Policy. Schedule 3.21 of the Seller Disclosure Schedule sets forth a complete and correct list as of June 30, 2010 of all claims filed by any Company Group Entity against any Insurance Policy in the last sixty (60) months. To the Knowledge of Seller, there is no such pending claim under any Insurance Policy as to which any insurer has disputed or denied liability. Neither Seller, any Seller Group Entity, nor any Company Group Entity has received any written notice of a material increase in the premium for any Insurance Policy that has not yet been effected.

3.22 Real Property.

(a) Schedule 3.22(a) of the Seller Disclosure Schedule sets forth a complete and correct list as of the date of this Agreement of each parcel of real property owned by each Company Group Entity (collectively, the "Owned Real Property"), including a street address, complete legal description or such description (Section, Block and Lot numbers or other legal description) as may be necessary to identify each parcel. The Company, or other Company Group Entity, as applicable, is the sole and exclusive legal owner of, and is in possession of, all Owned Real Property, including the buildings, structures, fixtures and Improvements situated thereon and appurtenances thereto, including such right of ways and easements running toward the benefit or burden of such Owned Real Property, in each case free and clear of all Encumbrances. Seller has made available to Purchaser all material Company Contracts pertaining to the Owned Real Property and are legally valid and binding and in full force and effect, and neither Seller, its Affiliates, nor any Company Group Entity has received any written notice of default, offset, counterclaim or defense asserted thereunder since the date falling three (3) years prior to the date of this Agreement.

(b) Schedule 3.22(b) of the Seller Disclosure Schedule sets forth a complete and correct list as of the date of this Agreement of all permits and leasehold or subleasehold estates and other rights to use or occupy any land, buildings or structures, Improvements, fixtures, or other interest in real property held by each Company Group Entity (collectively, the "Leased Real Property" and together with the Owned Real Property, the "Real Property") (including the street address of each Leased Real Property and the name of the lessor) and a list as of the date of this Agreement of all Real Property Leases and other material Company Contracts relating to each Real Property Lease). The Company or other Company Group Entity, as applicable, has been in lawful possession of the premises covered by each Real Property Lease since the commencement of the original term of such Real Property Lease. Company has delivered or made available to Purchaser complete and correct copies of each Real Property Lease and each material Company Contract relating to each Real Property Lease. All Real Property Leases are valid and in full force and effect in accordance with their respective terms and there exists no material default thereunder or, to the Knowledge of Seller, occurrence or condition which is reasonably likely to result in a material default thereunder or termination thereof.

- (c) All buildings, structures, fixtures, building systems and equipment, and all components thereof, including the roof, foundation, load-bearing walls and other structural elements thereof, heating, ventilation, air conditioning, mechanical, electrical, plumbing and other building systems, environmental control, remediation and abatement systems, sewer, storm and waste water systems, irrigation and other water distribution systems, parking facilities, fire protection, security and surveillance systems, and telecommunications, computer, wiring and cable installations, included in the Real Property (the "*Improvements*") are in good condition and repair (ordinary wear and tear excepted) and suitable in all material respects for the operation of the Current Business. There are no material structural deficiencies or material latent defects affecting any of the Improvements and, to the Knowledge of Seller, there are no facts or conditions affecting any of the Improvements that are reasonably likely to, individually or in the aggregate, interfere in any material respect with the use or occupancy of the Improvements or any portion thereof by the Company Group Entities in the operation of the Current Business.
- (d) There are no outstanding options or other contractual rights to purchase, lease or use, or rights of first refusal to purchase the Owned Real Property or any portion thereof or interests therein or contracts relating to the right to receive any portion of the income or profits from the sale, operation or development thereof.
- (e) No Company Group Entity nor any sublessee of any Company Group Entity under any of the Real Property Leases has made any material alterations, additions or improvement to the Leased Real Property that are required to be removed (or of which any landlord or sublandlord could require removal) at termination of the applicable lease term.
- (f) Since the date falling three (3) years prior to the date of this Agreement, neither Seller, any Seller Group Entity, nor any Company Group Entity has received written notice from any Governmental Authority or other Person that the use and occupancy of any of the Real Property, as currently used and occupied, and the conduct of the Current Business thereon, violates in any material respect any deed restrictions or applicable Legal Requirements, consisting of building codes, zoning, subdivision or other land use or similar laws. Neither Seller, any Seller Group Entity, nor any Company Group Entity has received written notice of any material zoning or other land-use regulation proceeding, or any material change in any applicable Legal Requirements, which to the Knowledge of Seller, is reasonably likely to affect the use and operation of the Real Property as currently used and operated in any material respect, other than violation that was cured or any notice that was withdrawn.

- (g) There are no eminent domain, condemnation or other similar proceedings pending or, to the Knowledge of Seller, threatened affecting any portion of the Owned Real Property. To the Knowledge of Seller, there are no eminent domain, condemnation or similar proceedings pending or threatened affecting any portion of the Leased Real Property.
- (h) The Company Group Entities have all material licenses, certificates of occupancy, permits, authorizations and approvals required to operate the Current Business and utilize the Real Property and the Leased Real Property to the extent required in the conduct of the Current Business on such Real Property. The Company Group Entities have all material easements and utility rights required by Legal Requirements or by the normal use and operation of the Real Property to the extent required in the conduct of the Current Business on such Real Property or the Leased Real Property.
- (i) No Company Group Entity has legally obligated itself to sell, assign or sublease or otherwise transfer or convey any Owned Real Property or its interest under any Real Property Lease to any Person, except as may be set forth in a Company Contract pertaining to the Leased Real Property identified in Schedule 3.22(b) of the Seller Disclosure Schedule. No Company Group Entity has legally obligated itself to place any Encumbrance or Improvement on any Real Property, except as may be set forth in a Company Contract affecting the Leased Real Property as identified in Schedule 3.22(b) of the Seller Disclosure Schedule. No Company Group Entity has hypothecated or assigned any rents or income from the Real Property, or any portion thereof, in any manner to any third party.
- (j) <u>Schedule 3.22(j)</u> of the Seller Disclosure Schedule sets forth a complete and correct list of all Leased Real Property where the lessor is a Seller Group Entity, the average monthly rent payable and the applicable average monthly rent exemption thereunder for the twelve (12) month period ending on March 31, 2010.
- 3.23 **Personal Property**. The Company Group Entities have good and valid title to the material items of tangible personal property owned by them, and have legal right to use all other material items of tangible personal property used by them pursuant to the terms of Company Contracts governing the possession or use of such material items of tangible personal property, other than those disposed of in the ordinary course of business since March 31, 2010, in each case free and clear of all Encumbrances. Such material items of tangible personal property are in good operating condition and repair (subject to normal wear and tear).

3.24 Customers and Suppliers.

(a) <u>Schedule 3.24(a)</u> of the Seller Disclosure Schedule sets forth a complete and correct list of the twenty (20) largest customers of the Business, determined on the basis of sales revenues based on data managed by the Sales & Marketing Division of the Company, for each of the fiscal years ended March 31, 2009 and March 31, 2010 and sets forth the revenues received from each such customer during the applicable period.

- (b) Schedule 3.24(b) of the Seller Disclosure Schedule sets forth a complete and correct list of:
- (i) the twenty (20) largest suppliers of the Business, determined on the basis of costs of items purchased based on data managed by the Procurement Department of the Company for each of the fiscal years ended March 31, 2009 and March 31, 2010 and sets forth the amounts paid to each such supplier during the applicable period; and
 - (ii) all sole source material suppliers of the Business as of the date of this Agreement.
- (c) No Company Group Entity is restricted (whether by Contract, Legal Requirement or otherwise) in any material respect from selling, licensing, distributing products or otherwise conducting the Current Business or any portion thereof with any customer or potential customer, in any geographic area, during any period of time or in any segment of the market. There is no material purchase commitment which provides that any supplier will be the exclusive supplier of or distributor for any product, component or service utilized in the Business by any Company Group Entity. There is no material purchase commitment requiring any Company Group Entity to purchase the entire output of a supplier.
- (d) Since the date falling three (3) years prior to the date of this Agreement, no Company Group Entity, Seller or Seller Group Company has received any written notice or otherwise has Knowledge, that any current customer or supplier identified or required to be listed on Schedule 3.24(a) of the Seller Disclosure Schedule or Schedule 3.24(b) of the Seller Disclosure Schedule intends or is reasonably likely to cease dealing with any Company Group Entity, intends or is reasonably likely to otherwise materially reduce the volume of business transacted by such Person with any Company Group Entity or otherwise is materially dissatisfied with the service any Company Group Entity provides such Person.
- 3.25 **Bank Accounts**. Schedule 3.25 of the Seller Disclosure Schedule lists the name and location of each bank or other institution in which any Company Group Entity has any deposit account or safe deposit box and all account numbers as of the date of this Agreement. As of the date of this Agreement, all of the Company Group Entities' cash (other than a reasonable amount of petty cash held by each Company Group Entity) is in the bank accounts described above, and the Company or other Company Group Entity, as applicable, has full legal and beneficial interest in all cash, cash equivalents and other financial assets deposited in such accounts, free and clear of any Encumbrances.
- 3.26 **Company Transaction Expenses**. Except as otherwise provided by the Transaction Agreements, there are no fees, costs, commissions or expenses (including fees, costs and expenses of legal counsel, accountants, investment bankers, brokers or other representatives and consultants and appraisal fees, costs and expenses), special bonuses, severance or other similar items of compensation (discretionary or otherwise) incurred by the Company or any other Company Group Entity at or prior to the Closing in connection with transactions contemplated by this Agreement and the consummation of the transactions contemplated hereby or payable at or prior to the Closing to any employee or director of any Company Group Entity in connection with or arising out of the transactions contemplated hereby (collectively, the "Company Transaction Expenses") or reasonably expected to be so incurred by any Company Group Entity.

- 3.27 **Transactions with Affiliates**. No Seller Group Entity (a) owns, directly or indirectly, any debt, equity or other ownership interest in any Entity with which any Company Group Entity has a material business relationship, other than (i) any Entity in which any Seller Group Entity, individually or in the aggregate together with any other Seller Group Entity, holds an investment (whether in the form of debt, equity, other ownership interest or any combination thereof) of <u>less</u> than one hundred million Yen (¥100,000,000); or (ii) any publicly-traded Entity in which such Seller Group Entity owns less than one percent (1%) of the issued and outstanding capital stock, (b) as at May 31, 2010, had any Indebtedness owed by any Seller Group Entity to any Company Group Entity; and (c) has any direct interest in any material asset, property or other material right primarily used for the Business, other than Business IP Assets, that will not be owned by Company Group Entities after the Closing.
- 3.28 **Brokers**. No Company Group Entity has retained any broker or finder or incurred any liability or obligation for any brokerage fees, commissions or finders fees with respect to this Agreement or the Transaction.
- 3.29 **Product Specifications and Warranties.** Since the date falling three (3) years prior to the date of this Agreement, no Company Product marketed as of such date, and no facility in which any such Company Product has been developed, manufactured, stored or shipped, has been disqualified by any then shipping customers, other than any notice that has been withdrawn. Since the date falling three (3) years prior to the date of this Agreement, no Company Group Entity has received any written notice that any Company Product currently marketed has been rejected by any currently shipping customer due to a failure to meet product specifications. Schedule 3.29 of the Seller Disclosure Schedule sets forth a correct and complete statement of the Company Product warranty return experience by month of the Company Group Entities, on a segment by segment basis, and specifying in each case where failure rates for any such segment exceeded one percent (1%) per month (on average as measured over a consecutive three month period) of the installed base for such segment, and any material concessions, compensation for settlements or credits granted in connection with such warranty returns, for the two year period ended May 31, 2010; including any currently pending customer claims for compensation which have yet to be settled.
- 3.30 **Product Liability**. No Company Group Entity has any material liability that has been asserted by a third party on or prior to the date of this Agreement (and, to the Knowledge of Seller, there are no facts or circumstances that are reasonably likely to give rise to any future material liability) arising out of any injury to individuals or property as a result of the ownership, possession, or use of any Company Product.
- 3.31 **Foreign Exchange Notification**. All of the information provided by Seller to Purchaser, pursuant to reasonable requests of Purchaser, regarding the "restricted business" of the Company Group Entities for purposes of Purchaser's filing of a prior notification under the Foreign Exchange and Foreign Trade Law of Japan is true and accurate.

- 3.32 **Demerger.** Seller has duly transferred to Company, in compliance with Legal Requirements and pursuant to the Company Demerger, those Contracts that license to a Seller Group Entity or Company Group Entity any Intellectual Property Rights (other than Patents) that are used by Company in the Business set forth on Schedule 3.32(a) of the Seller Disclosure Schedule, when delivered pursuant to Section 6.13. The Contracts set forth on Schedule 3.32(b) of the Seller Disclosure Schedule, when delivered pursuant to Section 6.13, sets forth a complete and correct list of Contracts existing as of the effective date of the Company Demerger that license to a Seller Group Entity or Company Group Entity any Intellectual Property Rights (other than Patents) that are used by Company in the Business but that were not transferred to the Company in connection with the Company Demerger ("Demerger Non-Transferred Contracts").
- 3.33 **Service Providers**. Schedule 3.33 of the Seller Disclosure Schedule, when delivered pursuant to Section 6.13, sets forth each service provider that provides services indirectly to the Company Group Entities through contractual arrangements with Seller Group Entities and the nature of the services provided.
- 3.34 **SSV Union Approval**. Neither the consent nor approval of the SSV Union is required for the implementation of the Restructuring Plan as it relates to SSV pursuant to current applicable Legal Requirements.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES REGARDING SELLER

Seller represents and warrants to Purchaser as follows:

- 4.1 **Ownership of Purchased Shares**. Seller is the record and beneficial owner of all right, title and interest, in and to all of the Purchased Shares, free and clear of all Encumbrances. The delivery to Purchaser of the Purchased Shares as contemplated in this Agreement will transfer to Purchaser valid title to the Purchased Shares, free and clear of any and all Encumbrances.
- 4.2 **Authority; Validity of Contemplated Transactions**. Seller and any Seller Group Entity as applicable, has all requisite corporate power and authority to execute and deliver this Agreement and all other Transaction Agreements to which such Seller Group Entity is a party and to carry out the provisions of this Agreement and such other Transaction Agreements. The execution, delivery and performance by Seller and any Seller Group Entity as applicable of this Agreement and such other Transaction Agreements have been approved by all requisite action on the part of Seller or applicable Seller Group Entity. This Agreement has been duly and validly executed and delivered by Seller. Each of this Agreement and such other Transaction Agreements to which Seller or applicable Seller Group Entity is a party constitutes, or upon execution and delivery by Seller or applicable Seller Group Entity and each of the other parties thereto, will constitute, the legal, valid and binding obligation of Seller or Seller Group Entity as applicable, enforceable against Seller or applicable Seller Group Entity in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles related to or limiting creditors' rights generally and by general principles of equity.

- 4.3 **Proceedings**. There are no Proceedings pending or, to the Knowledge of Seller, being overtly threatened against Seller (a) challenging or seeking to restrain, or prohibit the Transaction or (b) preventing Seller from performing its obligations under this Agreement or any Transaction Agreements to which it is a party.
- 4.4 **Brokers**. Other than Nikko Cordial Securities Inc., the fees of which shall be paid solely by Seller, Seller has not retained any broker or finder or incurred any Liability or obligation for any brokerage fees, commissions or finders fees with respect to this Agreement or the Transaction.

4.5 Securities Act Matters.

- (a) Illiquid Shares; Restriction on Resales. Seller acknowledges that the Issued Shares issued pursuant to this Agreement have not been, and when issued, will not be registered under the Securities Act, FIEL, the securities laws of any state of the United States or the securities laws of any other country. To the extent that the Securities Act or FIEL is applicable, the Issued Shares may not be sold, transferred or otherwise disposed of without registration under the Securities Act or FIEL (as applicable) or unless an exemption from registration is available. In the absence of an effective registration statement covering such shares, to the extent that the Securities Act or FIEL applies, Seller acknowledges that it may sell, transfer or otherwise dispose of such shares only in a transaction for which an exemption from registration under the Securities Act or FIEL is available. Seller further understands that there is no assurance that any exemption from registration under the Securities Act or FIEL will be available, that such exemption will allow such Seller to dispose of or otherwise transfer any or all of the Issued Shares under the circumstances, in the amounts or at the times such Seller might propose. Seller understands and acknowledges that the certificate(s) evidencing the Issued Shares to be issued or delivered pursuant to this Agreement shall bear Purchaser's standard form of Securities Act restrictive legend, as well as legends describing any applicable contractual restrictions. Seller understands that it may have to bear the economic risk of an investment in Issued Shares to be received by it pursuant to the transactions contemplated hereby for an indefinite period of time.
- (b) **Accredited Investor; Investment Intent.** Seller is an "accredited investor" (as defined in Rule 501(a) under the Securities Act). Seller, either alone or with its representative, is capable of evaluating the risks and merits of owning the Issued Shares. Seller is acquiring the Issued Shares for investment only and not with a view to or intention of or in connection with any resale or distribution of such shares or any interest therein in violation of the Securities Act.
- (c) **Unregistered Securities**. Seller has been advised and acknowledges that in issuing the Issued Shares to Seller pursuant hereto, Purchaser Parent is relying upon the exemption from registration provided by Section 4(2) of the Securities Act and Regulation D promulgated thereunder and/or upon Regulation S promulgated under the Securities Act.

ARTICLE V

REPRESENTATIONS AND WARRANTIES REGARDING PURCHASER AND PURCHASER PARENT

Each of Purchaser and Purchaser Parent represents and warrants as follows:

- 5.1 **Organization and Good Standing**. Each of Purchaser and Purchaser Parent is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of organization.
- 5.2 **Authority; Binding Nature of Agreements**. Each of Purchaser, Purchaser Parent and Affiliates of Purchaser that are party to any of the Transaction Agreements (each a "*Purchaser Transaction Agreement Party*") has all requisite corporate power and authority to execute and deliver this Agreement and all other Transaction Agreements to which it is a party and to carry out the provisions of this Agreement and such other Transaction Agreements. The execution, delivery and performance by each Purchaser Transaction Agreement Party, as applicable, of this Agreement and such other Transaction Agreements have been approved by all requisite action on the part of such Purchaser Transaction Agreement Party. This Agreement has been duly and validly executed and delivered by Purchaser and Purchaser Parent. Each of this Agreement and the other Transaction Agreements constitutes, or upon execution and delivery by each Purchaser Transaction Agreement Party, as the case may be, will constitute, the legal, valid and binding obligation of such Purchaser Transaction Agreement Party, as the case may be, enforceable against such Purchaser Transaction Agreement Party in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles related to or limiting creditors' rights generally and by general principles of equity.
- 5.3 **Purchaser Parent Capital Structure**. The authorized capital stock of Purchaser Parent consists of 750,000,000 Purchaser Parent Shares. At the close of business on July 2, 2010, (i) 431,046,132 Purchaser Parent Shares were issued and outstanding; (ii) 48,066,127 Purchaser Parent Shares were held by Purchaser Parent in its treasury; (iii) 85,620,294 Purchaser Parent Shares were reserved and available for issuance pursuant to the Purchaser Parent stock option, ESPP and RSU plans, and 44,476,194 Purchaser Parent Shares were subject to outstanding Purchaser Parent options and full value awards; (iv) 5,300,000 Purchaser Parent Shares were reserved and available for issuance pursuant to outstanding warrants of the Purchaser Parent and (v) 108,000,000 were reserved and available for issuance pursuant to convertible notes of the Purchaser Parent. Except as set forth in this Section 5.3, at the close of business on July 2, 2010, no shares of capital stock or other voting securities or equity interests of Purchaser Parent were issued, reserved for issuance or outstanding. Except as set forth in this Section 5.3 and except for the Issued Shares, (i) there are not issued, reserved for issuance or outstanding (A) any shares of capital stock or other voting securities or equity interests of Purchaser Parent; (B) any securities of Purchaser Parent convertible into or exchangeable or exercisable for shares of capital stock or other voting securities or equity interests of Purchaser Parent; or (C) any warrants, calls, options or other rights to acquire from Purchaser Parent, and no obligation of Purchaser Parent; and (ii) there are not any outstanding obligations of Purchaser Parent to repurchase, redeem or otherwise acquire any such securities or to issue, deliver or sell, or cause to be issued, delivered or sold, any such securities. The Issued Shares, when issued as contemplated herein, will be duly authorized, validly issued, fully paid and non-assessable, free and clear of any Encumbrances and will not be issued in vio

- 5.4 **No Conflicts; Required Consents**. The execution, delivery and performance of this Agreement or any other Transaction Agreement by each applicable Purchaser Transaction Agreement Party do not and will not (with or without notice or lapse of time):
- (a) violate or result in any breach of (i) any of the provisions of the Charter Documents of such Purchaser Transaction Agreement Party; (ii) any resolutions adopted by the stockholders, or board of directors or committees thereof of such Purchaser Transaction Agreement Party; (iii) any of the terms or requirements of any material Governmental Approval held by such Purchaser Transaction Agreement Party or that otherwise relates to such Purchaser Transaction Agreement Party's business; or (iv) any provision of a material Contract to which such Purchaser Transaction Agreement Party is a party;
- (b) give any Governmental Authority or other Person the right to (i) challenge the Transaction; (ii) exercise any remedy or obtain any relief under any Legal Requirement or any Order to which such Purchaser Transaction Agreement Party t or any of its assets is subject; or (iii) declare a default of, exercise any remedy under, accelerate the performance of, cancel, terminate or modify any Contract to which such Purchaser Transaction Agreement Party is a party; or
- (c) except for applicable requirements, if any, under any Antitrust Law, require such Purchaser Transaction Agreement Party to obtain any Consent or make or deliver any filing or notice to a Governmental Authority.

5.5 SEC Filings; Financial Statements.

(a) Purchaser Parent has delivered or made available to Seller accurate and complete copies of all registration statements, proxy statements and other statements, reports, schedules, forms and other documents filed by the Company with the U.S. Securities and Exchange Commission (the "SEC") since January 1, 2009, and all amendments thereto (the "SEC Documents"). All statements, reports, schedules, forms and other documents required to have been filed by Purchaser Parent with the SEC have been so filed on a timely basis. As of the time it was filed with the SEC (or, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing): (i) each of the SEC Documents complied in all material respects with the applicable requirements of the Securities Act or the U.S. Securities Exchange Act of 1934, as amended (as the case may be); and (ii) none of the SEC Documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

- (b) The financial statements (including any related notes) contained in the SEC Documents: (i) complied as to form in all material respects with the published rules and regulations of the SEC applicable thereto; (ii) were prepared in accordance with Local GAAP in the United States applied on a consistent basis throughout the periods covered (except as may be indicated in the notes to such financial statements and except that the unaudited financial statements may not contain footnotes), and (iii) fairly present the consolidated financial position of Purchaser Parent and its consolidated Subsidiaries as of the respective dates thereof and the consolidated results of operations and cash flows of Purchaser Parent and its consolidated Subsidiaries for the periods covered thereby.
- 5.6 **Brokers**. Other than GCA Savvian Advisors, LLC, the fees of which will be paid solely by Purchaser, neither Purchaser nor Purchaser Parent has retained any broker or finder or incurred any liability or obligation for any brokerage fees, commissions or finders fees with respect to this Agreement or the Transaction.
- 5.7 **Proceedings**. There are no Proceedings pending or, to the Knowledge of Purchaser, threatened against or affecting Purchaser Parent or Purchaser (a) challenging or seeking to restrain, delay or prohibit the Transaction (b) preventing Purchaser or Purchaser Parent from performing its obligations under this Agreement or any Transaction Agreements to which it is a party.
- 5.8 Non-Reliance of Purchaser and Purchaser Parent. Except for the specific representations and warranties expressly made by Seller in Article III and Article IV or any Transaction Agreement, (a) each of Purchaser and Purchaser Parent acknowledges and agrees that: (i) Seller is not making and has not made any representation or warranty, expressed or implied, at law or in equity, in respect of the Business, any Company Group Entity, any Seller Group Entity or any of their respective businesses, assets, liabilities, operations, prospects, or condition (financial or otherwise), including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any liabilities, the prospects of the Business, the effectiveness or the success of any operations, or the accuracy or completeness of any documents, projections, material or other information (financial or otherwise) regarding the Business, any Company Group Entity or any Seller Group Entity furnished or made available to Purchaser, Purchaser Parent or any of their respective Representatives (whether in any "data room," "virtual data room," management presentation or in any other form) in expectation of, or in connection with, the Transactions, or in respect of any other matter or thing whatsoever; and (ii) no Representative or employee of any Seller Group Entity or any Company Group Entity has any authority, express or implied, to make any representations, warranties or agreements not specifically set forth in this Agreement and subject to the limited remedies herein provided; (b) each of Purchaser and Purchaser Parent specifically disclaims that it is relying upon or has relied upon any such other representations or warranties that may have been made by any Person, and acknowledges and agrees that Seller has specifically disclaimed and does hereby specifically disclaim any such other representation or warranty made by any Person; and (c) each of Purchaser and Purchaser Parent is acquiring the Purchased Shares subject onl

ARTICLE VI

PRE-CLOSING COVENANTS

- 6.1 **Company's Conduct of the Business Prior to the Closing.** From and after the date hereof and prior to the earlier of the Closing Date and the date this Agreement is terminated in accordance with its terms (the "*Pre-Closing Period*"), except (w) to the extent required to comply with applicable Legal Requirements after having provided Seller with prior written notice, (x) to the extent necessary to comply with any obligation of Seller under this Agreement, including complying with or giving effect to the Restructuring Plan and/or the Reorganization, (y) with the prior written consent of Purchaser or Purchaser Parent (which consent shall not be unreasonably withheld, conditioned or delayed) or (z) as set forth on <u>Schedule 6.2</u>, Seller shall cause the Company Group Entities to:
 - (a) Conduct the Business in the ordinary course of business (including Patent prosecution);
 - (b) Pay all of its material Taxes in a manner consistent with current practices;
 - (c) Maintain insurance coverage consistent with its insurance coverage in existence as of the date of this Agreement; and
- (d) Use commercially reasonable efforts to maintain good relationships with employees, licensors, licensees, suppliers, contractors, distributors, customers, and others having business dealings with the Company Group Entities.
- 6.2 **Restrictions on Company's Conduct of the Business Prior to the Closing**. During the Pre-Closing Period, except (w) to the extent required to comply with applicable Legal Requirements after having provided Seller with prior written notice, (x) to the extent necessary to comply with any obligation or covenant of Seller under this Agreement, including complying with or giving effect to the Restructuring Plan and/or the Reorganization, or (y) with the prior written consent of Purchaser or Purchaser Parent (which consent shall not be unreasonably withheld, conditioned or delayed) or (z) as set forth on <u>Schedule 6.2</u>, Seller shall cause the Company Group Entities not to:
 - (a) Acquire by merging or consolidating with, or by purchasing any equity securities or assets of, or by any other manner, any business or any Entity;
- (b) Sell, transfer, lease, license, assign, option or otherwise encumber any of its assets (including any of the Business IP Assets) with a value of greater than fifty million yen (¥50,000,000) or that are material to the Business, except in the ordinary course of business;
- (c) Provide promotions, coupons, discounts or price increases or decreases with respect to the customers, suppliers or distributors of any Company Group Entity, except in a manner consistent with current practices;
 - (d) Enter into any Material Contracts, except on commercially reasonable terms in the ordinary course of business;

- (e) Materially change or announce any material change to the Company Products or any services sold by any Company Group Entity, other than customized changes to any Company Product for a particular customer;
 - (f) Terminate or amend in any material manner any Material Contract or material Governmental Approval;
- (g) Commence any Proceeding involving amounts in excess of fifty million yen (¥50,000,000), other than for (i) the routine collection of Accounts Receivable or (ii) injunctive relief on the grounds that any Company Group Entity has suffered immediate and irreparable harm not compensable in money damages;
- (h) Declare, authorize or pay any dividends on, make any other distributions with respect to, or redeem, repurchase or otherwise acquire any of its capital stock;
- (i) Issue, sell, contract to issue or sell, pledge, dispose of, grant, encumber, split, reclassify or authorize the issuance, sale, pledge, disposition, grant, split, reclassification or encumbrance of (1) any shares of capital stock or equity or similar interest in any Company Group Entity or (2) any options, warrants, convertible securities or other rights of any kind to acquire any shares of such capital stock or equity or similar interest in any Company Group Entity;
- (j) Purchase, lease, license or otherwise acquire assets in a transaction in an amount exceeding fifty million yen (¥50,000,000), except for supplies or raw material acquired by any Company Group Entity in the ordinary course of business;
- (k) Make any individual capital expenditure in excess of fifty million Yen (¥50,000,000) (other than capital expenditures required to be made under Section 6.12);
 - (1) Incur any Indebtedness exceeding fifty million Yen (¥50,000,000);
- (m) Provide any credit, loan, advance, guaranty, endorsement, indemnity, warranty or mortgage to any Person, including any of the customers, stockholders, officers, employees or directors of Company, other than in the ordinary course of business;
- (n) Materially change its credit practices, accounting methods or practices or standards used to maintain its books, accounts or business records, except as required by Local GAAP;
- (o) Materially defer or postpone making payments of any Accounts Payable, accrued expenses or other obligations unless such deferral or postponement is due to a good faith dispute as to liability or amount or is in accordance with past practices as to the timing of payments or other obligations;
- (p) Write-off as uncollectible, any Accounts Receivable, except write-offs in the ordinary course of business consistent with past practice charged to applicable reserves, or accelerate the collection of any Accounts Receivable, unless such acceleration is in accordance with the Company's past practices as to the timing of such collections;

- (q) Knowingly incur any Liability in an amount exceeding fifty million yen (¥50,000,000), except in the ordinary course of business;
- (r) Amend its Charter Documents;
- (s) Accelerate, materially amend or change the period of exercisability or vesting of any options or rights to acquire any equity or similar ownership interest in Company or other rights granted under any Plans, except as specifically contemplated in this Agreement or as required under the terms of such Plans;
- (t) Hire any new employee except consistent with current practices; terminate any officer (or, with respect to an Entity in Japan, any Person who is a *shikkouyakuin* or of higher position) or director; increase the annual level of compensation of any existing employee except for regular, scheduled compensation increases in the ordinary course of business; establish or adopt any Plan; or grant any bonuses, benefits or other forms of direct or indirect compensation to any employee, officer, director or consultant other than in the ordinary course of business;
- (u) Make any severance payments to any employee, officer or director, except payments made pursuant to written agreements or applicable work rules outstanding as of the date of this Agreement and listed in Schedule 6.2;
- (v) Make, amend, or revoke any material election relating to Taxes; adopt or change any accounting period relating to Taxes; file any material amendment to any Tax Return; enter into any Tax sharing, allocation, indemnity or similar agreement; enter into any closing agreement; settle or compromise any material claim or assessment relating to Taxes; surrender the right to any material refund of Tax; consent to any extension or waiver of the limitations period applicable to any Taxes or Tax Returns; take any other action outside the ordinary course of business that is reasonably likely to materially increase the tax liability of the Company Group Entities, taken as a whole.
 - (w) Enter into any Contract or agree, in writing or otherwise, to take any of the actions described in Section 6.2(a) through (u).

6.3 Certain Notifications.

- (a) During the Pre-Closing Period, Seller shall promptly notify Purchaser in writing of its becoming aware of any:
 - (i) Fact, circumstance or event that is reasonably likely to have a Company Material Adverse Effect;
- (ii) Fact, circumstance, event, or action by any Company Group Entity or Seller Group Entity the existence, occurrence, or taking of which would result in the condition set forth in <u>Section 9.1(a)</u> not being satisfied;
 - (iii) Material breach of any covenant or obligation of Seller hereunder; and

- (iv) Fact, circumstance or event which will result in, or is reasonably likely to result in, any of the conditions in <u>Section 9.1</u> or <u>9.3</u> not being satisfied.
 - (b) During the Pre-Closing Period, Purchaser and Purchaser Parent shall promptly notify Seller in writing of its becoming aware of any:
- (i) Fact, circumstance, event, or action by Purchaser or Purchaser Parent the existence, occurrence, or taking of which would result in the condition set forth in <u>Section 9.2(a)</u> not being satisfied;
 - (ii) Material breach of any covenant or obligation of Purchaser or Purchaser Parent hereunder; and
- (iii) Fact, circumstance or event which will result in, or is reasonably likely to result in, any of the conditions in <u>Section 9.2</u> or <u>9.3</u> not being satisfied.
- 6.4 Access to Information. During the Pre-Closing Period, Seller shall, and shall cause the Company Group Entities to, (i) permit Purchaser and its Representatives to have reasonable access at all reasonable times after receiving reasonable advance notice from Purchaser, and in a manner so as not to unreasonably interfere with the normal business operations of Seller or any Company Group Entity, to all premises, properties, personnel, books, records (including Tax records), Contracts, and documents of or pertaining to the Business; (ii) furnish Purchaser and/or Purchaser's Representative with all financial, operating and other data and information related to any Company Group Entity or the Business (including copies thereof), as Purchaser may reasonably request; and (iii) otherwise cooperate and assist, to the extent reasonably requested by Purchaser, with Purchaser's investigation of the Business *provided*, *however*, that Seller may restrict or otherwise prohibit access to any documents or information to the extent that Seller or Company determines in good faith that any applicable Legal Requirement requires it to do so. No information or knowledge obtained in any investigation pursuant to this Section 6.4 shall affect or be deemed to modify any representation or warranty contained herein or any Transaction Agreement or the conditions to the obligations of the parties to consummate the Transaction. Purchaser shall, and Purchaser Parent shall cause Purchaser to, provide a certificate to Seller upon completion of its due diligence investigation of the Company Group Entities and the Business, indicating (i) that it has completed such due diligence review and (ii) whether or not it is, acting reasonably and in good faith, satisfied with the results thereof. If Purchaser delivers a certificate that it is so satisfied, the condition precedent set forth in Section 9.1(n) shall be deemed waived by Purchaser's diligence hereunder.
- 6.5 **Reasonable Best Efforts**. During the Pre-Closing Period, (a) Seller shall use its reasonable best efforts to cause the conditions set forth in Sections 9.1 and 9.3 to be fulfilled and satisfied as soon as practicable following the execution and delivery of this Agreement and to consummate the transactions contemplated by this Agreement and (b) each of Purchaser and Purchaser Parent shall use its reasonable best efforts to cause the conditions set forth in Sections 9.2 and 9.3 to be fulfilled and satisfied as soon as practicable following the execution and delivery of this Agreement and to consummate the transactions contemplated by this Agreement.

6.6 Consents.

- (a) Unless otherwise agreed to by Purchaser and Purchaser Parent, Seller shall, and shall cause each of the Company Group Entities to, use its commercially reasonable efforts to obtain all Consents that are listed on Schedule 3.4 of the Seller Disclosure Schedule, all Consents with respect to Demerger Non-Transferred Contracts and any other consents otherwise needed to consummate the Transaction; provided, however, that in no event shall Seller or any Company Group Entity be required to make any payment to the relevant Consent counterparty or any Affiliate thereof to obtain any such Consent. Purchaser and Purchaser Parent shall use its commercially reasonable efforts to assist Seller and the Company Group Entities as applicable in its efforts to obtain any Consent referred to in the preceding sentence; provided, however, that in no event shall Purchaser or Purchaser Parent be required to make any payment to the relevant Consent counterparty or any Affiliate thereof to assist in obtaining any such Consent.
- (b) Purchaser and/or Purchaser Parent (as applicable), and Seller shall file, and to the extent applicable, Seller shall cause Company to file, the notification report, and all other documents to be filed in connection therewith, required by the HSR Act and the notification rules promulgated thereunder with the United States Federal Trade Commission and the Antitrust Division of the United States Department of Justice, as well as any other filings required under the Antitrust Laws of any other jurisdiction as reasonably determined by Purchaser, including any notification report or other documents required to be filed by the Anti-Monopoly Law of the People's Republic of China (the "AML") and the Anti-Monopoly Act of Japan (the "AMA"), as soon as practicable following the date hereof, but in any event within ten (10) Business Days for any filings required under the HSR Act, AML, AMA or any other Antitrust Law. All fees and expenses, including any applicable filing fees, in connection with any notification or other antitrust filing required in connection with the Transaction shall be borne equally by Purchaser and Seller.
- (c) Purchaser, Purchaser Parent and Seller shall furnish, (and Seller, to the extent applicable, shall cause Company to furnish), promptly to any requesting Governmental Authority any additional information requested in connection with any antitrust notification filed in such jurisdiction. Purchaser and Purchaser Parent shall use, and Seller shall, to the extent applicable, cause Company to use, commercially reasonable efforts to furnish to the other all information required for any application or other filing to be made pursuant to any applicable Legal Requirement in connection with the transactions contemplated by this Agreement. Subject to applicable Legal Requirements, each party shall promptly inform the other party of any oral communication with, and provide copies of written communications with, any Government Authority with jurisdiction over the enforcement of any applicable Antitrust Laws ("Government Antitrust Entity") regarding any such filings or any such transaction. Subject to applicable Legal Requirements, no party hereto shall independently participate in any formal meeting with any Government Antitrust Entity in respect of any such filings, investigation, or other inquiry without giving the other party prior notice of the meeting and, to the extent permitted by such Government Antitrust Entity, the opportunity to attend and participate. Subject to applicable Legal Requirements and subject to all applicable privileges (including the attorney client privilege), Purchaser Parent shall consult and cooperate, and Seller shall cause Company to consult and cooperate, with one another in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any party hereto relating to Proceedings under any Antitrust Law. Company or Purchaser Parent may, as it deems advisable and necessary, reasonably designate any competitively sensitive material provided to the other party under this Section 6.6(c) as "outside counsel only."

- (d) Subject to Section 6.6(e), each party shall use its commercially reasonable efforts to cause the applicable waiting period under each Antitrust Law where a notification or filing has been made to terminate or expire, and to cause any required approval or other Consent under each such Antitrust Law to be obtained, at the earliest possible date after the notification is made.
 - (e) Notwithstanding anything to the contrary in this Section 6.6 or elsewhere in this Agreement, it is expressly understood and agreed that:
- (i) Neither Purchaser, Purchaser Parent nor Seller shall be required to agree to any material concessions in connection with the obtainment of any Consent, or otherwise incur any material Liability or obligation of any kind, nor shall Seller be under any obligation to cause any Seller Group Entity or Company Group Entity to do any of the foregoing in this paragraph (e)(i),
- (ii) Neither Purchaser, Purchaser Parent nor Seller shall be under any obligation to make proposals, execute or carry out agreements or submit to orders providing for a Divestiture in order to obtain any Consent required under applicable Antitrust Laws, nor shall Seller be under any obligation to cause any Seller Group Entity or Company Group Entity to do any of the foregoing in this paragraph (e)(ii),
- (iii) Neither Purchaser, Purchaser Parent nor Seller shall have any obligation to litigate, contest or resist any Proceeding instituted (or threatened to be instituted) challenging any transaction contemplated by this Agreement as violative of any Antitrust Law ("Antitrust Proceeding"),
- (iv) Neither Purchaser, Purchaser Parent nor Seller shall have any obligation to take any action to have an Antitrust Order vacated, lifted, reversed or overturned,
 - (v) Seller shall cause Company not to conduct or agree to conduct a Divestiture without the prior written consent of Purchaser Parent.
- (f) Without limiting the generality of the undertakings pursuant to Section 6.6(e), it is expressly understood and agreed that Seller shall, with the prior consent of Purchaser, cause Company to use its commercially reasonable efforts to prevent the entry of any Antitrust Order; provided, however, that, Purchaser and its counsel shall be responsible for all such discussions with any Government Antitrust Entity (after consultation with Company and its counsel) to the maximum extent permitted by Legal Requirement and except as required by any Government Antitrust Entity.

6.7 Indebtedness and Intercompany Arrangements.

(a) Indebtedness.

- (i) Each of Seller and Purchaser Parent shall use its commercially reasonable efforts (without being required to pay any additional fees, premiums or other amounts) to have the DBJ Loans, and, if requested by Purchaser, the BOT Loan and the Shoko Loan, remain in full force and effect from and after Closing on the same terms as those terms in effect as of the date hereof, or refinanced on terms substantially economically equivalent in the aggregate thereto.
 - (ii) Seller shall ensure that all guarantees set forth in Schedule 3.10(c) of the Seller Disclosure Schedule shall be dealt with as set forth therein.
- (iii) No later than July 31, 2010, Seller shall provide to Purchaser a complete and correct list of all Indebtedness (other than Intercompany Indebtedness) of any Company Group Entity, including all Capital Leases, as of the date of this Agreement.

(b) Intercompany Arrangements:

- (i) Prior to the Closing, Seller shall cause (x) any Indebtedness owed by any Seller Group Entity to any Company Group Entity to be repaid in full and (y) all Deposits made by any Company Group Entity to any Seller Group Entity for cash management purposes to be returned to the Company Group Entity that made such Deposit.
- (ii) Prior to the Closing, Seller shall execute a release substantially in form and substance as attached hereto as <u>Exhibit E</u> (the "*Intercompany Release*").
- (iii) No later than July 31, 2010, Seller shall provide to Purchaser a complete and correct list of all Intercompany Indebtedness as of the date of this Agreement.
- (iv) No later than October 21, 2010, Seller shall provide to Purchaser a complete and correct list of all Indebtedness owed by any Seller Group Entity to a Company Group Entity, as at September 30, 2010. If the Closing does not occur on or prior to November 21, 2010, Seller shall provide an updated list, as of the last day of each subsequent month prior to Closing, no later than the twenty-first (21st) day of the following month, each of which shall set forth all Indebtedness owed by any Seller Group Entity to a Company Group Entity as of the last day of the previous month.

6.8 Financial Statements and Periodic Financial Information Seller shall promptly deliver to Purchaser copies of:

(a) During the Pre-Closing Period, the monthly management accounts of the Company Group Entities, which shall include a balance sheet, statement of income and statement of cash flows, each on a management accounting basis (the "*Monthly Management Accounts*"), prepared in a manner consistent with the Company's past practice and Relevant Books and Records, and in accordance with the Company's internal policies, as they are finalized between the date of this Agreement and the Closing Date, but in any event no later than the fifteenth (15th) day following the last day of each such month, *provided* that the statement of cash flows may be provided no later than the 21st day following the last day of each such month, and *provided further* that the Monthly Management Accounts that would otherwise be required to be finalized by August 15 under this Section 6.8(a) shall be finalized by August 22.

(b) During the Pre-Closing Period, the monthly financial statements of the Closing Company Group Entities (the "Monthly Financial Statements"), prepared in accordance with J-GAAP applied on a consistent basis throughout the periods covered, as they are finalized between the date of this Agreement and the Closing Date, but in any event no later than sixty (60) days following the last day of each such month (the "Monthly Financial Statement Date"), together with schedules setting forth in reasonable detail (including the calculation thereof), (i) Net Working Capital of the Closing Company Group Entities as at the relevant Monthly Financial Statement Date and a forecast of Actual Closing Net Working Capital, (ii) the aggregate amount of Indebtedness of the Closing Company Group Entities to or in favor of third parties (other than the Seller Group Entities) as at the relevant Monthly Financial Statement Date and a forecast of Closing Outstanding Third Party Indebtedness, (iii) the aggregate amount of Intercompany Indebtedness of the Company Group Entities as at the relevant Monthly Financial Statement Date and a forecast of Closing Intercompany Indebtedness, (iv) the amount of any Company Portion Seller Restructuring Plan Costs and Seller Incurred Restructuring Plan Costs incurred during such month, the aggregate amount of each such costs incurred since the date of this Agreement and a forecast of the aggregate amount of each such costs that will be incurred as of the Closing, in each instance assuming that the Closing is the Assumed Closing Date and (v) a reconciliation between the Monthly Financial Statements and the Monthly Management Accounts. Notwithstanding the foregoing, the statement of cash flows, and reconciliation thereof to the statement of cash flows delivered pursuant to Section 6.8(a), may be provided within two (2) weeks following the delivery of the relevant balance sheet and statement of income under this Section 6.8(b).

(c) Unaudited consolidated balance sheets, and the related unaudited consolidated statements of operations of the Pro Forma Company Group Entities and in the case of (iii), (iv) and (v) below, related unaudited consolidated statements of cash flow of the Pro Forma Company Group Entities and footnotes thereto (the "Interim Financial Statements") as of the end of, and for, (i) the quarter ended March 31, 2010, (ii) the six (6) months ended June 30, 2010, (iii) the six (6) months ended September 30, 2010 and 2009, (iv) the nine (9) months ended December 31, 2010 and 2009, in each case, prepared in accordance with U.S. GAAP and SAS 100 Review, and each such accounting standard applied on a consistent basis throughout the periods covered, as they are finalized, but in any event no later than October 31, 2010 (in the case of (i) and (ii)), December 15, 2010 (in the case of (iii) and(iv)) or March 16, 2011 (in the case of (v)), provided, however, that Seller's obligation to deliver any Interim Financial Statements under this Section 6.8(c) shall surve the Closing until such obligations are satisfied, and provided further, that notwithstanding the foregoing, Seller's obligation to deliver any Interim Financial Statements under this Section 6.8(c) shall terminate for any financial statements that would cover a period or a portion of a period after the Closing Date. Purchaser shall and shall cause the Company Group Entities to cooperate with Seller and its Representatives in order to enable Seller to fulfill any of the foregoing obligations, including providing necessary access to the Relevant Books and Records.

- 6.9 **Preparation of Pro Forma Audited Financial Statements of Company Group Entities**. Promptly after the date of this Agreement, Seller shall use its commercially reasonable efforts to cause to be prepared audited consolidated balance sheets, and the related consolidated statements of operations, consolidated statement of changes in stockholders' equity, and consolidated statement of cash flows, of the Pro Forma Company Group Entities as of and for the fiscal years ended March 31 in each of 2009 and 2010, together with the notes thereto, which accounts shall have been prepared in accordance with U.S. GAAP and U.S. GAAS, applied on a consistent basis throughout the periods covered, and audited by an international accounting firm of international reputation ("Audited Pro Forma Financial Statements"). Seller shall use its commercially reasonable efforts to deliver the Audited Pro Forma Financial Statements no later than the Closing.
- 6.10 **Conduct Prior to the Closing**. Seller agrees that during the Pre-Closing Period, unless Purchaser shall otherwise consent in writing (which Purchaser may refuse to do in its sole and absolute discretion) or except as expressly contemplated by this Agreement, Seller shall not sell, dispose of, mortgage, pledge or otherwise transfer or encumber any interest in any of the Purchased Shares.
- 6.11 **Capital Expenditures.** Prior to the Closing Date, except as otherwise agreed by Purchaser or Purchaser Parent, Seller shall, or shall cause the Company Group Entities to, make not less than eighty percent (80%) of the budgeted capital expenditures set forth, during the quarterly period set forth, on <u>Schedule 6.11</u>, and shall provide to Purchaser written confirmation upon making each such capital expenditures. Prior to the Closing Date, except as otherwise agreed by Purchaser or Purchaser Parent, Seller shall, or shall cause the Company Group Entities to, pay for the budgeted capital expenditures actually made in accordance with past practices and in any event no later than six (6) months after the commitment for such capital expenditure was made, and shall provide to Purchaser written confirmation upon such payment.

6.12 Certain Intellectual Property Matters.

- (a) On or before August 31, 2010, Seller shall provide to Purchaser a complete and correct list (as of June 30, 2010) of all Company IP Assets that are issued Patents, registered Trademarks, and registered Copyrights (including any pending applications for any of the foregoing). Such list will identify the jurisdictions in which each such item has been issued or registered or in which any such application for such issuance or registration has been filed, any applicable registration or application number, and any co- or joint owner or managing division (□□□□)(kanri busho).
- (b) On or before September 30, 2010, Seller shall provide to Purchaser a complete and correct list (to the Knowledge of Seller as of such date) of all actions that must be taken by the Company Group Entities within one hundred twenty (120) days following such date to maintain, perfect, preserve, or renew their rights in the Registered IP Assets, including the payment of any registration, maintenance or renewal fees and the filing of any documents, certificates, or responses to office actions.
- (c) During the Pre-Closing Period, Seller shall notify Purchaser if Seller obtains Knowledge of any misappropriation of any Trade Secrets owned by any Company Group Entity (or to be assigned to any Company Group Entity pursuant to the IP Assignment Agreement) that are material to the Business.

6.13 Updated Seller Disclosure Schedules.

- (a) No later than fifteen (15) days after the date hereof, Seller shall provide to Purchaser complete and correct copies of (i) updated Schedule 3.14(a) (Active Employees), Schedule 3.14(b) (Active Employees Requiring a Work Permit), Schedule 3.21 (Insurance Policies and Claims Filed Thereunder) and Schedule 3.27(b) (Indebtedness) of the Seller Disclosure Schedule, in each case providing the information described in the corresponding provision of Article III as of the date hereof but relating solely to matters occurring after the date set forth in the relevant Section to which such Schedule relates and (ii) an English translation of Schedule 3.32(b) (Demerger Non-Transferred IP Contracts) of the Seller Disclosure Schedule.
 - (b) No later than September 1, 2010, Seller shall provide to Purchaser a complete and correct copy of Schedule 3.33 of the Seller Disclosure Schedule.
- (c) No later than fifteen (15) days after the date hereof, Seller shall provide to Purchaser an updated <u>Schedule 6.11</u> to provide actual capital expenditures committed but unpaid as of the end of Q1/2010, anticipated cash expenditures for Q2/2010 and Q3/2010 (based on actual capital expenditures committed but unpaid as of the end of Q1/2010) and anticipated cash expenditures for Q4/2010 and Q1/2011 (based on budgeted capital expenditures Q2/2010 and Q3/2010).
 - 6.14 Third Party Patent License. Seller shall abide by and comply with the covenants set forth in Schedule 6.14.
- 6.15 **Recapitalization of Seller Loan Receivables**. Upon Purchaser's request, Seller shall discuss in good faith any proposal by Purchaser of the recapitalization of the Seller Loan Receivables.

6.16 July Disclosed Documents.

- (a) Between July 9, 2010 and the date hereof, Seller has disclosed a number of documents to the VDR that have not been reviewed by Purchaser or its Representatives (such documents, collectively, the "July Disclosed Documents"). With respect to the July Disclosed Documents, the parties agree as follows:
- (i) if any of the July Disclosed Documents are Material Contracts that require the consent of any counterparty thereto in connection with the Transaction or consummation thereof, Purchaser shall have the unilateral right, acting in good faith, to add any such Material Contract to the list of Contract Consents pursuant to Section 9.1(c)(iv) provided that Purchaser notifies Seller on or prior to September 1, 2010; and
- (ii) if Purchaser reasonably determines acting in good faith, that certain action should be taken prior to Closing with respect to any such July Disclosed Documents, including, without limitation, the termination or amendment of any Contract included in the July Disclosed Documents, then the closing conditions set forth in <u>Section 9.1</u> shall be expanded to include the completion of any such action prior to the Closing Date *provided* that Purchaser notifies Seller on or prior to September 1, 2010.

- (b) Seller hereby agrees that Purchaser shall have the right to update <u>Schedule 9.1(c)(iv)</u> and <u>Schedule 9.1(p)</u> in order to incorporate any changes or modifications contemplated by paragraphs (a)(i) and (a)(ii) above.
- (c) Seller hereby agrees to reimburse Purchaser fifty percent (50%) of the legal fees and expenses of Purchaser Parent's legal advisors with respect to its due diligence review of the July Disclosed Documents; *provided* that Seller's portion of such fees and expenses do not exceed two hundred fifty thousand dollars (\$250,000).
- 6.17 **Specified Employees.** Within thirty (30) days from the date hereof, Purchaser shall deliver to Seller a list of Active Employees divided into certain categories of general functionality, such as finance, sales, human resources and R&D/engineering (the "*Specified Employee List*").

ARTICLE VII

ADDITIONAL AGREEMENTS

- 7.1 **Intellectual Property License Agreement**. On or before the Closing, Company and Seller shall enter into cross-license agreement, substantially in the form attached hereto as <u>Exhibit F</u> (the "*Intellectual Property License Agreement*").
- 7.2 **Trademark License Agreement.** On or before the Closing, Seller and Company shall enter into a trademark license agreement, substantially in form and substance as attached hereto as Exhibit G (the "*Trademark License Agreement*").
- 7.3 **Transition Services Agreement.** On or before the Closing, Company and Seller shall enter into a transition services agreement, substantially in the form attached hereto as <u>Exhibit H-1</u> (the "*Transition Services Agreement*"). The terms and conditions applicable to such services will be set forth on schedules to be attached to the Transition Services Agreement, which, in the case of each such service that has been identified by Purchaser and Seller as of the date hereof, shall be in accordance with the terms described in attached <u>Exhibit H-2</u> and which, in all cases, shall otherwise in form and substance reasonably satisfactory to Purchaser and Seller.
- 7.4 **IP Assignment Agreement(s)**. On or before the Closing, Purchaser or its designated Affiliate and Seller shall enter into an intellectual property assignment agreement, substantially in form and substance attached hereto as <u>Exhibit I</u> (the "**IP Assignment Agreement**").

7.5 **Leases.** On or before the Closing:

(a) Purchaser and Seller shall enter into good faith negotiations for the lease by Company (or another Company Group Entity designated by Purchaser) of certain real property and premises (including, fixtures and equipment thereon) owned by a Seller Group Entity and used by a Company Group Entity as of the date hereof and as of the Closing Date, including those properties set forth on <u>Schedule 7.5(a)</u>, pursuant to lease agreements, each on the terms and conditions set forth in <u>Exhibit J-1</u> or as otherwise mutually agreed by the parties.

- (b) Purchaser and Seller shall enter into good faith negotiations for the sublease by Company (or another Company Group Entity designated by Purchaser) of certain real property parcels leased by or shared with a Seller Group Entity as of the date hereof and as of the Closing Date, including those properties set forth on Schedule 7.5(b), pursuant to a master sublease agreement on terms and conditions set forth in Exhibit J-2 or as otherwise mutually agreed by the parties.
- (c) Seller shall no later than thirty (30) days prior to the Closing provide to Purchaser a proposal for the consolidation of the Company's headquarters. Purchaser and Purchaser Parent shall thereafter discuss possible consolidation of Company's headquarters.
- 7.6 **Service Provider Agreements.** From the date of this Agreement until the Closing, Seller shall use its commercially reasonable efforts to cause each service provider set forth on Schedule 7.6(a) to provide, directly through a service agreement with a Company Group Entity designated by Purchaser, or indirectly through Seller, services on the same terms as those in effect as of the date hereof, or on terms substantially economically equivalent in the aggregate thereto, or such other terms reasonably acceptable to Purchaser. From the date of this Agreement until the Closing, Seller shall use its commercially reasonable efforts to cause each service provider set forth on Schedule 7.6(b) to provide directly through a service agreement with a Company Group Entity designated by Purchaser such services on the same terms as those in effect as of the date hereof, or on terms substantially economically equivalent in the aggregate thereto, or such other terms reasonably acceptable to Purchaser. After delivery of Schedule 3.33 by Seller and prior to the Closing Date, Purchaser shall, after consultation with Seller, update Schedule 7.6(a) and Schedule 7.6(b) to allocate the service providers set forth on Schedule 3.33 of the Seller Disclosure Schedule between Schedule 7.6(a) and Schedule 7.6(b).
- 7.7 **Secondment Agreement**. At the Closing, certain of the Seller Group Entities and certain of the Company Group Entities shall enter into six (6) secondment agreements as contemplated in <u>Annex III</u> and substantially in the forms attached hereto as <u>Exhibit K</u> (the "**Secondment Agreements**").
- 7.8 **No Solicitation of Competing Transactions**. Until the earlier of (a) the Closing and (b) the termination of this Agreement pursuant to its terms, Seller shall not, and they shall cause the other Company Group Entities and Seller Group Entities and the Representatives of the Company Group Entities and Seller Group Entities and Parent not to, directly or indirectly, (i) initiate, solicit or encourage any inquiries, or make any statements to third parties that may reasonably be expected to lead to any proposal concerning the sale of any Company Group Entity or the Business (whether by way of merger, purchase of capital shares, purchase of assets or otherwise) or a sale of any assets of any Company Group Entity or the Business or transaction the consummation of which would be inconsistent with or interfere with or prevent or delay, in any way whatsoever, the consummation of the Transaction (a "Competing Transaction"); or (ii) hold any discussions or enter into any agreements with, or provide any information or respond to, any third party concerning a proposed Competing Transaction or cooperate in any way with, agree to, assist or participate in, solicit, consider, entertain, facilitate or encourage any effort or attempt by any third party to do or seek any of the foregoing. If at any time prior to the earlier of (x) the Closing and (y) the termination of this Agreement pursuant to its terms, Seller or Company is approached in any manner by a third Person concerning a Competing Transaction (a "Competing Party"), Seller shall promptly, and in any event within twenty-four hours of such contact, inform Purchaser regarding such contact and furnish Purchaser with a copy of any inquiry or proposal, or, if not in writing, a description thereof, including the name of such Competing Party, and Seller shall keep Purchaser informed of the status and details of any future notices, requests, correspondence or communications related thereto, subject to the terms of any confidentiality agreement in effect between Seller and s

7.9 Non-Competition and Non-Solicitation Agreement.

- (a) Non-Competition.
- (i) For a period commencing on the Closing Date and ending on the date three (3) years following the Closing Date, Seller shall not and shall cause the other Seller Group Entities not to, directly or indirectly, anywhere in the world: (x) engage in, own, manage, carry on, promote, participate in or assist in, any Prohibited Business; (y) own or purchase any interest in, manage, operate or finance any Person that engages in any Prohibited Business or (z) enter into any marketing, manufacturing, designing, distribution, sales, service, support or other similar agreement with any Person if such agreement would support or assist such Person's Prohibited Business.
- (ii) Notwithstanding the foregoing, the obligations set forth in Section 7.9(a)(i) shall not apply to (1) any transaction solely among Parent and its Subsidiaries including Seller Group Entities; (2) any business conducted by OS Electronics Co., Ltd. and its Subsidiaries (collectively, "OS"); (3) any passive investment of up to 5% of the issued and outstanding ownership interests in a single Person; (4) the acquisition of any Person that engages in any Prohibited Business, provided that Seller Group Entity acquirer divests such acquired Prohibited Business within six (6) months of such acquisition; (5) holding of the Issued Shares; (6) any Contract or transaction involving or relating to the manufacture, design, sale, service, supply or procurement for or to any Seller Group Entity of semiconductor products or any parts or components thereof or any service related thereto.
- (iii) In addition, in the event that Seller merges into or with Parent, or ceases to exist as the result of any other business combination with Parent, Section 7.9(a)(i) shall continue to apply only to such Entities, departments, divisions or other business groups which belonged to a Seller Group Entity at any time between the date hereof and the effective date of such merger or other business combination.
- (b) Non-Solicitation of Employees of the Business. For a period commencing on the Closing Date and ending on the date five (5) years following the Closing Date (the "Nonsolicitation Period"), Seller shall not, and shall cause each of the Seller Group Entities not to, directly or indirectly, whether through a third Person or otherwise, recruit, solicit or knowingly induce any individual who (i) is employed by or is a consultant or has a similar contractual relationship with Purchaser or any Company Group Entity as of the date hereof or (ii) is an Eligible Business Employee (excluding any Seconded Employee who was not transferred to a Company Group Entity and returned to a Seller Group Entity after such Seconded Employee's secondment to a Company Group Entity), to accept employment with or enter into a consulting or similar relationship with any Person, other than Purchaser or any Company Group Entity, or terminate any such relationship with Purchaser or any Company Group Entity; provided, however, that the prohibition in this Section 7.9(b) shall not be deemed to have been violated by: (v) any Seller Group Entity's general solicitation of employment not specifically directed at individuals described above, including advertising of open positions, participating in job fairs, or other generalized forms of solicitation with respect to employment opportunities; (w) the implementation of Restructuring Plan; (x) any Seller Group Entity's solicitation of employment through any third party, provided that such third party has not been instructed to focus on any of the persons set forth in subsections (i) and (ii) of this Section 7.9(b); or (y) any Seller Group Entity's solicitation for employment directed at any individual who has no longer been employed or seconded to Purchaser or any Company Group Entity for a period of not less than six (6) months as of the date of such solicitation. Additionally, without Purchaser's prior written consent, during the Nonsolicitation Period, Seller shall not, and shall cause each of th

- (c) Non-Solicitation of Seller Employees. During the Nonsolicitation Period, Purchaser and Purchaser Parent shall not, and shall cause their Affiliates and the Company Group Entities to not, directly or indirectly, whether through a third Person or otherwise, recruit, solicit or knowingly induce any Person who is employed by Seller or any of its Affiliates as of the date hereof and is not an Eligible Business Employee, to accept employment with or enter into a consulting relationship with Purchaser or its Affiliates; provided, however, that the prohibition in the foregoing clause shall not be deemed to have been violated by (w) Purchaser's, its Affiliate's or any Company Group Entities' general solicitation of employment not specifically directed at employees of Seller or its Affiliates, including advertising of open positions, participating in job fairs, or other generalized forms of solicitation with respect to employment opportunities, or (x) any Purchaser's, its Affiliate's or any Company Group Entities' solicitation of employment through any third party, provided that such third party has not been instructed to focus on the employees of Seller or its Affiliates.
- (d) Non-Solicitation of Customers and Suppliers. During the Nonsolicitation Period, Seller shall not and shall cause the other Seller Group Entities not to, directly or indirectly, (i) call upon, for its own account or for the account of any other Person, any customer of the Company Group Entities as of the Closing Date or (ii) call upon for the account of any other Person any supplier or vendor of the Company Group Entities with respect to the Company Products as of the Closing Date, for the purpose of soliciting or selling products or services of any Prohibited Business. During the Nonsolicitation Period, Seller shall not, and shall cause the other Seller Group Entities to not, directly or indirectly, knowingly induce any customer, supplier or vendor of the Company Group Entities as of the Closing Date to cease doing business related to the Business as of the Closing Date, in whole or in part with, or otherwise knowingly interfere with the Business as of the Closing Date of, the Company Group Entities.

(e) <u>Scope</u>. The covenants contained in <u>Sections 7.9(a)</u> through (c) shall be construed as a series of separate covenants, one for each county, city, state and country throughout the world. Except for geographic coverage, each such separate covenant shall be deemed identical in terms to the covenant contained in <u>Sections 7.9(a)</u> through (c). If, in any Proceeding, a court refuses to enforce any of such separate covenants (or any part thereof), then such unenforceable covenant (or such part) shall be eliminated from this Agreement to the extent necessary to permit the remaining separate covenants (or portions thereof) to be enforced. In the event that the provisions of this <u>Section 7.9</u> are deemed to exceed the time, geographic or other scope limitations permitted by applicable Legal Requirement, then such provisions shall be reformed to the maximum time, geographic or scope limitations, as the case may be, permitted by applicable Legal Requirements.

Each party acknowledges that (i) the goodwill associated with the Business and customer relationships prior to the Closing are an integral component of the value of Company to Purchaser and are reflected in the payments to Seller from Purchaser contemplated by this Agreement, (ii) Seller's agreements as set forth herein are necessary to preserve the value of the Business, including their goodwill and customer relationships, for Purchaser following the Closing, and (iii) Purchaser's agreements as set forth herein are necessary to preserve the value of the Transaction for Seller and its Affiliates following the Closing. Each party also further acknowledges that the limitations of time, geographic scope and scope of activity agreed to in this Section 7.9 are reasonable.

7.10 Confidentiality.

(a) Each party agrees that, for a period of five (5) years from and after the Closing, it will not, and will use reasonable efforts to ensure that its Affiliates will not, through any action or inaction, use (except as contemplated by, or in connection with the performance and/or enforcement of, any Transaction Agreement), or disclose to any other Person, any Confidential Information relating to the other party (it being understood that following the Closing this Section 7.10 shall apply to the use or disclosure of any Confidential Information of any Company Group Entity or the Business by the Seller Group Entities and after the Closing shall not apply with respect to the use or disclosure of the Confidential Information of any Company Group Entity or the Business by Purchaser and its Affiliates); provided, however, that the foregoing prohibitions shall not apply to (i) subject to Section 7.10(b), disclosures that are required by any Legal Requirement (including any rule or regulation of the SEC or of a stock exchange which may require such disclosure) or by a Governmental Authority; (ii) information that is ascertainable or obtained from public or published information or is otherwise publicly known through no wrongful act of the using or disclosing party; (iii) information received from a Person not known after reasonably inquiry to the using or disclosing party to be under an obligation to keep such information confidential; (iv) information independently developed by the using or disclosing party without use of the other party's information; (v) information that was rightfully known by the disclosing party before receipt from the other party; and (vi) information disclosed to or filed with any Person for the purpose of obtaining consents to, or the financing of, the transactions contemplated by this Agreement. Notwithstanding anything herein to the contrary, each party may (without prior notification to, or approval or consent by, any other party) disclose to taxing authorities and/or to such party's repres

- (b) Notwithstanding Section 7.10(a), in the event a party is required to disclose Confidential Information of another party (in such event, such other party is a "Nondisclosing Party," and the party required to disclose is the "Disclosing Party") pursuant to any Legal Requirement, and would otherwise be prohibited from doing so under this Section 7.10, the Disclosing Party shall: (i) promptly notify the Nondisclosing Party of the existence, terms and circumstances surrounding such requirement; (ii) consult with the Nondisclosing Party on the advisability of taking legally available steps to resist or narrow such request; and (iii) if disclosure of such Confidential Information is required, furnish only that portion of the Confidential Information which the Disclosing Party is legally compelled to disclose and advise the Nondisclosing Party reasonably in advance of such disclosure so that the Nondisclosing Party may seek an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Confidential Information. The Disclosing Party shall not oppose actions by the Nondisclosing Party to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Confidential Information.
- (c) The provisions of this Section 7.10 shall not limit or restrict in any way any additional confidentiality obligations applicable to the parties and their Affiliates under any of the other Transaction Agreements.
- 7.11 **Public Announcements**. No party shall make, or cause or permit to be made, any press release or public announcement, or otherwise communicate with any news media, in respect of this Agreement or the Transaction without the prior written consent of the other parties, and the parties shall cooperate as to the timing and contents of any such press release or public announcement; *provided*, *however*, that any party and any Affiliate of a party may, without the prior consent of the other parties, (a) make such press release, public disclosure, securities law filings, or public announcement as may be required in such party's good faith opinion under applicable Legal Requirements, including stock exchange rules and regulations (including the filing of Forms 8-K announcing the Transaction and the filing of this Agreement with the Securities and Exchange Commission as may be required under applicable Legal Requirements) and (b) issue or make any press release or public announcement that is consistent with and not more expansive than press releases or other public announcements that were previously made in compliance with the terms of this Section 7.11.
- 7.12 **Current Public Information**. Purchaser Parent shall ensure that the condition set forth in paragraph (c) of Rule 144 of the Securities Act ("*Rule 144*") is satisfied at all times during the period commencing on the Closing Date and ending on the second (2nd) anniversary of the Closing Date, except to the extent the failure of Purchaser Parent to comply with the conditions set forth in paragraph (c) of Rule 144 results from the failure of the Company Group Entities to timely provide correct and complete financial information to Purchaser Parent.
- 7.13 **Purchaser Parent's Obligations**. Purchaser Parent shall cause Purchaser and, to the extent applicable, Purchaser's Affiliates, to satisfy and perform, and Purchaser Parent shall ensure the full satisfaction and performance by Purchaser and, to the extent applicable, Purchaser's Affiliates, of all of the obligations of Purchaser and, to the extent applicable, Purchaser's Affiliates, under this Agreement and each of the other Transaction Agreements.

7.14 **Benefit of Business Assets**. Seller shall, and shall cause Seller Group Entities to, ensure that the Company Group Entities own or have the right to use all Business Assets and Services through the Transaction Agreements as of the Closing. Without limiting the generality of the foregoing, if, through inadvertence or for any other reason, after the Closing, the Company Group Entities do not have the benefit of any Business Assets and Services held by a Seller Group Entity as of the Closing Date, Seller shall, and shall cause the Seller Group Entities to, transfer or make available, at no charge for assets and pursuant to the Transitional Services Agreement for services, such Business Assets and Services following the Closing to any Company Group Entity designated by Purchaser or Company.

7.15 Financial Arrangements.

- (a) Securitization of Receivables.
- (i) The parties agree that the Securitization of Receivables Transactions (and terms and conditions thereunder), in each case existing as of the date hereof, shall remain in full force and effect with respect to the Stage 1 Securitization Entities for a period of six (6) months from the Closing Date, after which such Securitization of Receivables Transactions shall terminate.
- (ii) The parties agree that the Securitization of Receivables Transactions (and terms and conditions thereunder), in each case existing as of the date hereof, shall remain in full force and effect with respect to the Stage 2 Securitization Entities until the date falling twelve (12) months after the Closing Date, after which such Securitization of Receivables Transactions shall terminate.
- (b) <u>Supplier Alternative Payment Systems</u>. Prior to the Closing Date, Seller shall cause the Company Supplier Alternative Payment System and the SSMC Supplier Alternative Payment System, in each case as in effect as of the date hereof, to be continued (other than the SSMC Supplier Alternative Payment System as it relates to Resona Banking Corporation, which Seller shall use its commercially reasonable efforts to be continued) from and after the Closing on such same terms for the applicable period set forth as follows:
- (i) Company: The Company Supplier Alternative Payment System shall be continued until the date falling twelve (12) months after the Closing Date.
- (ii) SSMC: The SSMC Supplier Alternative Payment Systems shall be continued until the date falling twelve (12) months after the Closing Date, as it relates to Sumitomo Mitsui Banking Corporation ("SMBC"), and until the date falling ninety (90) days after the Closing Date, as it relates to Resona Banking Corporation. If requested by SMBC, Purchaser and Purchaser Parent shall provide a memorandum agreeing to the termination of the SSMC Supplier Alternative Payment System as it relates to SMBC as of twelve (12) month anniversary of the Closing Date, and shall provide quarterly financial statements of Company and New SSMC during such twelve (12) month period.
- (c) Any guarantees or other similar instruments provided by any Seller Group Entity as of the date hereof in respect of the SSMC Supplier Alternative Payment System described above shall not be withdrawn and shall remain in full force and effect for the applicable period after the Closing Date as set forth in Section 7.15(b)(ii).

- (d) Between the date hereof and the Closing Date, Seller shall use its commercially reasonable efforts to cause that all suppliers of KSS transition to a promissory note system and any KSS Supplier Alternative Payment System shall, as of the Closing, be terminated to the extent that such transition to a promissory note system is completed.
- (e) Between the date hereof and the Closing Date, Seller shall use commercially reasonable efforts to enable the Company Group Entities to continue after the Closing Date the same banking relationships of the Company Group Entities as in place as of the date hereof. In furtherance of the foregoing, upon the request of Purchaser from time to time, Seller shall use commercially reasonable efforts to introduce Purchaser to such banks and arrange meetings in respect thereof.
- 7.16 **Transaction with Seller Group Entities**. The parties agree that agreements in effect as of the Closing Date between any Company Group Entity, on the one hand, and any Seller Group Entity, on the other hand, listed on Schedule 7.16(a), shall terminate as of the Closing Date and listed on Schedule 7.16(b), shall continue in effect. Seller and Purchaser shall negotiate in good faith and mutually agree on each of Schedule 7.16(a) and Schedule 7.16(b) prior to the Closing Date, and no party shall take any position inconsistent with the Transaction Agreements.
- 7.17 **Net Operating Losses**. Seller shall not use or carry back any net operating loss or other tax loss, or make any election that would limit or eliminate Purchaser's ability to use the net operating loss or other tax loss; *provided*, *however*, that Seller may use (a) up to two hundred million Yen (\fomation 200,000,000) of such net operating losses to offset taxable income of any Subsidiary of Seller arising in the current fiscal year and (ii) up to nineteen billion five hundred million Yen (\footnote{19},500,000,000) to offset any taxable income arising as a result of the Reorganization contemplated by Section 9.1(f)(ii).
- 7.18 **Reorganization.** Seller shall, and shall cause SSMC and New KSS to, effect the Reorganization in accordance with applicable Legal Requirements, abide by the terms of the SSMC Demerger Agreement and the KSS Demerger Agreement, as applicable, and conduct any and all processes, including processes in connection with the perfection of rights, using its commercially reasonable efforts to procure Consents from counterparties with respect to agreements having termination rights or are otherwise affected by the Reorganization and using its commercially reasonable efforts to procure Consents from counterparties with respect to agreements governed by the laws of a jurisdiction other than Japan necessary to transfer the demerged business as respectively contemplated by the SSMC Demerger Agreement and the KSS Demerger Agreement. No later than fifteen (15) days after the date hereof, Seller shall deliver to Purchaser the lists of specific assets, liabilities and Contracts to be transferred under the KSS Demerger Agreement in accordance with the format provided in Attachment 1(Details of Transferred Rights and Obligations) of the form of demerger agreement in Schedule 9.1(f)(i), which shall be agreed to by Purchaser, and subject to updates and revisions as mutually agreed by Seller and Purchaser prior to the execution of such demerger agreement. Seller shall cause New SSMC to conduct the demerged business and KSS to conduct the non-demerged business in the same manner as they were operated prior to the effective date of the respective demergers, including the transfer and/or reacquisition of permits/approvals, change in registrations with respect to Registered IP Assets and other assets, transfer and/or reacquisition of any insurance covering the transferred business or assets and submitting any notices and applications or filings required with Governmental Authorities. Seller shall bear all costs and expenses incurred in connection with or related to the Reorganization.

- 7.19 **Hazardous Material Waste.** Seller and Purchaser shall mutually cooperate to identify, promptly after the date hereof but no later than forty-five (45) days from the date hereof, all Hazardous Materials waste existing on any Real Property. Seller shall, and shall cause each of the Seller Group Entities to, use its commercially reasonable efforts to remove, at its sole cost and expense, all such Hazardous Materials waste, including Polychlorinated Biphenyl (PCB) waste, from such Real Property prior to Closing, apply promptly after the date hereof for any Governmental Approval required therefor, and shall in any event remove all such Hazardous Materials waste, including PCB waste, from such Real Property no later than the later of (i) three (3) months after the Closing Date and (ii) the date three (3) months after the receipt of Governmental Approval to remove such Hazardous Materials waste (the "*Removal Date*"). To the extent any Hazardous Materials waste are to be removed after the Closing Date, Purchaser Parent and Purchaser shall reasonably cooperate in the removal of such Hazardous Materials waste.
- 7.20 **SNA Assignment and Assumption Agreement**. On or before Closing, Seller shall cause SNA, and Purchaser shall or shall cause its designated Affiliate, to enter into an assignment and assumption agreement, substantially in the form attached hereto as Exhibit L (the "**Transferred SNA Business Assignment and Assumption**").
- 7.21 **SCE Assignment and Assumption Agreement.** On or before Closing, Seller shall cause SCE, and Purchaser shall or shall cause its designated Affiliate, to enter into an assignment and assumption agreement duly executed by SCE, substantially in the form attached hereto as <u>Exhibit M</u> (the "*Transferred SCE Business Assignment and Assumption*").
- 7.22 **Transfer of SSMPE Interest**. At Closing, Seller shall cause SANYO Marketing Corporation to sell all of its ownership interests in SSMPE ("SSMPE Ownership Interests") to an Entity in the Philippines to be designated by Purchaser, and Purchaser or Purchaser Parent shall cause such Entity in the Philippines to purchase from SANYO Marketing the SSMPE Ownership Interests, and the SSMPE Ownership Interests shall, upon such transfer, be deemed to constitute part of the Purchased Business hereunder.
- 7.23 **SLTI**. Seller shall bear and reimburse Purchaser for any costs and expenses arising after the Closing Date in connection with the dissolution of SLTI, including any severance or other amounts owed to any employee of SLTI.

ARTICLE VIII

EMPLOYEE MATTERS

8.1 Employee Matters. The parties agree to certain matters regarding the Active Employees as set forth in Annex III.

8.2 Restructuring Plan.

- (a) Attached hereto as Annex IV is (i) the restructuring plan formulated by the parties for the Company Group Entities and of the Business (the "Restructuring Plan"), and (ii) a schedule (the "Seller Restructuring Plan Costs Schedule") setting forth eleven billion two hundred million Yen (\(\xxi\)11,200,000,000,000) in restructuring costs (the "Seller Restructuring Plan Costs"), the timing and amount of each of the Seller Restructuring Plan Costs to be incurred in connection with carrying out apart of the Restructuring Plan and the party that will incur or assume such Seller Restructuring Plan Costs. Such portion of Seller Restructuring Plan Costs that is designated to be incurred by a Company Group Entity in the Seller Restructuring Costs Schedule shall be referred to as "Company Portion Seller Restructuring Plan Costs". For the avoidance of doubt, any further developments or amendments made to the Restructuring Plan pursuant to Section 8.2(b) shall not affect the Seller Restructuring Plan Costs Schedule provided herein.
 - (b) The parties acknowledge that the Restructuring Plan may be further developed and amended from time to time as mutually agreed by the parties.
- (c) As soon as practicable after the date hereof, Seller shall, or shall cause the other Seller Group Entities or Company to, submit and discuss the Restructuring Plan, to the Japan Unions and SET Union, and shall use commercially reasonable efforts to obtain all necessary consents or approvals prior to the Closing. Purchaser and Purchaser Parent shall reasonably assist Seller or the other Seller Group Entities to obtain such consents or approvals, including providing information reasonably requested by such unions. Prior to the Closing, Seller shall promptly notify Purchaser and Purchaser Parent in the event that the consent or approval of the SSV Union is required or necessary for the implementation of the Restructuring Plan as it pertains to SSV, including as a result of any change in applicable Legal Requirement (a "Change of Law").

ARTICLE IX

CONDITIONS TO CLOSING

- 9.1 **Conditions to Purchaser's Obligation to Close.** The obligations of Purchaser to consummate the Transaction shall be subject to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by Purchaser in writing:
- (a) <u>Representations and Warranties</u>. The representations and warranties made by Seller in this Agreement shall be true and correct in all material respects on the Closing Date with the same force and effect as if this Agreement had been executed on and as of the Closing Date, except that any such representations and warranties that expressly relate to a specified date shall be true and correct in all material respects only as of such date.

- (b) <u>Covenants</u>. Seller shall have duly performed in all material respects all of the obligations, agreements and covenants to be performed or complied with by it in this Agreement on or prior to the Closing Date (including agreements of Seller to cause the Seller Group Entities or Company Group Entities to take or refrain from taking certain actions).
 - (c) <u>Deliveries</u>. Seller shall have delivered or have caused to be delivered to Purchaser:
- (i) <u>Transfer Consents</u>: Consents in substantially the form attached hereto as <u>Exhibit N</u> ("*Transfer Consents*"), duly executed by each Transferred Employee;
- (ii) <u>Certificate of Representations and Warranties</u>. A certificate executed on behalf of Seller by its Representative Director or other duly authorized officer of Seller, certifying the matters in <u>Sections 9.1(a)</u> and (b);
- (iii) <u>Transaction Agreements</u>. Each of the Transaction Agreements (together with complete schedules thereto) to which any Seller Group Entity or any Company Group Entity is a party duly executed by the applicable Seller Group Entity or the applicable Company Group Entity, as the case my be;
- (iv) Consents. Consents, in form and substance reasonably satisfactory to Purchaser, required in connection with the Transaction pursuant to each of the Contracts listed on Schedule 9.1(c)(iv) (the "Contract Consents") as may be updated by Purchaser (x) pursuant to Section 6.16(a)(i), (y) no later than September 1, 2010 to include any Contract listed in Schedule 3.13(d) of the Seller Disclosure Schedule or Schedule 3.13(l) of the Seller Disclosure Schedule that Purchaser reasonably determines is material to the Business or (z) no later than September 1, 2010 to include any Contract of SSMC that Purchaser reasonably determines is material to the Business and requires consent of the counterparty in connection with the SSMC Demerger, in each case duly executed by the counter-party to each such Contract;
- (v) <u>Relevant Books and Records</u>. The Relevant Books and Records to the extent in the physical possession of a Seller Group Entity as of the Closing, *provided*, *however*, that to the extent that any such Relevant Books and Records are not primarily related, but are otherwise material, to the Business, Seller may deliver copies of such Relevant Books and Records;
- (vi) <u>Resignations</u>. To the extent that Purchaser has notified Seller in writing at least ten (10) Business Days prior to the Closing Date that any director of a Company Group Entity resign as of the Closing, evidence satisfactory to Purchaser of the resignation of such director effective as of the Closing; provided however, the requirement to deliver evidence of any such resignation shall be waived by Purchaser if Seller agrees to indemnify Purchaser of any Damages arising from the removal of such director;

- (vii) <u>Company and Seller Representative Director's Certificates</u>. Certificates duly executed by a Representative Director or other duly authorized officer of each of Company and Seller respectively, and each attaching correct and complete copies of Company's and Seller's respective commercial register, articles of incorporation, regulations of the board of directors (except in the case of Seller), and board resolutions indicating board approval of this Agreement, any applicable Transaction Agreements and the Transaction (including the transfer of the Purchased Shares from Seller to Purchaser);
- (viii) Other Documentation. Such other certificates, instruments or documents that are (A) reasonably requested by Purchaser in writing a reasonable time prior to the Closing Date and (B) required to transfer the Purchased Business in accordance with the terms hereof and consummate the Transaction, and to vest in Purchaser title to the Purchased Business, free and clear of all Encumbrances (other than licenses in the case of Transferred IP Assets); and
 - (ix) Financial Statements. The Audited Pro Forma Financial Statements contemplated by Section 6.9.
- (d) No Proceedings. Since the date of this Agreement, no Proceeding by a Person other than a Governmental Authority that has a reasonable likelihood of success as reasonably determined by Purchaser shall have been commenced and be continuing (a) involving any challenge to, or seeking material Damages or other material relief in connection with, the Transaction or (b) that is reasonably likely to have the effect of preventing or making illegal the Transaction.
 - (e) No Company Material Adverse Effect. No event that has had a Company Material Adverse Effect shall have occurred since the date hereof.
- (f) <u>Reorganization</u>. Seller shall have completed, or have caused KSS and SSMC to complete, prior to the Closing Date, the following transactions (collectively, the "**Reorganization**") on terms set forth in <u>Section 9.1(f)(i)</u>, <u>Section 9.1(f)(ii)</u> and <u>Section 7.18</u>, and Seller shall have delivered evidence of the completion of the Reorganization to Purchaser:
- (i) <u>Amorton Spin-Out</u>. A split off of KSS's Amorton business and all KSS employees into a new entity wholly owned by Seller ("*New KSS*") in accordance with an "absorb-type demerger" (*kyushu bunkatsu*) agreement that is substantially in the form of <u>Schedule 9.1(f)(i)</u>.
- (ii) <u>SSMC Spin-Out</u>. A split off of SSMC's entire business (including all the assets and liabilities, but excluding Gifu Employees and Gunma Employees) into a new entity wholly owned by Company ("*New SSMC*") in accordance with an "absorb-type demerger" (*kyushu bunkatsu*) agreement that is substantially in the form of <u>Schedule 9.1(f)(ii)-1</u>, and a sale and purchase of all of the shares of SSMC from Company to Seller or its designee pursuant to a share purchase agreement that is substantially in the form of <u>Schedule 9.1(f)(ii)-2</u>.
- (g) <u>Audited Pro Forma Financial Statements</u>. The 2009 Audited Pro Forma Financial Statements shall not be materially different in the aggregate in the reasonable determination of Purchaser from the 2009 Unaudited Financial Statements, but taking into account that (i) the 2009 Audited Pro Forma Financial Statements reflect the Reorganization and the other transactions contemplated by this Agreement, and the 2009 Unaudited Financial Statements do not and (ii) the 2009 Audited Pro Forma Fiancial Statements are prepared under U.S. GAAP and the 2009 Unaudited Financial Statements are prepared under J-GAAP.

- (h) Closing Cash and Cash Equivalents Balance. The Closing Cash and Cash Equivalents shall be no less than ten billion Yen (¥10,000,000,000).
- (i) <u>Bank Debt</u>. The aggregate amount (including principal and interest thereon) of Closing Outstanding Bank Debt shall not exceed twelve billion seven hundred million Yen (¥12,700,000,000).
- (j) <u>DBJ Loan</u>. The DBJ Loan shall either (i) remain in full force and effect on the same terms as those terms in effect as of the date hereof, or (ii) have been refinanced on terms reasonably acceptable to Purchaser or on terms substantially economically equivalent in the aggregate to the terms in effect as of the date hereof.
- (k) $\underline{\text{Other Indebtedness}}$. The aggregate amount of all Closing Outstanding Third Party Indebtedness shall not exceed one hundred fifty million Yen (\$150,000,000).
 - (1) Closing Intercompany Indebtedness. There shall be no Closing Intercompany Indebtedness.
- (m) <u>Due Diligence</u>. Purchaser shall have completed its due diligence review of the Business and the Company Group Entities, with results reasonably satisfactory to Purchaser.
- (n) <u>Service Provider Agreements</u>. Seller shall have made any necessary arrangements that enable such Company Group Entities to receive services from each service provider as set forth on <u>Schedule 7.6(a)</u> and <u>Schedule 7.6(b)</u>, directly from such service provider or indirectly through Seller (in the case of service providers set forth on <u>Schedule 7.6(b)</u>), on the same terms as those in effect as of the date hereof, or on terms substantially economically equivalent in the aggregate thereto or otherwise reasonably acceptable to Purchaser; *provided*, *however*, that if such Company Group Entities are not able to directly receive services from a service provider set forth in <u>Schedule 7.6(b)</u>, but Seller has made necessary arrangements that enable such Company Group Entities to receive such service indirectly through Seller, to the same extent and on the same terms as those in effect between Seller and such service provider as of the date hereof, or on terms substantially economically equivalent in the aggregate thereto or otherwise reasonably acceptable to Purchaser then the receipt of such services directly from such service provider shall not be a condition to Purchaser's obligation to consummate the Transaction pursuant to this <u>Section 9.1(n)</u> with respect to such service provider.
- (o) <u>Supplier Alternative Payment Arrangements</u>. Seller shall have delivered a certificate to Purchaser certifying that the SSMC Alternative Payment System as it relates to SMBC shall continue for a twelve (12) month period following the Closing Date.
 - (p) Additional Matters. Each matter set forth on Schedule 9.1(p), as such schedule may be updated pursuant to Section 6.16(b).

- 9.2 **Conditions to Seller's Obligation to Close.** The obligations of Seller to consummate the Transaction shall be subject to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by Seller in writing:
- (a) <u>Representations and Warranties</u>. The representations and warranties made by Purchaser and Purchaser Parent in this Agreement shall be true and correct in all material respects, on the Closing Date with the same force and effect as if this Agreement had been executed on and as of the Closing Date, except that any such representations and warranties that expressly relate to a specified date shall be true and correct in all material respects only as of such date.
- (b) <u>Covenants</u>. Purchaser and Purchaser Parent shall have duly performed all of the agreements and covenants and satisfied all of the conditions to be performed or complied with Purchaser or Purchaser Parent on or prior to the Closing Date.
 - (c) <u>Deliveries</u>. Purchaser shall have delivered or have caused to be delivered to Seller:
- (i) <u>Certificate of Representations and Warranties</u>. Certificates duly executed on behalf of Purchaser and Purchaser Parent by a duly authorized officer of Purchaser and Purchaser Parent, respectively, each certifying that the conditions set forth in <u>Sections 9.2(a)</u> and (b) have been duly satisfied;
 - (ii) Transaction Agreements. Duly executed copies of each of the Transaction Agreements to which Purchaser or any of its Affiliates is a party;
- (iii) New SSMC Support Letter. Support letter from Purchaser to New SSMC in connection with the SSMC Demerger in substantially the same form as Exhibit O; and
- (iv) <u>Purchaser Representative and Purchaser Parent Representative Certificates</u>. Certificates duly executed by a duly authorized officer of each of Purchaser and Purchaser Parent respectively, and each attaching correct and complete copies of Purchaser's certificate of formation and Purchaser Parent's articles of incorporation, respectively, regulations of the board of directors or other governing body, and resolutions of the board or other governing body approving this Agreement, any applicable Transaction Agreements and the Transaction.
- (d) <u>Securities Filings</u>. Purchaser Parent shall be current in all filings with the U.S. Securities and Exchange Commission on and as of the Closing Date and shall have delivered to Seller a certificate executed on behalf of Purchaser Parent by a duly authorized officer of Purchaser Parent to this effect.
- (e) <u>Foreign Exchange Notification</u>. Purchaser shall have filed with the relevant Governmental Authorities a prior notification of inbound direct investment under the Foreign Exchange and Foreign Trade Law of Japan, and the relevant waiting period (and any extension thereof) shall have expired.

- 9.3 **Conditions to Obligations of Each Party to Close**. The respective obligations of each party to this Agreement to consummate the Transaction shall be subject to the satisfaction, on or prior to the Closing Date, of each of the following condition(s), any of which may be waived by Purchaser or Seller, as applicable, in writing:
- (a) No Legal Impediments to Closing. There shall not be in effect any Order issued by any Governmental Authority preventing the consummation of the Transaction, awarding material Damages in connection with the Transaction, or requiring a Divestiture that would adversely affect the Business in any material respect. Since the date of this Agreement, no Proceeding by a Governmental Authority shall have been commenced (and be continuing) or threatened (i) involving any challenge to, or seeking material Damages or other material relief in connection with, the Transaction or (ii) that is reasonably likely to have the effect of preventing or making illegal the Transaction. There shall not be any Legal Requirement prohibiting Seller from selling or Purchaser from owning, operating or controlling the Business or that makes performance of any material post-Closing obligations under this Agreement or the consummation of the Transaction illegal.
- (b) <u>Antitrust Waiting Periods</u>. The waiting periods (and any extensions thereof) under the HSR Act, AML, AMA and any other filings required under any other applicable Antitrust Law shall have expired or been terminated and all filings required to be made prior to the Closing Date with, and all consents, approvals, permits and authorizations required under applicable Antitrust Laws to be obtained prior to the Closing Date from Governmental Authorities in connection with the execution and delivery of this Agreement and the consummation of the Transaction will have been made or obtained (as the case may be).

ARTICLE X

POST-CLOSING OBLIGATIONS

- 10.1 **Accounting and Tax**. If, after the Closing, Seller request access to or copies of the books and records of any Company Group Entity or such other cooperation as may be necessary for the settlement of accounts, including making corrections to previously settled accounts, or for dealing with tax inspections, Purchaser shall cause the Company Group Entities to promptly provide such requested access and cooperation.
- 10.2 **Environment**. Purchaser shall, and shall cause the Company Group Entities to, cooperate with Seller, at Seller's sole cost and expense, when Seller takes reasonable measures or countermeasures in connection with the contamination of the soil or groundwater or in connection with other environmental problems existing as of the Closing Date at the Seller's Tokyo factory (1-1, Sakata 1-chome, Oizumi-machi, Oura-gun, Gunma) or at the Gifu site (180 Omori, Anpachi-cho, Anpachi-gun, Gifu). Notwithstanding the foregoing, any and all liabilities in connection with such environmental problems shall be the sole responsibility of Seller and indemnifiable by Seller pursuant to Section 12.2.

ARTICLE XI

TERMINATION

- 11.1 **Circumstances for Termination**. At any time prior to the Closing, this Agreement may be terminated (without prejudice to any other remedies that may be available at law or in equity):
 - (a) by the mutual written consent of Purchaser and Seller;
- (b) by either Purchaser or Seller (i) upon any breach of the other party (including Purchaser Parent, if Seller is the terminating party) of any representation, warranty or covenant, which breach shall not have been cured within twenty (20) Business Days of being notified in writing thereof by the non-breaching party and which would result in the failure of any condition precedent set forth in Section 9.1(a) and 9.1(b) or 9.2(a) and 9.2(b), as the case may be, not being satisfied and (ii) the terminating party is not, on the date of termination, in material breach of any material provision of this Agreement;
- (c) by either Purchaser or Seller if the Closing has not occurred on or prior to March 31, 2011 for any reason; *provided, however*, that the right to terminate this Agreement under this <u>Section 11.1(c)</u> shall not be available to any party whose action or failure to act has been a principal cause of or resulted in the failure of the Closing to occur on or before such date or who is, on the date of termination, in material breach of this Agreement.
- 11.2 **Effect of Termination**. If this Agreement is terminated in accordance with Section 11.1, all obligations of the parties hereunder shall terminate, except for the obligations set forth in this Article XI and Article XIII; provided, however, that nothing herein shall relieve any party from liability for (a) the willful breach of any of its representations, warranties, covenants or agreements set forth in this Agreement (other than the obligation described in (b) below) or (b) a breach of such party's obligation to consummate the Transaction in accordance with the terms and subject to the conditions hereof.

ARTICLE XII

INDEMNIFICATION

- 12.1 Survival of Representations, Warranties and Special Indemnities.
- (a) All representations and warranties of Seller in this Agreement or any other Transaction Agreement shall survive the Closing until the date falling on the second (2nd) anniversary of the Closing Date (the "Escrow Release Date"); provided, however, that the Specified Reps shall survive the Closing until the date falling on the fifth (5th) anniversary of the Closing Date.
- (b) The indemnities of Seller set forth in Section 12.2(a)(iii) through Section 12.2(a)(xv) (each, a "Special Indemnity" and collectively, the "Special Indemnities") shall survive the Closing and remain in full force and effect until the following applicable dates:
- (i) the Special Indemnity in <u>Section 12.2(a)(iv)</u> (Taxes) shall survive until the date falling ninety (90) days after the expiration of the applicable statute of limitations:

- (ii) the Special Indemnity in <u>Section 12.2(a)(xv</u>) (Reorganization) shall survive until the date falling six (6) months after the Closing Date, unless there is Proceeding challenging the validity of the Reorganization prior to such date, in which case such Special Indemnity shall survive in respect of such challenge;
- (iii) the Special Indemnities in Sections 12.2(a)(iii) (Purchase Price Adjustment), 12.2(a)(v) (Pre-Closing Warranties), 12.2(a)(vi) (Amorton Business), 12.2(a)(viii) (Third Party Non-Patent Licenses), 12.2(a)(x) (Company Transaction Expenses), 12.2(a)(xiii) (Employee Inventions) and Section 12.2(a)(xiv) (Company Demerger), shall survive until the fifth (5th) anniversary of the Closing Date;
- (iv) the Special Indemnities in Section 12.2(a)(vii) (Patent Licenses) and 12.2(a)(ix) (Known Claims) shall survive until the sixth (6th) anniversary of the Closing Date;
 - (v) the Special Indemnity in Section 12.2(a)(xi) (Environmental) shall survive until the tenth (10th) anniversary of the Closing Date; and
- (vi) the Special Indemnity in <u>Section 12.2(a)(xii)</u> (Gas Contracts) shall survive until the date that is one (1) year after the term of the applicable contract referenced in <u>Section 12.2(a)(xii)</u>.
- (c) All representations and warranties of Purchaser or Purchaser Parent in this Agreement or any other Transaction Agreement shall survive the Closing until the date falling on the Escrow Release Date, *provided*, *however*, that the Specified Reps of Purchaser shall survive the Closing until the date falling five (5) years after the Closing Date and the representations and warranties set forth in <u>Section 5.8</u> shall survive the Closing indefinitely without limitation.

12.2 Indemnification by Seller.

- (a) Subject to the limitations set forth in this <u>Article XII</u>, from and after the Closing, Seller shall indemnify, defend and hold harmless Purchaser Parent, its Subsidiaries and its Representatives (collectively, "*Purchaser Parties*") from any and all liabilities, losses, loss of value, damages, Taxes, debts, obligations, costs or expenses, interest, awards, judgments, settlements, Orders, fines and penalties (including reasonable attorneys' fees, costs and expenses), whether or not involving a third-party claim (collectively, "*Damages*"), arising out of or resulting from:
- (i) any breach of a representation or warranty of Seller contained in this Agreement, in any Schedule or certificate delivered pursuant to this Agreement or in any other Transaction Agreement;
 - (ii) any breach of any covenant of Seller contained in this Agreement;

(iii) any Purchase Price Adjustment Amount payable by Seller as contemplated by <u>Section 1.4</u> to the extent not paid within the time period required in <u>Section 1.4</u>;

- (iv) any Pre-Closing Taxes;
- (v) any Pre-Closing Warranties;
- (vi) the operation and ownership of the Amorton Business;
- (vii) the matters set forth on Schedule 12.2(a)(vii) (Patent Licenses);
- (viii) the matters set forth on Schedule 12.2(a)(viii) (Third Party Non-Patent Licenses);
- (ix) the matters set forth on Schedule 12.2(a)(ix) (Known Claims);
- (x) any Company Transaction Expenses;
- (xi) any of: (A) the presence of any Hazardous Materials on any Property as of the Closing Date, (B) the violation of any Environmental and Safety Law by the Company Group Entities or the Seller Group Entities prior to the Closing Date, (C) any Liability under any Environmental and Safety Law based on the conduct of the Company Group Entities or Seller Group Entities prior to the Closing Date, or (D) any Liability of any predecessor of any of the Company Group Entities or Seller Group Entities arising prior to the Closing Date under any Environmental and Safety Law, which Liability was assumed by the Company Group Entities by Contract or was imposed upon any of them pursuant to any applicable Legal Requirement;
- (xii) the termination of any Gas Contracts prior to the expiration of the term (not taking into account any early termination rights that a party may have) as set forth in each respective Gas Contract;
- (xiii) any royalties, compensation or other fees that become payable prior to the Closing to any employee or former employee of a Seller Group Entity or a Company Group Entity, or that became payable to such individual as of Closing in connection with the Transaction, under the internal rules of the Seller Group Entities, as applicable, in connection with Intellectual Property Rights invented by such individual prior to the Closing;
- (xiv) any failure to transfer any Contracts that were to be transferred from Seller to Company pursuant to the Company Demerger (including those Contracts set forth on Schedule 3.32(a)) of the Seller Disclosure Schedule; and
 - (xv) the invalidity of the KSS Demerger or the SSMC Demerger.
- (b) Materiality standards or qualifications in any representation, warranty or covenant shall only be taken into account in determining whether a breach of or default in connection with such representation, warranty or covenant (or failure of any representation or warranty to be true and correct) exists, and shall not be taken into account in determining the amount of any Damages with respect to such breach, default or failure to be true and correct.

- (c) Since following the Closing, Company will be owned by Purchaser, the parties agree that any recovery by Purchaser after the Closing pursuant to this <u>Article XII</u> shall be against Seller, who will have no right of reimbursement, contribution or other recovery against Company or other Company Group Entities.
- (d) Subject to Section 12.5(e), the representations, warranties, covenants and obligations of Seller, and the rights and remedies that may be exercised by the Purchaser Parties based on such representations, warranties, covenants and obligations, will not be limited or affected by any investigation conducted by any of the Purchaser Parties with respect to, or any knowledge acquired (or capable of being acquired) by the Purchaser Parties at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to, the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or obligation. The waiver by Purchaser of any provision of this Agreement or any Transaction Agreement will not affect or limit the provisions of this Agricle XII.
- (e) Without limiting the effect of any other limitation set forth in Section 12.5 or elsewhere in this Agreement, if at any time after the date of this Agreement and prior to the Closing Date, (i) Seller discloses in writing to Purchaser and Purchaser Parent a matter occurring after the date of this Agreement that would constitute a breach of or inaccuracy in the representations and warranties of Seller in this Agreement, (ii) such inaccuracy or breach resulted in any condition set forth in Section 9.1 or 9.3 not being satisfied, and (iii) Purchaser and Purchaser Parent elect to waive the unsatisfied condition and consummate the Transaction, then no representation or warranty of Seller in this Agreement shall be deemed to have been breached in respect of such matter to the extent of such written disclosure by Seller.
 - (f) Indemnification Escrow Period; Distribution upon Escrow Release Date.
- (i) The Escrow Account shall be held in escrow by the Escrow Agent as security in favor of Purchaser for any Purchase Price Adjustment Amount payable by Seller in favor of Purchaser pursuant to Section 1.4 and as security in respect of Seller's indemnification obligations pursuant to this Article XII. Subject to the following requirements, the Escrow Account shall be in existence immediately following the Closing and shall terminate at 5:00 p.m., Pacific time, on the Escrow Release Date (the "Escrow Period").
- (ii) Promptly after the date falling twelve (12) months after the Closing Date, the Escrow Agent shall release from the Escrow Account and pay to Seller an amount of cash equal to two billion one hundred fifty million Yen (¥2,150,000,000) less (i) any amount paid from the Escrow Account for satisfaction of any Damages, and (ii) such amount of the Escrow Account set forth in any Claims Notice delivered to the Escrow Agent with respect to any Asserted Liability that remains unresolved as of such date.

(iii) Except as set forth below, promptly after the Escrow Release Date the Escrow Agent shall release from the Escrow Account and pay to Seller an aggregate amount of cash equal to the remaining portion of the Escrow Amount after satisfaction of any Damages. Notwithstanding anything in this Agreement to the contrary, the Escrow Period shall not terminate and the Escrow Release Date shall be extended with respect to such amount of the Escrow Account set forth in any Claims Notice delivered to the Escrow Agent prior to termination of the Escrow Period. As soon as any such Damages and claims has been resolved, the Escrow Agent shall release from the Escrow Account and pay to Seller an aggregate amount of cash equal to the remaining portion of the Escrow Amount not required to satisfy any other such unresolved Damages or claims.

12.3 Indemnification by Purchaser

- (a) Subject to the limitations set forth in this <u>Article XII</u>, from and after the Closing, Purchaser and Purchaser Parent shall jointly and severally indemnify, defend and hold harmless Seller, its Affiliates and its Representatives (collectively, "Seller Parties") from any and all Damages, arising out of, relating to or resulting from:
- (i) any breach of a representation or warranty of Purchaser or Purchaser Parent contained in this Agreement, in any certificate delivered pursuant to this Agreement or in any other Transaction Agreement (without taking into account any "materiality," "material adverse effect" or similar qualifiers included therein);
 - (ii) any breach of any covenant of Purchaser or Purchaser Parent contained in this Agreement;
 - (iii) any purchase price adjustments contemplated by Section 1.4; and
- (iv) any guarantee by any Seller Group Entity for the benefit of any Company Group Entity or the Business, that continues or is provided after the Closing, in each case, solely to the extent such Damages arise or result from any non-payment by the relevant Company Group Entity after the Closing Date.
- (b) The representations, warranties, covenants and obligations of Purchaser and Purchaser Parent, and the rights and remedies that may be exercised by the Seller Parties based on such representations, warranties, covenants and obligations, will not be limited or affected by any investigation conducted by the Seller Parties with respect to, or any knowledge acquired (or capable of being acquired) by any of the Seller Parties at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to, the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or obligation. The waiver by Seller of any provision of this Agreement will not affect or limit the provisions of this Article XII.

12.4 Procedures for Indemnification.

- (a) Notice. Promptly after any Purchaser Party or Seller Party, as the case may be, entitled to indemnification hereunder (an "Indemnified Party")

 (i) becomes aware of circumstances that have resulted in, or which are reasonably likely to result in, Damages for which the Indemnified Party may be entitled to indemnification ("Potential Indemnification Event") for or (ii) receives written notice of any third-party demand, claim or circumstances which, with the lapse of time, the giving of notice or both, reasonably could give rise to a claim or the commencement (or threatened commencement) of any Proceeding that may result in Damages for which the Indemnified Party may be indemnified hereunder (the demand, claim or Proceeding referred to in this clause (ii), an "Asserted Liability"), the Indemnified Party shall give notice thereof (a "Claims Notice") to any other party or parties who may become obligated to provide indemnification hereunder (an "Indemnifying Party"). The Claims Notice shall describe the Asserted Liability or Potential Indemnification Event in reasonable detail to the extent known, and shall indicate the amount (estimated, if necessary and if reasonably capable of estimation) of the Damages that have been, or that may be, suffered by the Indemnified Party (the "Claimed Amount") and the specific representations and warranties or covenants alleged to be breached, or any other clauses of Section 12.2 under which indemnification is being sought. The Claims Notice may be amended on one or more occasions with respect to the amount of the Asserted Liability or the Claimed Amount at any time prior to final resolution of the obligation to indemnify relating to the Asserted Liability, the Claimed Amount or Potential Indemnification Event. If a Claims Notice is not provided promptly as required by this Section 12.4, the Indemnified Party nonetheless shall be entitled to indemnification by the Indemnifying Party except to the extent the Indemnifying Party is actually materially prejudiced by such late receip
- (b) <u>Opportunity to Contest</u>. The Indemnified Party shall proceed with the defense of any Asserted Liability. The Indemnified Party shall have the right to settle, adjust or compromise such Asserted Claim; *provided, however*, that if the Indemnified Party settles, adjusts or compromises any such Asserted Liability without the consent of the Indemnifying Party, such settlement, adjustment or compromise shall not be conclusive evidence of the amount of Damages incurred by the Indemnified Party in connection with such Asserted Liability; and *provided, further*, that if the Indemnifying Party consents to any such settlement, adjustment or compromise, such consent shall not be evidence that such Asserted Claim is subject to indemnification by the Indemnifying Party under this <u>Article XII</u> and shall not be taken into account by any arbitrator or other trier of fact in determining whether the Indemnifying Party has any liability with respect to such settlement, adjustment or compromise, the Asserted Liability or any facts relating to the Asserted Liability. The Indemnifying Party may participate, at its own expense, in the defense of an Asserted Liability. The Indemnifying Party and the Indemnified Party shall reasonably cooperate with each other as to all Asserted Liabilities, shall make reasonably available to each other, as reasonably requested, all information, records, and documents until the termination of any Asserted Liability. The Indemnifying Party and the Indemnified Party also shall make reasonably available to each other, as reasonably requested, their respective personnel, agents, and other representatives who are responsible for preparing or maintaining information, records, or other documents, or who may have particular knowledge with respect to any Asserted Liability.
- (c) Notwithstanding the foregoing, the provisions of <u>Schedule 12.4(c)</u> shall apply with respect to the special indemnities set forth in (i) <u>Section 12.2(a)</u> (<u>vii)</u> (Patent Licenses), (ii) Section <u>12.2(a)(viii)</u> (Third Party Non-Patent Licenses) and (iii) <u>Section 12.2(a)(ix)</u> (Known Claims), solely relating to item 1 of <u>Schedule 12.2(a)(ix)</u>.

12.5 Limitations on Indemnification.

- (a) No amount shall be payable by Seller to any Purchaser Party pursuant to Section 12.2(a) (other than Section 12.2(a)(ii)) unless the aggregate amount of Damages indemnifiable by Seller under Section 12.2(a) (except for Section 12.2(a)(ii)) exceeds an amount (the "Seller Basket") equal to three hundred twenty-four million Yen (¥324,000,000), at which point Seller shall become liable for all such Damages, including the Damages reflected in the Seller Basket.
 - (b) Notwithstanding anything to the contrary contained in this Agreement:
- (i) the maximum amount of aggregate indemnifiable Damages which may be recovered from Seller under <u>Section 12.2(a)(i)</u> (other than with respect to a Specified Rep) shall be an amount equal to four billion nine hundred million Yen (¥4,900,000,000);
- (ii) the maximum amount of aggregate indemnifiable Damages which may be recovered from Seller under Section 12.2(a)(i) with respect to a Specified Rep (other than a Fundamental Rep) shall be an amount equal to eleven billion six hundred million Yen (¥11,600,000,000);
- (iii) the maximum amount of aggregate indemnifiable Damages which may be recovered from Seller under Sections 12.2(a)(y) (Pre-Closing Warranties), (vi) (Amorton Business), (vii) (Third Party Licenses), (x) (Company Transaction Expenses), (xii) (Gas Contracts), (xiii) (Employee Inventions), and (xiv) (Company Demerger), shall be an amount equal to fifty percent (50%) of the Purchase Price as may be adjusted pursuant to Section 1.4;
- (iv) the maximum amount of aggregate indemnifiable Damages which may be recovered from Seller under (1) Section 12.2(a)(i) with respect to a Fundamental Rep, (2) pursuant to any of Sections 12.2(a)(iii) (Purchase Price Adjustment), (iv) (Taxes), (vii) (Patent Licenses), (ix) (Known Claims), (xi) (Environmental) and (xv) (Reorganization) and (3) Section 12.2(a)(ii) in respect of any breach of any pre-Closing covenant that did not constitute a Willful Breach, shall be an amount equal to the Purchase Price as may be adjusted pursuant to Section 1.4
 - (v) Notwithstanding the foregoing, in no event shall
 - (x) the aggregate amount of Damages recoverable from Seller by the Purchaser Parties under the subsections of Section 12.2(a) referred to in Sections 12.5(b)(ii) and 12.5(b)(ii) exceed eleven billion six hundred million Yen (\xi11,600,000,000),
 - (y) the aggregate amount of Damages recoverable from Seller by the Purchaser Parties under the subsections of Section 12.2(a) referred to in Sections 12.5(b)(ii), 12.5(b)(iii) and 12.5(b)(iii) exceed an amount equal to fifty percent (50%) of the Purchase Price as it may be adjusted pursuant to Section 1.4, and
 - (z) the aggregate amount of Damages recoverable from Seller by the Purchaser Parties under the subsections of Section 12.2(a) referred to in Sections 12.5(b)(ii), 12.5(b)(ii), 12.5(b)(iii) and 12.5(b)(iv) exceed an amount equal to one hundred percent (100%) of the Purchase Price as it may be adjusted pursuant to Section 1.4.

- (c) The limitations set forth in subsections (a) and (b) of this Section 12.5 shall not apply to (i) any Seller indemnification obligation arising out of, relating to or resulting from fraud by Seller with the intent to deceive or intentional misrepresentation by Seller with the intent to deceive or (ii) any claim pursuant to Section 12.2(a)(ii) with respect to (A) a Willful Breach by Seller of any pre-Closing covenant or (B) breach of any covenant required to be performed by Seller after the Closing.
- (d) Notwithstanding anything to the contrary contained in this Agreement, for purposes of this <u>Article XII</u>, "Damages" shall exclude, and no Indemnified Party shall be entitled to recover pursuant to this <u>Article XII</u>, any remote, unforeseeable or punitive damages.
- (e) The amount of any Damages that are subject to indemnification under this <u>Article XII</u> shall be calculated net of the amount of any insurance proceeds or reimbursements actually received by any Indemnified Party or any Affiliate of any Indemnified Party in connection with such Damages or any of the events or circumstances giving rise or otherwise related to such Damages. Each Indemnified Party shall seek, and shall cause each of its Affiliates to seek, recovery under all insurance policies covering any Damages to the same extent as it would if such Damages were not subject to indemnification hereunder. In the event that an insurance recovery is made by an Indemnified Party or any Affiliate of an Indemnified Party with respect to any Damages for which such Indemnified Party has already been indemnified hereunder, then a refund equal to the aggregate amount of the insurance recovery shall be made promptly to the Indemnifying Party.
- (f) To the extent that an Indemnified Party is entitled to indemnification pursuant to this <u>Article XII</u>, the Indemnifying Party shall be entitled to exercise, and shall be subrogated to, any rights and remedies (including rights of indemnity, rights of contribution and other rights of recovery) that the Indemnified Party or any of the Indemnified Party's Affiliates may have against any other Person with respect to any Damages, circumstances or matter to which such indemnification is directly or indirectly related. The Indemnified Party shall assign all rights to pursue any claims against, or otherwise recover amounts from, such other Person, and otherwise take any action that the Indemnifying Party may reasonably request for the purpose of enabling the Indemnifying Party to perfect or exercise the right of subrogation of the Indemnifying Party under this <u>Section 12.5(f)</u>.
- (g) Promptly after an Indemnified Party becomes aware of any event or circumstance that could reasonably be expected to constitute or give rise to any breach of any representation, warranty or covenant of the Indemnifying Party contained in this Agreement or any other claim for indemnification pursuant to this Article XII, the Indemnified Party shall take all reasonable steps to mitigate and minimize all Damages that may result from or relate to such breach or claim.
- (h) No party shall be able to bring an indemnification claim pursuant to Section 12.2(a)(ii) or Section 12.3(a)(ii), as applicable, with respect to a breach of (i) any pre-Closing covenant unless it has delivered a Claims Notice with respect to such breach prior the second (2nd) anniversary of the Closing Date; and (ii) a covenant required to be performed after the Closing that has a stated duration unless it has delivered a Claims Notice with respect to such breach prior to the later of (A) the third (3nd) anniversary of the last day that such covenant is in effect under the Agreement; and (B) the fifth (5th) anniversary of the date on which a breach of such covenant occurs.

- (i) Solely with respect to indemnification for a breach of the representation and warranty of Seller contained in <u>Section 3.13(h)</u> or <u>Section 3.8(c)</u> (solely to the extent such a breach of <u>Section 3.8(c)</u> would also constitute a breach of <u>Section 3.13(h)</u>), Damages for which Seller shall indemnify Purchaser Parties shall be limited to Damages attributable to the period prior to the sixth (6th) anniversary of the Closing Date.
- 12.6 **Remedies Exclusive**. Subject to Section 15.10. from and after Closing, the right of an Indemnified Party to receive indemnification payments from the Indemnifying Party pursuant to this Article XII shall be the sole and exclusive right and remedy exercisable by the Indemnified Party with respect to any and all claims under, relating to or arising out of this Agreement or with respect to breaches of representations or warranties contained in any Transaction Agreement, and no Person shall have any other entitlement, remedy or recourse, whether in contract, tort or otherwise, it being agreed that all of such other remedies, entitlements and recourse are expressly waived and released by the parties hereto to the fullest extent permitted by applicable Legal Requirements, *provided, however*, that this Section 12.6 shall not preclude any party from asserting any other right or seeking any other remedies for fraud with the intent to deceive or for Willful Breach by such other party that committed such fraud with the intent to deceive or Willful Breach; and *provided further, however*, if an Indemnified Party is entitled to indemnification under more than one of the provisions of Section 12.2(a) or Section 12.2(b), as the case may be, such Indemnified Party may elect, in its sole discretion, the provision of Section 12.2(a) or Section 12.2(b), as the case may be, under which to seek indemnification.
- 12.7 **Nature of Indemnification Payments**. Any indemnification payment made by Seller or Purchaser pursuant to this <u>Article XII</u> shall be deemed by the parties as an adjustment to the Purchase Price.
- 12.8 **Offset.** If any claim as to which Purchaser or Seller, as applicable, has delivered a Claims Notice is pending or unresolved at the time that any payment is due from Purchaser or any Company Group Entity to Seller or any other Seller Group Entity, or from Seller or any other Seller Group Entity to Purchaser or any of its Affiliates, as applicable, pursuant to any of the Transaction Agreements or otherwise, Purchaser or Seller, as applicable, shall have the right, in addition to any other rights or remedies of Purchaser or Seller, as applicable (whether pursuant to this Agreement or applicable Legal Requirements) to withhold from such payment an amount equal to the amount of such claim until such matters are finally resolved in accordance with this Agreement. If it is finally determined that such claims are covered by this <a href="https://doi.org/10.1007/nn.10

ARTICLE XIII

TAX MATTERS

- 13.1 **Preparation and Filing of Tax Returns**. Purchaser shall prepare (or cause to be prepared) and file (or cause to be filed) each Tax Return required to be filed by Company or any other Company Group Entities after the Closing Date for a taxable period beginning before the Closing Date. To the extent any Tax shown as due on such Tax Return is payable by Seller (taking into account indemnification obligations hereunder), such Tax Return shall be provided to Seller at least thirty (30) days prior to the due date for filing such return (or, if required to be filed within thirty (30) days of the Closing Date, as soon as possible following the Closing Date); and Seller shall have the right to review and consent to such Tax Return, which consent shall not be unreasonably withheld, conditioned or delayed. The failure of Seller to propose any changes to any such Tax Return within fifteen (15) days of receipt thereof shall constitute consent. The Seller shall pay to Purchaser, on or before the due date thereof, the amount of Taxes shown as due on such Tax Returns that are payable by Seller (taking into account indemnification obligations hereunder).
- 13.2 **Amended Returns; Tax Elections**. Except consistently with a resolved audit, Seller may not amend (or permit the amendment of) a Tax Return of any Company Group Entity, or make, amend or revoke (or permit the making, amendment, or revocation of) any Tax election of any Company Group Entity, in each case, with respect to a taxable period beginning before the Closing Date without the prior written consent of Purchaser, which consent shall not be unreasonably withheld, conditioned or delayed. The failure of Purchaser to propose any changes to any such amendment or revocation of a Tax Return within thirty (30) days of receipt thereof shall constitute consent.
- 13.3 **Transfer Taxes**. All transfer, documentary, sales, use, stamp, registration, value-added and other similar Taxes and fees ("*Transfer Taxes*") incurred in connection with the transactions contemplated by this Agreement shall be borne by Seller and timely paid when due; *provided*, *however*, that if any Japanese consumption tax is imposed on the payment of the Purchase Price allocated to the Transferred IP Assets as a result of such Transferred IP Assets being transferred to a Japanese Entity designated by Purchaser Parent or Purchaser, such consumption tax shall be borne by Purchaser or its Affiliates, as applicable, and Purchaser shall pay to Seller an amount equal to such consumption tax within five (5) Business Days after the Allocation is determined as final in accordance with <u>Section 1.3(b)(ii)</u>. Except as otherwise required by applicable Tax law, Seller shall prepare and file, at its sole cost and expense, all necessary Tax Returns and other documentation with respect to all such Transfer Taxes.
- 13.4 **Straddle Period Allocations**. For purposes of this Agreement, Taxes of any Company Group Entity incurred with respect to a taxable period that includes but does not end on the day immediately preceding the Closing Date, shall be allocated to the portion of the taxable period ending on the day immediately preceding the Closing Date (i) except as provided in (ii) and (iii) below, to the extent feasible, on a specific identification basis, according to the date of the event or transaction giving rise to the Tax, and (ii) except as provided in (iii) below, with respect to periodically assessed ad valorem Taxes and Taxes not otherwise feasibly allocable to specific transactions or events, in proportion to the number of days in such taxable period, and (iii) in the case of any Tax based upon or related to income or receipts, in an amount equal to the Tax which would be payable if the relevant taxable period ended on the day immediately preceding the Closing Date.

- 13.5 **Termination of Tax Sharing Agreements**. All Tax sharing, allocation, indemnity or similar agreements with respect to or involving Company or any other Company Group Entity shall be terminated as of the Closing Date and, after the Closing Date, Seller shall indemnify Purchaser and hold Purchaser harmless for any liability, including successor liability, under any such Tax sharing allocation, indemnity or similar arrangement.
- 13.6 Cooperation, Access to Information, and Record Retention. If any of the Company Group Entities, Purchaser, or Seller receives any notice of a pending or threatened Tax assessment, adjustment, audit, litigation or other Proceeding ("Tax Audit") relating to any Company Group Entities, which may give rise to a liability of another party hereto (including but not limited to indemnification obligations hereunder), (i) the first party shall promptly notify such other party within ten (10) Business Days of receiving such notice; and (ii) both parties shall keep each other informed on a regular basis regarding the status thereof. If the Tax Audit relates to the period prior to the Closing, Seller may, upon providing, within fifteen (15) days of its receipt of notice, written notice to Company and Purchaser, assume the defense of any such Tax Audit, and thereafter conduct the defense thereof at its own expense. If Seller elects to elects to defend such Tax Audit, Purchaser shall, and shall cause Company Group Entities to, make available to Seller or its representatives all records and other materials reasonably required by Seller for use in contesting such Tax Audit, and cooperate fully with Seller in the defense thereof. The Company Group Entities, Purchaser, and Seller shall cooperate, and cause their representatives and Affiliates to cooperate, to the extent reasonably requested by any other party in connection with the preparation and filing of Tax Returns of any Company Group Entities. Such cooperation shall include the retention and (upon the other party's request) the provision of records and information which are reasonably relevant to any such Tax Return and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Company Group Entities, Purchaser and Seller shall (i) retain all books and records with respect to Taxes and Tax Returns of the Company Group Entities for any taxable period beginning before the Closing Date until the expiration of the statute of limitations (and, to the extent notified, any extensions thereof) with respect thereto; (ii) abide by all record retention agreements entered into with any Governmental Authority, with respect to Taxes; and (iii) give the other parties reasonable written notice prior to transferring, destroying or discarding any such books and records and, if another party so requests, allow such party to take possession of such books and records. The Company Group Entities, Purchaser, and Seller shall cooperate to obtain any certificate or other document from any Governmental Authority relating to Taxes.

13.7 **Tax Refund**. If, after the Closing, any Company Group Entity receives any Tax refund attributable to Pre-Closing Taxes, including any interest thereon, such Company Group Entity shall pay Seller the amount equal to the Tax refund, including any interest thereon by wire transfer of immediately available funds to the Seller Account.

ARTICLE XIV

ISSUED SHARES

14.1 Legends and Transfer of Issued Shares. Each certificate representing Issued Shares will be imprinted with a legend substantially in the following form:

The shares represented by this certificate have not been registered under the Securities Act of 1933, as amended, and may not be transferred without registration or an exemption therefrom. Further, the shares represented by this certificate are subject to a right of first offer in favor of ON Semiconductor Corporation "ON" in accordance with the terms of Section 14.3 of that certain Purchase Agreement dated as of July 15, 2010, by and among ON, Semiconductor Components Industries, LLC and SANYO Electric Co., Ltd., a copy of which is on file at the offices of ON.

14.2 Ownership of Issued Shares.

- (a) Seller shall not transfer or otherwise dispose of any of the Issued Shares for a period of two (2) years after the Closing Date; *provided*, *however*, that, subject to Section 14.3, Seller may transfer or otherwise dispose, of (i) up to one-half (1/2) of the Issued Shares prior to the first (1st) anniversary of the Closing Date, and (ii) up to all of the Issued Shares after the first (1st) anniversary of the Closing Date. Notwithstanding the foregoing, nothing in this Section 14.2 shall be deemed to limit Seller's ability to (i) merge or consolidate with another Entity or give effect to any similar business combination involving Seller, (ii) engage or participate in or unwind any hedging transaction or other similar transaction relating to any or all of the Issued Shares (including the pledging of the Issued Shares) or relating to Seller's ownership thereof or interest therein; (iii) participate in any merger, consolidation, reorganization or similar business combination involving Purchaser Parent in which Purchaser Parent Shares are exchanged or converted into cash or other property; (iv) tender or sell shares in connection with, or otherwise participate in, any tender or exchange offer or similar transaction with respect to Purchaser Parent Shares or (v) transfer or otherwise dispose of any or all of the Issued Shares after any event or situation that is or is reasonably likely to be a triggering event or situation for delisting of Purchaser Parent Shares under the rules of the relevant stock exchanges (the transactions referred to above being referred to as "Exempted Transactions").
- (b) If requested by Seller, Purchaser Parent shall cause the legend referred to in Section 14.1 to be removed from the certificates representing the Issued Shares in accordance with applicable securities laws.

14.3 Purchaser Parent's Right of First Offer

(a) Before any Issued Shares held by Seller may be transferred, whether prior to the second (2nd) anniversary of the Closing Date in accordance with Section 14.2(a) or thereafter (such shares, the "Offered Shares"), Seller shall first offer Purchaser Parent an opportunity to purchase the Offered Shares (the "ROFO Right"), and shall deliver to Purchaser Parent written notice setting forth the number of Offered Shares that Seller desires to sell or transfer (such notice, the "ROFO Notice"); provided, however, that the ROFO Right shall not apply to any transfer or other disposition of any Issued Shares in connection with an Exempted Transaction.

- (b) Purchaser Parent shall have the right, but not the obligation, to exercise its ROFO Right to purchase (or cause its Affiliate designee to purchase) all of the Offered Shares at the ROFO Price, provided, however, that if the fair market value of the Offered Shares is greater than fifty million Dollars (\$50,000,000), Purchaser Parent may purchase some, but no less than one-third (1/3), of the Offered Shares. Purchaser Parent shall exercise its ROFO Right by delivering to Seller, no later than ten (10) Business Days following the ROFO Notice Date (the "ROFO Exercise Period"), a written notice (the "ROFO Exercise Notice") setting forth the number of Offered Shares that Purchaser Parent elects to purchase in accordance with the immediately preceding sentence of this Section 14.3(b) (the "ROFO Shares"); provided, however, that Seller shall have the right to withdraw the ROFO Notice at any time on or prior to the date falling five (5) Business Days following the ROFO Notice Date and, if timely withdrawn, the ROFO Notice shall have no further force or effect. Promptly following receipt of the ROFO Exercise Notice (but in no event later than three (3) Business Days after the ROFO Price is determined), Seller and Purchaser Parent shall enter into a stock purchase agreement pursuant to which Seller shall provide Purchaser Parent with customary representations, warranties and indemnities in connection with such sale as to Seller's valid ownership of the ROFO Shares, its ability to convey title thereto free of all Encumbrances (other than those arising under applicable securities laws), and Seller's authority, power and right to enter into such binding stock purchase agreement and to consummate the sale and purchase thereunder without violating applicable Legal Requirements. Upon entering into such stock purchase agreement, Purchaser Parent (or its Affiliate designee) shall purchase the ROFO Shares from, and deliver the purchase price of the ROFO Shares to, Seller, and Seller shall sell the ROFO Shares and deliver the c
- (c) In the event Purchaser Parent does not timely elect to purchase some of the Offered Shares (such Offered Shares that Purchaser Parent has not timely elected to purchase, the "*Remaining Offered Shares*"), Seller may then sell the Remaining Offered Shares to any Person or Persons. In the event that such Remaining Offered Shares have not been sold within ninety (90) days after the earlier of the expiration of the ROFO Exercise Period or the date on which Purchaser Parent notifies Seller that it does not elect to exercise the ROFO Right, Seller may not sell such Remaining Offered Shares without first offering such Remaining Offered Shares to Purchaser Parent in the manner set forth in this Section 14.3.
- (d) In the event that Seller timely withdraws the ROFO Notice in accordance with <u>Section 14.3(b)</u>, Seller shall not transfer, sell or offer for transfer or sale any of the Issued Shares for a period of ninety (90) days following the date of the withdrawal of the ROFO Notice.
- (e) In the event that Seller intends to transfer any of the Issued Shares otherwise than pursuant to a registration statement filed in compliance with the Securities Act, Seller must, prior to any such transfer, furnish Purchaser Parent with a written opinion from counsel in form and substance reasonably satisfactory to Purchaser Parent to the effect that Seller may transfer such Issued Shares without registration under the Securities Act or other applicable securities laws.

ARTICLE XV

MISCELLANEOUS PROVISIONS

- 15.1 **Expenses**. Except as otherwise provided in Section 6.6(b) (Consents) and Section 13.3 (Transfer Taxes), whether or not the Transaction is consummated, each party shall pay its own fees, costs and expenses in connection with this Agreement and the Transaction (including the fees, costs and expenses of its advisers, accountants and legal counsel).
- 15.2 Interpretation. No party shall be considered the draftsperson, and this Agreement and the Transaction Agreements have been reviewed, negotiated and accepted by all parties and their attorneys, and any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement or any of the Transaction Agreements. Unless the context shall otherwise require, words using the singular or plural number shall also include the plural or singular number, respectively. The words "include," "includes" and "including" herein or in any Transaction Agreement shall be deemed to be followed by the phrase "without limitation" and the word "or" shall include the meaning "either or both." Unless the context shall otherwise require, any reference herein or in any Transaction Agreement to any agreement or other instrument or statute or regulation is to such agreement, instrument, statute or regulation as amended and supplemented from time to time (and, in the case of a statute or regulation, to any successor provision). Unless the context otherwise provides, all pronouns used herein shall be deemed to refer to the masculine, feminine or neuter gender as the context requires. All references in this Agreement or any other Transaction Agreement to "Section," "Subsection," "Article," "Exhibit," "Schedule" or "Annex" (or similar references) shall be deemed to be references to a section, subsection or article of, or exhibit to, the agreement in which they occur, unless the context otherwise requires. The table of contents and the captions and other headings contained in this Agreement or any Transaction Agreement as to the contents of particular articles, sections, paragraphs or other subdivisions contained herein or therein have been included for convenience of reference only and shall not, in any way, be construed as part of this Agreement or as limitations on the scope of the particular articles, sections, paragraphs or other subdivisions to which they refer and shall not affect the interpretation or meaning of this Agreement or any Transaction Agreement. The Seller Disclosure Schedule will be arranged in schedules corresponding to the lettered and numbered Sections and paragraphs contained in Article III or IV or Sections 6.1 or 6.2; provided, however, that any disclosure with respect to a lettered or numbered Section or paragraphs contained in Article III or IV or Sections 6.1 or 6.2 shall also be deemed to be disclosed with respect to any other lettered or numbered Sections or paragraphs in Article III or IV or Sections 6.1 or 6.2 if it is reasonably apparent on the face of such disclosure that such disclosure is applicable to the subject matter of such other lettered or numbered Sections or paragraphs in Article III or IV or Sections 6.1
- 15.3 **Further Assurances**. Each party agrees (a) to furnish upon request to each other party such further information; (b) to execute and deliver to each other party such other documents; and (c) to do such other acts and things, all as another party may reasonably request for the purpose of carrying out the intent of this Agreement and the Transaction.

15.4 Notices.

(a) All notices, requests, demands and other communications hereunder shall be either (i) delivered in person; (ii) sent by international courier service or other express commercial delivery service; or (iii) sent by facsimile with confirmation of receipt and, in each case, addressed as follows:

If to Seller: SANYO Electric Co., Ltd.

Corporate Planning Headquarters 5-5 Keihan-hondori 2-chome

Moriguchi City, Osaka 570-8677 Japan

Fax: +81-6-6994-7516

with copies to (which copy shall not constitute notice):

SANYO Electric Co., Ltd. Legal Headquarters 5-5 Keihan-hondori 2-chome

Moriguchi City, Osaka 570-8677 Japan

Fax: +81-6-6994-0992

If to Purchaser or Purchaser Parent: ON Semiconductor Corporation

5005 E. McDowell Road Phoenix, AZ 85008 U.S.A Fax: +1-602-244-5500 Attn: General Counsel

with copies to (which copy shall not constitute notice):

Morrison & Foerster LLP 425 Market Street

San Francisco, CA 94105 U.S.A

Fax: +1-415-268-7522

Attn: Eric McCrath, Esq. and Randy Laxer, Esq.

- (b) All notices, requests, instructions or documents given to any party in accordance with this <u>Section 15.4</u> shall be deemed to have been given on the date of mailing or transmission, whether delivered by hand, by international courier service, or by facsimile, with confirmation of receipt on such date.
 - (c) Any party may change its address specified for notices herein by designating a new address by notice given in accordance with this Section 15.4.
- 15.5 **Entire Agreement**. This Agreement, the Schedules, the Exhibits, the Annexes and the Transaction Agreements constitute the entire agreement between the parties relating to the subject matter hereof and thereof and supersede all prior oral and written understandings, all contemporaneous oral negotiations and discussions, and all other writings and agreements relating to the subject matter of this Agreement.
- 15.6 **Modifications**, **Amendments and Waivers**. This Agreement and any Transaction Agreements cannot be amended or changed nor any performance, term, or condition waived in whole or in part, except by a writing signed by the party against whom enforcement of the amendment, change or waiver is sought. No delay or failure on the part of any party in exercising any rights hereunder, and no partial or single exercise thereof, will constitute a waiver of such rights or of any other rights hereunder.
- 15.7 **Successors and Assigns**. This Agreement cannot be assigned by any party without the prior written consent of the other parties and any assignment in violation of this Section 15.7 shall be null and void; provided, however, that Purchaser may assign this Agreement (i) to Affiliates of Purchaser or (ii) following the Closing, in connection with any merger, sale of all or substantially all of Purchaser's assets or similar corporate transaction, including a sale or transfer of Company, its assets, or a significant portion thereof; provided further, however, that in case of any assignment of this Agreement, the assignor shall not be released from its obligations hereunder without the consent, in the case of Purchaser, of Seller, and in the case of Seller, of Purchaser. Upon any consolidation or merger in which a party is not the surviving corporation, or any sale, assignment, transfer, lease, conveyance or other disposition formed by such consolidation or into or with which such party is merged or the Entity to which such sale, assignment, transfer, lease, conveyance or other disposition is made shall succeed to, and be substituted for and may exercise every right and power of, and be obligated to perform every obligation of, such party with the same effect as if such successor Entity has been named as such party herein; provided, however, that the predecessor party in the case of a sale, assignment, transfer, lease, conveyance or other disposition shall not be released from any of the obligations hereunder.
- 15.8 **Governing Law**. This Agreement is to be construed in accordance with and governed by the laws of Japan, without giving effect to any choice of law rule that would cause the application of the laws of any jurisdiction other than the laws of the Japan to the rights and duties of the parties.

- 15.9 **Severability**. Should any one or more of the provisions of this Agreement be determined to be invalid, illegal or unenforceable, such invalid, illegal or unenforceable provisions shall be deemed severed herefrom, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby. The parties shall replace the invalid, illegal or unenforceable provisions with valid provisions, the economic effect of which comes as close as practicable to that of the invalid, illegal or unenforceable provisions.
- 15.10 **Specific Performance**. Each party agrees that irreparable harm, for which there will be no adequate remedy at law and for which the ascertainment of monetary damages would be difficult, would occur in the event any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Each party accordingly agrees that, notwithstanding <u>Section 12.6</u> but subject to <u>Section 15.12</u>, the other parties shall be entitled to specifically enforce this Agreement and to obtain an injunction or injunctions to prevent breaches of the provisions of this Agreement or any Transaction Agreement and to enforce specifically the terms and provisions hereof or thereof, in each instance without being required to post bond or other security and in addition to, and without having to prove the adequacy of, other remedies at law.
- 15.11 **Counterparts**. This Agreement may be executed in any number of counterparts which may be delivered by facsimile, each of which shall be deemed to be an original, but all of which counterparts shall together constitute one and the same instrument.
- 15.12 **Dispute Resolution**. Each party agrees that, upon written request of any other party, it shall use commercially reasonable efforts to settle amicably through good faith discussions any dispute or disagreement which may arise under or pursuant to this Agreement. In the event that any such dispute or disagreement cannot be resolved through such discussions, the parties agree that any dispute or disagreement which may arise under or pursuant to this Agreement or the Transaction Agreements or the transactions contemplated hereby or thereby shall be settled by the parties amicably through good faith discussions upon the written request of any party. In the event that any such dispute cannot be resolved through such discussions within a period of twenty (20) days after delivery of such written request, the dispute shall be finally settled by binding arbitration in the Republic of Singapore, using the English language and in accordance with the Rules of Arbitration of the International Chamber of Commerce then in effect. The arbitrator(s) shall have the authority to grant specific performance, and to allocate between the Purchaser and Seller the costs of arbitration in such equitable manner as the arbitrator(s) may determine, *provided*, *however*, that each party shall bear its own translation, transportation and travel costs. Judgment upon the award so rendered may be entered in any court of competent jurisdiction or application may be made to such court for judicial acceptance of any award and an order of enforcement, as the case may be.
- 15.13 **Attorneys' Fees, Costs and Expenses**. Subject to <u>Section 15.12</u>, in any Proceeding to enforce an arbitral award in connection with this Agreement, the prevailing party shall be entitled to recover reasonable attorneys' fees, costs and expenses incurred in connection with such enforcement Proceeding.
- 15.14 **No Third-Party Beneficiaries**. The terms and provisions of this Agreement are intended solely for the benefit of each party and their respective successors and permitted assigns, and the parties do not intend to confer third-party beneficiary rights upon any other Person.

[Signatures Follow On a Separate Page]

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed on its behalf by their respective officers thereunto duly authorized all as of the date first written above.

PURCHASER:

SEMICONDUCTOR COMPONENTS INDUSTRIES, LLC

By: Its sole member

ON Semiconductor Corporation

By: /s/ KEITH JACKSON

Name: Keith Jackson

Title: President and Chief Executive Officer

PURCHASER PARENT:

ON SEMICONDUCTOR CORPORATION

By: /s/ KEITH JACKSON

Name: Keith Jackson

Title: President and Chief Executive Officer

SELLER:

SANYO ELECTRIC CO., LTD.

By: /s/ SEIICHIRO SANO

Name: Seiichiro Sano Title: President

Signature Page – Purchase Agreement

EXHIBIT A

CERTAIN DEFINITIONS

- "Accounts Payable" means all accounts payable and other amounts owed to trade creditors of any Company Group Entity.
- "Accounts Receivable" means all accounts receivable, notes receivable and other monies due to any Company Group Entity for sales and deliveries of goods and performance of services (whether or not on the books of any Company Group Entity).
- "Active Employees" means the Company Group Employees and the Eligible Business Employees; and "Active Employee" means any Company Group Employee or Eligible Business Employee.
- "Actual Closing Net Working Capital" has the meaning specified in Section 1.4(b).
- "Actuary" has the meaning specified in Section 1.4(i).
- "Actuary Statement" has the meaning specified in Section 1.4(i).
- "Actuary Statement Dispute Notice" has the meaning specified in Section 1.4(1).
- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under common control with such Person. For the purposes of this definition, "control," "controlled by" and "under common control with," with respect to the relationship between or among two or more Persons, means (i) the ownership of a majority of the voting share capital of a Person or (ii) the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of a Person, whether through the ownership of voting securities, by agreement or otherwise.
- "Agreed Enterprise Value" means fifty-five billion Yen (¥55,000,000,000).
- "Agreement" means the Purchase Agreement to which this <u>Exhibit A</u> is attached (including the Seller Disclosure Schedule and all other schedules, exhibits, annexes and schedules attached hereto), as it may be amended from time to time, but does not include any other Transaction Agreement.
- "Allocation" has the meaning specified in Section 1.3(b)(ii).
- "AMA" has the meaning specified in Section 6.6(b).
- "AML" has the meaning specified in Section 6.6(b).
- "Amorton Business" means the amorphous silicon solar cells and amorphous solar cells business of KANTO SANYO Semiconductors Co., Ltd.

- "Amorton Employees" means the employees of KSS who are primarily engaged in the Amorton Business.
- "Antitrust Law" means the HSR Act, AML, AMA and any other federal, state or foreign statutes, rules, regulations, orders or decrees that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade.
- "Antitrust Order" means any decree, judgment, injunction or other order (whether temporary, preliminary or permanent) that restricts, prevents, prohibits or delays the consummation of any transaction contemplated by this Agreement under any Antitrust Law.
- "Antitrust Proceeding" has the meaning specified in Section 6.6(e)(iii).
- "Applicable Spot Rate" means the Tokyo spot rate in effect the date five (5) Business Days prior to the date hereof for the Yen equivalent of the local currency as applicable.
- "Asserted Liability" has the meaning specified in Section 12.4(a).
- "Assumed Closing Date" means November 1, 2010.
- "Audited Pro Forma Financial Statements" has the meaning specified in Section 6.9.
- "Bank Debt" means, collectively, the BOT Loan, the DBJ Loans and the Shoko Loan.
- "BOT" means the BOT LEASE Co. Ltd.
- "BOT Loan" means the Indebtedness of SSMC to BOT pursuant to a purchase agreement between SSMC and BOT.
- "Business" means the business of developing, designing, manufacturing, assembling, diagnosing, repairing, engaging in warranty services for, testing, distributing, marketing and selling, and conducting research with respect thereto, semiconductor products and parts and components therefor, as currently conducted by the Company Group Entities, including any technologies or products currently under development or design.
- "Business Assets and Services" means all the assets and services, wherever located and whether held, owned or leased by or provided to a Seller Group Entity or Company Group Entity, that are primarily used in the Business (other than the Business IP Assets and the Excluded Business Assets and Services).
- "Business Day" means any day other than (i) a Saturday or a Sunday or (ii) a day on which banking and savings and loan institutions are authorized or required by law to be closed in the United States or Japan.
- "Business IP Assets" means all Intellectual Property Rights owned, held, or in-licensed by any Company Group Entity or Seller Group Entity and used in the Business.

- "Capital Leases" means, with respect to any Person at any date, all capitalized lease obligations of such Person at such date.
- "Cash and Cash Equivalents" means cash, cash equivalents and marketable securities of the Company Group Entities, including cash and time deposits at banks and accrued interest receivable, as determined in accordance with J-GAAP, applied on a basis consistent with the basis used in the preparation of the Unaudited Financial Statements
- "Cash Consideration" has the meaning specified in Section 1.3(c)(i).
- "Charter Documents" of any Person means such Person's articles of incorporation, by-laws, certificate of formation, limited liability company agreement or equivalent governing or organizational documents, including all amendments thereto.
- "Claims Notice" has the meaning specified in Section 12.4(a).
- "Closing" has the meaning specified in Section 1.2.
- "Closing Adjustment Base Date" means the last day of the month immediately prior to the month in which the Closing takes place.
- "Closing Assumed Unfunded Benefit Liabilities" means the aggregate amount of the Unfunded Benefit Liabilities associated with Niigata Employees and SSDC Employees as of the Closing Adjustment Base Date.
- "Closing Balance Sheet" has the meaning specified in Section 1.4(b).
- "Closing Balance Sheet Dispute Notice" has the meaning specified in Section 1.4(b).
- "Closing Cash and Cash Equivalents" means Cash and Cash Equivalents of the Company Group Entities as at Closing Adjustment Base Date.
- "Closing Cash and Cash Equivalents Shortfall" means (i) zero or (ii) in the event that Purchaser elects to waive the Closing condition set forth in Section 9.1(k), the amount that ten billion Yen (¥10,000,000,000) exceeds Estimated Closing Cash and Cash Equivalents.
- "Closing Cash Payment" has the meaning specified in Section 1.3(c)(i).
- "Closing Company Group Entities" means Company and its Subsidiaries (including as a Subsidiary for this purpose, SSMPE, and SCE and SNA to the extent of the Transferred SCE Business and the Transferred SNA Business) existing as of the Closing Adjustment Base Date; provided, for the avoidance of doubt, the Closing Company Group Entities shall be determined after giving effect to the Reorganization.
- "Closing Date" has the meaning specified in Section 1.2.
- "Closing Factoring Liabilities" means the amount of the Factoring Liabilities as of Closing Adjustment Base Date.

- "Closing Intercompany Indebtedness" means the aggregate amount of Intercompany Indebtedness of the Company Group Entities as at the Closing Adjustment Base Date, other than the Permitted Intercompany Indebtedness.
- "Closing Net Working Capital" means Net Working Capital as of the Closing Adjustment Base Date.
- "Closing Net Working Capital Adjustment" has the meaning specified in Section 1.4(d).
- "Closing Net Working Capital Target" means thirty-six billion two hundred sixty-nine million Yen (¥36,269,000,000).
- "Closing Outstanding Bank Debt" means the aggregate amount of Bank Debt as at the Closing Adjustment Base Date.
- "Closing Outstanding Third Party Indebtedness" means the aggregate amount of Indebtedness (other than Bank Debt, Intercompany Indebtedness (including Permitted Intercompany Indebtedness), Identified Capital Leases, any Indebtedness of a Company Group Entity incurred to replace any Housing Loan Guarantee, factoring liabilities with third party financial institutions and Indebtedness of the Company Group Entities consented to by Purchaser pursuant to Section 6.2) as at the Closing Adjustment Base Date.
- "Closing Statement" has the meaning specified in Section 1.4(b).
- "Company" has the meaning specified in the preamble of this Agreement.
- "Company Contracts" has the meaning specified in Section 3.12(a).
- "Company Demerger" means the demerger of the Company from Seller effected as of July 3, 2006.
- "Company Group Employee" means any and all Persons employed by any Company Group Entity as of the date hereof, but, for the avoidance of doubt, shall not include Eligible Business Employees.
- "Company Group Entities" means Company and its Subsidiaries (including as a Subsidiary for this purpose, SSMPE, and SCE and SNA to the extent of the Transferred SCE Business and the Transferred SNA Business, but excluding the Amorton Business); provided, however, that for purposes of Sections 3.5 and 6.8(a), SCE and SNA shall not be deemed to be Subsidiaries of the Company, but Amorton Business shall be included within the Subsidiaries of Company.
- "Company IP Assets" means all Intellectual Property Rights that (a) any Company Group Entity owns immediately prior to the Closing Date or (b) are Transferred IP Assets.

"Company Material Adverse Effect" means any event, change or effect that, when taken individually or together with all other adverse events, changes and effects, is or is reasonably likely (a) to be materially adverse to the financial condition, business, or results of operations of the Company Group Entities taken as a whole or the Business as a whole or (b) to prevent consummation of the Transaction or otherwise to prevent Seller from performing its obligations under any of the Transaction Agreements; provided, however, that none of the following shall be deemed, either alone or in combination, to constitute, and there shall not be taken into account in determining whether there has been or will be a Company Material Adverse Effect, any adverse event, change or effect arising from or attributable to: (a) conditions affecting (i) the industry in which the Company Group Entities operate generally that do not affect the Company Group Entities in a materially disproportionate manner when compared to other companies operating in the same industry, or (ii) the economy or financial markets in any location where any Company Group Entity has operations or sales that do not affect the Company Group Entities in a materially disproportionate manner when compared to other companies operating in the same industry; (b) changes in Legal Requirements or Local GAAP or the interpretation thereof; (c) any acts of war or terrorism or any natural disaster; (d) the announcement or pendency of this Agreement, any other Transaction Agreement or the Transaction; (e) any action or inaction required to cause compliance with the terms of or the satisfaction of any condition set forth in, or any action or inaction required by, this Agreement or any Transaction Agreement; (f) any breach by Purchaser or Purchaser Parent of this Agreement; (g) any action or inaction by Seller or any Company Group Entity consented to by Purchaser or Purchaser Parent pursuant to the terms of this Agreement; (h) the failure by the Company Group Entities to meet

"Company Portion Seller Restructuring Plan Costs" has the meaning specified in Section 8.2(a).

"Company Products" mean all products and services manufactured, made, designed, maintained, supported, developed, sold, licensed, marketed, or otherwise distributed or provided (or planned or envisioned to be manufactured, made, designed, maintained, supported, developed, sold, licensed, marketed, or otherwise distributed or provided) by or for any Company Group Entity (including all versions and releases thereof, whether already distributed or provided, under development, planned or conceived, or otherwise), together with any related materials, information or data, including, without limitation, the names, numbers (e.g., part numbers) and packaging associated with such products and services.

"Company Shares" means all of the issued and outstanding capital stock of the Company.

"Company Supplier Alternative Payment System" means any arrangement under which Seller causes certain banks to pay the accounts payable arising from Company's purchase of certain parts and other goods from certain of Company's suppliers after Seller assumes such accounts payable from Company, and all Contracts entered into in connection with or otherwise related to such arrangements.

"Company Transaction Expenses" has the meaning specified in Section 3.26.

"Competing Party" has the meaning specified in Section 7.8.

"Competing Transaction" has the meaning specified in Section 7.8.

- "Confidential Information" means all Trade Secrets and other confidential and/or proprietary information of a Person, including information derived from reports, investigations, research, work in progress, codes, marketing and sales programs, manufacturing processes, financial projections, cost summaries, pricing formulae, contract analyses, financial information, projections, confidential filings with any state or federal agency, and all other confidential concepts, methods of doing business, ideas, materials or information prepared or performed for, by or on behalf of such Person by its employees, officers, directors, agents, Representatives, or consultants.
- "Consent" means any approval, consent, ratification, permission, waiver or authorization (including any Governmental Approval).
- "Contract" means any agreement, contract, obligation, promise, understanding, arrangement, commitment or undertaking of any nature (whether written or oral and whether express or implied), whether or not legally binding.
- "Contract Consents" has the meaning specified in Section 9.1(c)(iv).
- "Copyrights" means all copyrights, copyrights works, semiconductor topography and mask work rights, and applications for registration thereof, including all rights of authorship, publication, reproduction, distribution, performance and transformation, moral rights and all rights to register and obtain renewals and extensions of registrations, together with all other interests accruing by reason of international copyright, semiconductor topography and mask work conventions.
- "Current Business" means the Business as it is being conducted by the Company Group Entities on the date of this Agreement, including, the stage of development or design of any technologies or products that are under development or design by the Company Group Entities as of the date of this Agreement.
- "Damages" has the meaning specified in Section 12.2(a).
- "DBJ" means the Development Bank of Japan.
- "DBJ Loans" means the Indebtedness of SSMC under loan agreements between SSMC and DBJ.
- "Deposits" means security deposit, earnest money, advance or other deposits temporarily or provisionally provided by a Company Group Entity to a Seller Group Entity.
- "Determination Period" means the twenty (20) trading days immediately prior to the date of this Agreement.
- "Disclosing Party" has the meaning specified in Section 7.10(b).

- "Divestiture" means (1) the sale, license or other disposition or holding separate (through the establishment of a trust or otherwise) of any material assets or categories of assets of a party, its Affiliates or Subsidiaries; (2) the imposition of any material limitation or restriction on the ability of a party or any of its Affiliates to freely conduct their business or the ability of any Company Group Entity to conduct the Business; or (3) the holding separate of the Purchased Shares or any material limitation or regulation on the ability of Purchaser or any of its Affiliates to exercise full rights of ownership of the Purchased Shares.
- "Eligible Business Employee" means any person employed by any Seller Group Entity and primarily engaged in activities relating to the Business, including Seconded Employees, in each case, as of the date hereof.
- "Encumbrance" means any lien, pledge, hypothecation, charge, mortgage, security interest, encumbrance, deed of trust, equitable interest, claim, preference, license, encroachment, right of first refusal, community property interest, joint ownership interest, and any restriction on the use, transfer or possession of any asset or property or irregularities in title, in each case, other than Permitted Encumbrances.
- "Entity" means any corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust or company (including any limited liability company or joint stock company).
- "Environmental and Safety Laws" means any federal, state, local, administrative or foreign Legal Requirement that pertains to (i) the condition or protection of air, groundwater, surface water, drinking water, land or soil, surface or subsurface strata or medium, natural resources or other environmental media; (ii) the protection of human health and safety; or (iii) the generation, treatment, manufacturing, use, storage, handling, recycling, presence, Release, disposal, transportation or shipment of any Hazardous Materials.
- "Environmental Permits" means all permits, licenses, authorizations, certificates, approvals or registrations required by any Governmental Authority under any Environmental and Safety Laws.
- "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "Escrow Account" has the meaning specified in Section 1.3(d).
- "Escrow Agent" means a financial institution to be mutually agreed by the parties.
- "Escrow Agreement" means an escrow agreement to be entered into on customary terms and conditions among Seller, Purchaser and the Escrow Agent relating to the Escrow Amount.
- "Escrow Amount" has the meaning specified in Section 1.3(d).
- "Escrow Period" has the meaning specified in Section 12.1(f)(i).
- "Escrow Release Date" has the meaning specified in Section 12.1(a).
- "Estimated Closing Cash and Cash Equivalents" has the meaning specified in Section 1.4(a)(i).
- "Estimated Closing Outstanding Bank Debt" has the meaning specified in Section 1.4(a).

- "Estimated Closing Net Working Capital" has the meaning specified in Section 1.4(a)(i).
- "Estimated Closing Net Working Capital Adjustment Amount" has the meaning specified in Section 1.4(a)(vi).
- "Estimated Seller Restructuring Costs" means eleven billion two hundred million Yen (¥11,200,000,000).
- "Estimated Transferred Unfunded Benefit Liabilities" has the meaning specified in Section 1.4(a)(ii).
- "Examination" means any audit, formal examination, formal investigation, claim, request for information or ruling request by any Governmental Authority.
- "Exempted Transactions" has the meaning specified in Section 14.2(a).
- "Excluded Business Assets and Services" means: (a) any "Excluded Asset" identified as such in the Transferred SCE Business Assignment and Assumption or the Transferred SNA Business Assignment and Assumption; and (b) any asset or service that is subject to a service agreement referred to in Section 7.6.
- "Facilities" means all buildings and improvements on the Property.
- "Factoring Liabilities" means the amount of payables or other liabilities owed by Company overseas sales subsidiaries to any Seller Group Entity minus receivables or other assets credited to any Company Group Entity by any Seller Group Entity, arising from Securitization of Receivables Transactions.
- "FIEL" means the Financial Instruments and Exchange Law of Japan.
- "Final Resolution Date" has the meaning specified in Section 1.4(b).
- "Financial Statements" means the Unaudited Financial Statements, the Audited Pro Forma Financial Statements and the Interim Financial Statements.
- "Fundamental Reps" means the representations and warranties under Sections 3.1, 3.2(a), 4.1 and 4.1.
- "Gas Contracts" means agreements entered by any Seller Group Entity for the provision of natural gas (including hydrogen and nitrogen) for the operations of the SSMC Gifu facilities and/or the SSMC Gunma facilities.
- "Gifu and Gunma Employees" means those SSMC employees who are registered with the SSMC Gifu facilities or the SSMC Gunma facilities, regardless of the actual working location.
- "Government Antitrust Entity" has the meaning specified in Section 6.6(c).

"Governmental Approval" means any: permit, license, certificate, concession, approval, consent, ratification, permission, clearance, confirmation, exemption, waiver, franchise, certification, designation, rating, registration, variance, qualification, accreditation or authorization issued, granted, given or otherwise made available by any Governmental Authority.

"Governmental Authority" means any: (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or quasi governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, organization, unit, body or Entity and any court or other tribunal); or (d) multinational organization or body.

"Hazardous Materials" means any material, substance, waste, compound, pollutant or contaminant listed, defined, designated or classified as hazardous, toxic, flammable, explosive, reactive, corrosive, infectious, carcinogenic, mutagenic or radioactive or otherwise regulated by any Governmental Authority or under any Environmental and Safety Law, including petroleum or petroleum products (including crude oil) and any derivative or by-products thereof, natural gas, synthetic gas and any mixtures thereof, or any substance that is or contains friable asbestos, polychlorinated biphenyls, urea formaldehyde or lead.

"HSR Act" means the Hart Scott Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations thereunder.

"Identified Capital Leases" has the meaning specified in Section 3.10(a).

"Illustrative Purchase Price Calculation" means the illustrative calculation of the Purchase Price set forth on Annex II.

"Immigration Authorities" has the meaning specified in Section 3.14(b).

"Improvements" has the meaning specified in Section 3.22(c).

"Indebtedness" means, with respect to any Person at any date, without duplication: (i) all obligations of such Person for borrowed money or in respect of loans from third-party financing parties; (ii) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments; (iii) all obligations in respect of letters of credit, whether or not drawn, and bankers' acceptances or similar credit transactions, whether or not then due, issued for the account of such Person; (iv) all capitalized lease obligations of such Person; (v) all interest rate protection agreements of such Person (valued on a market quotation basis); (vi) all guarantees of such Person in connection with any of the foregoing; (vii) any debt-like obligation or financing-type arrangement in respect of the deferred purchase price of property or property received as of the Closing with respect to which such Person is liable, contingently or otherwise, as obligor or otherwise; (viii) any accrued interest, prepayment premiums or penalties or other similar costs or expenses related to any of the foregoing; provided, however, that with respect to any Company Group Entity or Seller Group Entity, the determination of "Indebtedness" shall be in accordance with J-GAAP; provided further, however, that Indebtedness does not include accounts payable to trade creditors and accrued expenses.

"Indemnified Party" has the meaning specified in Section 12.4(a).

- "Indemnifying Party" has the meaning specified in Section 12.4(a).
- "Independent Accounting Firm" has the meaning specified in Section 1.4(c).
- "Independent Actuary" has the meaning specified in Section 1.4(1).
- "Insurance Policies" has the meaning specified in Section 3.21.
- "Integration Period" has the meaning specified in Annex III.
- "Intellectual Property License Agreement" has the meaning specified in Section 7.1.
- "Intellectual Property Licenses" means (i) any grant to a Person of any right or authorization to use or practice or any covenant not to sue under or release from any claim of damages for infringement, unauthorized use or disclosure, or misappropriation of any Company IP Assets, and (ii) any grant to Company of any right or authorization to use or practice or any covenant not to sue under or release from any claim of damages for infringement, unauthorized use or disclosure, or misappropriation of any Person's Intellectual Property Rights or Technology which is necessary, useful, or otherwise related to the Business.
- "Intellectual Property Rights" means all right, title, and interest arising from or in respect of any of the following, whether protected, created or arising under the laws of the United States or any other jurisdiction: (i) all Patents; (ii) all Trademarks; (iii) all Copyrights; (iv) any Trade Secrets; and (v) all rights, benefits, privileges, causes of action, and remedies relating to any of the foregoing, whether before or hereafter accrued, including the exclusive rights to apply for and maintain all registrations, renewals, and extensions, to sue for all past, present, and future infringements, unauthorized uses or disclosures, or misappropriations of any rights relating thereto, and to settle and retain proceeds from any such actions.
- "Intercompany Indebtedness" means all Indebtedness owing by a Company Group Entity to a Seller Group Entity.
- "Intercompany Release" has the meaning specified in Section 6.7(b)(ii).
- "Interim Financial Statements" has the meaning specified in Section 6.8(c).
- "Inventory" means all finished goods, work in process for the production of finished goods, raw materials and other materials and supplies to be used or consumed by Company Group Entities in the production of its goods or the provision of services.
- "Investments" has the meaning specified in Section 3.25.
- "IP Assignment Agreement" has the meaning specified in Section 7.4.
- "Issued Shares" has the meaning specified in Section 1.3(c)(ii).
- "ITU Application" means any intent-to-use trademark application with respect to any Purchased IP Asset that is pending at any time between the date of the Agreement and the Closing.

- "Japan Unions" means all labor unions and other labor organizations whose approval is necessary to carry out and to complete the Restructuring Plan in Japan.
- "J-GAAP" means Japan generally accepted accounting principles in effect on the date on which they are to be applied pursuant to this Agreement, applied consistently throughout the relevant periods.
- "Knowhow" means all right, title and interest in all Technology, Trade Secrets, ideas, concepts, proprietary techniques, processes, business and other methodologies, scientific, technical, research, development, engineering and business information, and other knowhow, including all trade secret rights arising under any law, including common law, state law, federal law or laws of foreign countries, other than Copyrights, Patents and Trademarks.
- "Knowledge" or "knowledge" An individual shall be deemed to have "Knowledge" of a particular fact or other matter if: such individual is actually aware of such fact or other matter. Seller shall be deemed to have "Knowledge" of a particular fact or other matter if any director of Seller or Company or any of the individuals listed on Exhibit D has Knowledge of such fact or other matter.
- "KSS" means KANTO SANYO Semiconductors Co., Ltd.
- "KSS Employees" means employees, including those seconded to other Company Group Entities, of KSS other than those employees who are primarily engaged in the Amorton Business as of the date hereof.
- "KSS Supplier Alternative Payment System" means any arrangement by which a Seller Group Entity causes certain banks to pay the accounts payable of KSS arising from its purchase of certain parts and other goods from certain of its suppliers on its behalf, and all Contracts entered into in connection with or otherwise related to such arrangements.
- "KSS Transfer Date" means a date specified by Purchaser, after consultation with Seller, for the transfer of the KSS Employees from New KSS to KSS, expected (but not required) to be April 1, 2011, but in any event no later than June 30, 2012.
- "Latest Balance Sheet Date" means March 31, 2010.
- "Latest Financial Statement" has the meaning specified in Section 3.5(a).
- "Lease Agreements" shall mean the lease agreements, master sublease agreement and the guarantee agreements contemplated by Section 7.5.
- "Leased Real Property" has the meaning specified in Section 3.22(b).
- "Legal Requirement" means any federal, state, local, municipal, foreign or other law, statute, legislation, constitution, principle of common law, resolution, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, permit, ruling, directive, pronouncement, determination, decision, opinion or interpretation, in each case having the force of law enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority.

- "Liability" means any debt, obligation, liability or Tax of any nature (including any unmatured, unaccrued, contingent, conditional, joint or several liability), regardless of whether such debt, obligation, liability or Tax would be required to be disclosed on a balance sheet prepared in accordance with generally accepted accounting principles and regardless of whether such debt, obligation, or liability is immediately due and payable.
- "Local GAAP" means the generally accepted accounting principles applicable in a particular jurisdiction.
- "Material Contracts" has the meaning specified in Section 3.12(a).
- "Monthly Financial Statements" has the meaning specified in Section 6.8(b).
- "Net Working Capital" means, as of any date, the (a) the sum of the Company Group Entities' current assets (which shall include all Accounts Receivable) on a consolidated basis but which shall exclude Cash and Cash Equivalents and deferred tax assets, minus (b) the sum of the Company Group Entities' current liabilities (which shall include any Indebtedness accelerated as a result of the consummation of the Transaction but shall exclude Bank Debt, Intercompany Indebtedness, accounts payable facilities, notes payable facilities anddeferred tax liabilities), in each case as of such date. The foregoing shall be determined on a consolidated basis for the Company Group Entities and in accordance with J-GAAP, applied on a basis consistent with the basis used in the preparation of the Unaudited Financial Statements (except as otherwise provided with respect to the exclusion of certain Indebtedness in the immediately preceding sentence).
- "New KSS" has the meaning specified in Section 9.1(f)(i).
- "New SSMC" has the meaning specified in Section 9.1(f)(ii).
- "Niigata Employees" means those SSMC employees who are registered with the SSMC Niigata facility, regardless of actual working location.
- "No-Hire Employee" means any Person who is employed by or is a consultant or has a similar relationship with any Company Group Entity as of the date hereof or is an Eligible Business Employee (excluding those employees who did not transfer to a Company Group Entity) and either (i) holds a position as of the date hereof of manager (buchō) or above or (ii) is primarily engaged in a research and development, sales or engineering function.
- "Nondisclosing Party" has the meaning specified in Section 7.10(b).
- "Nonsolicitation Period" has the meaning specified in Section 7.8(b).
- "Off the Shelf Software" has the meaning set forth in Section 3.12(a)(viii).
- "Offered Shares" has the meaning specified in Section 14.3(a).

"Open Source Software" means any software, including programs, libraries, drivers, header files, APIs, and scripts, that is incorporated into, combined with, or distributed in conjunction with any Company Products, and that is, contains, or is derived in any manner (in whole or in part) from any software that is distributed as "free software" (as defined by the Free Software Foundation), "open source software" (i.e., software distributed under any license approved by the Open Source Initiative as set forth in www.opensource.org), or similar licensing or distribution terms, including any version of the following licenses and licenses substantially similar thereto: (i) the GNU General Public License; (ii) the GNU Lesser or Library General Public License; (iii) the Artistic License; (iv) the Mozilla Public License; (v) the Netscape Public License; (vi) the Sun Community Source License; (vii) the Sun Industry Standards License; (viii) the Common Public License; (ix) the BSD License; (x) the Apache License; (xi) the OpenSymphony Software License; and (xii) the MIT License.

"Order" means any temporary, preliminary or permanent order, judgment, injunction, edict, decree, ruling, pronouncement, determination, decision, opinion, verdict, sentence, stipulation, subpoena, writ or award issued, made, entered, rendered or otherwise put into effect by any court, administrative agency or other Governmental Authority or any arbitrator or arbitration panel.

"OS" has the meaning specified in Section 7.9(a)(i).

"Owned Real Property" has the meaning specified in Section 3.22(a).

"Parent" means Panasonic Corporation.

"party" and "parties" has the meaning specified in the preamble to this Agreement.

"Patent Claimant" means a Person claiming infringement, invalidity, or other violation of with Patent rights in connection with a Third Party Patent License.

"Patents" means all patent rights, title and interests in and to all letters patent and rights accorded under patent law systems, utility models, and applications therefor, including continuations, divisionals, continuations-in-part, reissues, reexaminations, substitutions, renewals, and extensions thereof, and patents issuing thereon.

"Permitted Encumbrances" means: (a) statutory liens of landlords, liens of carriers, warehousepersons, mechanics and material persons, and purchase money liens incurred in the ordinary course of business for sums (i) not yet due and payable, or (ii) being contested in good faith; (b) liens incurred or deposits made in connection with workers' compensation, unemployment insurance and other similar types of social security programs or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return of money bonds and similar obligations, in each case in the ordinary course of business, consistent with past practice; (c) easements, rights-of-way, restrictions and other similar charges or encumbrances, in each case, which do not interfere with the ordinary conduct of business of any Company Group Entity and do not materially detract from the value of the property upon which such encumbrance exists; (d) liens securing Taxes, assessments and governmental charges not yet due and payable; (e) leases or subleases and non-exclusive licenses or sublicenses granted in the ordinary course of the business of any Company Group Entity and any interest or title of a licensor under any license or sublicense; (f) banker's liens, rights of setoff and similar liens incurred on deposits made in the ordinary course of business; (g) restrictions on transfer of securities under applicable company law or securities laws and (h) any Encumbrance that does not materially interfere with the ownership or operation of the assets to which it relates.

- "Permitted Indebtedness" means Bank Debt and Permitted Intercompany Indebtedness.
- "Permitted Intercompany Indebtedness" means the Seller Loan Receivables and Closing Factoring Liabilities.
- "Person" mean any individual, Entity or Governmental Authority.
- "Personally-Identifiable Information" means any information that can be used to identify a specific individual as defined by applicable Legal Requirement in the relevant jurisdictions, such as the individual's name, address, telephone number, fax number, email address, credit card or financial account number, medical information, or health insurance information.
- "Plans" means every plan, fund, Contract, program and arrangement (whether written or not) for the benefit of present or former employees, including those intended to provide (i) medical, surgical, health care, hospitalization, dental, vision, workers' compensation, life insurance, death, disability, legal services, severance, sickness or accident benefits (whether or not defined in Section 3(1) of ERISA); (ii) pension, profit sharing, stock bonus, retirement, supplemental retirement or deferred compensation benefits (whether or not tax qualified and whether or not defined in Section 3(2) of ERISA); or (iii) salary continuation, unemployment, supplemental unemployment, severance, termination pay, change-in-control, vacation or holiday benefits (whether or not defined in Section 3(3) of ERISA) that is maintained by any of the Company Group Entities or Seller Group Entities for the benefit of the Active Employees and their beneficiaries and with respect to which the Company Group Entities could reasonably be expected to have any material direct or indirect Liability. The term "Plan" does not include any arrangement that has been terminated and completely wound up prior to the date of this Agreement and for which none of the Company Group Entities and Seller Group Entities has any present or potential material liability, nor does it include any employment, agency or consulting Contract or private pension, insurance, disability or other plan maintained by any employee.
- "Pre-Closing Taxes" means (i) any Taxes imposed on or payable by any Company Group Entity for taxable periods (or portions thereof) before the Closing Date (determine with respect to taxable periods that include but do not end on the Closing Date, pursuant to Section 13.4 (Straddle Period Allocations)), and (ii) any Taxes resulting from the Transaction.
- "Pre-Closing Warranties" means (i) the actual cost to Purchaser or any Company Group Entity to repair, replace or provide customer credit, as set forth in the applicable customer agreement, for products sold or shipped by Seller, its Affiliates or any Company Group Entity prior to the Closing that fail to conform to the warranty applicable to such products and which are returned to Purchaser or any Company Group Entity; (ii) the actual transportation costs paid by Purchaser or any Company Group Entity for shipment of a repaired or replaced product to the customer of such returned product; and (iii) labor costs actually incurred by Purchaser or any Company Group Entity for the installation of such repaired or replaced product.

- "Privacy Policies" has the meaning specified in Section 3.13(n).
- "Proceeding" means any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), prosecution, formal contest, hearing, formal inquiry, formal inquest, audit, formal examination or formal investigation commenced, brought, conducted or heard at law or in equity or before any Governmental Authority or any arbitrator or arbitration panel.
- "Pro Forma Company Group Entities" means the Company and its Subsidiaries as shall exist at the Closing and the Amorton Business (including as a Subsidiary for this purpose, SSMPE, and SNA to the extent of the Transferred SNA Business), subject to adjustments to be agreed upon in writing by Seller and Purchaser Parent.
- "Prohibited Business" means any activity or other business that is competitive or in competition with the Business of the Company Group Entities, other than the Amorton photovoltaic device business, Laser diode business (but not the Laser driver business), LED business (but not the LED driver business), ISB (Integrated System in Board) business, designing of ASIC (application specific integrated circuits) for digital camera, and designing of semiconductor integrated circuit and its module for the development of for power management modules and tuner modules.
- "Projected Benefit Obligations" means, in respect of any employee as of a specified date, the aggregate of the obligations of a Company Group Entity or Seller Group Entity, as applicable, or of a fund that has recourse to a Company Group Entity or Seller Group Entity, for the benefits in respect of such employee, which benefits accrue during such employee's period of service as determined based on a formula stipulated in employment regulations, employee contracts, the relevant Company Group Entity or Seller Group Entity employee pension fund or applicable Legal Requirement, and which are payable after termination of such employee's employment on a voluntary or involuntary basis. The determination of Projected Benefit Obligations shall be based on certain assumptions that are consistent with assumptions used in the Company's Audited Pro Forma Financial Statements for the fiscal year ended March 31, 2010, and consistent with the terms set forth in Annex I. For the avoidance of doubt, Projected Benefit Obligations do not include any additional liabilities that may arise in connection with layoffs or other terminations by a Company Group Entity or Seller Group Entity.
- "Property" means all real property leased or owned by any Company Group Entity.
- "Purchase Price" has the meaning specified in Section 1.3(a).
- "Purchase Price Adjustment Amount" has the meaning specified in Section 1.4(h).
- "Purchased Business" has the meaning specified in Section 1.1.
- "Purchased Shares" means the Company Shares.

- "Purchaser" has the meaning specified in the preamble of this Agreement.
- "Purchaser Parent" has the meaning specified in the preamble of this Agreement.
- "Purchaser Parent Share Price" means six hundred and five point ninety-two Yen (¥605.92) per Purchaser Parent Share; provided, however, that if during the period commencing on the date hereof and ending on the Closing Date, the outstanding Purchaser Parent Shares are changed into a different number or class of shares by reason of any stock split, division or subdivision of shares, stock dividend, reverse stock split, consolidation, reclassification, recapitalization or other similar transaction, or if a stock dividend is declared by Purchaser Parent during such period, or a record date with respect to any such event shall occur during such period, then the Purchaser Parent Share Price shall be adjusted to the extent appropriate to provide the same economic effect as would have resulted pursuant to this Agreement had no such action been taken in respect of the Purchaser Parent Shares.
- "Purchaser Parent Shares" has the meaning specified in Section 1.3(c)(ii).
- "Purchaser Parties" has the meaning specified in Section 12.2(a).
- "Real Property" has the meaning specified in Section 3.22(b).
- "Real Property Leases" means all leases, subleases, licenses, concessions and other agreements or Contracts (written or oral), including all amendments, extensions, renewals, guaranties, and other agreements with respect thereto, pursuant to which any Company Group Entity holds any Leased Real Property, including the right to all security deposits and other amounts and instruments deposited by or on behalf of any Company Group Entity.
- "Registered IP Assets" means issued Patents, registered Trademarks, and registered Copyrights, including any pending applications for any of the foregoing.
- "*Release*" means any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, migrating, dumping, or disposing of Hazardous Materials (including the abandonment or discarding of barrels, containers or other closed receptacles containing Hazardous Materials) into the environment, whether intentional or unintentional, negligent or non-negligent, sudden or non-sudden, accidental or non-accidental.
- "Relevant Books and Records" means all original books of account, ledgers, general, financial and accounting records, files, papers, customers' and suppliers' lists, sales and promotional literature, customer and supplier correspondence and other data, records and documents primarily related to, or primarily used by any of the Seller Group Entities and the Company Group Entities in connection with, the Business whether or not physically held at a Company Group Entity as of the date hereof.
- "Remaining Offered Shares" has the meaning specified in Section 14.3(c).
- "Reorganization" has the meaning specified in Section 9.1(f).

- "Representatives" means, in respect of any party, officers, directors, employees, attorneys, accountants, advisors, agents, and subsidiaries of such party.
- "*Resolutions*" means, collectively, all minutes and other records of all meetings and other proceedings (including any actions taken by written consent or otherwise without a meeting) of the shareholders of a Company Group Entity, a Company Group Entity's Board of Directors and all committees thereof.
- "Restricted Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; (b) used by Company Group Entities primarily in the Business as conducted immediately prior to the Closing Date; and (c) assignable only with the consent of the Third-Party IP Owners, subject to obtaining the consent of such Third-Party IP Owners.
- "Restricted Assigned Patents" means Patents listed on Exhibit H to the Intellectual Property Assignment Agreement, which are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; and (b) assignable only with the consent of the Third-Party IP Owners.
- "Restricted IP Assets" means all Business IP Assets any Seller Group Entity owns jointly with a third party or parties that may not be assigned without the consent of such third party or parties.
- "Restructuring Plan" has the meaning specified in Section 8.2(a).
- "ROFO Exercise Notice" has the meaning specified in Section 14.3(b).
- "ROFO Exercise Period" has the meaning specified in Section 14.3(b).
- "ROFO Notice" has the meaning specified in Section 14.3(a).
- "ROFO Notice Date" means the date on which Seller delivers the ROFO Notice.
- "ROFO Price" means the average trading price of Purchaser Parent's common stock for the twenty (20) day period commencing nine (9) days prior to the ROFO Notice Date and ending ten (10) days following the ROFO Notice Date.
- "ROFO Right" has the meaning specified in Section 14.3(b).
- "ROFO Shares" has the meaning specified in Section 14.3(b).
- "Sales and Marketing Personnel" means Eligible Business Employees primarily engaged in sales and marketing activities for or relating to the Business of the Company Group Entities.
- "SAS 100" means Statement of Auditing Standards 100: Objective and General Principles Governing an Audit of Financial Statements.

- "SCE" means SANYO Component Europe GmbH.
- "SCE Employees" means the Active Employees employed by SCE.
- "Seconded Employees" means those employees of any Seller Group Entity who are seconded to any Company Group Entity.
- "Secondment Agreements" has the meaning specified in Section 7.7.
- "Securities Act" means the Securities Act of 1933, as amended.
- "Securitization of Receivables Transactions" means such arrangements under which any of Company Group Entity Products are supplied outside of Japan via a Seller Group Entity.
- "Seller" has the meaning specified in the preamble of this Agreement.
- "Seller Account" has the meaning specified in Section 1.3(c)(i).
- "Seller Basket" has the meaning specified in Section 12.5(a).
- "Seller Disclosure Schedule" means the disclosure schedule delivered by Seller to Purchaser on the date hereof.
- "Seller Group Entities" means Seller and its Subsidiaries other than the Company Group Entities.
- "Seller Incurred Restructuring Plan Costs" has the meaning specified in Section 1.4(a)(vii).
- "Seller IP Assets" means all Business IP Assets any Seller Group Entity owns or has a license to, other than Restricted IP Assets.
- "Seller Loan Receivables" means the outstanding amount of certain loans made by Seller to Company pursuant to a loan agreement dated December 26, 2008, a complete and correct copy of which is attached hereto as Exhibit B.
- "Seller Parties" has the meaning specified in Section 12.3(a).
- "Seller Pension Plans" means pension plans of Seller.
- "Seller Restructuring Plan Costs" has the meaning specified in Section 8.2(a).
- "Seller Restructuring Plan Costs Schedule" has the meaning specified in Section 8.2(a).
- "SET" means SANYO Electric (Taichung) Co., Ltd.
- "SET Union" means the labor union of SET.
- "Shareholders' Register Request" has the meaning specified in Section 2.1(a).

- "Shoko" means The Shoko Chukin Bank, Ltd.
- "Shoko Loan" means the Indebtedness of KSS pursuant to a loan agreement between KSS and Shoko.
- "SNA" means SANYO North American Corporation.
- "SNA Employees" means the Active Employees employed by SNA.
- "Specified Employee" means those Active Employees listed on the Specified Employee List.
- "Specified Employee List" has the meaning specified in Section 6.17.
- "Specified Rep" means (i) with respect to Seller, the representations and warranties of Seller contained in Section 3.1 (Organization, Good Standing, Qualification), Section 3.2(a) (Company Group Entities), Section 3.3(c) (Capitalization), Section 3.8 (Ownership and Sufficiency of Business Assets), Section 3.13 (Intellectual Property), Section 3.15 (Employee Benefits), Section 3.19 (Environmental Matters), Section 3.20 (Taxes), Section 4.1 (Ownership of Purchased Shares) and Section 4.2 (Authority: Validity of Contemplated Transaction) and (ii) with respect to Purchaser, the representations and warranties of Purchaser contained in Article V, except for Section 5.5 (SEC Filings, Financial Statements) and Section 5.6 (Proceedings).
- "SSDC" means SANYO Semicon Device Sales Co., Ltd.
- "SSDC Employees" means employees of SSDC.
- "SSMC" means SANYO Semiconductor Manufacturing Co., Ltd.
- "SSMC Employees" mean Niigata Employees, Gifu and Gunma Employees.
- "SSMC Supplier Alternative Payment System" means any arrangement under which a Seller Group Entity causes certain banks to pay the accounts payable of SSMC arising from its purchase of certain parts and goods from certain of its suppliers on its behalf, and all Contracts entered into in connection with or otherwise related to such arrangements.
- "SSMC Transfer Date" means a date specified by Purchaser, after consultation with Seller, for the transfer of the Gifu and Gunma Employees from SSMC to New SSMC, expected (but not required) to be in the fourth quarter of calendar year 2011, but in any event no later than June 30, 2012.
- "SSMPE" means SSMP Estate Corporation.
- "SSMPE Ownership Interests" has the meaning specified in Section 7.22.
- "SSV Union" means the labor union of SANYO Semiconductor (Vietnam) Co., Ltd.
- "Stage 1 Securitization Entities" means the Entities set forth on Schedule 7.15(a)(i).

- "Stage 2 Securitization Entities" means each of the Company Group Entities other than the Stage 1 Securitization Entities.
- "Stock Consideration" has the meaning specified in Section 1.3(c)(ii).
- "Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association, or other business entity of which (i) if a corporation or limited liability company in Japan, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers, or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof or (ii) if a limited liability company (outside of Japan), partnership, association, or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more Subsidiaries of that Person or a combination thereof and for this purpose, a Person or Persons own a majority ownership interest in such a business entity (other than a corporation) if such Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be or control any managing director or general partner of such business entity (other than a corporation).
- "Supplier Alternative Payment System" means the Company Supplier Alternative Payment System and the SSMC Supplier Alternative Payment System.
- "Target Number of Transferring Gifu and Gunma Employee" has meaning specified in Annex III.
- "Target Number of Transferring KSS Employee" has the meaning specified in Annex III.
- "Tax" (and, with correlative meaning, "Taxes") means (a) any federal, state, local or foreign income, alternative or add-on minimum income, ad valorem, business license, capital, custom, disability, documentary, employment, environmental, excise, franchise, gains, gross income, gross receipts, import, license, occupation, payroll, personal property, premium, profits, property transfer, real property, recording, registration, sales, services, severance, social security, stamp, transfer, unemployment, unemployment insurance, use, value added, wage, windfall profit or withholding tax, custom, duty, levy or other governmental assessment, charge or fee in the nature of a tax (whether payable directly or by withholding); (b) any liability for Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by contract or otherwise; and (c) any estimated Tax, interest, fines, penalties or additions to Tax with respect to amounts referred to in clauses (a) or (b) hereof.
- "*Tax Return*" means any return, report, estimate, certificate, declaration of estimated tax, claim for refund, information statement or other document filed with or submitted, or required to be filed with or submitted with or to any governmental Tax authority with respect to the determination, assessment, collection or payment of any Taxes, including information returns, statements (whether or not attached to a return).

"Technology" means, collectively, all technology, formulae, recipes, specifications (including information regarding materials, ingredients, tools, apparatus, sources, vendors), procedures, processes, methods, techniques, ideas, creations, inventions and discoveries (whether patentable or unpatentable and whether or not reduced to practice), improvements, know-how, research and development, technical data, designs, models, algorithms, subroutines, software, works of authorship, copyrightable subject matter, and other similar materials, recordings, graphs, drawings, reports, analyses, and other writings, confidential or proprietary information, and general intangibles of like nature, and all tangible embodiments of the foregoing, in any form whether or not specifically listed herein.

"Technology Systems" has the meaning specified in Section 3.13(m).

"Third-Party IP Owners" means a third party (other than a Company Group Entity or Seller Group Entity) that owns a joint Patent or Copyright together with one or more Seller Group Entities.

"Third Party License" means a license agreement between any Company Group Entity or Seller Group Entity and a third party which grants any Company Group Entity a right to use such third party Intellectual Property Rights prior to the Closing Date.

"Third Party Non-Patent License" shall mean any Contract that licenses to a Seller Group Entity or Company Group Entity any Intellectual Property Rights (other than Patents) that were used by the Company in the Business.

"Third Party Non-Transferred Non-Patent License" shall mean any Contract that licenses to a Seller Group Entity or Company Group Entity any Intellectual Property Rights (other than Patents) that are used by any Company Group Entity in the Business, that was not transferred from Seller to Company pursuant to the Company Demerger or otherwise transferred from Seller to Company prior to Closing.

"Third Party Patent License" means those Contracts set forth in Schedule 3.13(0) of the Seller Disclosure Schedule.

"Third Party Purchaser" has the meaning specified in Section 14.3(c).

"Trademarks" means all trademarks, service marks, trade names, service names, brand names, trade dress, together with the goodwill associated with any of the foregoing, and all applications, ITU Applications, registrations and renewals thereof.

"Trademark License Agreement" has the meaning specified in Section 7.2.

"Trade Secrets" means any trade secrets or similar forms of protection for confidential information, including invention disclosures, formulae, recipes, specifications (including information regarding materials, ingredients, tools, apparatus, sources, vendors), procedures, processes, methods, techniques, ideas, creations, inventions and discoveries (whether patentable or unpatentable and whether or not reduced to practice), improvements, know-how, research and development, technical data, designs, models, algorithms, subroutines and similar confidential information.

- "Transaction" means, collectively, the transactions contemplated by the Transaction Agreements, and the execution, delivery and performance thereof.
- "Transaction Agreements" means (i) this Agreement; (ii) the Escrow Agreement; (iii) the Secondment Agreements; (iv) the IP Assignment Agreement; (v) the Intellectual Property License Agreement; (vi) the Transition Services Agreement; (vii) the Lease Agreements; (viii) the Intercompany Release; (ix) Transferred SCE Business Assignment and Assumption; (x) Transferred SNA Business Assignment and Assumption.
- "Transfer Consents" has the meaning specified in Section 9.1(c)(i).
- "Transferred Employee" means each employee of Seller that will transfer to the Company as at the Closing.
- "Transfer Offered Employees" means any employee of Seller primarily engaged in Business as of the date hereof.
- "Transferred Gifu and Gunma Employees" means those Gifu Employees and Gunma Employees who transfer from SSMC to New SSMC as of the SSMC Transfer Date.
- "Transferred IP Assets" means, collectively, the Transferred Patents, the Transferred Copyrights and Know-How and the Transferred Trademarks, each as defined under the Intellectual Property Assignment Agreement.
- "Transferred KSS Employees" means those KSS Employees who transfer to KSS from New KSS as of the KSS Transfer Date.
- "Transferred SCE Business" means all of the assets and liabilities of SCE to be transferred under the Transferred SCE Business Assignment and Assumption.
- "Transferred SNA Business" means all of the assets and liabilities of SCE to be transferred under the Transferred SNA Business Assignment and Assumption.
- "Transferred SCE Business Assignment and Assumption" has the meaning specified in Section 7.21.
- "Transferred SNA Business Assignment and Assumption" has the meaning specified in Section 7.20.
- "Transferred Unfunded Benefit Liabilities" means the aggregate of (i) the Unfunded Benefit Liabilities associated with the Transferred Employees, (ii) the Unfunded Benefit Liabilities associated with Transferred KSS Employees.
- "Transfer Taxes" has the meaning specified in Section 13.3.
- "Transition Services Agreement" has the meaning specified in Section 7.3.

- "Unassigned Third Party Patent License" means those Third Party Patent Licenses for which Seller does not obtain the necessary Consent to assign such Third Party Patent License to Company (or its designated Company Group Entity).
- "Unaudited Financial Statements" has the meaning specified in Section 3.5(a).
- "Unfunded Benefit Liabilities" means, in respect of any employee as of a specified date, the unfunded portion of the Projected Benefit Obligations, as calculated in accordance with those provision of J-GAAP and applicable to Japanese corporations with greater than three hundred (300) employees, with respect to such employee as of such specified date, less the market value of any pension fund assets associated therewith as of such specified date. The determination of Unfunded Benefit Liabilities shall be based on certain assumptions that are consistent with assumptions used in the Company's Audited Pro Forma Financial Statements for the fiscal year ended March 31, 2010, and consistent with the terms set forth in Annex I.
- "U.S. GAAP" means generally accepted accounting principles in the United States of America in effect on the date on which they are to be applied pursuant to this Agreement, applied consistently throughout the relevant periods.
- "U.S. GAAS" means U.S. Generally Accepted Auditing Standards.
- "VDR" means that certain online virtual data room to which Seller has disclosed certain information regarding the Seller Group Entities, Company Group Entities and the Business.
- "Willful Breach" means, in respect of the breach of a covenant by a party, (a) such party shall have willfully breached such covenant and (b) any of such party's directors or executive officers had actual knowledge, at the time of Seller's breach of such covenant, that Seller was breaching such covenant.
- "Work Permit" has the meaning specified in Section 3.14(b).
- "Yen" means Japanese Yen.

INTELLECTUAL PROPERTY LICENSE AGREEMENT

by and between

SANYO SEMICONDUCTOR CO., LTD.

a Japanese corporation

and

SANYO ELECTRIC CO., LTD.

a Japanese corporation

Dated as of [__], 2010

TABLE OF CONTENTS

		Page
ARTICLE I	PATENT LICENSES AND COVENANTS	1
1.1	License of Licensed Patents	1
1.2	License of Remaining Semiconductor Patents	2
1.3	Covenant Not to Sue	2
1.4	Covenant to Sue Last	3
1.5	Remedies	3
1.6	No Exhaustion of Seller Patents	4
1.7	Binding Effect of Covenants	4
ARTICLE II	COPYRIGHT AND KNOW-HOW LICENSES	4
2.1	License of Retained Copyrights and Know-How	4
ARTICLE III	ADDITIONAL TERMS	5
3.1	No Other Rights	5
3.2	Subsidiaries	5
3.3	Limitations	5
3.4	Further Assurances	6
ARTICLE IV	CONFIDENTIALITY	6
4.1	Company Obligations	6
4.2	Exclusions	7
4.3	Required Disclosures	7
4.4	No Limitation on Use	7
4.5	Survival	7
ARTICLE V	ASSIGNMENT AND TRANSFERABILITY	7
5.1	Permitted Assignment, Merger, and Change of Control	7
5.2	No Other Assignment	8
ARTICLE VI	NO WARRANTY, INDEMNIFICATION AND LIMITED LIABILITY	8
6.1	No Warranties	8
6.2	Indemnification	8
6.3	Limitation of Liability	8
ARTICLE VII	TERM	8
7.1	Term	8

9

7.2

Termination

TABLE OF CONTENTS

(continued)

		Page
ARTICLE VIII	MISCELLANEOUS	9
8.1	Purchase Agreement Terms	9
8.2	Injunctive Relief	9
8.3	Notices	9
<u>Exhibits</u>		
Exhibit A:	Certain Defined Terms	A-1

INTELLECTUAL PROPERTY LICENSE AGREEMENT

This INTELLECTUAL PROPERTY LICENSE AGREEMENT (the "*Agreement*") dated as of [__], 2010 ("*Effective Date*"), is entered into between Sanyo Electric Co., Ltd., a company organized under the laws of Japan ("*Seller*") and Sanyo Semiconductor Co., Ltd., a company organized under the laws of Japan ("*Company*"). Each of Seller and Company are sometimes hereinafter referred to as a "*Party*" and collectively as the "*Parties*." Certain capitalized terms used in this Agreement are defined in Exhibit A or in the Purchase Agreement (if not defined in Exhibit A).

RECITALS

- A. WHEREAS, Seller, ON Semiconductor Corporation ("Purchaser Parent"), and SCI LLC ("Purchaser") have entered into that certain Purchase Agreement, dated as of July 15, 2010 ("Purchase Agreement"), pursuant to which Purchaser will purchase from Seller, and Seller will sell to Purchaser, all of the issued and paid up share capital of Company, and Purchaser, Seller and Company will enter into certain other transactions, all on the terms and subject to the conditions set forth in the Purchase Agreement;
- B. WHEREAS, in connection with the Transaction, the Parties or their affiliates have entered into the Intellectual Property Assignment Agreement and the Trademark License Agreement; and
- C. WHEREAS, in connection with the Transaction, Seller wishes to grant to Company, and Company wishes to receive, licenses with respect to intellectual property rights and assets retained by Seller after the Transaction;

NOW, THEREFORE, the Parties hereby agree as follows:

AGREEMENT

ARTICLE I

PATENT LICENSES AND COVENANTS

- 1.1 License of Licensed Patents. Seller, on behalf of itself and all Seller Group Entities, hereby grants Company and Company Group Entities a non-exclusive, non-transferable (except as set forth herein), irrevocable, perpetual, non-sublicenseable, worldwide, fully paid-up and royalty-free license, under the Licensed Patents (but excluding Systems-Level Claims):
- (a) to make, Have Made, use, lease, sell, offer for sale, import, modify, develop, export and otherwise commercially exploit (directly or indirectly though multiple tiers, and including the right to have such rights exercised by third parties solely on behalf of and solely for the benefit of Company Group Entities) and dispose of Semiconductor Products;

Note to draft: The party to this agreement may change depending on tax and structuring issues, such as possible use of an IP holding company.

- (b) to develop and use (but not make or Have Made) Manufacturing Apparatus and to sell used (but not new) Manufacturing Apparatus;
- (c) to use any processes or methods covered by Licensed Patents (but excluding Systems-Level Claims), to manufacture Semiconductor Products; and
- (d) to provide or perform Semiconductor Services, including to use any processes or methods covered by Licensed Patents (but excluding Systems-Level Claims).
- **1.2 License of Remaining Semiconductor Patents**. Seller, on behalf of itself and all Seller Group Entities, hereby grants Company and Company Group Entities a non-exclusive, non-transferable (except as set forth herein), irrevocable, perpetual, non-sublicenseable, worldwide, fully paid-up, royalty-free license, under the Remaining Semiconductor Patents:
- (a) to make, Have Made, use, lease, sell, offer for sale, import, modify, develop, export and otherwise commercially exploit (directly or indirectly though multiple tiers, and including the right to have such rights exercised by third parties solely on behalf of and solely for the benefit of Company Group Entities) and dispose of Current Company Products; and
 - (b) to develop and use (but not make or Have Made) Manufacturing Apparatus and to sell used (but not new) Manufacturing Apparatus;
 - (c) to use any processes or methods covered by Remaining Semiconductor Patents to manufacture Current Company Products; and
- (d) to provide or perform Semiconductor Services, to the extent such Semiconductor Services are provided by Company or Company Group Entities as of the Closing Date, including to use any processes or methods covered by Licensed Patents.
- **1.3 Covenant Not to Sue.** Subject to the conditions and limitations set forth in Section 1.6, and without limiting the generality of the license granted in Section 1.2, Seller, on behalf of itself and all Seller Group Entities, hereby irrevocably and perpetually (subject only to Section 1.6) covenants that neither it nor any Seller Group Entity will assert (by way of litigation or any other Proceeding), anywhere in the world, any of the Remaining Semiconductor Patents (but excluding Systems-Level Claims):
- (a) against Company or any Company Group Entity for: (i) making, Having Made, using, leasing, selling, offering for sale, importing, modifying, developing, exporting or otherwise commercially exploiting (directly or indirectly through multiple tiers, including having such rights exercised by third parties solely on behalf of and solely for the benefit of Company Group Entities) or disposing of Semiconductor Products; (ii) developing or using Manufacturing Apparatus or selling used Manufacturing Apparatus, (iii) using any processes or methods covered by Remaining Semiconductor Patents (but excluding Systems-Level Claims) to manufacture Semiconductor Products; or (iv) providing or performing Semiconductor Services, including using any processes or methods covered by Remaining Semiconductor Patents (but excluding Systems-Level Claims); and

- (b) against Company's or any Company Group Entity's foundries or suppliers for manufacturing Semiconductor Products or otherwise providing Semiconductor Services on behalf of, and selling or otherwise distributing such Semiconductor Products to, Company or any Company Group Entity that are Have Made by such foundries or suppliers in accordance with the limitations to Have Made hereunder.
- 1.4 Covenant to Sue Last. Seller, on behalf of itself and all Seller Group Entities, hereby irrevocably and perpetually (subject only to Section 1.6) covenants that it and the Seller Group Entities will not assert (by way of litigation or any other Proceeding), anywhere in the world, any claim of infringement or other violation of any CTSL Patents or any Systems-Level Claims of any Licensed Patents against Company or any Company Group Entity with respect to any Semiconductor Products or Semiconductor Services unless and until (a) Seller and the Seller Group Entities have exhausted all of their rights and remedies under such CTSL Patents against Company's and the Company Group Entities' Customers in a final non-appealable decision that includes one or more findings of infringement of such CTSL Patents by such products or services and (b) Seller and the Seller Group Entities have diligently sought to collect the monetary damages awarded for such infringement but have been unable to do so.
- **1.5 Remedies.** Notwithstanding and without limitation to Sections 1.3 and 1.4, in the event that Seller or any Seller Group Entity brings an action against Company or any Company Group Entity for infringement of the Remaining Semiconductor Patents or the CTSL Patents, as the case may be, the sole and exclusive remedy available to Seller and the Seller Group Entities will be recovery of reasonable royalties based on the revenue received by Company and the Company Group Entities (*e.g.*, no injunctive or other equitable relief, enhanced damages for willfulness, lost profits, interest, and attorneys' fees), and Seller and the Seller Group Entities will not seek any remedies other than such reasonable royalties.
- 1.6 No Exhaustion of Seller Patents. The Parties acknowledge and agree that (a) both Parties intend that the covenants set forth in Sections 1.3 and 1.4 above do not and shall not (whether expressly, impliedly, by estoppel, by operation of law, or otherwise) exhaust or create any implied licenses under any of the Remaining Semiconductor Patents, CTSL Patents, or Systems-Level Claims of Licensed Patents, and (b) the covenants set forth in Sections 1.3 and 1.4 above do not authorize the sale of any product or service to any third party that claims or contends in any Proceeding, anywhere in the world, that the covenants set forth in Sections 1.3 and 1.4 above, in whole or in part, do exhaust, or create any implied licenses that would result in immunity for the use, sale or manufacture of products that incorporate Semiconductor Products under any Remaining Semiconductor Patents, CTSL Patents, or Systems-Level Claims of Licensed Patents. The Parties further agree and intend that such covenants may not be circumvented by any litigation or similar legal, regulatory or administrative proceeding based, in whole or in part, on any theory or doctrine of extra-territorial patent infringement or indirect infringement. If, despite the Parties' joint intent, it is adjudicated in a final non-appealable decision in any country or jurisdiction that either the covenant set forth in Section 1.3 or the covenant set forth in Section 1.4 (or the combination of such covenants) exhausts, or creates any implied license that results in immunity for the use, sale or manufacture of products that incorporate Semiconductor Products under, any of the Remaining Semiconductor Patents, CTSL Patents, or Systems-Level Claims of Licensed Patents, then the relevant portion of such covenant (set forth in either Section 1.3 or Section 1.4 as the case may be) will be deemed to be null and void from inception, but only with respect to such country or jurisdiction where such final non-appealable decision is issued and takes effect. After any such final,

1.7 Binding Effect of Covenants. The licenses and covenants in Sections 1.1, 1.2, 1.3, and 1.4 are intended to run with the Remaining Semiconductor Patents and CTSL Patents, as the case may be, irrespective of any future transfers, assignments and licenses, and intended to be binding on any future assignee or successor in interest to the rights, title, and/or interest of Seller and the Seller Group Entities therein. Seller shall ensure that such assignee or successor in interest agrees to assume and is bound by these licenses and covenants. The Parties acknowledge and agree that the Licensed Patents, Remaining Semiconductor Patents, and CTSL Patents are "intellectual property" as defined in Section 101(35A) of the United States Bankruptcy Code, as may be amended from time to time (the "Code"), and rights under such intellectual property have been granted hereunder in a contemporaneous exchange for value. The Parties further acknowledge and agree that if Seller or any Seller Group Entity: (i) becomes insolvent or generally fails to pay, or admits in writing its inability to pay, its debts as they become due; (ii) applies for or consents to the appointment of a trustee, receiver or other custodian for it, or makes a general assignment for the benefit of its creditors; (iii) commences, or has commenced against it, any bankruptcy, reorganization, debt arrangement, or other case or proceeding under any bankruptcy or insolvency law, or any dissolution or liquidation proceedings; or (iv) elects to reject, or a trustee on behalf of it elects to reject, this Agreement or any agreement supplementary hereto (including, without limitation, the covenants set forth in Sections 1.3 and 1.4 above), pursuant to Section 365 of the Code, or if this Agreement or any agreement supplementary hereto (including, without limitation, the covenants set forth in Sections 1.3 and 1.4 above) shall be governed by Section 365(n) of the Code, and Company and the Company Group Entities will retain and may elect to fully exercise their rights un

ARTICLE II

COPYRIGHT AND KNOW-HOW LICENSES

2.1 License of Retained Copyrights and Know-How. Seller, on behalf of itself and all Seller Group Entities, hereby grants Company and all Company Group Entities a non-exclusive, non-transferable (except as set forth herein), irrevocable, perpetual, sublicenseable, worldwide, fully paid-up and royalty-free right and license, under all of the Intellectual Property Rights of any kind and character throughout the world (other than Patents and Trademarks) owned, managed or controlled by Seller and all Seller Group Entities, to all Retained Copyrights and Know-How for any use or purpose whatsoever, now known or hereafter discovered. This license includes the right to use, execute, reproduce, prepare derivative works based on, distribute, transmit (internally and externally), disclose and otherwise exploit, in any form, the Retained Copyrights and Know-How and derivative works based thereon, and to grant revocable or irrevocable sublicenses to any third party (directly and indirectly through multiple tiers) to do any or all of the foregoing.

ARTICLE III

ADDITIONAL TERMS

- **3.1 No Other Rights**. No other rights are granted hereunder, by implication, estoppel, statute or otherwise, except as expressly provided herein. The licenses and covenants granted under Article I expressly do not apply to any portion of a product manufactured or assembled by Company and Company Group Entities using designs, specifications, and/or working drawings furnished by or on behalf of, or owned by, a third party (excluding library tools and standard cells), and the license and covenants granted hereunder shall not extend to any third parties providing Have Made services for Company to the extent that such third parties incorporate their own designs, specifications, and/or working drawings. For avoidance of doubt, Purchaser Parent (as well as any Subsidiary of Purchaser Parent that is not Company or a Subsidiary of Company) is a third party.
- **3.2 Subsidiaries.** With respect to the licenses in Articles I and II, in the event an entity becomes a Subsidiary of either Party during the term of this Agreement, such new Subsidiary, upon becoming a Subsidiary, shall receive the same licenses and covenants as granted hereunder. In the event a Subsidiary of either Party ceases to be a Subsidiary during or after the term of this Agreement, the licenses and covenants granted hereunder to such Subsidiary shall automatically terminate forthwith. However, the licenses and covenants granted to the other Party and its Subsidiaries hereunder shall continue after such entity ceases to be a Subsidiary (even if such entity owns any of the Intellectual Property Rights that are subject to such licenses and covenants), and the Party that was formerly the parent of such Subsidiary shall ensure that such Subsidiary agrees to assume and be bound by such licenses and covenants with respect to any such Intellectual Property Rights.
- 3.3 Limitations. The licenses in Articles I and II shall not include any right or license to any Licensed Patents, Remaining Semiconductor Patents or Retained Copyrights and Know-How that cannot be licensed without the consent of, or the obligation to make payments to, a Third-Party IP Owner in order to grant such licenses or rights (except for payments to Seller's Subsidiaries or payments to any Person for inventions made while employed by Seller or any of its Subsidiaries). Without limitation to Seller's indemnification obligations under the Purchase Agreement, at Company's written request, Seller shall use reasonable efforts to obtain consent to grant such a license and, if Seller is able to obtain such consent (or such consent is not necessary) and Company agrees to pay all such payments on behalf of Seller and fully indemnify Seller and its Subsidiaries for any failure or delay by Company in making such payments, Seller shall grant such a license. In the event of, and notwithstanding, any failure by Seller or a Seller Group Entity to obtain the consent to license such Licensed Patents, Remaining Semiconductor Patents or Retained Copyrights and Know-How to Company or a Company Group Entity from the Third-Party IP Owner(s), Seller, on behalf of itself and all Seller Group Entities, hereby covenants not to commence, voluntarily join, or cooperate in any Proceeding brought by any Third-Party IP Owner against Company or a Company Group Entity for any alleged infringement, violation or misappropriation of such Licensed Patents or Retained Copyrights and Know-How.

3.4 Further Assurances. Upon Company's reasonable request, Seller will, and will cause all Seller Group Entities and its and their Representatives to, execute, verify, acknowledge and deliver all such further papers, in recordable form, as may be necessary to record, vest, secure and perfect the rights and interests of Company in and to the Licensed Patents, Remaining Semiconductor Patents, CTSL Patents and Retained Copyrights and Know-How.

ARTICLE IV

CONFIDENTIALITY

4.1 Company Obligations. Company shall, and shall cause the other Company Group Entities to, hold the Seller Confidential Information in confidence and not disclose any Seller Confidential Information to any third party, other than to: (a) its employees, agents, consultants, Subsidiaries and other Affiliates who need to know such information and who are bound by restrictions regarding disclosure of such Seller Confidential Information no less restrictive than those set forth herein; and (b) sublicensees, manufacturers, suppliers, contractors, distributors, and customers who are bound in writing or otherwise by restrictions regarding disclosure of such Seller Confidential Information no less restrictive than those set forth herein. The Company Group Entities shall take the same degree of care that they use to protect their own confidential and proprietary information and materials of similar nature and importance (but in no event less than reasonable care) to protect the confidentiality and avoid the unauthorized use, disclosure, publication or dissemination of the Seller Confidential Information. Notwithstanding anything herein to the contrary, the Company Group Entities may (without prior notification to, or approval or consent by, Seller) disclose to taxing authorities and/or to the Company's and Company Group Entities' representatives (including, outside counsel and advisors) any Seller Confidential Information that is required to be disclosed in connection with Company and the Company Group Entities' tax filings, reports, claims, audits, or litigation.

4.2 Exclusions. The obligations set forth in Section 4.1 with respect to Seller Confidential Information shall not apply to any Seller Confidential Information to the extent that such Seller Confidential Information: (a) is independently developed by Company or any Company Group Entity after the Closing Date, or by Purchaser Parent or its Affiliates that are not Company Group Entities at any time, in each case without the use of Seller Confidential Information; (b) is or becomes (or has already become) publicly available through no wrongful act of the Company or Company Group Entities; (c) is rightfully received by Company from a third party without obligation of confidentiality; (d) is disclosed with the consent of Seller or any of its Subsidiaries after the Closing Date; or (e) subject to Section 4.3 below, is disclosed in order to comply with any Legal Requirement (including any rule or regulation of the SEC or of a stock exchange which may require such disclosure) or with any order, rule or decree issued by a Governmental Authority. The inherent disclosure of any such Seller Confidential Information in either: (x) the use, lease, sale or other distribution of any present or future product or provision of any present or future service of Company or any of its Subsidiaries; (y) the documentation relating to such product or service reasonably made publicly available; or (z) the documentation relating to all domestic and foreign patent and copyright applications and registrations (and all patents and copyrights that issue therefrom and, to the extent applicable, all divisions, continuations, continuations-in-part, reexaminations, substitutions, resisues, extensions and renewals of such applications, registrations, copyrights and patents), shall not be deemed to be an unauthorized disclosure or publication of such information.

- **4.3 Required Disclosures.** Notwithstanding Section 4.2, in the event Company or any Company Group Entity is required to disclose Seller Confidential Information pursuant to any Legal Requirement, and would otherwise be prohibited from doing so under Section 4.1, Company or such Company Group Entity shall: (i) promptly notify Seller of the existence, terms and circumstances surrounding such requirement; (ii) consult with Seller on the advisability of taking legally available steps to resist or narrow such request; and (iii) if disclosure of such Seller Confidential Information is required, furnish only that portion of the Seller Confidential Information which Company or such Company Group Entity is legally compelled to disclose. Company or such Company Group Entity shall not oppose actions by Seller to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Seller Confidential Information.
- **4.4 No Limitation on Use.** Pursuant to the license granted in <u>Section 2.1</u>, Company and its Subsidiaries will be free to use the Seller Confidential Information for any purpose, including use in the development, manufacture, marketing, and maintenance of its products and services.
 - **4.5 Survival.** The obligations of this <u>Article IV</u> shall survive any termination or expiration of this Agreement.

ARTICLE V

ASSIGNMENT AND TRANSFERABILITY

5.1 Permitted Assignment, Merger, and Change of Control. This Agreement may be assigned or transferred by Seller to Panasonic or a Subsidiary of Panasonic upon written notice to Company, provided that Panasonic or such Subsidiary of Panasonic shall agree in writing to be bound by the terms and conditions of this Agreement. Company may assign or transfer this Agreement or its rights hereunder solely in connection with a corporate merger or consolidation, sale of all or substantially all of its assets, or Change of Control of Company as set forth below. In the event of any such assignment or transfer (other than an assignment or transfer resulting from the merger or consolidation of Company with or into a Subsidiary of Company), any merger or consolidation of Company with or into Purchaser Parent or any Subsidiary of Purchaser Parent that is not a Subsidiary of Company (regardless of which entity is the surviving entity), or any Change of Control of Company, Company shall (a) provide written notice to Seller within a reasonable period of time after such assignment or transfer, setting forth the name of the assignee, the entity with or into which Company will be merged or consolidated, or the new controlling entity of Company, as the case may be, (b) the assignee, the surviving entity of such merger or consolidation, or the new controlling entity of Company, as the case may be, must agree in writing to be bound by the terms and conditions of this Agreement, and (c) all of the rights and licenses granted herein pursuant to Article I (Patent Licenses and Covenants) to Company and its Subsidiaries will be limited to use with Semiconductor Products, Manufacturing Apparatus, and Semiconductor Services, in each case of Company Group Entities, that were (i) in existence immediately prior to the consummation of such assignment, merger, consolidation, or Change of Control, and (ii) under development immediately prior to the consummation of such assignment, merger, consolidation, or Change of Control and disclosed in cont

5.2 No Other Assignment. Except as set forth in Section 5.1, this Agreement shall inure solely to the benefit of the Parties hereto and their Subsidiaries, and may not otherwise be assigned or otherwise transferred, in whole or in part, by a Party, voluntarily or involuntarily, by merger, division, operation of law, corporate reorganization, or otherwise, without the prior written consent of the other Party. Any such purported assignment or transfer shall be null and void.

ARTICLE VI

NO WARRANTY, INDEMNIFICATION AND LIMITED LIABILITY

- **6.1 No Warranties.** EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THE PURCHASE AGREEMENT, THE PARTIES MAKE NO REPRESENTATIONS OR WARRANTIES, WHETHER EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE OR NON-INFRINGEMENT.
- **6.2 Indemnification.** Each Party (the "*Indemnifying Party*") shall defend, indemnify, and hold harmless the other Party, its Subsidiaries, and its and their Representatives (collectively, "*Indemnified Parties*") from any and all damages, actions, claims, fines, judgments, costs, losses, liabilities and expenses (including reasonable legal fees and litigation expenses) suffered by Indemnified Parties arising out of or relating to a breach by the Indemnifying Party or its Subsidiaries of any of its covenants or other provisions herein.
- **6.3 Limitation of Liability**. Except as may be expressly set forth in a Transaction Agreement and with respect to third party damages, costs, liabilities, and settlements payable pursuant to indemnification obligations under Section 6.2, neither Party nor any of its Subsidiaries or Representatives shall be liable for any indirect, punitive, special, incidental or consequential damages, or damages for loss of profits, business interruption or otherwise, arising from or relating to this Agreement, even if such Party, its Subsidiaries, or their Representatives are expressly advised of the possibility of such damages. The foregoing limitation of liability and exclusion of certain damages shall apply regardless of the failure of essential purpose of any remedies available to either Party.

ARTICLE VII

TERM

7.1 Term. All licenses, covenants, and grants of other rights under this Agreement shall be effective as of the Closing Date. The term of each granted license and covenant shall be for the life of the applicable Intellectual Property Right (except as expressly provided in Section 1.6). This Agreement shall continue in full force and effect until the expiration of the last to expire of the Intellectual Property Right licensed or for which a covenant is granted hereunder.

7.2 Termination. This Agreement, and the rights, licenses and covenants granted hereunder (except as expressly provided in <u>Section 1.6</u>), shall not be terminable by either Party or their respective Subsidiaries for any reason.

ARTICLE VIII

MISCELLANEOUS

- **8.1** Purchase Agreement Terms. The following provisions of the Purchase Agreement are hereby incorporated, mutatis mutandis, by this reference: 15.2, 15.3, 15.6, 15.8, 15.9, 15.11, 15.12, 15.13.
- **8.2 Injunctive Relief**. Notwithstanding anything to the contrary in this Agreement or the Purchase Agreement, Seller may seek temporary or preliminary injunctive relief in any court of competent jurisdiction in order to protect and preserve its Intellectual Property Rights and Seller Confidential Information and enforce its rights therein.
- **8.3 Notices.** All notices, requests, demands and other communications hereunder shall be either (a) delivered in person, (b) sent by overnight courier service or other express commercial delivery service, or (c) sent by facsimile with confirmation of receipt and, in each case, addressed as follows:

If to Company:	
with copies to (which copy shall not constitute notice):	
with copies to (winch copy shall not constitute notice).	
If to Seller:	
ii to Jenei.	
	_
with copies to (which copy shall not constitute notice):	: <u></u>

(a) All notices, requests, instructions or documents given to either Party in accordance with this <u>Section 8.3</u> shall be deemed to have been given on the date of mailing or transmission, whether delivered by hand, by overnight courier service, or by facsimile, with confirmation of receipt on such date.

(b) Either Party hereto may change its address specified for notices herein by designating a new address by notice given in accordance with this Section 8.3.

[Signatures Follow On a Separate Page]

SANYO ELECTRIC CO., LTD.
By:
Name:
Title:
SANYO SEMICONDUCTOR CO., LTD. L

[Signature Page to Intellectual Property Licence Agreement]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the date and year first above written.

EXHIBIT A

Certain Defined Terms

CONFIDENTIAL

EXHIBIT A

CERTAIN DEFINED TERMS

- "Assignment Documents" has the meaning set forth in Section 1.9 of the Intellectual Property Assignment Agreement.
- "Brand Licenses" means, collectively: (a) that certain Brand License Agreement, by and between Licensor and Licensee, dated April 1, 2008; (b) that certain Brand License Agreement, by and between Licensor and SSMC, dated April 1, 2008; (c) that certain Brand License Agreement, by and between Licensor and Sanyo LSI Design System Soft Co., Ltd., dated April 1, 2008; and (d) that certain Brand License Agreement, by and between Licensor and KSS dated April 1, 2008, dated April 1, 2008.
- "Code" has the meaning set forth in Section 1.7 of the Intellectual Property License Agreement and Section 2.6 of the Intellectual Property Assignment Agreement, as applicable.
- "Company Confidential Information" means all non-public Company Copyrights and Know-How.
- "Company Jointly Owned Patents" means Patents that are jointly owned by one or more Seller Group Entities and one or more Company Group Entities immediately following the Closing Date.
- "Company Names" has the meaning set forth in Section 1.1(b) of the Trademark License Agreement.
- "Company Copyrights and Know-How" means, collectively, (a) the Transferred Copyrights and Know-How and (b) Copyrights and Know-How that one or more Company Group Entities own as of the Closing Date.
- "Company Patents" means (a) the Transferred Patents, (b) all Patents that any of the Company Group Entities own as of the Closing Date (the "Company-Owned Patents"), and (c) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of the Company-Owned Patents.
- "Corporate Identity" means, collectively, letterhead, business cards, corporate signs, stationery, packaging slips, invoices, product datasheets, product labels, or any other materials that would normally identify a company or imply a corporate identity.

- "CTSL Patents" means all Patents, other than Licensed Patents, owned by Seller and the Seller Group Entities as of the Closing Date, and all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Patents, that would be infringed (absent a license) by making, using or selling Semiconductor Products (whether on a standalone basis or not), by using Manufacturing Apparatus to manufacture Semiconductor Products, or by providing Semiconductor Services. CTSL Patents include any Restricted Assigned Patents for which Seller has not obtained consent for assignment to HoldCo to the extent such Restricted Assigned Patents otherwise meet the definition set forth in the preceding sentence.
- "Current Company Products" means all Semiconductor Products that the Company Group Entities are selling, offering to sell, distributing, or making as of the Closing Date, including any minor improvements to the same that do not involve changes to existing features or functionality or the addition of new features or functionality.
- "Customer" means a third party (other than a Distributor or an End User) that purchases a Semiconductor Product (or a product containing one or more Semiconductor Products) manufactured by, or on behalf of, a Company Group Entity.
- "Distributor" means a third party that purchases a Semiconductor Product (or a product containing one or more Semiconductor Products) manufactured by, or on behalf of, a Company Group Entity and resells such Semiconductor Product (or such product containing one or more such Semiconductor Products, as the case may be) without modification or combination with any other component, part or product.
- "End User" means a third party that purchases a product containing one or more Semiconductor Products made by, or on behalf of, a Company Group Entity for such third party's own use and not for resale.
- "Effective Date" shall have the meaning set forth in the applicable preamble.
- "Filing" means the submission of any documentation, application, filing, registration or the like required to perfect or enforce, the Parties' interest in the Company Jointly Owned Patents under statutory Patent protection mechanisms, including, without limitation, any correspondence or other communication with any patent office or other Governmental Authorities with respect thereto.
- "Freely Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) solely owned by any Seller Group Entities, (ii) jointly owned by one or more Seller Group Entities and one or more Company Group Entities, (iii) jointly owned by one or more Seller Group Entities, Company Group Entities and Third-Party IP Owners, or (iv) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners; (b) primarily used in the Business by Company Group Entities as conducted immediately prior to the Closing Date; and (c) assignable without the consent of the Third-Party IP Owners (if any).
- "Freely Assigned Patents" means Patents that are listed on Exhibit F to the Intellectual Property Assignment Agreement.

- "Have Made" means the right to have a third party make a product or practice a method licensed under this Agreement for the use and benefit of a Party to this Agreement or its Subsidiary, provided that the designs, specifications, and/or working drawings for such products are supplied to such third party by, and originate with, the Party exercising such right or its Subsidiary (or with a Party's or its Subsidiaries' joint development with one or more third parties under terms of an agreement providing for the joint development of such designs, specifications, or working drawings).
- "HIC" means a hybrid integrated circuit, which is a chipset consisting of two or more integrated circuits that are packaged together onto a single device.
- "Know-How" means all right, title and interest in all Technology, Trade Secrets, ideas, concepts, proprietary techniques, processes, business and other methodologies, scientific, technical, research, development, engineering and business information, and other knowhow, including all rights arising under any law, including common law, state law, federal law or laws of foreign countries, other than Copyrights, Patents and Trademarks.
- "Licensed Forms" means the forms that incorporate either (i) the Licensed Marks, or (ii) the word "SANYO" or "□□" (as the case may be) set forth in Exhibit C to the Trademark License Agreement.
- "Licensed Marks" means the Trademarks set forth in Exhibit B to the Trademark License Agreement.
- "Licensed Patents" means (a) the Patents of Seller that are not Transferred Patents and are listed in <u>Exhibit B</u> to the Intellectual Property License Agreement, including any Restricted Assigned Patents for which Seller has not obtained consent for assignment, but has obtained consent to license to Company on the terms set forth in the Intellectual Property License Agreement ("Seller Licensed Patents"), and (b) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of the Seller Licensed Patents.
- "Manufacturing Apparatus" means any instrumentality or aggregate instrumentality primarily designed for use in the development and fabrication of Semiconductor Products.
- "Marked Products" has the meaning set forth in Section 2.2 of the Trademark License Agreement.
- "Name Change Procedures" has the meaning set forth in Section 1.1(b) of the Trademark License Agreement.
- "Panasonic" means Panasonic Corporation.
- "Parties" shall have the meaning set forth in the applicable preamble.
- "Party" shall have the meaning set forth in the applicable preamble.
- "Purchase Agreement" shall have the meaning set forth in the applicable recital.

- "Remaining Semiconductor Patents" means all Patents, other than the Licensed Patents, owned by Seller and the Seller Group Entities as of the Closing Date, and all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Patents, that would be infringed (absent a license) by making, using or selling Semiconductor Products (solely on a standalone basis), by using Manufacturing Apparatus to manufacture Semiconductor Products, or by providing Semiconductor Services. Remaining Semiconductor Patents include any Restricted Assigned Patents for which Seller has not obtained consent for assignment to HoldCo to the extent such Restricted Assigned Patents otherwise meet the definition set forth in the preceding sentence.
- "Restricted Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; (b) primarily used in the Business by Company Group Entities as conducted immediately prior to the Closing Date; and (c) assignable only with the consent of the Third-Party IP Owners, subject to obtaining the consent of such Third-Party IP Owners.
- "Restricted Assigned Patents" means Patents listed on Exhibit H to the Intellectual Property Assignment Agreement which are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; and (b) assignable only with the consent of the Third-Party IP Owners.
- "Retained Copyrights and Know-How" means the Copyrights and Know-How of Seller Group Entities that are not Transferred Copyrights and Know-How and that are used in the Business as currently conducted or planned to be conducted, including any Restricted Assigned Copyrights and Know-How for which Seller has not obtained consent for assignment to HoldCo.
- "Seller Confidential Information" means all non-public Retained Copyrights and Know-How.
- "Seller Licensed Products" means all products other than Semiconductor Products. For the avoidance of doubt, ISB ("Integrated System in Board") products, photovoltaic devices, LED diodes (but not drivers), and laser diodes (but not drivers) are each Seller Licensed Products. For the further avoidance of doubt, (i) ASICs (Application Specific Integrated Circuits) for digital camera products, and (ii) semiconductor devices, integrated circuits and HICs for the development of power management modules and tuner modules are not Seller Licensed Products.
- "Seller Licensed Services" means, collectively, (i) a service that is not a Semiconductor Service, (ii) designing of ASICs (Application Specific Integrated Circuits) for digital camera products, and (iii) designing of semiconductor devices, integrated circuits and HICs for the development of power management modules and tuner modules.

"Semiconductor Product" means any semiconductor device, integrated circuit or HIC (hybrid integrated circuit) on a standalone basis (i.e., a semiconductor device, integrated circuit or HIC alone, and not as part of a combination with other devices that are not semiconductor devices), whereby semiconductor devices include, but are not limited to, semiconductor device dies, semiconductor wafers or packaged semiconductors (including but not limited to supporting or packaging means or structure and software, firmware and microcode intended to be embedded or implemented therewith). Semiconductor devices do not lose their character as such whether or not part of an assemblage of packaged semiconductors or other devices. Semiconductor Products do not include ISB ("Integrated System in Board") products, photovoltaic devices, LED diodes, or laser diodes. For the avoidance of doubt, LED drivers and laser drivers are Semiconductor Products.

"Semiconductor Services" means the design, manufacture, testing, assembly, fabrication, and development of Semiconductor Products, including the provision of foundry services related thereto.

"Systems-Level Claim" means a claim of a Patent comprising multiple elements whereby at least one element is (or is embodied in) a Semiconductor Product and at least one of the other elements (which is not, and is not embodied in, such Semiconductor Product) is <u>not</u> a standard element technically essential to the functioning of the Semiconductor Product as part of any system or device. For avoidance of doubt, the definition of Systems-Level Claim is not intended to be construed or interpreted by reference to whether the authorized and unconditional sale of any particular Semiconductor Product would exhaust the claim in question. Attachment 1 to this Exhibit A provides one set of examples of Systems-Level Claims, but is not intended to modify or limit the meaning or scope of this definition.

"Third-Party IP Owner" mean a third party (i.e., not a Company Group Entity or Seller Group Entity) that owns a Patent, Copyright, or Know-How jointly with one or more Seller Group Entities.

"Transferred Circuit Layout Design Rights" means the circuit layout design rights set forth in Exhibit E to the Intellectual Property Assignment Agreement.

"Transferred Copyrights and Know-How" means (a) Freely Assigned Copyrights and Know-How, (b) any Restricted Assigned Copyrights and Know-How for which Seller obtains consent for assignment and (c) Transferred Circuit Layout Design Rights.

"Transferred Patents" means (a) Freely Assigned Patents, (b) any Restricted Assigned Patents for which Seller obtains consent for assignment to HoldCo, (c) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Freely Assigned Patents that are exclusively owned by Seller Group Entities immediately prior to the Closing Date, (d) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Restricted Assigned Patents that are exclusively owned by Seller Group Entities when Seller obtains consent for assignment of the Restricted Assigned Patents to HoldCo (provided that consent covers the assignment of such continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals as well), and (e) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of Restricted Assigned Patents for which consent to the assignment thereof to HoldCo is not required.

"*Transferred Trademarks*" means, collectively, the Trademarks set forth in Exhibit G to the Intellectual Property Assignment Agreement and any common law trademarks owned immediately prior to the Closing Date by a Seller Group Entity and exclusively used in connection with the Business.

TRADEMARK LICENSE AGREEMENT

by and between

SANYO SEMICONDUCTOR CO., LTD.

a Japanese corporation

and

SANYO ELECTRIC CO., LTD.

a Japanese corporation

Dated as of [__], 2010

TABLE OF CONTENTS

		Page
ARTICLE I	LICENSE	1
1.1	License Grant	1
1.2	Restrictions	2
1.3	Maintenance, Renewal and Enforcement	3
1.4	Reservation of Rights	3
1.5	Business Name and Corporate Identity	3
1.6	Customs	4
ARTICLE II	QUALITY CONTROL	4
2.1	Trademark Guidelines	4
2.2	Quality Standard	4
2.3	Quality Control Inspections	4
2.4	Subsidiary Compliance	4
2.5	Existing Inventory	5
ARTICLE III	TERM AND TERMINATION	5
3.1	Term	5
3.2	Termination	5
3.3	Effect of Termination	6
3.4	Termination of Pre-Existing Agreements	6
3.5	Parts	7
ARTICLE IV	NO WARRANTY, INDEMNITY AND LIMITED LIABILITY	7
4.1	Warranties	7
4.2	Indemnity	7
4.3	Limitation of Liability	7
ARTICLE V	MISCELLANEOUS	7
5.1	Purchase Agreement Terms	7
5.2	Assignment	7
5.3	Injunctive Relief	7
5.4	Notices	8
<u>Exhibits</u>		

A-1

Exhibit A:

Certain Defined Terms

TRADEMARK LICENSE AGREEMENT

This TRADEMARK LICENSE AGREEMENT (the "Agreement") dated as of [__], 2010 (the "Effective Date"), is entered into by and between Sanyo Electric Co., Ltd., a company organized under the laws of Japan ("Licensor") and Sanyo Semiconductor Co., Ltd. a company organized under the laws of Japan ("Licensee"). Each of Licensor and Licensee are sometimes hereinafter referred to as a "Party" and collectively as the "Parties." Certain capitalized terms used in this Agreement are defined in Exhibit A or in the Purchase Agreement (if not defined in Exhibit A).

RECITALS

- A. WHEREAS, Licensor, ON Semiconductor Corporation (the "Purchaser Parent") and SCI LLC ("Purchaser") have entered into that certain Purchase Agreement, dated as of July 15, 2010 ("Purchase Agreement"), pursuant to which Purchaser will purchase from Licensor, and Licensor will sell to Purchaser, all of the issued and paid up share capital of Licensee, and Purchaser, Licensor, and Licensee will enter into certain other transactions, all on the terms and subject to the conditions set forth in the Purchase Agreement;
- B. WHEREAS, in connection with the Transaction, the Parties or their affiliates have entered into the Intellectual Property Assignment Agreement and the Intellectual Property License Agreement;
 - C. WHEREAS, Licensor is the owner of all right, title and interest in and to the Licensed Marks;
- C. WHEREAS, Licensee wishes to license from Licensor the right to use the Licensed Marks in connection with the Business, and Licensor has agreed to license to Licensee the Licensed Marks for such purposes;

NOW, THEREFORE, in consideration of the mutual representations, warranties, covenants and conditions contained herein, and other good and valuable consideration, the sufficiency of which is hereby acknowledged, the Parties hereby agree as follows:

ARTICLE I

LICENSE

1.1 License Grant.

(a) Effective as of the Closing Date, Licensor hereby grants to Licensee and its Subsidiaries (but only as long as they remain Subsidiaries of Licensee) under the Trademark rights that Licensor has or may have in the Licensed Marks, a worldwide, non-transferable (except as set forth in Section 5.2), non-sublicenseable, non-exclusive, fully paid-up and royalty-free license, during the term of this Agreement, to: (i) use the Licensed Marks, including in the Licensed Forms, in connection with the marketing, promotion, advertisement, lease, sale, and other distribution of semiconductor products and services other than ISB (Integrated System in Board) products, photovoltaic devices, LED diodes (for clarity, LED drivers are not LED diodes), and laser diodes (for clarity, laser drivers are not laser diodes), and (ii) otherwise use Licensed Marks in a manner that is substantially similar to a manner used by the Licensee or its Subsidiaries prior to the Closing Date.

(b) Licensor authorizes Licensee and its Subsidiaries (but only as long as they remain Subsidiaries of Licensee) to use (including in marketing, sales, and
advertisements), as their respective company names and Corporate Identities, (i) the company names of Company Group Entities used on the Closing Date, (ii) "Sanyo
Semiconductor" with nothing preceding it and optionally with "Inc.", "Co.", "Ltd.", "Corp." or the like (but nothing else) following it (and, for the avoidance of doubt, not
"Sanyo" alone), and/or (iii) variations or abbreviations of "Sanyo Semiconductor" that have been approved in advance and in writing by Licensor (all of the foregoing,
"Company Names"), including in the Licensed Forms, during the term of this Agreement as long as the Company Group Entities are primarily engaged in the Business,
but in no event in connection with marketing, promotion, advertisement, lease, sale, or other distribution of ISB (Integrated System in Board) products, photovoltaic
devices, LED diodes (for clarity, LED drivers are not LED diodes), and laser diodes (for clarity, laser drivers are not laser diodes). Licensee shall (i) complete the
procedures (including the registrations) to change Company Names to new names which do not contain "SANYO" or " 🗆 " (or any words similar thereto, including
words that contain formatives or derivatives thereof) or any Licensed Mark, and (ii) cease to use the Corporate Identity containing "SANYO" or " 🗆 🗀" (or any words
similar thereto, including words that contain formatives or derivatives thereof) or any Licensed Mark, prior to the expiration of this Agreement (collectively, the "Name
Change Procedures"). In the event that either Party terminates this Agreement early pursuant to Section 3.2(a) or Section 3.2(c), Licensee shall complete the Name
Change Procedures as soon as reasonably practicable.

1.2 Restrictions.

- (a) Subject to Section 1.4, Licensee shall not use, reproduce or display (or authorize the use, reproduction or display of) the Licensed Marks in a manner that requires a license other than as permitted by this Agreement, and shall not take any action to change or alter any of the Licensed Marks, for example, by modifying its shape or color, adding decorative designs, or deforming it.
- (b) Licensee shall not challenge the validity or effectiveness of a Licensed Mark, nor shall Licensee challenge Licensor's ownership of the Licensed Marks or the enforceability of Licensor's rights therein.
- (c) Licensee shall not register any trademarks which are identical or substantially similar to the Licensed Marks or a combination of any Licensed Mark with any trademark of Licensee or any of its Affiliates.
 - (d) Licensee shall not use or register any trademarks which are confusingly similar to the Licensed Marks.
 - (e) Licensee shall not use the Licensed Marks in any manner that damages the value of the Licensed Marks or the goodwill or reputation of Licensor.

1.3 Maintenance, Renewal and Enforcement.

- (a) Licensor shall diligently maintain its rights in and to the Licensed Marks during the term of this Agreement, subject to Section 1.3(c) of this Agreement.
- (b) Licensee shall notify Licensor of any actual material infringements of any of Licensor's Trademark rights in and to the Licensed Marks of which Licensee becomes aware.
- (c) Licensor shall have the sole and exclusive right, but shall not be obliged, to bring actions for past, present and future infringements of its rights in and to the Licensed Marks. Licensor shall be entitled to retain damages and other monies awarded or otherwise paid in connection with any such action. If any of the Company Group Entities becomes involved in a formal dispute with a third party in relation to Licensee's or its Subsidiaries' use of the Licensed Marks, Licensee shall promptly notify Licensor in writing, and the relevant Company Group Entity shall resolve such dispute on its own responsibility and at its own expense, and Licensee shall indemnify and hold harmless Licensor and its Subsidiaries from any liability, loss, damage, cost or expense arising from any such dispute or any claim asserted against Licensor or any of its Subsidiaries arising from Licensee's or its Subsidiaries' use of the Licensed Marks (including, but not limited to, product liability, warranty, and similar claims), except where the dispute or claim occurred due to Licensor's breach of this Agreement or other Transaction Agreement (including any breach of the representations and warranties set out in the Purchase Agreement), or where the dispute or claim relates to unaltered Inventory existing as of the Closing Date.

 Notwithstanding the preceding sentence, Licensor may, with Licensee's consent (which will not be unreasonably withheld), take the initiative in handling and resolving such dispute or claim, and in such case, (i) Licensee shall ensure that the Company Group Entities cooperate with Licensor in accordance with Licensor's reasonable requests, and (ii) any costs incurred in respect of the defense of such dispute or claim shall be borne by Licensor.
- 1.4 Reservation of Rights. Licensee acknowledges that Licensor retains all right, title and interest in and to the Licensed Marks, and that Licensee has not acquired any right, title or interest in or to the Licensed Marks, except the right to use such Licensed Marks as set forth in this Agreement. Use of the Licensed Marks by Licensee, and all goodwill associated with such use, shall inure to the benefit of Licensor. To the extent that, notwithstanding the foregoing, any rights or interests in the Licensed Marks accrue to or vest in Licensee or its Subsidiaries as a matter of applicable law, Licensee shall assign (and ensure that its Subsidiaries assign) all such rights and interests to Licensor or, if such rights or interests cannot be assigned as a matter of applicable law, Licensor shall not (and shall ensure that its Subsidiaries do not) assert or exercise such rights or interests in any manner. Except for the rights expressly granted to Licensee and its Subsidiaries herein (including, for the avoidance of doubt, this Article I and Section 2.5), no other rights or interests in any Trademarks or other Intellectual Property Rights of Licensor are granted to Licensee or its Subsidiaries (by implication, estoppel, or otherwise) under this Agreement.
- 1.5 <u>Business Name and Corporate Identity</u>. Notwithstanding anything to the contrary in this Agreement, including in <u>Sections 1.2(a)</u>, <u>1.2(d)</u>, and <u>2.1</u>, or in Licensor's trademark guidelines described in <u>Section 2.1</u>, Licensee shall have the right to: (a) use the Company Names as their respective company names and Corporate Identities pursuant to <u>Section 1.1(b)</u>; and (b) use the Licensed Marks in the Licensed Forms pursuant to <u>Section 1.1(a)</u>.

1.6 <u>Customs</u>. Licensor shall, upon request of Licensee, instruct the relevant Government Authorities to remove any restrictions, controls or prohibitions on import, export, or commerce with respect to Licensed Products bearing the Licensed Marks in accordance with this Agreement that would otherwise be restricted, controlled, or prohibited as a result of the use of a Licensed Form not currently used.

ARTICLE II

QUALITY CONTROL

- 2.1 <u>Trademark Guidelines</u>. Subject to <u>Section 2.5</u>, use of the Licensed Marks by Licensee hereunder shall be in accordance with the provisions of <u>Article I</u>, this <u>Article II</u>, any reasonable instructions given by Licensor and Licensor's then-current reasonable trademark guidelines that are applied equally to all of Licensor's Subsidiaries, including the Sanyo CI Design Manual, as such guidelines may be provided to Licensee and amended from time to time by Licensor upon reasonable prior written notice to Licensee, provided, however, that such amendments are commercially reasonable and Licensee is provided with a reasonable period of time in which to comply with any such amendment.
- 2.2 <u>Quality Standard</u>. Subject to <u>Section 2.5</u>, with respect to products manufactured, marketed, distributed, or sold by Licensee or its Subsidiaries which bear a Licensed Mark on the product itself or its packaging ("**Marked Products**"), Licensee shall maintain a standard of quality which (i) would reasonably be expected to be carried by a product made or marketed under the Licensed Mark and (ii) is at least equal in material respects in quality to similar Company Products (if any) bearing such Licensed Mark prior to the Closing Date.
- 2.3 Quality Control Inspections. At Licensor's reasonable request from time to time (but no more than once per year), Licensee shall furnish to Licensor samples of Marked Products and samples of marketing, advertising, and promotional materials containing the Licensed Marks, in order for Licensor to confirm that Licensee is adhering to requirements set forth in this Agreement. In addition, from time to time (but no more than once per year) Licensor may, upon reasonable advance notice to Licensee, inspect Licensee's and its Subsidiaries' facilities and records to confirm that Licensee and its Subsidiaries are complying with this Agreement and are implementing adequate quality control processes in the manufacturing, testing, and packaging of Marked Products. Any such inspection will be done in a manner reasonably designed to minimize any disruption to the normal business operations of Licensee and its Subsidiaries. Licensee will cooperate with Licensor in performing any such inspection with the mutual goal of ensuring that all uses of the Licensed Marks are consistent with the reputation for high quality symbolized by the Licensed Marks.
- 2.4 <u>Subsidiary Compliance</u>. To the extent a Subsidiary of Licensee exercises the license rights set forth in <u>Section 1.1</u>, Licensee shall ensure that such Subsidiary complies with the terms and conditions of this Agreement.

2.5 Existing Inventory. Notwithstanding anything to the contrary, the requirements of Sections 2.1 and 2.2 shall not apply to unaltered Inventory of Company Products or to unaltered stocks of packaging, promotional material, or any other documents and materials bearing a Licensed Mark, in each case that exist as of the Closing Date. In addition, Licensee and its Subsidiaries (but only as long as they remain Subsidiaries of Licensee) may sell any unaltered Inventory existing as of the Closing Date and bearing a Trademark of Licensor or any of its Subsidiaries not otherwise licensed hereunder, provided that Licensee and its Subsidiaries comply with any reasonable instructions from Licensor regarding the use of such Trademarks owned by Licensor and its Subsidiaries on such Inventory (e.g., instructions to remove labels containing certain Trademarks that Licensor has decided not to continue using); provided, however, that such reasonable instructions are applied equally to all of Licensor's Subsidiaries, and provided further that Licensor shall bear any reasonable out-of-pocket costs or expenses incurred by Licensee and/or its Subsidiaries in complying with such reasonable instructions (e.g., the cost of removing labels).

ARTICLE III

TERM AND TERMINATION

3.1 <u>Term</u>. The term of this Agreement shall commence on the Closing Date and shall continue in full force and effect for three (3) years, unless earlier terminated as provided in this <u>Article III</u>.

3.2 Termination.

- (a) Either Party may terminate this Agreement upon written notice to the other Party if the other Party materially breaches any term or condition of this Agreement and fails to cure such breach within seventy-five (75) days following written notice specifying such breach.
- (b) Licensee may terminate this Agreement at any time by giving Licensor thirty (30) days' prior written notice that Licensee has ceased using the Licensed Marks and notifying Licensor of Licensee's intent to terminate this Agreement.
 - (c) Licensor may terminate this Agreement at any time by giving Licensee written notice if any of the following events occur:
 - (i) After the Closing, Purchaser Parent (directly or indirectly) no longer holds the majority (i.e., in excess of fifty percent (50%)) of the voting rights in Licensee.
 - (ii) Licensee files an application for bankruptcy, special liquidation proceedings, corporate reorganization proceedings or civil rehabilitation proceedings, or an application for any of the above procedures is filed against Licensee that has not been dismissed within sixty (60) days thereof;
 - (iii) Licensee commences a liquidation proceeding;
 - (iv) Licensee passes a resolution to dissolve or wind up;

- (v) Licensee becomes insolvent; or
- (vi) Licensee makes an assignment for the benefit of its creditors.

3.3 Effect of Termination.

- (a) Upon expiration or earlier termination of this Agreement, the rights and licenses granted hereunder shall immediately terminate, except that Sections 1.3(c), 1.4, 3.3, 3.4, and 3.5, Articles IV and V, and Exhibit A shall survive, and Licensee shall on its own responsibility and at its own expense:
 - (i) immediately cease to use the Licensed Marks;
 - (ii) re-label any products or materials (including, but not limited to, any Inventory of Company Products) which bear the Licensed Marks held by any of the Company Group Entities or remove the Licensed Marks from such products or materials; and
 - (iii) destroy or return to Licensor, in accordance with Licensor's reasonable instructions, any and all confidential information which was disclosed by Licensor under the terms of this Agreement (including, but not limited to, any manuals or guidelines relating to the Licensed Marks), and any reproduction or copies of such confidential information.
 - (iv) Notwithstanding the foregoing, Licensee and its Subsidiaries may, for six (6) months after the termination or expiration of this Agreement (other than termination by Licensor pursuant to Section 3.2(c)), continue to use the Licensed Marks under the terms of and in compliance with this Agreement in connection with the sale of Inventory of Company Products bearing a Licensed Mark and existing prior to the expiration or termination of this Agreement. Neither Party shall be liable to the other Party for damages of any kind solely as a result of terminating this Agreement in accordance with its terms, and any such termination of this Agreement by a Party will be without prejudice to any other right or remedy of such Party under this Agreement or applicable law.
- 3.4 <u>Termination of Pre-Existing Agreements</u>. The Brand Licenses are hereby terminated in their entirety, effective as of the Closing Date and notwithstanding anything to the contrary in the Brand Licenses. For avoidance of doubt, if any provision of any Brand License indicates that any provisions, terms, rights or obligations in or under such Brand License will remain in effect or otherwise survive the termination of such Brand License, then such provision shall be disregarded.

3.5 <u>Parts</u>. If any Legal Requirement requires Licensee and/or any of its Subsidiaries to retain replacement, maintenance and/or spare parts which bear any Licensed Mark after the termination or expiration of this Agreement, Licensee may provide written notice to Licensor within thirty (30) days of the expiration or termination of this Agreement (other than termination by Licensor pursuant to <u>Section 3.2(c)</u>) setting forth: (a) the model name and item number of such replacement, maintenance and repair parts; (b) the quantity of such replacement, maintenance and repair parts in respect of each item number; and (c) the period of and reasons for retention of such replacement, maintenance and repair parts (*i.e.*, the applicable Legal Requirement) in respect of each item number. To the extent Licensee provides such notice, Licensee and its Subsidiaries shall be permitted to continue to supply such replacement, maintenance and repair parts to their customers, but only for a period of time which is necessary to comply with the applicable Legal Requirement, as agreed by the Parties.

ARTICLE IV

NO WARRANTY, INDEMNITY AND LIMITED LIABILITY

- 4.1 Warranties. EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THE PURCHASE AGREEMENT, THE PARTIES MAKE NO REPRESENTATIONS OR WARRANTIES, WHETHER EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE OR NON-INFRINGEMENT.
- 4.2 <u>Indemnity</u>. Each Party (the "*Indemnifying Party*") shall defend, indemnify, and hold harmless the other Party, its Subsidiaries, and its and their Representatives (collectively, "*Indemnified Parties*") from any and all damages, actions, claims, fines, judgments, costs, losses, liabilities and expenses (including reasonable legal fees and litigation expenses) suffered by Indemnified Parties arising out of or relating to a breach by the Indemnifying Party or its Subsidiaries of any of its covenants or other provisions herein.
- 4.3 <u>Limitation of Liability</u>. Except with respect to third party damages, costs, liabilities, and settlements payable pursuant to each Party's indemnification obligations pursuant to <u>Sections 1.3(c)</u> and <u>4.2</u>, neither Party nor any of its Subsidiaries or Representatives shall be liable for any indirect, punitive, special, incidental or consequential damages, or damages for loss of profits, business interruption or otherwise, arising from or relating to this Agreement, even if such Party, its Subsidiaries, or its Representatives are expressly advised of the possibility of such damages. The foregoing limitation of liability and exclusion of certain damages shall apply regardless of the failure of essential purpose of any remedies available to either Party, but shall not restrict Licensor's ability to recover damages based on loss of goodwill.

ARTICLE V

MISCELLANEOUS

- 5.1 <u>Purchase Agreement Terms</u>. The following provisions of the Purchase Agreement are hereby incorporated, mutatis mutandis, by this reference: 15.2, 15.3, 15.6, 15.8, 15.9, 15.11, 15.12 and 15.13.
- 5.2 <u>Assignment</u>. This Agreement cannot be assigned by either Party without the prior written consent of the other Party hereto and any assignment in violation of this <u>Section 5.2</u> shall be null and void.
- 5.3 <u>Injunctive Relief</u>. Notwithstanding anything to the contrary in this Agreement or the Purchase Agreement, Licensor may seek temporary or preliminary injunctive relief in any court of competent jurisdiction in order to protect and preserve the Licensed Marks and enforce its rights therein.

•
_
<u>.</u>
•
· •
•
•
•
his <u>Section 5.4</u> shall be deemed to have been given on the date confirmation of receipt on such date.

(b) Either Party hereto may change its address specified for notices herein by designating a new address by notice given in accordance with this Section 5.4.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed as of the date and year first above written.		
SANYO ELECTRIC CO., LTD.		
В	sy:	
	Name: Title:	
S	ANYO SEMICONDUCTOR CO., LTD.	

By:

Name: Title:

[Signature Page to Trademark License Agreement]

EXHIBIT A

Certain Defined Terms

- "Assignment Documents" has the meaning set forth in Section 1.9 of the Intellectual Property Assignment Agreement.
- "Brand Licenses" means, collectively: (a) that certain Brand License Agreement, by and between Licensor and Licensee, dated April 1, 2008; (b) that certain Brand License Agreement, by and between Licensor and SSMC, dated April 1, 2008; (c) that certain Brand License Agreement, by and between Licensor and Sanyo LSI Design System Soft Co., Ltd., dated April 1, 2008; and (d) that certain Brand License Agreement, by and between Licensor and KSS dated April 1, 2008, dated April 1, 2008.
- "Code" has the meaning set forth in Section 1.7 of the Intellectual Property License Agreement and Section 2.6 of the Intellectual Property Assignment Agreement, as applicable.
- "Company Confidential Information" means all non-public Company Copyrights and Know-How.
- "Company Jointly Owned Patents" means Patents that are jointly owned by one or more Seller Group Entities and one or more Company Group Entities immediately following the Closing Date.
- "Company Names" has the meaning set forth in Section 1.1(b) of the Trademark License Agreement.
- "Company Copyrights and Know-How" means, collectively, (a) the Transferred Copyrights and Know-How and (b) Copyrights and Know-How that one or more Company Group Entities own as of the Closing Date.
- "Company Patents" means (a) the Transferred Patents, (b) all Patents that any of the Company Group Entities own as of the Closing Date (the "Company-Owned Patents"), and (c) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of the Company-Owned Patents.
- "Corporate Identity" means, collectively, letterhead, business cards, corporate signs, stationery, packaging slips, invoices, product datasheets, product labels, or any other materials that would normally identify a company or imply a corporate identity.
- "CTSL Patents" means all Patents, other than Licensed Patents, owned by Seller and the Seller Group Entities as of the Closing Date, and all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Patents, that would be infringed (absent a license) by making, using or selling Semiconductor Products (whether on a standalone basis or not), by using Manufacturing Apparatus to manufacture Semiconductor Products, or by providing Semiconductor Services. CTSL Patents include any Restricted Assigned Patents for which Seller has not obtained consent for assignment to HoldCo to the extent such Restricted Assigned Patents otherwise meet the definition set forth in the preceding sentence.

- "Current Company Products" means all Semiconductor Products that the Company Group Entities are selling, offering to sell, distributing, or making as of the Closing Date, including any minor improvements to the same that do not involve changes to existing features or functionality or the addition of new features or functionality.
- "Customer" means a third party (other than a Distributor or an End User) that purchases a Semiconductor Product (or a product containing one or more Semiconductor Products) manufactured by, or on behalf of, a Company Group Entity.
- "Distributor" means a third party that purchases a Semiconductor Product (or a product containing one or more Semiconductor Products) manufactured by, or on behalf of, a Company Group Entity and resells such Semiconductor Product (or such product containing one or more such Semiconductor Products, as the case may be) without modification or combination with any other component, part or product.
- "End User" means a third party that purchases a product containing one or more Semiconductor Products made by, or on behalf of, a Company Group Entity for such third party's own use and not for resale.
- "Effective Date" shall have the meaning set forth in the applicable preamble.
- "Filing" means the submission of any documentation, application, filing, registration or the like required to perfect or enforce, the Parties' interest in the Company Jointly Owned Patents under statutory Patent protection mechanisms, including, without limitation, any correspondence or other communication with any patent office or other Governmental Authorities with respect thereto.
- "Freely Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) solely owned by any Seller Group Entities, (ii) jointly owned by one or more Seller Group Entities and one or more Company Group Entities, (iii) jointly owned by one or more Seller Group Entities, Company Group Entities and Third-Party IP Owners, or (iv) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners; (b) primarily used in the Business by Company Group Entities as conducted immediately prior to the Closing Date; and (c) assignable without the consent of the Third-Party IP Owners (if any).
- "Freely Assigned Patents" means Patents that are listed on Exhibit F to the Intellectual Property Assignment Agreement.
- "Have Made" means the right to have a third party make a product or practice a method licensed under this Agreement for the use and benefit of a Party to this Agreement or its Subsidiary, provided that the designs, specifications, and/or working drawings for such products are supplied to such third party by, and originate with, the Party exercising such right or its Subsidiary (or with a Party's or its Subsidiaries' joint development with one or more third parties under terms of an agreement providing for the joint development of such designs, specifications, or working drawings).

- "HIC" means a hybrid integrated circuit, which is a chipset consisting of two or more integrated circuits that are packaged together onto a single device.
- "Know-How" means all right, title and interest in all Technology, Trade Secrets, ideas, concepts, proprietary techniques, processes, business and other methodologies, scientific, technical, research, development, engineering and business information, and other knowhow, including all rights arising under any law, including common law, state law, federal law or laws of foreign countries, other than Copyrights, Patents and Trademarks.
- "Licensed Forms" means the forms that incorporate either (i) the Licensed Marks, or (ii) the word "SANYO" or "\[\] " (as the case may be) set forth in Exhibit C to the Trademark License Agreement.
- "Licensed Marks" means the Trademarks set forth in Exhibit B to the Trademark License Agreement.
- "Licensed Patents" means (a) the Patents of Seller that are not Transferred Patents and are listed in Exhibit B to the Intellectual Property License Agreement, including any Restricted Assigned Patents for which Seller has not obtained consent for assignment, but has obtained consent to license to Company on the terms set forth in the Intellectual Property License Agreement ("Seller Licensed Patents"), and (b) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of the Seller Licensed Patents.
- "Manufacturing Apparatus" means any instrumentality or aggregate instrumentality primarily designed for use in the development and fabrication of Semiconductor Products.
- "Marked Products" has the meaning set forth in Section 2.2 of the Trademark License Agreement.
- "Name Change Procedures" has the meaning set forth in Section 1.1(b) of the Trademark License Agreement.
- "Panasonic" means Panasonic Corporation.
- "Parties" shall have the meaning set forth in the applicable preamble.
- "Party" shall have the meaning set forth in the applicable preamble.
- "Purchase Agreement" shall have the meaning set forth in the applicable recital.
- "Remaining Semiconductor Patents" means all Patents, other than the Licensed Patents, owned by Seller and the Seller Group Entities as of the Closing Date, and all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Patents, that would be infringed (absent a license) by making, using or selling Semiconductor Products (solely on a standalone basis), by using Manufacturing Apparatus to manufacture Semiconductor Products, or by providing Semiconductor Services. Remaining Semiconductor Patents include any Restricted Assigned Patents for which Seller has not obtained consent for assignment to HoldCo to the extent such Restricted Assigned Patents otherwise meet the definition set forth in the preceding sentence.

- "Restricted Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; (b) primarily used in the Business by Company Group Entities as conducted immediately prior to the Closing Date; and (c) assignable only with the consent of the Third-Party IP Owners, subject to obtaining the consent of such Third-Party IP Owners.
- "Restricted Assigned Patents" means Patents listed on Exhibit H to the Intellectual Property Assignment Agreement which are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; and (b) assignable only with the consent of the Third-Party IP Owners.
- "Retained Copyrights and Know-How" means the Copyrights and Know-How of Seller Group Entities that are not Transferred Copyrights and Know-How and that are used in the Business as currently conducted or planned to be conducted, including any Restricted Assigned Copyrights and Know-How for which Seller has not obtained consent for assignment to HoldCo.
- "Seller Confidential Information" means all non-public Retained Copyrights and Know-How.
- "Seller Licensed Products" means all products other than Semiconductor Products. For the avoidance of doubt, ISB ("Integrated System in Board") products, photovoltaic devices, LED diodes (but not drivers), and laser diodes (but not drivers) are each Seller Licensed Products. For the further avoidance of doubt, (i) ASICs (Application Specific Integrated Circuits) for digital camera products, and (ii) semiconductor devices, integrated circuits and HICs for the development of power management modules and tuner modules are not Seller Licensed Products.
- "Seller Licensed Services" means, collectively, (i) a service that is not a Semiconductor Service, (ii) designing of ASICs (Application Specific Integrated Circuits) for digital camera products, and (iii) designing of semiconductor devices, integrated circuits and HICs for the development of power management modules and tuner modules.
- "Semiconductor Product" means any semiconductor device, integrated circuit or HIC (hybrid integrated circuit) on a standalone basis (i.e., a semiconductor device, integrated circuit or HIC alone, and not as part of a combination with other devices that are not semiconductor devices), whereby semiconductor devices include, but are not limited to, semiconductor device dies, semiconductor wafers or packaged semiconductors (including but not limited to supporting or packaging means or structure and software, firmware and microcode intended to be embedded or implemented therewith). Semiconductor devices do not lose their character as such whether or not part of an assemblage of packaged semiconductors or other devices. Semiconductor Products do not include ISB ("Integrated System in Board") products, photovoltaic devices, LED diodes, or laser diodes. For the avoidance of doubt, LED drivers and laser drivers are Semiconductor Products.

"Semiconductor Services" means the design, manufacture, testing, assembly, fabrication, and development of Semiconductor Products, including the provision of foundry services related thereto.

"Systems-Level Claim" means a claim of a Patent comprising multiple elements whereby at least one element is (or is embodied in) a Semiconductor Product and at least one of the other elements (which is not, and is not embodied in, such Semiconductor Product) is <u>not</u> a standard element technically essential to the functioning of the Semiconductor Product as part of any system or device. For avoidance of doubt, the definition of Systems-Level Claim is not intended to be construed or interpreted by reference to whether the authorized and unconditional sale of any particular Semiconductor Product would exhaust the claim in question. Attachment 1 to this Exhibit A provides one set of examples of Systems-Level Claims, but is not intended to modify or limit the meaning or scope of this definition.

"Third-Party IP Owner" mean a third party (i.e., not a Company Group Entity or Seller Group Entity) that owns a Patent, Copyright, or Know-How jointly with one or more Seller Group Entities.

"Transferred Circuit Layout Design Rights" means the circuit layout design rights set forth in Exhibit E to the Intellectual Property Assignment Agreement.

"Transferred Copyrights and Know-How" means (a) Freely Assigned Copyrights and Know-How, (b) any Restricted Assigned Copyrights and Know-How for which Seller obtains consent for assignment and (c) Transferred Circuit Layout Design Rights.

"Transferred Patents" means (a) Freely Assigned Patents, (b) any Restricted Assigned Patents for which Seller obtains consent for assignment to HoldCo, (c) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Freely Assigned Patents that are exclusively owned by Seller Group Entities immediately prior to the Closing Date, (d) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Restricted Assigned Patents that are exclusively owned by Seller Group Entities when Seller obtains consent for assignment of the Restricted Assigned Patents to HoldCo (provided that consent covers the assignment of such continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals as well), and (e) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of Restricted Assigned Patents for which consent to the assignment thereof to HoldCo is not required.

"Transferred Trademarks" means, collectively, the Trademarks set forth in <u>Exhibit G</u> to the Intellectual Property Assignment Agreement and any common law trademarks owned immediately prior to the Closing Date by a Seller Group Entity and exclusively used in connection with the Business.

TRANSITION SERVICES AGREEMENT

by and between

SANYO ELECTRIC CO., LTD.,

a Japanese corporation,

and

SANYO SEMICONDUCTOR CO., LTD.,

a Japanese corporation

Dated as of ______, 2010

TABLE OF CONTENTS

ARTICLE I	AGREEMENT TO PROVIDE AND ACCEPT THE SERVICES	2
1.1	Overview	2
1.2	Additional Services to be Provided	
1.3	Service Description Schedules	
1.4	Service Level Objectives	4
1.5	Monthly Progress Reports	4
1.6	Base Service Levels	4
1.7	Access	4
1.8	Retention of Records; Verification	į
1.9	Service Providers	į
1.10	Designated Representatives; Services Managers	į
1.11	Cooperation	(
1.12	Reduction or Termination of a Service	(
1.13	Intellectual Property	(
ARTICLE II	PAYMENT	
2.1	Payment	
2.2	Prices	8
2.3	Taxes	8
2.4	Further Information	8
ARTICLE III	CONFIDENTIALITY	9
3.1	Confidentiality	9
3.2	Exclusions	9
ARTICLE IV	WARRANTIES; COVENANTS	9
4.1	By the Company	9
4.2	By the Seller	10
ARTICLE V	LIABILITIES	1
5.1	Obligation to Re-Perform	1
5.2	Indemnity	1
ARTICLE VI	TERM AND TERMINATION	13

13

6.1

Term; Extension

TABLE OF CONTENTS

(continued)

6.2	Termination	13
6.3	Return of Data; Confidential Information	13
6.4	Effect of Termination	13
ARTICLE VII	MISCELLANEOUS PROVISIONS	13
7.1	Relationship of the Parties	13
7.2	Interpretation	14
7.3	Further Assurances	14
7.4	Notices	14
7.5	Entire Agreement	15
7.6	Modifications, Amendments and Waivers	15
7.7	Successors and Assigns	16
7.8	Governing Law	16
7.9	Severability	16
7.10	Specific Performance	17
7.11	Counterparts	17
7.12	Dispute Resolution	17
7.13	Attorneys' Fees, Costs and Expenses	17
7.14	No Third Party Beneficiaries	17
ARTICLE VIII	DEFINED TERMS	18

18

TRANSITION SERVICES AGREEMENT

THIS TRANSITION SERVICES AGREEMENT (this "Agreement") is made and entered into this [—] day of [—], 2010 (the "Effective Date"), by and between SANYO Electric Co., Ltd., a company organized under the laws of Japan (the "Seller"), and SANYO Semiconductor Co., Ltd., a company organized under the laws of Japan (the "Company"). Each of the Seller and the Company are hereinafter referred to as a "Party" and collectively as the "Parties". Capitalized terms used in this Agreement are defined in Article VIII.

RECITALS

WHEREAS, **ON** Semiconductor Corporation, a Delaware corporation (the "*Purchaser Parent*"), Semiconductor Components Industries, LLC, a Delaware limited liability company (the "*Purchaser*"), the Company and the Seller are parties to a Purchase Agreement, dated July 15th, 2010 (the "*Purchase Agreement*"), pursuant to which the Purchaser will acquire from the Seller the Purchased Business (as defined in the Purchase Agreement) as of the Closing Date;

WHEREAS, to enable the Company Group Entities to conduct the Business in all material respects as it was conducted as of the signing date of the Purchase Agreement (the "Signing Date") and immediately prior to the closing under the Purchase Agreement (the "Closing"), in compliance with Legal Requirements, Governmental Approvals and Contracts applicable thereto (as in effect as of the Signing Date and immediately prior to the Closing), the Seller has agreed to provide, and to cause other Service Providers to provide, to the Company Group Entities certain transitional services with respect to the Business for the periods and on the terms and subject to the conditions set forth in this Agreement to support the Company Group Entities while they are making their own arrangements to perform such services independently from a systems and business process perspective;

WHEREAS, the Seller has agreed to provide, and to cause other Service Providers to provide, such services to the Company Group Entities on the understanding that Company Group Entities will complete this transition as soon as reasonably practicable after the Closing; and

WHEREAS, it is a condition to the Closing that the Parties enter into this Agreement.

NOW, THEREFORE, in consideration of the foregoing recitals and mutual representations, warranties, covenants and promises contained herein, the adequacy and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

AGREEMENT

ARTICLE I

AGREEMENT TO PROVIDE AND ACCEPT THE SERVICES

1.1 Overview.

- (a) On the terms and subject to the conditions contained in this Agreement, and to enable the Company Group Entities to conduct the Business in all material respects as it was conducted as of the Signing Date and immediately prior to the Closing, in compliance with Legal Requirements, Governmental Approvals and Contracts applicable thereto (as in effect as of the Signing Date and immediately prior to the Closing) and to support the Company Group Entities while they are making their own arrangements to perform the Services (as defined below) independently from a systems and business process perspective, the Seller, directly or indirectly through Service Providers, shall provide the services (collectively, the "Services" and, individually, a "Service") specified on the applicable service schedule or schedules attached to this Agreement (each a "Service Description Schedule")¹ to the Company Group Entities for the period of time specified thereon (such period in respect of each such Service, a "Transition Service Period"). Each of the Services shall be provided in accordance with the terms, limitations and conditions of this Agreement and the applicable Service Description Schedule. In the event of a conflict between the terms of this Agreement and the terms of any Service Description Schedule, the terms of the Service Description Schedule shall control as to the Services covered thereby, other than with respect to Article V, which Article shall govern in the event of a conflict between the terms of this Agreement and the terms of any Service Description Schedule.
- (b) The Services shall exclude (i) the provision of any funding or financial accommodation (except for certain arrangements which are expressly provided for in the Purchase Agreement or any Service Description Schedule) and (ii) any service that is directly provided by any third party (other than a Seller Group Entity) to any Company Group Entity under a Contract between such third party and such Company Group Entity (for example, any services which are directly provided by NTT Data Sanyo System Corporation ("NDSS") to the Company pursuant to the agreement dated the Closing Date between NDSS and the Company).
- (c) In addition, on the terms and subject to the conditions contained in this Agreement and the relevant Service Description Schedules, the Services shall include assisting and cooperating with the Company Group Entities, to the extent the Seller or any other Seller Group Entity is reasonably capable of it, in building and installing an information technology system to be used by the Company Group Entities after the expiration or termination of the relevant Transition Service Periods to independently operate financial and other systems and processes from a systems and business process perspective; *provided* that any costs and expenses related to the building and installing of an information technology system of a reasonable standard (but excluding any additional costs and expenses incurred as a result of accommodating any requests made by the Purchaser or the Purchaser Parent which are based on reasons or circumstances solely attributable to them (such as the preparation for the SAS70 representations)) that are in excess of \footnote{8}10,000,000 in the aggregate shall be borne equally by the Seller and the Company (that is, fifty percent (50%) of any such excess costs and expenses shall be borne by each of the Seller and the Company).

Note to Form of this Agreement: The final forms of Service Description Schedules shall be negotiated in good faith by the Parties during the period between the Signing Date and the Closing Date. The forms of Service Description Schedules attached as Exhibit J-2 to the Purchase Agreement do not reflect an exhaustive list of the Services to be provided under this Agreement and instead reflect those Services that have been identified by the Parties as of the Signing Date. As indicated on such forms, the final forms of such Service Description Schedules are subject to further good faith negotiation by the Parties and additional Service Description Schedules may be negotiated by the Parties in good faith prior to Closing (as well as after the Closing pursuant to Section 1.2.

- (d) Except as otherwise expressly set forth in a Service Description Schedule for a particular type of Service, the quantity of each type of Service to be provided by the Service Providers shall be that which the Service Recipient reasonably requires, from time to time, for the operation of the Business in the ordinary course consistent with the operation of the Business prior to the Signing Date and the Closing.
- (e) The Company shall use, and shall cause each of the relevant Company Group Entities to use, its or their respective best efforts to make its or their own arrangements to perform the Services independently from a systems and business process perspective (i) as soon as reasonably practicable after the Closing and (ii) in accordance with the prioritization schedule to be mutually agreed in good faith by the Seller and the Company from time to time.
- (f) Unless otherwise set forth in a Service Description Schedule, the Transition Service Period for each Service shall commence on the Closing Date and terminate as of the date on which the Seller and the Purchaser Parent mutually agree in good faith that such Service can be performed by the Company or any other Company Group Entity independently from a systems and business process perspective.
- 1.2 Additional Services to be Provided. If, within the one (1) year period after the Effective Date, a Company Group Entity reasonably determines that it needs a Service Provider to provide any Additional Services, then the Company will notify the Seller in writing of its need to obtain such Additional Services, providing a description in reasonable detail of such Additional Services. If such Additional Services fall within the scope of clause (a) of the definition of such term set forth in Article VIIII, then the Seller shall provide, or shall cause the applicable Seller Group Entity to provide, such Additional Services. If such Additional Services fall within the scope of clause (b) of the definition of such term set forth in Article VIII, then the Seller shall provide, or shall cause the applicable Seller Group Entity to provide, such Additional Services. Any Additional Services provided under this Section 1.2 shall be treated as Services provided in accordance with the terms and conditions of this Agreement for all purposes hereof and shall be specified in applicable Service Description Schedules. Any Additional Services within the scope of clause (a) of the definition of such term shall be invoiced based on the historical pass through allocation method utilized by the Parties during the Comparison Period for the same or similar services. Any Additional Services falling within the scope of Clause (b) of the definition of such term shall be invoiced based on actual personnel costs of the individuals providing such Additional Services on a pass through basis.

1.3 Service Description Schedules.

(a) Each Service Description Schedule shall describe: (i) a summary of the Services to be provided, (ii) the scope of such Services, (iii) the Transition Service Period, (iv) the Price for such Services, (v) any specific Service Level Objectives for such Services, (vi) the persons to whom disputes may be escalated, (vii) the notice period required to terminate each Service under <u>Section 1.12</u> and (viii) such other items as the Parties may agree in good faith.

- (b) The Parties acknowledge and agree that (i) it may not be practicable to describe each and every aspect of a particular Service in detail and (ii) therefore, a Service, when generally agreed upon by the Parties and set forth on a Service Description Schedule, shall be provided in accordance with the applicable terms and conditions of this Agreement, consistent with the Parties' past practices, even where all aspects of the Service are not described in detail. If any services, functions, responsibilities or other components of work for any Service are not described in a Service Description Schedule, but are reasonably required for proper performance or provision of the Service, they shall be deemed to be Services for all purposes hereof. Any such Services, functions, responsibilities or other components shall be billed based on the historical pass through allocation method utilized by the Parties during the Comparison Period for the same or similar services, functions, responsibilities or other components.
- **1.4 Service Level Objectives.** Each Service Provider shall meet or exceed the Service Level Objectives in performing relevant Services. A Service Recipient may notify the Seller if the actual performance of a Service by a Service Provider does not satisfy the applicable Service Level Objective for that Service ("*Variances*"). The Seller shall, and shall cause such Service Provider to, as promptly as practicable, review and repair in accordance with <u>Section 5.1</u> any Variance identified by a Service Recipient, and shall contact the Company's Designated Representative with respect thereto as promptly as reasonably necessary in light of the nature and criticality of the Variance identified to insure that the Business can continue to operate, in all material respects, as it did prior to the Signing Date and the Closing Date.
- **1.5 Monthly Progress Reports**. At the beginning of each calendar month commencing in the fourth calendar month after the Closing Date, the Company shall provide to the Seller a monthly report setting forth the progress made by the Company Group Entities in making their own arrangements to perform the Services independently from a systems and business process perspective in accordance with <u>Section 1.1(e)</u>. Such reports shall set forth details which are reasonably requested by the Seller; *provided* that such requests shall not impose any unreasonable burden upon any Company Group Entity.
- **1.6 Base Service Levels.** In addition to any Service Level Objective, each Service Provider shall perform the Services in a timely, professional and workmanlike manner consistent with the manner in which the Seller (or the relevant Service Provider) provided the same or similar services to the Company Group Entities as of the Signing Date and immediately prior to the Closing and in accordance with (a) this Agreement, including the applicable Service Description Schedule, (b) applicable Legal Requirements and Governmental Approvals and (c) no less than the same degree of care, skill, quality and timeliness as it exercised in performing the same or similar services for the Business prior to the Closing Date (and the Comparison Period).
- 1.7 Access. Subject to the confidentiality provisions in this Agreement and other information security and similar policies, each Party shall make available to the other Party such information and materials concerning the Company Group Entities and the Business, in compliance with Legal Requirements, Governmental Approvals and Contracts applicable thereto (as in effect as of the Signing Date and immediately prior to the Closing), reasonably requested to enable such other Party to provide or receive the applicable Services, as the case may be. Each Party shall provide to the other Party reasonable access to its premises (and the systems, software and networks located therein), to the extent necessary or advisable for such other Party to provide or receive the applicable Services.

1.8 Retention of Records; Verification. Each Service Provider shall keep accurate and detailed books and records of (a) all Services that it provided under this Agreement, (b) all Prices charged for such Services and (c) all Invoices sent and payments made under this Agreement. The Company may, at its own expense, upon reasonable advance notice and during regular business hours, access and review the books and records of the Service Providers maintained in connection with this Agreement to verify their compliance with this Agreement and the accuracy of the Prices charged under this Agreement.

1.9 Service Providers.

- (a) The Seller shall: (i) supervise the performance of each Service Provider to ensure that such Service Provider complies with any applicable obligations under this Agreement; (ii) impose on each Service Provider the same confidentiality obligations that apply to the Seller under this Agreement; (iii) remain responsible for the complete performance of each Service Provider in accordance with the terms and conditions of this Agreement (and any Service Provider's noncompliance with any terms and conditions hereof); and (iv) ensure that employees and contractors of each Service Provider providing Services are (x) suitably knowledgeable, experienced and skilled therefor, and (y) comply with all security, health, safety and similar policies and regulations of the Service Recipient in effect as of the Signing Date and immediately prior to the Closing.
- (b) In the event that the Company reasonably believes that any employee or third party contractor of a Service Provider is neither meeting the foregoing requirements nor providing satisfactory Services and on such basis requests that such employee or third party contractor be replaced, the Seller and the Company shall discuss such matter in good faith and shall reasonably agree upon a means of improving such performance.
- (c) Each Party understands that the other Party may have contracted, and may in the future contract with third parties to provide Services in connection with all or a portion of the Services, including any Service reduced or terminated pursuant to <u>Section 1.12</u>. Each Party reserves the right to continue to contract with third parties to provide such Services or to enter into new contract relationships for such Services.
- **1.10 Designated Representatives; Services Managers**. Each Party shall designate a single point of contact (a "Designated Representative") responsible for the implementation of this Agreement. Each Party shall also designate individuals to oversee each Service who shall have sufficient knowledge, experience and skill with the relevant Services (a "Service Manager"). Each Party's initial Designated Representative shall be as specified in Exhibit A, and each Party's initial Services Manager shall be as specified in the applicable Service Description Schedule. Either Party may replace any of its Designated Representatives or Service Managers with an individual of comparable knowledge, experience and skill by notifying the other Party in writing of such new appointment.

1.11 Cooperation. In the event that (a) there is nonperformance of any Service or a breach of Section 1.1(e) as a result of a Force Majeure Event, (b) the provision of a Service or compliance with Section 1.1(e) would violate applicable Legal Requirements or Governmental Approvals or (c) the provision of a Service or compliance with Section 1.1(e) requires the consent, approval or agreement of a third party ("Consent") which has not been obtained, then the Parties shall work together in good faith (at their own expense) to arrange for an alternative means for the Service Recipient to obtain the Services so affected or for the Company or the relevant Company Group Entity to comply with Section 1.1(e). Failure to perform or provide any of the Services due to the reason set forth in clause (a) or (b) above shall not constitute a breach of this Agreement or an event subject to indemnification by the Seller under Section 5.2.

1.12 Reduction or Termination of a Service. In addition to any early termination rights set forth in a Service Description Schedule:

- (a) In the event that the Company wishes to terminate all or part of, or reduce levels of, a Service or any separately priced portion of a Service, prior to termination of the Transition Service Period, the Company may do so at any time, without obligation or penalty; *provided* that it either (i) gives the Seller at least thirty (30) days prior written notice of such election to terminate (or such other notice period as may be set forth in the relevant Service Description Schedule), specifying the Service affected thereby in reasonable detail, or (ii) promptly reimburses the applicable Service Provider for any reasonable, documented out-of-pocket cancellation costs incurred by such Service Provider resulting from such reduction or early termination by the Company.
- (b) Notwithstanding anything to the contrary in <u>Section 1.11</u>, the Company may immediately terminate all or part of, without obligation or penalty, a Service, any separately priced portion of a Service or this Agreement in its entirety, prior to termination of the Transition Service Period, upon written notice to the Seller, in the event a Service has, for a period longer than fourteen (14) days, been interrupted or has not been provided in accordance with the Service Level Objectives.

1.13 Intellectual Property.

(a) Each Seller Entity Group which is a Service Provider hereby grants to each applicable Service Recipient a non-exclusive, sublicensable, perpetual, irrevocable, royalty-free, full paid up, worldwide right and license to use, copy, edit, format, modify, translate and create Derivative Technology of the source and object code versions of the Licensed Software; *provided*, *however*, that (i) the purpose and application of the above license shall be limited to the building, installation, operation, ownership, use and disposal of an information technology system to be owned or used by the Company Group Entities and which are required by them after the expiration or termination of the relevant Transition Service Periods so that the Company Group Entities are able to independently operate financial and other systems and processes from a systems and business process perspective, and (ii) notwithstanding anything to the contrary contained in this Agreement, with respect to Licensed Software from a third party, (x) the license granted hereunder and (y) the use and ownership of the above information technology system by the Company Group Entities shall be limited to the scope of the rights under the original license granted by the original licensed to the relevant Service Provider in relation to the relevant Licensed Software, and the extent to which such rights can be lawfully and validly licensed or sublicensed to the Company Group Entities.

(b) Except as otherwise expressly set forth in a Service Description Schedule, each Service Provider acknowledges that it shall not, by reason of the provision of the Services under this Agreement, acquire any right, title or interest (including any license rights or rights of use) in any Intellectual Property Rights of any Service Recipient that are (i) owned by any Service Recipient prior to delivery of the Services to such Service Recipient or (ii) developed, conceived, created and/or acquired by any Service Recipient thereafter in connection with the Services or otherwise.

ARTICLE II

PAYMENT

2.1 Payment.

(a) <u>Invoices</u>. Promptly after the end of each calendar month, each Service Provider or, if notified by the Seller in advance (and in any event the Seller shall provide any such notice to the relevant Service Recipient no later than thirty (30) days before the relevant Invoice is due and payable), the Seller, shall deliver to the relevant Service Recipient an invoice ("*Invoice*") containing (a) a description of the Services provided by the relevant Service Provider to the Service Recipient during that month, in sufficient detail to identify the Services to which the Invoice pertains, and (b) the Price for such Services. Except as otherwise specified in the applicable Service Description Schedule, the applicable Service Recipient shall pay all amounts due under such Invoice in Japanese Yen within forty-five (45) calendar days of the date on which such Invoice is received by means of wire transfer of immediately available funds to the bank account in Japan designated by the Service Provider or, if so notified by the Seller in advance, the Seller, from time to time. If the Price for a particular Service is stated in the relevant Service Schedule as an annual fee, then the Service Provider or, if applicable, the Seller shall invoice such fee in twelve (12) equal monthly installments. Service Providers shall directly or indirectly through the Seller provide such information in their possession with respect to Invoices as Service Recipients may reasonably request from time to time for purposes of understanding or confirming invoiced amounts, and shall make their personnel available to answer such questions with respect thereto.

(b) Expense Reimbursement. In addition to paying the applicable Price for the Services, each Service Recipient shall promptly reimburse the applicable Service Provider the actual amount of reasonable and documented out-of-pocket costs or expenses incurred during the relevant calendar month in connection with providing Services (but not including any overhead costs or expense mark-ups) and described in reasonable detail in a relevant Invoice; provided that (i) in addition to any such Invoices, the Seller shall deliver to the Company, promptly after the end of each calendar month (and in any event by the 15th day of the next succeeding calendar month), a written report ("Monthly Expense Report") describing in reasonable detail all such costs and expenses incurred during such calendar month by all of the Service Providers, and (ii) unless otherwise stated in a Service Description Schedule, any costs or expenses exceeding ¥500,000 shall require the prior written approval of the Company. Any request for such prior written approval shall be sent by the Service Manager of the relevant Service Provider by email to the person specified in Exhibit B and, if such cost or expense is approved, such person shall inform such Service Manager by reply email (which shall constitute the prior written approval contemplated by the immediately preceding sentence). The Company may replace the person specified in Exhibit B by notifying the Seller in writing of such new appointment.

2.2 Prices. Unless otherwise stated in a Service Description Schedule, Prices shall be calculated on a "cost" basis by historical pass through allocation method, with cost being calculated as (a) in cases where the Service Provider is the Seller or a Seller Subsidiary, the Service Provider's direct actual cost plus, where applicable, an administrative charge reasonably similar in value to the administrative charges historically charged to the Service Recipient for similar services (provided that any such administrative charge is charged by the Service Provider (which is the Seller or a Subsidiary of the Seller) to all relevant Seller Group Companies in a similar manner) to provide such Service and (b) in cases where the Service Provider is a third party, the actual payments for such Service due and paid to such third party by the Seller or a Seller Subsidiary, consistent with past practice. With respect to (b) above, for the avoidance of doubt, if the prices charged by the third party Service Provider to any Seller Group Entity are increased for both such Services and the same or similar services to be provided by such third party Service Provider to any Seller Group Entity, the Prices paid by the Company in respect of those Services shall reflect such increased price.

2.3 Taxes. The amounts set forth as the applicable consideration with respect to each Service in the Service Description Schedules do not include any sales tax, value added tax, goods and services tax or similar tax (each, a "*Tax*") and any such amounts required to be paid by any Service Provider in connection with this Agreement or the performance of this Agreement shall be reimbursed by the applicable Service Recipient, and such reimbursement shall be in addition to the amounts required to be paid by the Service Recipient as set forth in the Service Description Schedules. To the extent that such amounts paid by a Service Recipient are taxed as net income (or other similar tax) of a Service Provider, no Service Recipient shall be responsible for any such resulting income (or other similar) taxes.

2.4 Further Information.

- (a) From time to time upon the reasonable written request by the Company or any Service Recipient, the Seller shall, and shall cause the relevant Service Provider to, (i) provide such information in its possession with respect to any Invoice, Monthly Expense Report or Price for the Services provided by such Service Provider for the purpose of supporting the Prices and out-of-pocket costs or expenses represented in any such Invoice, and (ii) make its personnel available to answer such questions as the Company or any Service Recipient may reasonably ask for such purpose.
- (b) In the event that a Service Recipient disputes the amount of any Invoice from a Service Provider, such Service Recipient shall nevertheless pay whatever portion is undisputed at the time when payment is due pursuant to this Agreement.

ARTICLE III

CONFIDENTIALITY

3.1 Confidentiality. Each Party agrees that, during the term for which any Service is provided hereunder, and for a period of five (5) years after the last day of the latest Transition Service Period, it will not, and will use reasonable efforts to ensure that its Affiliates will not, through any action or inaction, use (except as contemplated by this Agreement), or disclose to any other Person, any Confidential Information relating to the other Party (it being understood that following the Closing this Article III shall apply to the use or disclosure of any Confidential Information of any Company Group Entity or the Business by the Seller Group Entities, and shall not apply to the use or disclosure of the Confidential Information of any Company Group Entity or the Business by the Purchaser and its Affiliates); *provided, however*, that the foregoing prohibitions shall not apply to (a) subject to Section 3.2, disclosures that are required by any Legal Requirement (including any rule or regulation of the SEC (as defined in the Purchase Agreement) or of a stock exchange which may require such disclosure) or by a Governmental Authority; (b) information that is ascertainable or obtained from public or published information or is otherwise publicly known through no wrongful act of the using or disclosing Party; and (c) information disclosed to or filed with any Person for the purpose of obtaining consents to, or the financing of, the transactions contemplated by this Agreement. Notwithstanding anything herein to the contrary, each Party may (without prior notification to, or approval or consent by, the other Party) disclose to taxing authorities and/or to such Party's representatives (including outside counsel and advisors) any confidential or non-public information that is required to be disclosed in connection with such Party's tax filings, reports, claims, audits, or litigation.

3.2 Exclusions. Notwithstanding Section 3.1, in the event a Party is required to disclose Confidential Information of the other Party (in such event, such other Party is a "Nondisclosing Party," and the Party required to disclose is the "Disclosing Party") pursuant to any Legal Requirement, and would otherwise be prohibited from doing so under this Article III, the Disclosing Party shall: (a) promptly notify the Nondisclosing Party of the existence, terms and circumstances surrounding such requirement; (b) consult with the Nondisclosing Party on the advisability of taking legally available steps to resist or narrow such request; and (c) if disclosure of such Confidential Information is required, furnish only that portion of the Confidential Information which the Disclosing Party is legally compelled to disclose and advise the Nondisclosing Party reasonably in advance of such disclosure so that the Nondisclosing Party may seek an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Confidential Information. The Disclosing Party shall not oppose actions by the Nondisclosing Party to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Confidential Information.

ARTICLE IV

WARRANTIES; COVENANTS

4.1 By the Company. The Company represents and warrants that the Company has all requisite corporate power and authority to execute, deliver and perform its obligations under this Agreement.

4.2 By the Seller.

- (a) The Seller represents and warrants as of the Effective Date that:
- (i) the Seller has obtained the Consent of any third party reasonably necessary in order for each Service Recipient to use the respective Services and Licensed Software (or any equipment or software owned by or leased or licensed to the respective Service Provider);
- (ii) neither the Seller nor any Service Provider has previously granted, any rights in the Licensed Software to any third party that are inconsistent with the rights granted to any Service Recipient hereunder; and
- (iii) the Seller and each Service Provider have properly compensated all personnel of the Seller Group Entities in connection with the provision of any Service, Licensed Software or Work Product to any Service Recipient.

(b) The Seller covenants that:

- (i) the Services, Licensed Software and Work Product shall comply with all Legal Requirements applicable to such Services;
- (ii) the Services, Licensed Software and Work Product shall conform to any specifications or other requirements set forth in the Service Description Schedules;
- (iii) the Services, Licensed Software and Work Product shall not infringe, violate or misappropriate the proprietary rights, including any Intellectual Property Rights, of any third party;
- (iv) neither the Seller nor any Service Provider shall grant, any rights in the Licensed Software to any third party that are inconsistent with the rights granted to any Service Recipient hereunder;
- (v) the Seller and each Service Provider shall properly compensate, all personnel of the Seller Group Entities in connection with the provision of any Service, Licensed Software or Work Product to any Service Recipient;
 - (vi) [Further representations, warranties and covenants to be considered].

ARTICLE V

LIABILITIES

5.1 Obligation to Re-Perform. In the event of any error, inaccuracy or defect in the result or provision of any Service, the Seller shall, or shall cause the applicable Service Provider to, as promptly as practicable and in any event within fifteen (15) Business Days (*provided*, *however*, that, in the case where such fifteen (15) Business Day period is impracticable for correction of such error, inaccuracy or defect, and when the Seller so reasonably requests, such period may be extended subject to the consent of the Company, which consent shall not be unreasonably withheld) (or such other period mutually agreed by the Parties), after receiving a request therefor, correct in all material respects such error, inaccuracy or defect or re-perform in all material respects such Service at the request of the Company and at the sole expense of the Seller, the applicable Seller Subsidiary or such other Service Provider.

5.2 Indemnity.

- (a) The Seller shall indemnify, defend and hold harmless the Company, its Affiliates and its and their directors, officers, employees, managers, agents, customers and representatives (collectively, the "Company Indemnified Parties") from and against any Losses incurred by a Company Indemnified Party arising out of (i) any breach of this Agreement by the Seller, including the failure of the Seller to perform any of its covenants or agreements contained in this Agreement (but excluding the breach by the Seller of any of its representations and warranties set forth in Section 4.2(a) of this Agreement, which is covered by Section 12.2(a)(i) of the Purchase Agreement), (ii) any negligence or willful misconduct of the Seller or any violation of any applicable Legal Requirement by the Seller in connection with this Agreement, (iii) any claim or threat by a third party alleging that any Service, Licensed Software, Work Product or Confidential Information (and the exercise of the rights granted in this Agreement with respect thereto) supplied by any Service Provider infringes, misappropriates or violates the Intellectual Property Rights of any third party or (iv) any claim or threat by a third party alleging that the Seller failed to obtain such third party's Consent in order for each Service Recipient to use the respective Services and Licensed Software (or any equipment or software owned by or leased or licensed to the respective Service Provider). For clarity, all breaches, failures, acts, omissions, negligence, willful misconduct or violations of Legal Requirements of a Service Provider related to the Services shall be deemed acts and omissions of the Seller.
- (b) The indemnification given by the Seller under this Agreement shall survive until the applicable date which is set forth below, and shall cease upon such applicable date:
 - (i) the indemnification under Sections 5.2(a)(i) and 5.2(a)(ii) shall survive, with respect to each breach of this Agreement, act of negligence, misconduct or violation of Legal Requirement, until the third (3rd) anniversary of the termination or expiration date of the Transition Services Period concerning the relevant Services of such breach of this Agreement, act of negligence, misconduct or violation of a Legal Requirement;
 - (ii) the indemnification under Section 5.2(a)(iii) shall survive until the third (3rd) anniversary of the date on which the relevant Services, Licensed Software, Work Product or Confidential Information have been supplied to the relevant Service Recipients; and
 - (iii) the indemnification under Section 5.2(a)(iv) shall survive until the second (2nd) anniversary of the Effective Date.

- (c) The sum of the Losses subject to indemnification by the Seller incurred by the Company Indemnified Parties pursuant to <u>Section 5.2(a)(iii)</u> and those incurred pursuant to <u>Section 5.2(a)(iv)</u> shall not exceed one (1) billion Yen (¥1,000,000,000).
- (d) The sum of the Losses subject to indemnification by the Seller incurred by the Company Indemnified Parties pursuant to Section 5.2(a)(i) and those incurred pursuant to Section 5.2(a)(ii) caused by any Service Provider which is a Seller Group Entity (including the Seller) (but excluding the willful breach of this Agreement, an act of willful or gross negligence, willful misconduct or willful violation of a Legal Requirement by any Service Provider which is a Seller Group Entity (including the Seller)) shall not exceed seven (7) billion Yen (¥7,000,000,000).
- (e) There shall be no cap on the amount of Losses incurred by the Company Indemnified Parties subject to indemnification by the Seller pursuant to Section 5.2(a)(ii) or Section 5.2(a)(iii) with respect to a willful breach of this Agreement, an act of willful or gross negligence, willful misconduct or willful violation of a Legal Requirement, by any Service Provider which is a Seller Group Entity (including the Seller).
- (f) The Losses subject to indemnification by the Seller incurred by the Company Indemnified Parties pursuant to Section 5.2(a)(ii) and those incurred pursuant to Section 5.2(a)(ii), caused by a third party Service Provider shall be the total amount reimbursable by the Seller from such third party Service Provider concerning such Losses under a Contract between the Seller and such third party Service Provider.
- (g) A Company Indemnified Party shall give prompt written notice to the Seller of any Loss for which indemnification may be required under this Agreement; *provided, however*, that the failure of the Company Indemnified Party to provide such notice shall not affect the Seller's obligations under this Section 5.2(g) if such failure does not materially prejudice the Seller. The Seller shall be entitled to assume the defense and control of any third party Claim at its own cost and expense; *provided, however*, that the Company Indemnified Party (as applicable) shall have the right to be represented by its own counsel at its own cost in such matters. Neither the Seller nor the Company Indemnified Party may concede, settle or compromise any third party Claim without the consent of the other Party, such consent not to be unreasonably withheld or delayed. Each Party shall reasonably cooperate with the other Party and its counsel in the course of the defense of any third party Claim, such cooperation to include using reasonable efforts to provide or make available documents, information and witnesses.
- (h) Promptly after the Company Indemnified Party becomes aware of any event or circumstance that could reasonably be expected to constitute or give rise to any breach of any representation, warranty or covenant of the Seller which is contained in this Agreement or any other claim for indemnification pursuant to this Article V, the Company shall, and shall cause each Company Indemnified Party to, take all reasonable steps to mitigate and minimize all Losses that may result from or relate to such breach or claim.

ARTICLE VI

TERM AND TERMINATION

- **6.1 Term; Extension.** Unless earlier terminated in accordance with Section 6.2, this Agreement shall be in effect from the Effective Date until the expiration of the latest Transition Service Period. If the Service Recipient reasonably believes that the need for a Service will continue beyond the Transition Service Period in respect of such Service, then, at the Service Recipient's written request not less than sixty (60) days prior to the end of such Transition Service Period, the Parties shall seek in good faith to agree on terms for the extension of such Transition Service Period.
 - **6.2 Termination**. Notwithstanding anything to the contrary in <u>Section 6.1</u>:

and

- (a) this Agreement shall terminate on the date on which all Services or this Agreement in its entirety has been terminated in accordance with Section 1.12;
- (b) in the event either Party (or any Service Provider or any Service Recipient, as the case may be) materially breaches any other term of this Agreement, and fails to cure such breach within thirty (30) days of written notice thereof from the other Party, the non-breaching Party may immediately terminate the applicable Service or this Agreement in its entirety.
- **6.3 Return of Data; Confidential Information.** Upon any termination or expiration of this Agreement, each Party shall, and shall cause, in case of the Seller, each Service Provider and, in case of the Company, each Service Recipient, to, deliver to the other Party (a) all records and data (including backup tapes, records and related information) received, computed, developed, processed and stored by it hereunder, and (b) all Confidential Information of such other Party; *provided* that, in lieu of delivering all of the foregoing to the other Party, the relevant delivering Party may certify in writing that it has destroyed, or has caused each Service Provider or Service Recipient, as the case may be, to destroy, all of the foregoing.
- **6.4 Effect of Termination**. Upon termination of this Agreement, the Service Providers shall no longer be obligated to provide any Services and the Service Recipients shall no longer be obligated to pay for any Services. Section 6.3, this Section 6.4 and Article II (with respect to Services performed prior to termination or expiration of this Agreement), Article III (to the extent provided therein), Article V, Article VIII and Article VIII shall survive any termination or expiration of this Agreement.

ARTICLE VII

MISCELLANEOUS PROVISIONS

7.1 Relationship of the Parties. Nothing in this Agreement shall be construed as creating a partnership, joint venture or other association of any kind between the Parties, or to constitute either Party as the agent of the other Party for any purpose. For purpose of this Section 7.1, "Party" and "Parties" shall include any Affiliate of the Seller of the Company.

7.2 Interpretation. Neither Party shall be considered the draftsperson of this Agreement. Unless the context shall otherwise require, words using the singular or plural number shall also include the plural or singular number, respectively. The word "or" shall include the meaning "either or both." Unless the context otherwise provides, all pronouns used in this Agreement shall be deemed to refer to the masculine, feminine or neuter gender as the context requires. All references in this Agreement to "Section," "Article," "Recital," "Exhibit" or "Schedule" (or similar references) shall be deemed to be references to a section, article or recital of, or exhibit or schedule to, this Agreement, unless the context otherwise requires. The table of contents and the captions and other headings contained in this Agreement as to the contents of particular articles, sections, paragraphs or other subdivisions contained herein have been included for convenience of reference only and shall not, in any way, be construed as part of this Agreement or as limitations on the scope of the particular articles, sections, paragraphs or other subdivisions to which they refer and shall not affect the interpretation or meaning of this Agreement.

7.3 Further Assurances. Each Party agrees, and to cause each Service Provider or each Service Recipient, as the case may be, (a) to furnish upon request to the other Party such further information; and (b) to execute and deliver to the other Party such other documents; all as the other Party may reasonably request for the purpose of carrying out the intent of this Agreement and the Transaction.

7.4 Notices

7.4 Hodecs.		
	r communications under this Agreement shall be either (i) delivered in person; (ii) sent by overnight co (iii) sent by facsimile with confirmation of receipt and, in each case, addressed as follows:	ırier
If to the Seller (or any Service Provider):		
	14	

with copies to (which copies shall not constitute notice):	
	· -
If to the Company (or any Service Recipient):	
	: - -
with coming to (a shiph coming shall make	
with copies to (which copies shall not constitute notice):	_
	· · -
	 -

- (b) All notices, requests, instructions or documents given to either Party (or any Service Provider or Service Recipient, as the case may be) in accordance with this Section 7.4 shall be deemed to have been given on the date of mailing or transmission, whether delivered by hand, by overnight courier service, or by facsimile, with confirmation of receipt on such date.
 - (c) Either Party may change its address specified for notices herein by designating a new address by notice given in accordance with this Section 7.4.
- **7.5 Entire Agreement**. This Agreement, the Service Description Schedules, the Exhibits, the Purchase Agreement and the Transaction Agreements constitute the entire agreement between the Parties relating to the subject matter hereof and thereof and supersede all prior oral and written understandings, all contemporaneous oral negotiations and discussions, and all other writings and agreements relating to the subject matter of this Agreement.
- **7.6 Modifications, Amendments and Waivers**. This Agreement cannot be amended or changed nor any performance, term or condition waived in whole or in part, except by a writing signed by the Party against whom enforcement of the amendment, change or waiver is sought that expressly states that it is a modification, amendment or waiver hereof and expressly refers to the provisions to be altered or waived. No delay or failure on the part of either Party in exercising any rights hereunder, and no partial or single exercise thereof, shall constitute a waiver of such rights or of any other rights under this Agreement.

7.7 Successors and Assigns.

- (a) This Agreement cannot be assigned by the Company without the prior written consent of the Seller (which consent shall not be unreasonably withheld) and any assignment in violation of this Section 7.7 shall be null and void; provided, however, that the Company may assign this Agreement to one or more Affiliates of the Purchaser, in connection with any merger, sale or transfer of all or a substantial portion of the Company's assets or similar corporate transaction with or to such Affiliates without the consent of the Seller. In the event of any such assignment, the Company shall provide prompt notice thereof to the Seller. Notwithstanding the foregoing, if any assignment satisfying the conditions of the foregoing proviso takes place, the defined terms "Company", "Company Group Entity" and "Company Group Entities" for purposes of any Service Description Schedule or the definition of "Additional Services" shall mean the Company and its Subsidiaries as they exist immediately prior to giving effect to such assignment so that the Services to be performed by the Service Providers are not materially increased or expanded as a result of such assignment.
- (b) This Agreement cannot be assigned by the Seller without the prior written consent of the Company (which consent shall not be unreasonably withheld) and any assignment in violation of this Section 7.7 shall be null and void; provided, further, that the Seller shall not be released from its obligations hereunder without the consent of the Company. Upon any consolidation or merger in which the Seller is not the surviving corporation, or any sale, assignment, transfer, lease, conveyance or other disposition of all or substantially all of the assets of the Seller, the successor corporation formed by such consolidation or into or with which the Seller is merged or the entity to which such sale, assignment, transfer, lease, conveyance or other disposition is made shall succeed to, and be substituted for and may exercise every right and power of, and be obligated to perform every obligation of, the Seller with the same effect as if such successor entity has been named as the Seller herein; provided, however, that the predecessor Seller in the case of a sale, assignment, transfer, lease, conveyance or other disposition shall not be released from any of the obligations hereunder.
- **7.8 Governing Law**. This Agreement is to be construed in accordance with and governed by the laws of Japan, without giving effect to any choice of law rule that would cause the application of the laws of any jurisdiction other than the laws of Japan to the rights and duties of the Parties.
- **7.9 Severability**. Should any one or more of the provisions of this Agreement be determined to be invalid, illegal or unenforceable, such invalid, illegal or unenforceable provisions shall be deemed severed from this Agreement, and the validity, legality and enforceability of the remaining provisions of this Agreement shall not in any way be affected or impaired thereby. The Parties shall replace the invalid, illegal or unenforceable provisions with valid provisions, the economic effect of which comes as close as practicable to that of the invalid, illegal or unenforceable provisions.

- **7.10 Specific Performance**. Each Party agrees that irreparable harm, for which there shall be no adequate remedy at law and for which the ascertainment of monetary damages would be difficult, would occur in the event any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Each Party accordingly agrees that, subject to Section 7.12, the other Party shall be entitled to specifically enforce this Agreement and to obtain an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically the terms and provisions hereof, in each instance without being required to post bond or other security and in addition to, and without having to prove the adequacy of, other remedies at law.
- **7.11 Counterparts.** This Agreement may be executed in any number of counterparts which may be delivered by facsimile, each of which shall be deemed to be an original, but all of which counterparts shall together constitute one and the same instrument.
- 7.12 Dispute Resolution. The Parties agree that, in the event that a dispute or disagreement arises under or pursuant to this Agreement or the services contemplated hereby, then, prior to either Party instituting any action or claim or pursuing any other remedy hereunder, the Parties shall first attempt to settle such dispute amicably through good faith discussions upon the written request of either Party. Unless otherwise agreed in writing, each Service Provider shall continue to provide all Services and the Parties shall honor all other commitments under this Agreement during the course of any dispute pursuant to the provisions of this Section 7.12. In the event that any such dispute cannot be resolved through such discussions within a period of twenty (20) days after delivery of such written request, the dispute shall be finally settled by binding arbitration in the Republic of Singapore, using the English language and in accordance with the Rules of Arbitration of the International Chamber of Commerce then in effect. The arbitrator(s) shall have the authority to grant specific performance, and to allocate between the Seller and the Company the costs of arbitration in such equitable manner as the arbitrator(s) may determine, provided, however, that each Party shall bear its own translation, transportation and travel costs. Judgment upon the award so rendered may be entered in any court of competent jurisdiction or application may be made to such court for judicial acceptance of any award and an order of enforcement, as the case may be. THE PARTIES AGREE TO WAIVE OBJECTION TO VENUE AND CONSENT TO THE PERSONAL JURISDICTION OF THE COURTS OF THE REPUBLIC SINGAPORE IN ANY ACTION TO ENFORCE THIS ARBITRATION CLAUSE OR ANY ORDER OR AWARD OF THE ARBITRATIONS PANEL OR THE PROVISIONAL OR INTERIM REMEDIES PROVIDED FOR HEREIN.
- **7.13 Attorneys' Fees, Costs and Expenses**. Subject to Section 7.12, in any Proceeding to enforce an arbitral award in connection with this Agreement, the prevailing Party shall be entitled to recover reasonable attorneys' fees, costs and expenses incurred in connection with such enforcement Proceeding.
- **7.14 No Third Party Beneficiaries.** The terms and provisions of this Agreement are intended solely for the benefit of each Party and their respective successors and permitted assigns, and the Parties do not intend to confer third party beneficiary rights upon any other Person.

ARTICLE VIII

DEFINED TERMS

Except as otherwise provided herein, capitalized terms used in this Agreement shall have the meanings set forth below. Capitalized terms used in any Service Description Schedule but not otherwise defined therein, shall have the meanings assigned to them in this Agreement.

"Additional Services" mean, other than those Services covered by any Service Description Schedule, (a) any services that the Seller (or a Seller Subsidiary, NDSS or a third party) provided to any Company Group Entity as of the Signing Date and immediately prior to the Closing, including any reasonable variation of such services, which do not cause any unreasonable burden to the Service Provider or (b) any services that (i) the relevant Service Provider is reasonably capable of providing such Additional Services and (ii) the Company Group Entity reasonably believes it needs to receive from a Seller Group Entity (if it would be commercially reasonable to arrange for a third party (other than a Seller Group Entity) to provide certain services without any additional historical or institutional knowledge of the relationship between the Seller Group Entities and the Company Group Entities or how the Business was conducted in all material respects by the Company Group Entities as of the Signing Date and immediately prior to the Closing, which knowledge the personnel of the Seller Group Entities already have, then it shall be deemed that the relevant Company Group Entity does not need to receive such services from any Seller Group Entity) in order to (x) conduct the Business in all material respects as it was conducted as of the Signing Date and immediately prior to the Closing, in compliance with Legal Requirements, Governmental Approvals and Contracts (as in effect as of the Signing Date and immediately prior to the Closing), (y) make its own arrangements to perform any of the Services independently or efficiently from a systems and business process perspective, or (2) comply, and to enable the Purchaser Parent to comply, with applicable Legal Requirements and Governmental Approvals.

"Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under common control with such Person. For the purposes of this definition, "control," "controlled by" and "under common control with," with respect to the relationship between or among two or more Persons, means (i) the ownership of a majority of the voting share capital of a Person or (ii) the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of a Person, whether through the ownership of voting securities, by agreement or otherwise.

"Agreement" has the meaning specified in the first sentence of this Agreement.

"Amorton Business" means the amorphous silicon solar cells and amorphous solar cells business of KANTO SANYO Semiconductors Co., Ltd.

"Business" means the business of developing, designing, manufacturing, assembling, diagnosing, repairing, engaging in warranty services for, testing, distributing, marketing and selling, and conducting research with respect thereto, semiconductor products and parts and components therefor, as currently conducted by the Company Group Entities, including any technologies or products currently under development or design. For the avoidance of doubt, "Business" includes, for purposes of this Agreement, the information technology, human resources, finance and other systems used by or necessary for the Company or any other Company Group Entity to conduct the Business, as an operational matter, in all material respects as it was conducted as of the Signing Date and immediately prior to the Closing in compliance with Legal Requirements, Governmental Approvals and Contracts applicable thereto.

- "Business Day" means any day other than (i) a Saturday or a Sunday, or (ii) a day on which banking and savings and loan institutions are authorized or required by law to be closed in Japan or the country where the relevant Service is being provided.
- "Claim" means any litigation, suit, complaint, demand or legal, administrative, arbitral or criminal proceeding, information, request or order, in each case in the context of legal, arbitral or governmental proceedings.
 - "Closing" has the meaning specified in the Recitals.
 - "Closing Date" means the date of the Closing.
 - "Company" has the meaning specified in the first sentence of this Agreement.
- "Company Group Entities" means the Company and its Subsidiaries (including as a Subsidiary for this purpose, SSMPE, and SCE and SNA to the extent of the Transferred SCE Business and the Transferred SNA Business, but excluding the Amorton Business).
 - "Company Indemnified Parties" has the meaning specified in Section 5.2(a).
 - "Comparison Period" means the one (1)-year period prior to the Closing Date.
- "Confidential Information" means all Trade Secrets and other confidential and/or proprietary information of a Person, including information derived from reports, investigations, research, work in progress, codes, marketing and sales programs, manufacturing processes, financial projections, cost summaries, pricing formulae, contract analyses, financial information, projections, confidential filings with any state or federal agency, and all other confidential concepts, methods of doing business, ideas, materials or information prepared or performed for, by or on behalf of such Person by its employees, officers, directors, agents, Representatives or consultants.
 - "Consent" has the meaning specified in Section 1.11.
- "Contract" means any legally binding agreement, contract, consensual obligation, promise, understanding, arrangement, commitment or undertaking of any nature (whether written or oral and whether express or implied).
- "Copyrights" means all copyrights, copyrightable works, semiconductor topography and mask work rights, and applications for registration thereof, including all rights of authorship, publication, reproduction, distribution, performance and transformation, moral rights and all rights to register and obtain renewals and extensions of registrations, together with all other interests accruing by reason of international copyright, semiconductor topography and mask work conventions.

- "Derivative Technology" means: (a) for copyrightable or copyrighted material, any translation (including translation into other computer languages), portation, modification, correction, addition, extension, upgrade, improvement, compilation, abridgment or other form in which an existing work may be recast, transformed or adapted; (b) for patentable or patented material, any improvement thereon; and (c) for material protected by trade secret, any new material derived from such existing trade secret material, including new material that may be protected by copyright, patent and/or trade secret.
 - "Designated Representative" has the meaning specified in Section 1.10.
 - "Disclosing Party" has the meaning specified in Section 3.2.
 - "Effective Date" has the meaning specified in the first sentence of this Agreement.
- "Entity" means any corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust or company (including any limited liability company or joint stock company).
- "Force Majeure Event" means any act, event or occurrence beyond a Party's control, despite its reasonable efforts to prevent, avoid, delay or mitigate such acts, events or occurrences, which directly impairs such Party's ability to perform its obligations under this Agreement, including acts of God, fires, floods, storms, landslides, epidemics, lightning, earthquakes, drought, blight, famine, quarantine, blockade, governmental acts or inaction, order or injunctions, war, insurrection, terrorist activities and civil strife.
- "Governmental Approval" means any: permit, license, certificate, concession, approval, consent, ratification, permission, clearance, confirmation, exemption, waiver, franchise, certification, designation, rating, registration, variance, qualification, accreditation or authorization issued, granted, given or otherwise made available by any Governmental Authority.
- "Governmental Authority" means any: (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or quasi governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, organization, unit, body or Entity and any court or other tribunal); or (d) multinational organization or body.
- "Intellectual Property Rights" means all right, title, and interest arising from or in respect of any of the following, whether protected, created or arising under the laws of the United States or any other jurisdiction: (i) all Patents; (ii) all Trademarks; (iii) all Copyrights; (iv) any Trade Secrets; and (v) all rights, benefits, privileges, causes of action, and remedies relating to any of the foregoing, whether before or hereafter accrued, including the exclusive rights to apply for and maintain all registrations, renewals, and extensions, to sue for all past, present, and future infringements, unauthorized uses or disclosures, or misappropriations of any rights relating thereto, and to settle and retain proceeds from any such actions.

- "Invoice" has the meaning specified in Section 2.1(a).
- "Japanese Yen" or "\widetilde{Y}" means the currency of Japan.
- "*J-GAAP*" means Japan generally accepted accounting principles in effect on the date on which they are to be applied pursuant to this Agreement, applied consistently throughout the relevant periods.
- "Legal Requirement" means any federal, state, local, municipal, foreign or other law, statute, legislation, constitution, principle of common law, resolution, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, permit, ruling, directive, pronouncement, determination, decision, opinion or interpretation, in each case having the force of law enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority.
- "*Liability*" means any debt, obligation, liability or Tax of any nature (including any unmatured, unaccrued, contingent, conditional, joint or several liability), regardless of whether such debt, obligation, liability or Tax would be required to be disclosed on a balance sheet prepared in accordance with generally accepted accounting principles and regardless of whether such debt, obligation, or liability is immediately due and payable.
- "*Licensed Software*" means a Service Provider's proprietary software product(s) identified in <u>Exhibit C</u>, together with all end user manuals, reference manuals, installation guides and other documentation therefor, and any update, upgrade, patch, new version or other modification to a software program that corrects an error and/or improves performance of such software program or that adds substantial new functionality to such software program.
- "Losse" means any and all losses, Liabilities, damages, deficiencies (including fines, penalties and administrative, criminal or civil judgments and settlements), costs (including court costs, the costs of retaining expert witnesses and costs resulting from the defense of other claims indemnifiable under the terms of this Agreement) and expenses (including reasonable attorneys', accountants' and consultants' fees and expenses resulting from the defense of claims indemnifiable under the terms of this Agreement.
 - "Monthly Expense Report" has the meaning specified in Section 2.1(b).
 - "NDSS" has the meaning specified in Section 1.1(b).
 - "Nondisclosing Party" has the meaning specified in Section 3.2.
 - "Party" or "Parties" has the meaning specified in the preamble of this Agreement.
- "Patents" means all patent rights, title and interests in and to all letters patent and rights accorded under patent law systems, utility models, and applications therefor, including continuations, divisionals, continuations-in-part, reissues, reexaminations, substitutions, renewals, and extensions thereof, and patents issuing thereon.
 - "Person" means any individual, Entity or Governmental Authority.

- "Price" means a price for a given Service as specified in the applicable Service Description Schedule.
- "Proceeding" means any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), prosecution, formal contest, hearing, formal inquiry, formal inquest, audit, formal examination or formal investigation commenced, brought, conducted or heard at law or in equity or before any Governmental Authority or any arbitrator or arbitration panel.
 - "Purchase Agreement" has the meaning specified in the Recitals.
 - "Purchaser" has the meaning specified in the Recitals.
 - "Purchaser Parent" has the meaning specified in the Recitals.
 - "Representatives" means, in respect of any Party, officers, directors, employees, attorneys, accountants, advisors, agents, and Subsidiaries of such Party.
 - "SCE" means SANYO Component Europe GmbH.
 - "Seller" has the meaning specified in the first sentence of this Agreement.
 - "Seller Group Entities" means Seller and its Subsidiaries other than the Company Group Entities.
 - "Seller Subsidiary" means a Subsidiary of the Seller.
 - "Service" or "Services" has the meaning specified in Section 1.1(a).
 - "Service Description Schedule" has the meaning specified in Section 1.1(a).
- "Service Level Objectives" means the specific performance metrics measuring the performance by the Seller, or the other Service Providers, of the Services under this Agreement. Specific Service Level Objectives for Services may be set forth in the corresponding Service Description Schedule. Where none is set forth, the Seller and the other Service Providers shall perform such Services in such a manner and in such level or amount as is similar in all material respects to the manner and amount that the Seller or the other Service Providers performed such Services prior to the Closing Date (and any relevant the Comparison Period) in connection with the Business.
 - "Service Manager" has the meaning specified in Section 1.10.
- "Service Provider" means, with respect to any of the Services, (i) the Seller, (ii) a Seller Subsidiary or (iii) a third party specified as providing services under the applicable Service Description Schedule to the extent such third party was routinely used by the Seller to provide the same or similar services to a Company Group Entity prior to the Closing Date.

- "Service Recipient" means, with respect to any of the Services, (i) the Company and (ii) any other Company Group Entity specified as receiving services under the applicable Service Description Schedule.
 - "Signing Date" has the meaning specified in the Recitals.
 - "SNA" means SANYO North American Corporation.
 - "SSMPE" means SSMP Estate Corporation.
- "Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association, or other business entity of which (i) if a corporation or limited liability company in Japan, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers, or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof or (ii) if a limited liability company (outside of Japan), partnership, association, or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more Subsidiaries of that Person or a combination thereof and for this purpose, a Person or Persons own a majority ownership interest in such a business entity (other than a corporation) if such Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be or control any managing director or general partner of such business entity (other than a corporation).
 - "Tax" or "Taxes" has the meaning specified in Section 2.3.
- "Technology" means, collectively, all technology, formulae, recipes, specifications (including information regarding materials, ingredients, tools, apparatus, sources, vendors), procedures, processes, methods, techniques, ideas, creations, inventions and discoveries (whether patentable or unpatentable and whether or not reduced to practice), improvements, know-how, research and development, technical data, designs, models, algorithms, subroutines, software, works of authorship, copyrightable subject matter, and other similar materials, recordings, graphs, drawings, reports, analyses, and other writings, confidential or proprietary information, and general intangibles of like nature, and all tangible embodiments of the foregoing, in any form whether or not specifically listed herein.
- "*Trademarks*" means all trademarks, service marks, trade names, service names, brand names, trade dress, together with the goodwill associated with any of the foregoing, and all applications, ITU Applications, registrations and renewals thereof.
- "*Trade Secrets*" means any trade secrets or similar forms of protection for confidential information, including invention disclosures, formulae, recipes, specifications (including information regarding materials, ingredients, tools, apparatus, sources, vendors), procedures, processes, methods, techniques, ideas, creations, inventions and discoveries (whether patentable or unpatentable and whether or not reduced to practice), improvements, know-how, research and development, technical data, designs, models, algorithms, subroutines and similar confidential information.

- "Transaction" means, collectively, the transactions contemplated by the Transaction Agreements, and the execution, delivery and performance thereof.
- "Transaction Agreements" has the meaning specified in the Purchase Agreement.
- "Transferred SCE Business" means all of the assets and liabilities of SCE to be transferred under the Transferred SCE Business Assignment and Assumption.
- "Transferred SCE Business Assignment and Assumption" has the meaning specified in the Purchase Agreement.
- "Transferred SNA Business" means all of the assets and liabilities of SCE to be transferred under the Transferred SNA Business Assignment and Assumption.
- "Transferred SNA Business Assignment and Assumption" has the meaning specified in the Purchase Agreement.
- "Transition Service Period" has the meaning specified in Section 1.1(a).
- "Variances" has the meaning specified in Section 1.4.
- "Work Product" means all Technology and other creations (and any related improvements or modifications), that are conceived, created or otherwise developed by or for a Service Provider (alone or with others), or result from or are suggested by any work performed by or for a Service Provider (alone or with others), in connection with the Services. Except to the extent (if any) expressly set forth in a Services Description Schedule, Work Product shall include all deliverables and other materials delivered to a Service Recipient in connection with this Agreement.

[Remainder of page intentionally left blank]

Effective Date.	
	SANYO ELECTRIC CO., LTD.
	Ву:
	Name: Title:
	SANYO SEMICONDUCTOR CO., LTD.
	Ву:
	Name:

IN WITNESS WHEREOF, each Party has caused this Agreement to be executed on its behalf by its respective officer thereunto duly authorized all as of the

SIGNATURE PAGE TO TRANSITION SERVICES AGREEMENT

Title

CONFIDENTIAL

INTELLECTUAL PROPERTY ASSIGNMENT AGREEMENT

by and between [IP HOLD CO.]

a _____

and

SANYO ELECTRIC CO., LTD.

a Japanese corporation

Dated as of [__], 2010

TABLE OF CONTENTS

ARTICLE I	ASSIGNMENT OF TRANSFERRED IP ASSETS AND DELIVERY	1
1.1	Assignment of Transferred Trademarks	1
1.2	Assignment of Transferred Patents	2
1.3	Assignment of Transferred Copyrights and Know-How	2
1.4	Delivery of Tangible Business IP Assets	3
1.5	No Other Assignments	3
1.6	Allocation	3
1.7	Existing Licenses	3
1.8	Transition of Prosecution	4
1.9	Confirmatory Assignments	5
1.10	Assistance	5
1.11	No Suits with Third-Party IP Owners	6
ARTICLE II	GRANT-BACK PATENT LICENSE AND COVENANTS	6
2.1	Grant-back HoldCo Patent License to Seller	6
2.2	Covenant Not to Sue	7
2.3	Covenant to Sue Last	7
2.4	Remedies	8
2.5	No Exhaustion of Transferred Patents	8
2.6	Binding Effect of Covenants	8
ARTICLE III	GRANT-BACK COPYRIGHT AND KNOW-HOW LICENSE	9
3.1	Grant-back Company Copyrights and Know-How License to Seller	9
ARTICLE IV	ADDITIONAL TERMS	9
4.1	No Other Rights	9
4.2	Subsidiaries	10
4.3	Limitations	10
4.4	Further Assurances	11
4.5	Treatment of Patents Jointly Owned by Seller and Company	11
ARTICLE V	CONFIDENTIALITY	12
5.1	Seller Obligations	12

TABLE OF CONTENTS

(continued)

		Page
5.2	Exclusions	12
5.3	Required Disclosures	13
5.4	Limited Limitation on Use	13
5.5	Survival	13
ARTICLE VI	ASSIGNMENT AND TRANSFERABILITY	13
6.1	Permitted Assignment, Merger, and Change of Control	13
6.2	No Other Assignment	14
ARTICLE VII	NO WARRANTY, INDEMNIFICATION AND LIMITED LIABILITY	14
7.1	No Warranties	14
7.2	Indemnification	14
7.3	Limitation of Liability	14
ARTICLE VIII	TERM	15
8.1	Term	15
8.2	Termination	15
ARTICLE IX	MISCELLANEOUS	15
9.1	Purchase Agreement Terms	15
9.2	Injunctive Relief	15
9.3	Notices	15
Exhibits		

A-1

Exhibit A:

Certain Defined Terms

INTELLECTUAL PROPERTY ASSIGNMENT AGREEMENT

This INTELLECTUAL PROPERTY ASSIGNMENT AGREEMENT (the "Agreement") dated as	of [], 2010 ("Effective Date")	, is entered into between Sanyo Electric
Co., Ltd., a company organized under the laws of Japan ("Seller") and [IP HOLD CO.],¹ a	organized under the laws of	(" <i>HoldCo</i> "). Each of Seller and
HoldCo are sometimes hereinafter referred to as a "Party" and collectively as the "Parties." Certai	n capitalized terms used in this /	Agreement are defined in <u>Exhibit A</u> or ir
the Purchase Agreement (if not defined in Exhibit A).		

RECITALS

A. WHEREAS, Seller, ON Semiconductor Corporation ("*Purchaser Parent*"), and SCI LLC ("*Purchaser*") have entered into that certain Purchase Agreement, dated as of July 15, 2010 ("*Purchase Agreement*"), pursuant to which Purchaser will purchase from Seller, and Seller will sell to Purchaser, all of the issued and paid up share capital of Sanyo Semiconductor Co., Ltd. ("*Company*"), and certain other assets, including Intellectual Property Rights, and the Parties or their affiliates will enter into certain other transactions, all on the terms and subject to the conditions set forth in the Purchase Agreement;

B. WHEREAS, in connection with the Transaction, the Parties or their affiliates have entered into the Intellectual Property License Agreement and the Trademark License Agreement;

C. WHEREAS, as part of the Transaction, in consideration of a portion of the Purchase Price allocated to Intellectual Property Rights set forth in the Purchase Agreement, Seller wishes to transfer to HoldCo, and HoldCo wishes to receive, ownership of certain intellectual property rights and assets to effect the transfers contemplated by the Purchase Agreement;

D. WHEREAS, in connection with the Transaction, HoldCo wishes to grant-back to Seller, and Seller wishes to receive, licenses and covenants with respect to the above-mentioned intellectual property rights and assets transferred to HoldCo as part of the Transaction.

NOW, THEREFORE, the Parties hereby agree as follows:

AGREEMENT

ARTICLE I

ASSIGNMENT OF TRANSFERRED IP ASSETS AND DELIVERY

1.1 Assignment of Transferred Trademarks. Effective as of the Closing Date, in consideration of a portion of the Purchase Price allocated to Intellectual Property Rights set forth in the Purchase Agreement, Seller, on behalf of itself and all Seller Group Entities, hereby irrevocably assigns, conveys, sells, and transfers and agrees to assign, convey, sell, and transfer to HoldCo all of their rights, title and interest of every kind and character throughout the world in and to the Transferred Trademarks, to the maximum extent legally assignable, including, without limitation: (a) all national, foreign and state Trademarks, and all registrations, applications for registration and renewals and extensions, the exclusive right to apply for and maintain the Transferred Trademarks, and national, foreign and state counterparts of all of the foregoing, (b) all common law rights associated therewith, (c) all goodwill associated with the Transferred Trademarks and with the business to which the Transferred Trademarks pertain, and (d) all of their benefits, privileges, causes of action and remedies relating to any of the foregoing, whether before or hereafter accrued (including, without limitation, the right to sue for all past, present or future infringements or other violations of any rights in the Transferred Trademarks and to settle and retain proceeds from any such actions). Seller Group Entities retain no license or rights to use the Transferred Trademarks and agree not to challenge the validity of HoldCo's ownership in the Transferred Trademarks.

Note to draft: The party to this agreement may change depending on tax and structuring issues, such as possible use of an IP holding company.

1.2 Assignment of Transferred Patents. Effective as of the Closing Date, in consideration of a portion of the Purchase Price allocated to Intellectual Property Rights set forth in the Purchase Agreement, Seller, on behalf of itself and all Seller Group Entities hereby irrevocably assigns, conveys, sells, and transfers and agrees to assign, convey, sell and transfer to HoldCo, all of their rights, title and interest of every kind and character throughout the world in and to Transferred Patents (including any joint or co-ownership interest in Restricted Assigned Patents to the maximum extent of such interest, but subject to obtaining consent of any Third Party IP Owner whose consent is required for such assignment), to the maximum extent legally assignable, including, without limitation: (a) all national, foreign and PCT Patents, registrations, applications, continuations, renewals and extensions thereof, the exclusive right to apply for and maintain the Transferred Patents (other than Transferred Patents that are jointly owned with one or more third parties), and national, foreign and PCT counterparts of all of the foregoing, and (b) all of their benefits, privileges, causes of action and remedies relating to any of the foregoing, whether before or hereafter accrued (including, without limitation, the right to sue for all past, present or future infringements or other violations of any rights in the Transferred Patents and to settle and retain proceeds from any such actions). Except as set forth herein, Seller and the Seller Group Entities retain no license or rights to use the Transferred Patents and agree not to challenge the validity of HoldCo's ownership in the Transferred Patents.

1.3 Assignment of Transferred Copyrights and Know-How. Effective as of the Closing Date, to the extent not already transferred to Company pursuant to the Company Demerger, in consideration of a portion of the Purchase Price allocated to Intellectual Property Rights set forth in the Purchase Agreement, Seller, on behalf of itself and all Seller Group Entities, hereby irrevocably assigns, conveys, sells, and transfers and agrees to assign, convey, sell, and transfer to HoldCo, all of their rights, title and interest of every kind and character throughout the world (including all Intellectual Property Rights other than in Patents and Trademarks) in and to all Transferred Copyrights and Know-How (including, for the avoidance of doubt, any joint or co-ownership interest to the maximum extent of such interest, but subject to obtaining consent of any Third Party IP Owner whose consent is required for such assignment), to the maximum extent legally assignable, including, without limitation:

(a) all national, foreign and state registrations, applications, continuations, renewals, extensions and the exclusive right to apply for and maintain the Transferred Copyrights and Know-How (other than Transferred Copyrights and Know-How that are jointly owned with one or more Third Party IP Owners), and national, foreign and state counterparts of all of the foregoing, and (b) all of their benefits, privileges, causes of action and remedies relating to any of the foregoing, whether before or hereafter accrued (including, without limitation, the right to sue for all past, present or future infringements or other violations of any rights in the Transferred Copyrights and Know-How and to settle and retain proceeds from any such actions). Except as set forth herein, Seller and the Seller Group Entities retain no license or rights to use the Transferred Copyrights and Know-How and agree not to challenge the validity of HoldCo's ownership in the Transferred Copyrights and Know-How.

- 1.4 Delivery of Tangible Business IP Assets. Seller shall, and shall cause all Seller Group Entities to, deliver all of the Business IP Assets constituting tangible, including electronic, embodiments of the Business IP Assets ("Tangible Business IP Assets") through the delivery of facilities, personal property contained therein, and employees from Seller Group Entities to the Company Group Entities prior to or at the Closing Date, to the extent Seller may do so without breaching any third party inbound licenses. Without limiting the generality of the foregoing, with respect to any Tangible Business IP Assets that, through inadvertence or for any other reason (other than restrictions in third party inbound licenses), are not delivered via the delivery of facilities, personal property contained therein, and employees, upon Purchaser Parent's or HoldCo's reasonable request Seller shall, and shall cause Seller Group Entities to, promptly deliver, at no charge, all such assets following the Closing to those Company locations designated by Purchaser Parent or HoldCo to the extent Seller may do so without breaching any third party inbound licenses. To the extent that Seller or the Seller Group Entities inadvertently deliver any tangible, including electronic, embodiments of any Intellectual Property Rights that do not constitute Business IP Assets and that should not have otherwise been delivered to HoldCo or the Company Group Entities pursuant to any Transaction Agreement, upon Seller's reasonable request HoldCo shall, and shall cause Company Group Entities to, promptly return, at no charge, all such assets following the Closing to those locations designated by Seller to the extent HoldCo may do so without breaching any third party inbound licenses.
- **1.5 No Other Assignments**. Except as otherwise expressly set forth in another Transaction Agreement, the Parties acknowledge and agree that neither Seller nor any other Seller Group Entity will have any obligation to assign, convey, sell or transfer hereunder, or be deemed to have assigned, conveyed, sold, or transferred, any rights, title, or interest in any to any Intellectual Property Rights other than those expressly provided in <u>Sections 1.1</u>, 1.2, and 1.3.
 - 1.6 Allocation. The Parties shall use the procedures set forth in Exhibit B to finalize the lists of Transferred Patents and Licensed Patents.

1.7 Existing Licenses

(a) <u>Covenant to Terminate Existing Intercompany Licenses</u>. Prior to or as of the Closing Date, Seller shall, on behalf of itself and all Seller Group Entities and all Company Group Entities, cause all licenses and options to obtain licenses (including releases, non-assertions or covenants not to sue) under any Company IP Assets, that Seller, any Seller Group Entity or any Company Group Entity has granted to Seller or any of its Affiliates (other than to Company Group Entities) to be terminated and be of no further force or effect, excluding those licenses and options set forth in <u>Exhibit C</u>.

(b) Existing License Obligations. HoldCo hereby assumes and agrees to abide by, and shall cause its successors and assigns to assume and agree to abide by, the obligations (including without limitation licenses, options to obtain a license, releases, and covenants not to sue or assert) to which the Transferred IP Assets are subject immediately prior to the Closing Date that are contained in the agreements set forth in Schedule 3.14(g) of the Purchase Agreement that Seller or any Seller Group Entity has granted to any Person other than a Seller Group Entity (including Seller) ("License Obligations"). Without limiting the foregoing, upon HoldCo's reasonable written request, Seller will indicate, to the extent it may do so without breaching any contractual confidentiality obligations, whether or not Seller or any Seller Group Entity has granted any License Obligations under a particular item of Transferred IP Assets to any entity identified by HoldCo and if so, the scope of such License Obligations and other information reasonably needed by HoldCo to enforce its rights in such item of Transferred IP Assets. Both Parties will treat the existence of such request and response, Seller will treat the identity of the entity that is the subject of HoldCo's request, and HoldCo will treat the content of Seller's response as Confidential Information pursuant to the Purchase Agreement; provided, however, that HoldCo may use and disclose such Confidential Information to the extent reasonably necessary to enforce its rights in the Transferred IP Assets.

1.8 Transition of Prosecution. Without limiting the generality of Section 14.3 (Further Assurances) of the Purchase Agreement or Section 1.9 of this Agreement, on the Closing Date, Seller shall deliver to Company, to the extent not already delivered or in possession of Company pursuant to Section 1.4, the following documents: (a) a list of all outside counsel or agencies and Seller and Seller Group Entity personnel responsible for the prosecution and maintenance of registrations and applications within the Transferred IP Assets, and contact information for such counsel, agency or personnel (to the extent such personnel contact information can be disclosed without violating applicable Legal Requirements), and (b) the complete prosecution records and materials, including any invention disclosure forms, inventor notes, correspondence with the Governmental Authorities, and "file wrapper" for all registrations and applications within the Transferred IP Assets, to the extent that Seller or any Seller Group Entity has such documents in its possession immediately prior to the Closing Date (and shall, if requested by HoldCo, expressly instruct outside counsel to deliver the foregoing and consent to the retention of such outside counsel on behalf of HoldCo or a Company Group Entity in connection with prosecuting, maintaining, or otherwise protecting Transferred IP Assets) (for the avoidance of doubt, Seller will not be required to create new documents to satisfy its obligations under this clause (b) and will not be required to have governmental agencies deliver any documents to HoldCo under this clause (b)); provided, however, that Seller may maintain one archival copy of such prosecution records and materials, subject to Seller's and the Seller Group Entities' confidentiality obligations set forth in Section 5.1. With respect to each item of Transferred IP Assets that are Registered IP Assets, Seller agrees, for the greater of ninety (90) days after the Closing Date and ninety (90) days after the assignment from Seller to HoldCo of such item of Transferred IP Assets (where assignment is not legally effective until recorded), to: (i) continue to receive any notices or other correspondence from relevant Governmental Authorities, (ii) promptly forward such notices and correspondence in the ordinary course to HoldCo, and (iii) not abandon the prosecution of such Transferred IP Assets during such time unless at HoldCo's express written direction. Notwithstanding the foregoing, HoldCo shall be responsible for: (i) the prosecution of registrations and applications within the Transferred IP Assets, (ii) providing proper and timely instructions to Seller for the purpose of performing its obligations under this Section 1.8, and (iii) any out-of-pocket costs and expenses incurred by Seller due to its performance of the obligations set forth in the immediately preceding sentence.

1.9 Confirmatory Assignments. The Parties agree to have executed and filed with the Japan Patent Office, United States Patent and Trademark Office, and any other relevant Governmental Authority the confirmatory assignment(s) attached hereto as **Exhibit D** as are reasonably requested by HoldCo, customarily filed, or required by any Legal Requirement to record any Transferred IP Asset that is a Registered IP Asset and is assigned to HoldCo or any Company Group Entity pursuant to this Agreement ("**Assignment Documents**"). Except as otherwise set forth in **Section 1.8** and **Section 1.10**, any such procedures (such as registration) required for the assignment of the Transferred IP Assets from Seller and the Seller Group Entities to HoldCo shall be undertaken at HoldCo's expense and responsibility on or after the Closing Date. Seller shall deliver to HoldCo the Assignment Documents for all (i) Freely Assigned Patents and Freely Assigned Copyrights and Know-How on the Closing Date, and (ii) Restricted Assigned Patents and Restricted Assigned Patents and Restricted Assigned Copyrights and Know-How.

1.10 Assistance.

(a) <u>Assistance with Prosecution</u>. On and after the Closing Date upon HoldCo's request, Seller and the Seller Group Entities will, and will cause their Representatives to, cooperate with the prosecution of Transferred IP Assets, to the extent reasonably requested by HoldCo and customary or required by a Legal Requirement. Such requested assistance may include: (a) cooperating with HoldCo in the prosecution of applications on the Transferred IP Assets; (b) executing and delivering further papers, in recordable form, as may be customary or legally necessary to record, vest, secure, or perfect the rights and interests of HoldCo in and to Transferred IP Assets; (c) providing notices to third parties as to HoldCo's right and interest in and to the Transferred IP Assets as may be required by Legal Requirements; and (d) with their consent, providing HoldCo with contact information, if known, for former employees of Seller Group Entities.

(b) <u>Assistance with Enforcement</u>. On and after the Closing Date, HoldCo may, from time to time, request the assistance of Seller and the Seller Group Entities with respect to the enforcement of HoldCo's rights and interests in and to the Transferred IP Assets. Such requested assistance may include: (a) execution and delivery of further papers, in recordable form, as may be customary or legally necessary to protect or enforce the rights and interests of HoldCo in and to Transferred IP Assets; (b) obtaining the cooperation of current Seller Group Entity employees; and (c) with their consent, providing HoldCo with contact information, if known, for former employees of Seller Group Entities. Seller will consider such requests in this subsection (b) in good faith but may refuse any request that Seller considers to be unreasonably disruptive to the business operations of the Seller Group Entities, such as (without limitation) any request that would interfere with the ability of employees of the Seller Group Entities to perform their duties and responsibilities as employees of Seller Group Entities.

(c) <u>Costs and Expenses</u>. HoldCo shall reimburse Seller and the Seller Group Entities for reasonable pre-approved costs and expenses they incur in cooperating with and assisting HoldCo in connection with the prosecution and enforcement of the Transferred IP Assets under this <u>Section 1.10</u>, including compensation at a reasonable rate for any substantial amounts of time that employees of Seller Group Entities are unable to devote to their regular duties and responsibilities as a result of their efforts to assist HoldCo.

1.11 No Suits with Third-Party IP Owners. With respect to any Restricted Assigned Patents and Restricted Assigned Copyrights and Know-How, in the event of, and notwithstanding, any failure by Seller or a Seller Group Entity to obtain the consent to assign or license such Restricted Assigned Patents or Restricted Assigned Copyrights and Know-How to HoldCo or a Company Group Entity from the Third-Party IP Owner(s), Seller, on behalf of itself and all Seller Group Entities, hereby covenants not to commence, voluntarily join, or cooperate in any Proceeding brought by any Third-Party IP Owner against HoldCo or a Company Group Entity for any alleged infringement, violation or misappropriation of Restricted Assigned Patents or Restricted Assigned Copyrights and Know-How.

ARTICLE II

GRANT-BACK PATENT LICENSE AND COVENANTS

- **2.1 Grant-back HoldCo Patent License to Seller**. HoldCo, on behalf of itself and all Company Group Entities, hereby grants Seller and Seller Group Entities a non-exclusive, non-transferable (except as set forth herein), irrevocable, perpetual, non-sublicenseable, worldwide, fully paid-up and royalty-free license, under the Company Patents:
- (a) to make, Have Made, use, lease, sell, offer for sale, import, modify, develop, export and otherwise commercially exploit (directly or indirectly though multiple tiers, and to have such rights exercised by third parties solely on behalf of and solely for the benefit of Seller Group Entities) and dispose of Seller Licensed Products;
 - (b) to use any processes or methods covered by Company Patents to manufacture Seller Licensed Products;
 - (c) to provide or perform Seller Licensed Services, including to use any processes or methods covered by Company Patents; and
- (d) to design, develop and Have Made (but not to make) Semiconductor Products (including, but not limited to, (i) ASICs for digital camera products, and (ii) semiconductor devices, integrated circuits, and HICs for the development of power management modules and tuner modules) solely for use as incorporated into Seller Licensed Products (including, but not limited to, digital camera products, power management modules, and tuner modules), and to sell such Semiconductor Products on a standalone basis solely to another Seller Group Entity and solely for incorporation into Seller Licensed Products (for avoidance of doubt, in no event does this license include the right to sell Semiconductor Products on a standalone basis to any entity that is not a Seller Group Entity or for any use other than incorporation into Seller Licensed Products); provided, however, that, in each case, the licenses granted under Sections 2.1(a) and 2.1(b) (i) are not intended to limit in any manner, the terms and conditions of Section 7.9(a) (Non-Competition) of the Purchase Agreement and (ii) subject only to the limitation on Have Made rights with respect to Semiconductor Products set forth in Section 4.1 (and the definition of Have Made itself), are intended to cover every component and element of final products, including semiconductor devices, integrated circuits, HICs and modules, but solely if and as incorporated into and sold as part of such final products, regardless of the source of design or supply of such components and elements.

- **2.2 Covenant Not to Sue.** Commencing upon the expiration of the Noncompetition Period and subject to the conditions and limitations set forth in Section 2.5, HoldCo, on behalf of itself and all Company Group Entities, hereby irrevocably and perpetually (subject only to Section 2.5) covenants that neither it nor any Company Group Entity will assert (by way of litigation or any other Proceeding), anywhere in the world, any of the Company Patents (but excluding Systems-Level Claims):
- (a) against Seller or any Seller Group Entity for: (i) making, Having Made, using, leasing, selling, offering for sale, importing, modifying, developing, exporting or otherwise commercially exploiting (directly or indirectly through multiple tiers, including having such rights exercised by third parties solely on behalf of and solely for the benefit of Seller Group Entities) or disposing of Semiconductor Products; (ii) providing or performing Semiconductor Services, or (iii) using any processes or methods covered by Company Patents (but excluding Systems-Level Claims) to manufacture any Semiconductor Products or provide any Semiconductor Services; and
- (b) against Seller's or any Seller Group Entity's foundries or suppliers for manufacturing Semiconductor Products or otherwise providing Semiconductor Services on behalf of, and selling or otherwise distributing such Semiconductor Products to, Seller or any Seller Group Entity that are Have Made by such foundries or suppliers in accordance with the limitations to Have Made hereunder.

For the avoidance of doubt, this <u>Section 2.2</u> shall not limit HoldCo's and the Company Group Entities' right to assert the Company Patents with respect to Semiconductor Products and Semiconductor Services for any reason and against any Person (including, for the avoidance of doubt, against Seller and the Seller Group Entities) at any time during the Noncompetition Period.

2.3 Covenant to Sue Last. Commencing upon the expiration of the Noncompetition Period, HoldCo, on behalf of itself and all Company Group Entities, hereby irrevocably and perpetually (subject only to Section 2.5) covenants that it and the Company Group Entities will not assert (by way of litigation or any other Proceeding), anywhere in the world, any claim of infringement or other violation of any Company Patents against Seller or any Seller Group Entity with respect to any Semiconductor Products or Semiconductor Services unless and until (a) HoldCo and the Company Group Entities have exhausted all of their rights and remedies under such Company Patents against Seller's and the Seller Group Entities' Customers in a final non-appealable decision that includes one or more findings of infringement of such Company Patents by such products or services and (b) HoldCo and the Company Group Entities have diligently sought to collect the monetary damages awarded for such infringement but have been unable to do so.

2.4 Remedies. Notwithstanding and without limitation to Section 2.2 in the event that HoldCo or any Company Group Entity brings an action against Seller or any Seller Group Entity for infringement of the Company Patents with respect to (i) the making, Having Made, using, leasing, selling, offering for sale, importing, modifying, developing, exporting, or otherwise commercially exploiting or disposing of Semiconductor Products, or (ii) providing or performing Semiconductor Services, the sole and exclusive remedy available to HoldCo and the Company Group Entities will be recovery of reasonable royalties based on the revenue received by Seller and the Seller Group Entities (e.g. no injunctive or other equitable relief, enhanced damages for willfulness, lost profits, interest, and attorneys' fees), and HoldCo and the Company Group Entities will not seek any remedies other than such reasonable royalties.

2.5 No Exhaustion of Transferred Patents. The Parties acknowledge and agree that (a) both Parties intend that the covenants set forth in Sections 2.2 and 2.3 above do not and shall not (whether expressly, impliedly, by estoppel, by operation of law, or otherwise) exhaust or create any implied licenses under any of the Company Patents, and (b) the covenants set forth in Sections 2.2 and Section 2.3 above do not authorize the sale of any product or service to any third party that claims or contends in any Proceeding, anywhere in the world, that the covenants set forth in Sections 2.2 and Section 2.3 above, in whole or in part, do exhaust, or create any implied licenses that would result in immunity for the use, sale or manufacture of products that incorporate Seller Licensed Products or Semiconductor Products under, any Company Patents. The Parties further agree and intend that such covenants may not be circumvented by any litigation or similar legal, regulatory or administrative proceeding based, in whole or in part, on any theory or doctrine of extra-territorial patent infringement or indirect infringement. If, despite the Parties' joint intent, it is adjudicated in a final non-appealable decision in any country or jurisdiction that either the covenant set forth in Section 2.2 or the covenant set forth in Section 2.3 (or the combination of such covenants) exhausts, or creates any implied license that results in immunity for the use, sale or manufacture of products that incorporate Seller Licensed Products or Semiconductor Products, under any Company Patents, then the relevant portion of such covenant (set forth in either Section 2.2 or Section 2.3 as the case may be) will be deemed to be null and void from inception, but only with respect to such country or jurisdiction where such final non-appealable decision is issued and takes effect. After any such final, non-appealable decision that effectuates the Parties' above stated intent.

2.6 Binding Effect of Covenants. The licenses and covenants in Section 2.1, Section 2.2 and Section 2.3 are intended to run with the Company Patents irrespective of any future transfers, assignments and licenses, and intended to be binding on any future assignee or successor in interest to the rights, title, and/or interest of HoldCo and the Company Group Entities therein. HoldCo shall ensure that such assignee or successor in interest agrees to assume and is bound by these licenses and covenants. The Parties acknowledge and agree that the Company Patents are "intellectual property" as defined in Section 101(35A) of the United States Bankruptcy Code, as may be amended from time to time (the "Code"), and rights under such intellectual property have been granted hereunder in a contemporaneous exchange for value. The Parties further acknowledge and agree that if HoldCo or any Company Group Entity: (i) becomes insolvent or generally fails to pay, or admits in writing its inability to pay, its debts as they become due; (ii) applies for or consents to the appointment of a trustee, receiver or other custodian for it, or makes a general assignment for the benefit of its creditors; (iii) commences, or has commenced against it, any bankruptcy, reorganization, debt arrangement, or other case or proceeding under any bankruptcy or insolvency law, or any dissolution or liquidation proceedings; or (iv) elects to reject, or a trustee on behalf of it elects to reject, this Agreement or any agreement supplementary hereto (including, without limitation, the licenses and covenants set forth in Section 2.1, Section 2.2 and Section 2.3 above), pursuant to Section 365 of the Code, or if this Agreement or any agreement supplementary hereto (including, without limitation, the licenses and covenants set forth in Section 2.1 and Section 2.3 above) is deemed to be rejected pursuant to Section 365 of the Code for any reason, this Agreement, and any agreement supplementary hereto (including, for the avoidance of doubt, the licenses and cove

ARTICLE III

GRANT-BACK COPYRIGHT AND KNOW-HOW LICENSE

3.1 Grant-back Company Copyrights and Know-How License to Seller. HoldCo, on behalf of itself and all Company Group Entities, hereby grants Seller and Seller Group Entities a non- exclusive, non-transferable (except as set forth herein), irrevocable, perpetual, sublicenseable, nonexclusive, worldwide, fully paid-up and royalty-free right and license, under all of the Intellectual Property Rights of any kind and character throughout the world, (other than Patents and Trademarks) to all Company Copyrights and Know-How for any use or purpose whatsoever, now known or hereafter discovered excluding any use or purpose that constitutes, or is in connection with, a Prohibited Business. This license includes the right to use, execute, reproduce, prepare derivative works based on, distribute, transmit (internally and externally), disclose and otherwise exploit, in any form, the Company Copyrights and Know-How and derivative works based thereon, and to grant revocable or irrevocable sublicenses to any third party (directly and indirectly through multiple tiers) to do any or all of the foregoing excluding any use or purpose that constitutes, or is in connection with, a Prohibited Business.

ARTICLE IV

ADDITIONAL TERMS

4.1 No Other Rights. No other rights are granted hereunder, by implication, estoppel, statute or otherwise, except as expressly provided herein. The Have Made rights granted with respect to Semiconductor Products that are components of Seller Licensed Products under the license granted in Section 2.1, and with respect to Semiconductor Products under the covenants granted under Section 2.2 and Section 2.3, or any other right to have a right exercised by a third party with respect to Semiconductor Products, expressly do not apply to any portion of a Semiconductor Product used, leased, sold, offered for sale, imported, modified, exported or otherwise commercially exploited or assembled by Seller and Seller Group Entities where such portion uses or is based upon designs, specifications, and/or working drawings furnished by or on behalf of, or owned by, a third party (excluding library tools and standard cells). For the avoidance of doubt, and without limiting the generality of the foregoing, (a) the Have Made rights granted in this Agreement shall not extend to any third parties providing Semiconductor Products or Semiconductor Services for Seller to the extent that such third parties incorporate their own designs, specifications, and/or working drawings; and (b) if a third party supplies to a Seller Group Entity a Semiconductor Product that was not designed by or on behalf of any Seller Group Entity, then such third party supplier will not be licensed or otherwise authorized under the Have Made rights granted under the license granted in Section 2.1 but the Seller Group Entities' use, importation, offer for sale, and sale of such Semiconductor Product solely if and as incorporated into a Seller Licensed Product will be licensed under Section 2.1. For avoidance of doubt, Panasonic (as well as any Subsidiary of Panasonic that is not Seller or a Subsidiary of Seller) is a third party.

4.2 Subsidiaries. With respect to the licenses in Articles II and III, in the event an entity becomes a Subsidiary of either Party after the Effective Date, such new Subsidiary, upon becoming a Subsidiary, shall receive the same licenses and covenants as granted hereunder. In the event a Subsidiary of either Party ceases to be a Subsidiary during or after the term of this Agreement, the licenses and covenants granted hereunder to such Subsidiary shall automatically terminate forthwith. However, the licenses and covenants granted to the other Party and its Subsidiaries hereunder shall continue after such entity ceases to be a Subsidiary (even if such entity owns any of the Intellectual Property Rights that are subject to such licenses and covenants), and the Party that was formerly the parent of such Subsidiary shall ensure that such Subsidiary agrees to assume and be bound by such licenses and covenants with respect to any such Intellectual Property Rights.

4.3 Limitations. The licenses in Articles II and III shall not include any right or license to an item of Company Patents and Company Copyrights and Know-How (as the case may be) that cannot be licensed without the consent of, or the obligation to make payments, a Third-Party IP Owner in order to grant such licenses or rights (except for payments to HoldCo's or Company's Subsidiaries or payments to any Person for inventions made while employed by HoldCo or Company or any of its Subsidiaries after the Closing Date). However, at Seller's written request, HoldCo shall (and shall cause Company to) use reasonable efforts to obtain consent to grant such a license and, if HoldCo or Company is able to obtain such consent (or such consent is not necessary) and Seller agrees to pay all such payments on behalf of HoldCo and Company and fully indemnify HoldCo and Company and its Subsidiaries for any failure or delay by Seller in making such payments, HoldCo shall grant such a license. In the event of, and notwithstanding, any failure by HoldCo or Company to obtain the consent to license such Company Patents or Company Copyrights and Know-How to Seller Group Entity from the Third-Party IP Owner(s), HoldCo, on behalf of itself and Company and all Company Group Entities, hereby covenants not to commence, voluntarily join, or cooperate in any Proceeding brought by any Third-Party IP Owner against Seller or a Seller Group Entity for any alleged infringement, violation or misappropriation of such Company Patents or Company Copyrights and Know-How.

- **4.4 Further Assurances**. Upon Seller's reasonable request, HoldCo will, and will cause all Company Group Entities and its and their Representatives to, execute, verify, acknowledge and deliver all such further papers, in recordable form, as may be necessary to record, vest, secure and perfect the rights and interests of Seller in and to Company Patents and Company Copyrights and Know-How.
 - 4.5 Treatment of Patents Jointly Owned by Seller and Company. With respect to Company Jointly Owned Patents:
- (a) <u>Assignment to HoldCo.</u> Seller, on behalf of itself and the Seller Group Entities, hereby consents to the assignment of all Company Jointly Owned Patents by Company to HoldCo and shall take all necessary steps to effectuate such consent;
- (b) Filings for Certain Company Jointly Owned Patents. Seller, on the one hand, and Company or HoldCo (as the case may be) (for the purposes of this Section, Company and HoldCo are collectively referred to as a "party") on the other, shall mutually agree upon a Filing plan and share equally in the expenses of seeking and maintaining patent protection for the Company Jointly Owned Patents set forth on Exhibit I. The party preparing a Filing shall submit such Filing to the other party for its review and comment prior to any submission to any Governmental Authority. A Filing shall be deemed accepted by the receiving party if the receiving party does not provide a written notice of rejection to the submitting party within ten (10) business days after the submitting party's notice thereof. The parties will negotiate in good faith to determine an acceptable form, content, and time for the proposed Filing. Each party shall provide the other with copies of any correspondence, materials or other communications submitted to or received from a Governmental Authority or a third party relating to any Filing.
- (c) <u>Filings for All Other Company Jointly Owned Patents</u>. With respect to all other Company Jointly Owned Patents that are not set forth on <u>Exhibit I</u>, Seller will remain responsible for, and have sole control over, the Filings for all Company Jointly Owned Patents and HoldCo may consult with Seller on the status of and plans for any such Filing, at HoldCo's reasonable request. HoldCo will reimburse Seller for thirty percent (30%) of the expenses of seeking and maintaining patent protection for such Company Jointly Owned Patents. At HoldCo's reasonable request, Seller shall provide HoldCo with copies of any correspondence, materials or other communications submitted to or received from a Governmental Authority or a third party relating to any Filing for such a Company Jointly Owned Patent.
- (d) Enforcement. Seller may not assert, license or otherwise enforce its rights in the Company Jointly Owned Patents against any third party with respect to Semiconductor Products and Semiconductor Services without the prior written consent of Company or HoldCo (as the case may be). Company and HoldCo (as the case may be) may not assert, license or otherwise enforce their rights in the Company Jointly Owned Patents against any third party with respect to (i) ISB (Integrated System in Board) products and services and (ii) metal substrate for LED diode (but not driver) products and services without the prior written consent of Seller. Except as otherwise expressly provided in the immediately preceding two sentences, the Parties may assert, license or otherwise enforce the Company Jointly Owned Patents without consent of the other Party. The parties shall discuss in good faith the division of any proceeds resulting from any assertion, license or enforcement of Company Jointly Owned Patents.

(e) Other. The provisions and covenants in this Section 4.5 are intended to run with the Company Jointly Owned Patents irrespective of any future transfers, assignments and licenses of or under a party's joint ownership rights, and intended to be binding on any future assignee or successor in interest to the rights of a joint owner in the Company Jointly Owned Patents. Each Party, as a joint owner, shall ensure that such assignee or successor in interest agrees to assume and is bound by the provisions and covenants in this Section 4.5.

ARTICLE V

CONFIDENTIALITY

5.1 Seller Obligations. Seller shall, and shall cause the other Seller Group Entities to, hold the Company Confidential Information in confidence and not disclose any Company Confidential Information to any third party, other than to: (a) its employees, agents, consultants, Subsidiaries and other Affiliates who need to know such information and who are bound by restrictions regarding disclosure and use of such Company Confidential Information no less restrictive than those set forth herein; and (b) sublicensees, manufacturers, suppliers, contractors, distributors and customers who are bound in writing or otherwise by restrictions regarding disclosure and use of such Company Confidential Information no less restrictive than those set forth herein; provided, however, that in each case Seller shall not, and shall cause the other Seller Group Entities to not, disclose any Company Confidential Information in connection with a Prohibited Business. The Seller Group Entities shall take the same degree of care that they use to protect their own confidential and proprietary information and materials of similar nature and importance (but in no event less than reasonable care) to protect the confidentiality and avoid the unauthorized use, disclosure, publication or dissemination of the Company Confidential Information. Notwithstanding anything herein to the contrary, the Seller Group Entities may (without prior notification to, or approval or consent by, HoldCo) disclose to taxing authorities and/or to the Seller's and Seller Group Entities' representatives (including, outside counsel and advisors) any Company Confidential Information that is required to be disclosed in connection with Seller and the Seller Group Entities' tax filings, reports, claims, audits, or litigation.

5.2 Exclusions. The obligations set forth in Section 5.1 with respect to Company Confidential Information shall not apply to any Company Confidential Information to the extent that such Company Confidential Information: (a) is independently developed by Seller or its Affiliates after the Closing Date, or at any time by Panasonic or its Affiliates that are not Seller Group Entities or Company Group Entities, in each case without the use of Company Confidential Information; (b) is or becomes (or has already become) publicly available through no wrongful act of Seller or Seller Group Entities; (c) is rightfully received by Seller from a third party without obligation of confidentiality; (d) is disclosed with the consent of HoldCo or any of its Subsidiaries after the Closing Date; or (e) subject to Section 5.3 below, is disclosed in order to comply with any Legal Requirement (including any rule or regulation of the SEC or of a stock exchange which may require such disclosure) or with any order, rule or decree issued by a Governmental Authority. The inherent disclosure of any such Company Confidential Information in either: (x) the use, lease, sale or other distribution of any present or future product or provision of any present or future service of Seller or any of its Subsidiaries; (y) in the documentation relating to such product or service reasonably made publicly available; or (z) in the documentation relating to all domestic and foreign patent and copyright applications and registrations (and all patents and copyrights that issue therefrom and, to the extent applicable, all divisions, continuations, continuations-in-part, reexaminations, substitutions, reissues, extensions and renewals of such applications, registrations, copyrights and patents), shall not be deemed to be an unauthorized disclosure or publication of such information.

- **5.3 Required Disclosures**. Notwithstanding Section 5.2, in the event Seller or any Seller Group Entity is required to disclose Company Confidential Information pursuant to any Legal Requirement, and would otherwise be prohibited from doing so under Section 5.1, Seller or such Seller Group Entity shall: (i) promptly notify HoldCo of the existence, terms and circumstances surrounding such requirement; (ii) consult with HoldCo on the advisability of taking legally available steps to resist or narrow such request; and (iii) if disclosure of such Company Confidential Information is required, furnish only that portion of the Company Confidential Information which Seller or such Seller Group Entity is legally compelled to disclose. Seller or such Seller Group Entity shall not oppose actions by HoldCo to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Company Confidential Information.
- **5.4 Limited Limitation on Use.** Subject to the license granted in <u>Section 3.1</u>, Seller and its Subsidiaries will be free to use the Company Confidential Information for any purpose (other than any use or purpose that constitutes, or is in connection with, a Prohibited Business), including use in the development, manufacture, marketing, and maintenance of its products and services, provided that any such use of Company Confidential Information shall be subject to the terms and conditions of <u>Section 7.10(a)</u> (Non-Competition) of the Purchase Agreement.
 - **5.5 Survival**. The obligations of this <u>Article V</u> shall survive any termination or expiration of this Agreement.

ARTICLE VI

ASSIGNMENT AND TRANSFERABILITY

6.1 Permitted Assignment, Merger, and Change of Control. This Agreement may be assigned or transferred by HoldCo to Purchaser Parent or a Subsidiary of Purchaser Parent upon written notice to Seller, provided that Purchaser Parent or such Subsidiary of Purchaser Parent shall agree in writing to be bound by the terms and conditions of this Agreement. Seller may assign or transfer this Agreement or its rights hereunder solely in connection with a corporate merger or consolidation, sale of all or substantially all of its assets, or Change of Control of Seller as set forth below. In the event of any such assignment or transfer (other than an assignment or transfer resulting from the merger or consolidation of Seller with or into a Subsidiary of Seller), any merger or consolidation of Seller with or into Panasonic or any Subsidiary of Panasonic that is not a Subsidiary of Seller (regardless of which entity is the surviving entity), or any Change of Control of Seller, Seller shall (a) provide written notice to HoldCo within a reasonable period of time after such assignment or transfer, setting forth the name of the assignee, the entity with or into which Seller will be merged or consolidated, or the new controlling entity of Seller, as the case may be, (b) the assignee, the surviving entity of such merger or consolidation, or the new controlling entity of Seller, as the case may be, must agree in writing to be bound by the terms and conditions of this Agreement, and (c) all of the rights and licenses granted herein pursuant to Article II (Grant-Back Patent License and Covenants) to Seller and its Subsidiaries will be limited to use with Seller Licensed Products, Seller Licensed Services, Semiconductor Products, and Semiconductor Services, in each case of Seller Group Entities, that were (i) in existence immediately prior to the consummation of such assignment, merger, consolidation, or Change of Control and disclosed in contemporaneous, publicly available product roadmaps or similar documents.

6.2 No Other Assignment. Except as set forth in Section 6.1, this Agreement shall inure solely to the benefit of the Parties hereto and their Subsidiaries, and may not otherwise be assigned or otherwise transferred, in whole or in part, by a Party, voluntarily or involuntarily, by merger, division, operation of law, corporate reorganization, or otherwise, without the prior written consent of the other Party. Any such purported assignment or transfer shall be null and void.

ARTICLE VII

NO WARRANTY, INDEMNIFICATION AND LIMITED LIABILITY

- **7.1 No Warranties.** EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THE PURCHASE AGREEMENT, THE PARTIES MAKE NO OTHER REPRESENTATIONS OR WARRANTIES, WHETHER EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE OR NON-INFRINGEMENT.
- **7.2 Indemnification**. Each Party (the "*Indemnifying Party*") shall defend, indemnify, and hold harmless the other Party, its Subsidiaries, and its and their Representatives (collectively, "*Indemnified Parties*") from any and all damages, actions, claims, fines, judgments, costs, losses, liabilities and expenses (including reasonable legal fees and litigation expenses) suffered by Indemnified Parties arising out of or relating to a breach by the Indemnifying Party or its Subsidiaries of any of its covenants or other provisions herein.
- **7.3 Limitation of Liability**. Except as may be expressly set forth in a Transaction Agreement and with respect to third party damages, costs, liabilities, and settlements payable pursuant to indemnification obligations under Section 7.2, neither Party nor any of its Subsidiaries or Representatives shall be liable for any indirect, punitive, special, incidental or consequential damages, or damages for loss of profits, business interruption or otherwise, arising from or relating to this Agreement, even if such Party, its Subsidiaries, or their Representatives are expressly advised of the possibility of such damages. The foregoing limitation of liability and exclusion of certain damages shall apply regardless of the failure of essential purpose of any remedies available to either Party.

ARTICLE VIII

TERM

- **8.1 Term.** All licenses, assignments and grants of other rights (excluding the covenants set forth in <u>Sections 2.2</u> and 2.3) under this Agreement shall be effective as of the Closing Date. The covenants set forth in <u>Section 2.2</u> and 2.3 shall be effective upon the expiration of the Noncompetition Period. The term of each granted license and covenant shall continue for the life of the applicable Intellectual Property Right (except as expressly provided in <u>Section 2.5</u>). This Agreement shall continue in full force and effect until the expiration of the last to expire of the Intellectual Property Right licensed or covenanted hereunder.
- **8.2 Termination**. This Agreement, and the rights, licenses and covenants granted hereunder (except as expressly provided in <u>Section 2.5</u>), shall not be terminable by either Party or their respective Subsidiaries for any reason.

ARTICLE IX

MISCELLANEOUS

- **9.1 Purchase Agreement Terms**. The following provisions of the Purchase Agreement are hereby incorporated, mutatis mutandis, by this reference: 15.2, 15.3, 15.6, 15.8, 15.9, 15.11, 15.12, 15.13.
- **9.2 Injunctive Relief**. Notwithstanding anything to the contrary in this Agreement or the Purchase Agreement, each Party may seek temporary or preliminary injunctive relief in any court of competent jurisdiction in order to protect and preserve its Intellectual Property Rights and Company Confidential Information (in the case of HoldCo) and enforce its rights therein.
- **9.3 Notices**. All notices, requests, demands and other communications hereunder shall be either (a) delivered in person, (b) sent by overnight courier service or other express commercial delivery service, or (c) sent by facsimile with confirmation of receipt and, in each case, addressed as follows:

15

If to Hold Co:			
with copies to (which copy shall not constitute notice):			
with copies to (which copy shall not constitute notice).			
Y6			
If to Seller:			
with copies to (which copy shall not constitute notice):			

- (a) All notices, requests, instructions or documents given to either Party in accordance with this <u>Section 9.3</u> shall be deemed to have been given on the date of mailing or transmission, whether delivered by hand, by overnight courier service, or by facsimile, with confirmation of receipt on such date.
 - (b) Either Party hereto may change its address specified for notices herein by designating a new address by notice given in accordance with this

Section 9.3.

[Signatures Follow On a Separate Page]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be experience of the property of the parties have caused the parties have caused the parties have caused the parties of the parties have caused the parties have caused the parties have caused the parties have caused the parties have been property of the parties have caused the parties have caused the parties have caused the parties have been parties at the parties have caused the parties have caused the parties have caused the parties have been parties at the parties have caused the parties have caused the parties have caused the parties have been parties have been parties at the parties have caused the parties have been parties at the parties at the parties have been parties at the parties at	xecuted as of the date and year first above written.
SANYO ELECTRIC CO., LTD.	
By:	
Name:	
Title:	
[IP HOLD CO.]	
[]	
By:	
Name:	

Title:

[Signature Page to Intellectual Property Assignment Agreement]

EXHIBIT A

Certain Defined Terms

- "Assignment Documents" has the meaning set forth in Section 1.9 of the Intellectual Property Assignment Agreement.
- "Brand Licenses" means, collectively: (a) that certain Brand License Agreement, by and between Licensor and Licensee, dated April 1, 2008; (b) that certain Brand License Agreement, by and between Licensor and SSMC, dated April 1, 2008; (c) that certain Brand License Agreement, by and between Licensor and Sanyo LSI Design System Soft Co., Ltd., dated April 1, 2008; and (d) that certain Brand License Agreement, by and between Licensor and KSS dated April 1, 2008, dated April 1, 2008.
- "Code" has the meaning set forth in Section 1.7 of the Intellectual Property License Agreement and Section 2.6 of the Intellectual Property Assignment Agreement, as applicable.
- "Company Confidential Information" means all non-public Company Copyrights and Know-How.
- "Company Jointly Owned Patents" means Patents that are jointly owned by one or more Seller Group Entities and one or more Company Group Entities immediately following the Closing Date.
- "Company Names" has the meaning set forth in Section 1.1(b) of the Trademark License Agreement.
- "Company Copyrights and Know-How" means, collectively, (a) the Transferred Copyrights and Know-How and (b) Copyrights and Know-How that one or more Company Group Entities own as of the Closing Date.
- "Company Patents" means (a) the Transferred Patents, (b) all Patents that any of the Company Group Entities own as of the Closing Date (the "Company-Owned Patents"), and (c) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of the Company-Owned Patents.
- "Corporate Identity" means, collectively, letterhead, business cards, corporate signs, stationery, packaging slips, invoices, product datasheets, product labels, or any other materials that would normally identify a company or imply a corporate identity.
- "CTSL Patents" means all Patents, other than Licensed Patents, owned by Seller and the Seller Group Entities as of the Closing Date, and all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Patents, that would be infringed (absent a license) by making, using or selling Semiconductor Products (whether on a standalone basis or not), by using Manufacturing Apparatus to manufacture Semiconductor Products, or by providing Semiconductor Services. CTSL Patents include any Restricted Assigned Patents for which Seller has not obtained consent for assignment to HoldCo to the extent such Restricted Assigned Patents otherwise meet the definition set forth in the preceding sentence.

- "Current Company Products" means all Semiconductor Products that the Company Group Entities are selling, offering to sell, distributing, or making as of the Closing Date, including any minor improvements to the same that do not involve changes to existing features or functionality or the addition of new features or functionality.
- "Customer" means a third party (other than a Distributor or an End User) that purchases a Semiconductor Product (or a product containing one or more Semiconductor Products) manufactured by, or on behalf of, a Company Group Entity.
- "Distributor" means a third party that purchases a Semiconductor Product (or a product containing one or more Semiconductor Products) manufactured by, or on behalf of, a Company Group Entity and resells such Semiconductor Product (or such product containing one or more such Semiconductor Products, as the case may be) without modification or combination with any other component, part or product.
- "End User" means a third party that purchases a product containing one or more Semiconductor Products made by, or on behalf of, a Company Group Entity for such third party's own use and not for resale.
- "Effective Date" shall have the meaning set forth in the applicable preamble.
- "Filing" means the submission of any documentation, application, filing, registration or the like required to perfect or enforce, the Parties' interest in the Company Jointly Owned Patents under statutory Patent protection mechanisms, including, without limitation, any correspondence or other communication with any patent office or other Governmental Authorities with respect thereto.
- "Freely Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) solely owned by any Seller Group Entities, (ii) jointly owned by one or more Seller Group Entities and one or more Company Group Entities, (iii) jointly owned by one or more Seller Group Entities, Company Group Entities and Third-Party IP Owners, or (iv) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners; (b) primarily used in the Business by Company Group Entities as conducted immediately prior to the Closing Date; and (c) assignable without the consent of the Third-Party IP Owners (if any).
- "Freely Assigned Patents" means Patents that are listed on Exhibit F to the Intellectual Property Assignment Agreement.
- "Have Made" means the right to have a third party make a product or practice a method licensed under this Agreement for the use and benefit of a Party to this Agreement or its Subsidiary, provided that the designs, specifications, and/or working drawings for such products are supplied to such third party by, and originate with, the Party exercising such right or its Subsidiary (or with a Party's or its Subsidiaries' joint development with one or more third parties under terms of an agreement providing for the joint development of such designs, specifications, or working drawings).

- "HIC" means a hybrid integrated circuit, which is a chipset consisting of two or more integrated circuits that are packaged together onto a single device.
- "Know-How" means all right, title and interest in all Technology, Trade Secrets, ideas, concepts, proprietary techniques, processes, business and other methodologies, scientific, technical, research, development, engineering and business information, and other knowhow, including all rights arising under any law, including common law, state law, federal law or laws of foreign countries, other than Copyrights, Patents and Trademarks.
- "*Licensed Forms*" means the forms that incorporate either (i) the Licensed Marks, or (ii) the word "SANYO" or "\[\]\" (as the case may be) set forth in \(\frac{\text{Exhibit } C}{\text{to the Trademark License Agreement.}} \)
- "Licensed Marks" means the Trademarks set forth in Exhibit B to the Trademark License Agreement.
- "Licensed Patents" means (a) the Patents of Seller that are not Transferred Patents and are listed in Exhibit B to the Intellectual Property License Agreement, including any Restricted Assigned Patents for which Seller has not obtained consent for assignment, but has obtained consent to license to Company on the terms set forth in the Intellectual Property License Agreement ("Seller Licensed Patents"), and (b) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of the Seller Licensed Patents.
- "Manufacturing Apparatus" means any instrumentality or aggregate instrumentality primarily designed for use in the development and fabrication of Semiconductor Products.
- "Marked Products" has the meaning set forth in Section 2.2 of the Trademark License Agreement.
- "Name Change Procedures" has the meaning set forth in Section 1.1(b) of the Trademark License Agreement.
- "Panasonic" means Panasonic Corporation.
- "Parties" shall have the meaning set forth in the applicable preamble.
- "Party" shall have the meaning set forth in the applicable preamble.
- "Purchase Agreement" shall have the meaning set forth in the applicable recital.
- "Remaining Semiconductor Patents" means all Patents, other than the Licensed Patents, owned by Seller and the Seller Group Entities as of the Closing Date, and all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Patents, that would be infringed (absent a license) by making, using or selling Semiconductor Products (solely on a standalone basis), by using Manufacturing Apparatus to manufacture Semiconductor Products, or by providing Semiconductor Services. Remaining Semiconductor Patents include any Restricted Assigned Patents for which Seller has not obtained consent for assignment to HoldCo to the extent such Restricted Assigned Patents otherwise meet the definition set forth in the preceding sentence.

- "Restricted Assigned Copyrights and Know-How" means Copyrights and Know-How that are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; (b) primarily used in the Business by Company Group Entities as conducted immediately prior to the Closing Date; and (c) assignable only with the consent of the Third-Party IP Owners, subject to obtaining the consent of such Third-Party IP Owners.
- "Restricted Assigned Patents" means Patents listed on Exhibit H to the Intellectual Property Assignment Agreement which are (a) (i) jointly owned by one or more Seller Group Entities and one or more Third-Party IP Owners or (ii) jointly owned by one or more Seller Group Entities, one or more Company Group Entities, and one or more Third-Party IP Owners; and (b) assignable only with the consent of the Third-Party IP Owners.
- "Retained Copyrights and Know-How" means the Copyrights and Know-How of Seller Group Entities that are not Transferred Copyrights and Know-How and that are used in the Business as currently conducted or planned to be conducted, including any Restricted Assigned Copyrights and Know-How for which Seller has not obtained consent for assignment to HoldCo.
- "Seller Confidential Information" means all non-public Retained Copyrights and Know-How.
- "Seller Licensed Products" means all products other than Semiconductor Products. For the avoidance of doubt, ISB ("Integrated System in Board") products, photovoltaic devices, LED diodes (but not drivers), and laser diodes (but not drivers) are each Seller Licensed Products. For the further avoidance of doubt, (i) ASICs (Application Specific Integrated Circuits) for digital camera products, and (ii) semiconductor devices, integrated circuits and HICs for the development of power management modules and tuner modules are not Seller Licensed Products.
- "Seller Licensed Services" means, collectively, (i) a service that is not a Semiconductor Service, (ii) designing of ASICs (Application Specific Integrated Circuits) for digital camera products, and (iii) designing of semiconductor devices, integrated circuits and HICs for the development of power management modules and tuner modules.
- "Semiconductor Product" means any semiconductor device, integrated circuit or HIC (hybrid integrated circuit) on a standalone basis (i.e., a semiconductor device, integrated circuit or HIC alone, and not as part of a combination with other devices that are not semiconductor devices), whereby semiconductor devices include, but are not limited to, semiconductor device dies, semiconductor wafers or packaged semiconductors (including but not limited to supporting or packaging means or structure and software, firmware and microcode intended to be embedded or implemented therewith). Semiconductor devices do not lose their character as such whether or not part of an assemblage of packaged semiconductors or other devices. Semiconductor Products do not include ISB ("Integrated System in Board") products, photovoltaic devices, LED diodes, or laser diodes. For the avoidance of doubt, LED drivers and laser drivers are Semiconductor Products.

"Semiconductor Services" means the design, manufacture, testing, assembly, fabrication, and development of Semiconductor Products, including the provision of foundry services related thereto.

"Systems-Level Claim" means a claim of a Patent comprising multiple elements whereby at least one element is (or is embodied in) a Semiconductor Product and at least one of the other elements (which is not, and is not embodied in, such Semiconductor Product) is <u>not</u> a standard element technically essential to the functioning of the Semiconductor Product as part of any system or device. For avoidance of doubt, the definition of Systems-Level Claim is not intended to be construed or interpreted by reference to whether the authorized and unconditional sale of any particular Semiconductor Product would exhaust the claim in question. Attachment 1 to this Exhibit A provides one set of examples of Systems-Level Claims, but is not intended to modify or limit the meaning or scope of this definition.

"Third-Party IP Owner" mean a third party (i.e., not a Company Group Entity or Seller Group Entity) that owns a Patent, Copyright, or Know-How jointly with one or more Seller Group Entities.

"Transferred Circuit Layout Design Rights" means the circuit layout design rights set forth in Exhibit E to the Intellectual Property Assignment Agreement.

"Transferred Copyrights and Know-How" means (a) Freely Assigned Copyrights and Know-How, (b) any Restricted Assigned Copyrights and Know-How for which Seller obtains consent for assignment and (c) Transferred Circuit Layout Design Rights.

"Transferred Patents" means (a) Freely Assigned Patents, (b) any Restricted Assigned Patents for which Seller obtains consent for assignment to HoldCo, (c) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Freely Assigned Patents that are exclusively owned by Seller Group Entities immediately prior to the Closing Date, (d) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of such Restricted Assigned Patents that are exclusively owned by Seller Group Entities when Seller obtains consent for assignment of the Restricted Assigned Patents to HoldCo (provided that consent covers the assignment of such continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals as well), and (e) all continuations, continuations-in-part, divisionals, extensions, foreign counterparts, reissues, reexaminations, and renewals of Restricted Assigned Patents for which consent to the assignment thereof to HoldCo is not required.

"Transferred Trademarks" means, collectively, the Trademarks set forth in <u>Exhibit G</u> to the Intellectual Property Assignment Agreement and any common law trademarks owned immediately prior to the Closing Date by a Seller Group Entity and exclusively used in connection with the Business.

CERTIFICATION OF CHIEF EXECUTIVE OFFICER

I, Keith D. Jackson, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of ON Semiconductor Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure
 that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities,
 particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 4, 2010

/s/ KEITH D. JACKSON

Keith D. Jackson Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER

I, Donald A. Colvin, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of ON Semiconductor Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure
 that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities,
 particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 4, 2010

/s/ DONALD A. COLVIN

Donald A. Colvin Chief Financial Officer

Certification

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code), each of the undersigned officers of ON Semiconductor Corporation, a Delaware corporation ("Company"), does hereby certify, to such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the fiscal quarter ended October 1, 2010 ("Form 10-Q") of the Company fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: November 4, 2010 /s/ KEITH D. JACKSON

Dated: November 4, 2010

Keith D. Jackson

President and Chief Executive Officer

/s/ DONALD A. COLVIN

Donald A. Colvin
Executive Vice President and
Chief Financial Officer